

This report has been reproduced directly from the best available copy.

Copies of this document are available (while supplies last) upon written request to:

and the second second

Office of Fissile Materials Disposition, MD-4 Forrestal Building United States Department of Energy 1000 Independence Avenue, SW Washington, DC 20585

Attention: Storage and Disposition of Weapons-Usable Fissile Materials Final Programmatic Environmental Impact Statement/Volume II - Part A

Available to DOE and DOE contractors from the Office of Scientific and Technical Information, P.O. Box 62, Oak Ridge, TN 37831; telephone (615) 576-8401 for prices

Available to the public from the National Technical Information Services, 5285 Port Royal Road, Springfield, VA 22161.



COVER SHEET

RESPONSIBLE AGENCY: U.S. Department of Energy (DOE)

TITLE: Storage and Disposition of Weapons-Usable Fissile Materials Final Programmatic Environmental Impact Statement (DOE/EIS-0229)

CONTACTS:

I

environmental impact statement (PEIS), call (202) 586-4513 or fax (202) 586-4078 or contact: Mr. J. David Nulton Director Office of NEPA Compliance and Outreach Office of Fissile Materials Disposition U.S. Department of Energy 1000 Independence Ave., SW Washington, DC 20585 (202) 586-4513	 (NEPA) process, call (800) 472-2756 or contact: Ms. Carol Borgstrom Director Office of NEPA Policy and Assistance (EH-42) Office of Environment, Safety and Health U.S. Department of Energy 1000 Independence Ave., SW Washington, DC 20585 (202) 586-4600
---	--

ABSTRACT: This document analyzes the potential environmental consequences of alternatives for the long-term storage (up to 50 years), including storage until disposition, and disposition of weapons-usable fissile materials from U.S. nuclear weapon dismantlements under the responsibility of the DOE. Long-term storage of nonsurplus inventories of weapons-usable plutonium (Pu) and highly enriched uranium (HEU) are required for national defense purposes, while the disposition of surplus weapons-usable Pu is necessary in order to implement our national nonproliferation policy. In addition to the No Action Alternative, this PEIS assesses three storage alternatives (that is, upgrade at multiple sites, consolidation of Pu, and collocation of Pu and HEU) at six DOE candidate sites located across the country. These sites are Hanford Site, Nevada Test Site, Idaho National Engineering Laboratory, Pantex Plant, Oak Ridge Reservation, and Savannah River Site. Although they are not candidate sites for storage, Rocky Flats Environmental Technology Site (RFETS) and Los Alamos National Laboratory are assessed for the No Action Alternative. For the disposition of surplus Pu, three alternative categories (that is, deep borehole, immobilization, and reactor) with nine primary alternatives are assessed at several DOE and representative sites for analysis purposes. Evaluations of impacts on site infrastructure, water resources, air quality and noise, socioeconomics, waste management, public and occupational health and safety, and environmental justice are included in the assessment. The intersite transportation of nuclear and hazardous materials is also assessed. DOE's Preferred Alternative is identified in this Final PEIS. The Preferred Alternative for storage is a combination of No Action and Upgrade Alternatives for the various DOE sites, and phaseout of Pu storage at RFETS. The Preferred Alternative for disposition of surplus Pu is to pursue a disposition strategy involving a combination of immobilization and reactor alternatives, including vitrification, ceramic immobilization, and existing reactors.

PUBLIC INVOLVEMENT: The DOE issued a Draft PEIS on March 8, 1996, and held a formal public comment period on the Draft through June 7, 1996. In preparing the Final PEIS, DOE considered comments received via mail, fax, electronic bulletin board (Internet), and transcripts of messages recorded by telephone. In addition, comments and concerns were recorded by notetakers during interactive public meetings held during March and April 1996 in Denver, CO, Las Vegas, NV, Oak Ridge, TN, Richland, WA, Idaho Falls, ID, Washington, DC, Amarillo, TX, and North Augusta, SC. Comments received and DOE's responses to those comments are found in Volume IV of the Final PEIS.



Storage and Disposition of Weapons-Usable Fissile Materials Final Programmatic Environmental Impact Statement

Volume II - Part A

United States Department of Energy Office of Fissile Materials Disposition

December 1996

FOREWORD

This is the Storage and Disposition of Weapons-Usable Fissile Materials Final Programmatic Environmental Impact Statement (PEIS), prepared by the U.S. Department of Energy, Office of Fissile Materials Disposition. The document is composed of four volumes and a separate Summary. Changes made since the Draft PEIS are shown by change bar notation (vertical lines adjacent to the changes) in this Final PEIS for both text and tables. Deletion of one or more sentences is indicated by the phrase "Text deleted." in brackets. This Final PEIS includes the Preferred Alternative, which is a combination of alternatives. The Preferred Alternative is described in Section 1.6 and Chapter 2 of Volume I, and analyzed in Chapter 4 of Volume II. For all the alternatives, including the Preferred Alternative, a comparison of alternatives is presented in Section 2.5 of Volume I and a summary of impacts is presented in Section 4.6 of Volume II (Part B). Information from these sections is also presented in the Summary.

Volume I contains Chapters 1 through 3 of the PEIS. Chapter 1 includes a description of the history and background of the fissile materials disposition program, the purpose of and need for the proposed action, a summary of changes made to the Draft PEIS, and the Preferred Alternative. Chapter 2 gives a description of the proposed long-term storage and disposition alternatives, a description of how the alternatives were selected and why others were eliminated from further consideration, and a comparison of the alternatives in terms of their potential environmental impacts. Chapter 3 describes the affected environment at candidate long-term storage locations, and at sites and environmental settings for the disposition alternatives.

Volume II (Parts A and B) contains Chapters 4 through 10 of the PEIS. Chapter 4 describes the potential environmental impacts resulting from construction and operation of the proposed long-term storage and disposition alternatives, including the Preferred Alternative. Also contained in this chapter are intersite transportation impacts, a discussion of environmental justice issues, cumulative impacts due to the implementation of the proposed alternatives in addition to other actions at a site, avoided environmental impacts, and a summary of impacts. Chapter 5 provides a list of references used in the preparation of this document. Chapter 6 provides an index to the main text of the PEIS. Chapter 7 is a glossary of key terms used in the document. Chapter 8 is a list of preparers. Chapter 9 lists government agencies and organizations contacted during the preparation of this PEIS. Chapter 10 provides a distribution list for the document.

Volume III contains the appendices to this PEIS. Appendix A contains the fact sheet on the President's Nonproliferation and Export Control Policy, and the Joint Statement Between the United States and Russia on Nonproliferation. Appendix B provides specifications for key buildings within each facility complex analyzed in this PEIS. Appendix C describes requirements for construction and operation of the various facilities required to accomplish the storage and disposition activities essential to the alternatives described in this PEIS. Appendix D provides information on overall water usage for the storage and disposition facilities discussed in this PEIS. Appendix E gives a general overview of the Department of Energy (DOE) environmental restoration and waste management program, baseline waste management at DOE sites, and project-specific waste management activities associated with the proposed long-term storage and disposition alternatives. Appendix F provides detailed data supporting the air quality and noise analyses. Appendix G describes the methodology used for intersite transportation risk analysis and provides a summary of hazardous materials shipped to and from DOE sites, plus information on shipping containers. Appendix H evaluates various plutonium waste forms for potential disposal in a high-level waste repository. Appendix I describes operations of a Canadian Deuterium Uranium Reactor. Appendix J identifies the compliance requirements associated with the Proposed Action, as specified by the major Federal and State environmental, safety, and health statutes, regulations, and orders. Appendix K lists the scientific names of common nonthreatened and nonendangered animal and plant species identified in Chapters 3 and 4. Appendix L includes the supporting data used for assessing the No Action

TABLE OF CONTENTS

Table of Contents		i
Table of Contents	- 1	xi
List of Figures	,	xiii
List of Tables		xxix
List of Acronyms and Abbreviations		xxxvi
Chemicals and Units of Measure		
Metric Conversion Chart and Metric Prefixes	,	

.

, .

Volume II - Part A

 r_{i}^{i}

<u>, ۰</u>,

Char	oter 4		- - 11	,
Envi	rōnmen	tal Conse	quences	
41	Method	s for Asses	ssing Environmental Impacts	
-7.1	4.1.1	Land Res	ources	
	4.1.2	Site Infra	structure	
	4.1.3	Air Ouali	ty and Noise	
	4.1.4	Water Re	sources	
	4.1.5	Geology	and Soils	
	4.1.6	Biologica	l Resources	
	4.1.7	Cultural a	and Paleontological Resources	
	4.1.8	Socioeco	nomics	
	4.1.9	Public an	d Occupational Health and Safety	
	4.1.10	Waste M	anagement	
4.2	No Act	ion Alterna	ative and Long-Term Storage Alternatives	
	4.2.1	Hanford	Site	
		4.2.1.1	Land Resources	
		4.2.1.2	Site Infrastructure	
		4.2.1.3	Air Quality and Noise	
		4.2.1.4	Water Resources	
		4.2.1.5	Geology and Soils	
		4.2.1.6	Biological Resources	
		4.2.1.7	Cultural and Paleontological Resources	
		4.2.1.8	Socioeconomics	
		4.2.1.9	Public and Occupational Health and Safety	
		4.2.1.10	Waste Management	
	4.2.2	Nevada 🕻	Fest Site	
		4.2.2.1	Land Resources	
		4.2.2.2	Site Infrastructure	
		4.2.2.3	Air Quality and Noise	
		4.2.2.4	Water Resources	
		4.2.2.5	Geology and Soils	
		4.2.2.6	Biological Resources	
		4.2.2.7	Cultural and Paleontological Resources	
		4.2.2.8	Socioeconomics	
		4.2.2.9	Public and Occupational Health and Safety	
		4.2.2.10	Waste Management	

Alternative in the socioeconomics sections of this PEIS. Appendix M presents detailed information on the potential health risks associated with releases of radioactivity and hazardous chemicals from the proposed storage and disposition alternatives during normal operations and from postulated accidents. Appendix N describes different concepts for, and provides cost and benefit information on, the multipurpose reactor. Appendix O provides a description of facilities and operations for a can-in-canister approach to plutonium immobilization at the Savannah River Site in South Carolina. Appendix P describes the potential environmental impacts of using the Manzano Weapons Storage Area in New Mexico for the long-term storage of plutonium pits. Appendix Q identifies the potential health impacts from the storage of Rocky Flats Environmental Technology Site plutonium pits at the Pantex Plant in Texas. Appendix R discusses the aircraft crash and radioactive release probabilities for proposed storage and disposition facilities at Pantex Plant in Texas. A separate Classified Appendix was also prepared, which provides detailed analysis results for intersite transportation risks based on classified inventories of materials stored at DOE sites.

Volume IV (Parts A and B) is the Comment Response Document. It contains an overview of the public comment process, the comments received on the Draft PEIS during the public review period, and the DOE responses to those comments, including identifying changes made to the Draft PEIS in response to public comments.

The Summary provides a brief overview of the PEIS. It includes the purpose of and need for the Proposed Action, a description of the storage and disposition alternatives including the Preferred Alternative, and the potential environmental impacts resulting from these alternatives.

***** ----

4.2.3	Idaho Na	ational Engineering Laboratory
	4.2.3.1	Land Resources
	4.2.3.2	Site Infrastructure
	4.2.3.3	Air Ouality and Noise
	4.2.3.4	Water Resources
	4.2.3.5	Geology and Soils
	4.2.3.6	Biological Resources
	4.2.3.7	Cultural and Paleontological Resources
	4.2.3.8	Socioeconomics
	4.2.3.9	Public and Occupational Health and Safety
	4.2.3.10	Waste Management
4.2.4	Pantex P	Plant
	4.2.4.1	Land Resources
	4.2.4.2	Site Infrastructure
	4.2.4.3	Air Ouality and Noise
	4.2.4.4	Water Resources
	4.2.4.5	Geology and Soils
	4.2.4.6	Biological Resources
	4.2.4.7	Cultural and Paleontological Resources
	4.2.4.8	Socioeconomics
	4.2.4.9	Public and Occupational Health and Safety
	4.2.4.10	Waste Management
4.2.5	Oak Rid	ge Reservation
	4.2.5.1	Land Resources
	4.2.5.2	Site Infrastructure
	4.2.5.3	Air Quality and Noise
	4.2.5.4	Water Resources
	4.2.5.5	Geology and Soils
	4.2.5.6	Biological Resources
	4.2.5.7	Cultural and Paleontological Resources
	4.2.5.8	Socioeconomics
	4.2.5.9	Public and Occupational Health and Safety
	4.2.5.10	Waste Management
4.2.6	Savanna	h River Site
	4.2.6.1	Land Resources
	4.2.6.2	Site Infrastructure
	4.2.6.3	Air Quality and Noise
	4.2.6.4	Water Resources
	4.2.6.5	Geology and Soils'
	4.2.6.6	Biological Resources
	4.2.6.7	Cultural and Paleontological Resources
	4.2.6.8	Socioeconomics
	4.2.6.9	Public and Occupational Health and Safety
	4.2.6.10	Waste Management
4.2.7	Rocky F	lats Environmental Technology Site
	4.2.7.1	Land Resources
	4.2.7.2	Site Infrastructure
	4.2.7.3	Air Quality and Noise
	4.2.7.4	Water Resources
	4.2.7.5	Geology and Soils
	4.2.7.6	Biological Resources
	4.2.7.7	Cultural and Paleontological Resources 4-352

ł

ł

			4 9 5 9
	4.2.7.8	Socioeconomics	
	4.2.7.9	Public and Occupational Health and Safety	
	4.2.7.10	Waste Management	4–359
1.2.8	Los Alar	nos National Laboratory	
	4.2.8.1	Land Resources	
	4.2.8.2	Site Infrastructure	
	4.2.8.3	Air Quality and Noise	
	4.2.8.4	Water Resources	4–369
	4.2.8.5	Geology and Soils	4–371
	4.2.8.6	Biological Resources	4–372
	4.2.8.7	Cultural and Paleontological Resources	
	4.2.8.8	Socioeconomics	
	4.2.8.9	Public and Occupational Health and Safety	
	4.2.8.10	Waste Management	

Volume II - Part B

Į

4.3	Plutoni	ium Dispos	sition Alterna	atives and Related Activities	4–383
	4.3.1	Pit Disas	sembly/Con	version Facility	
		4.3.1.1	Land Resou	urces	
		4.3.1.2	Site Infrast	ructure	4–387
		4.3.1.3	Air Quality	and Noise	4–391
		4.3.1.4	Water Reso	ources	4–393
		4.3.1.5	Geology an	nd Soils	4-401
		4.3.1.6	Biological	Resources	4-402
		4.3.1.7	Cultural an	d Paleontological Resources	
		4.3.1.8	Socioecono	omics	
		4.3.1.9	Public and	Occupational Health and Safety	
		4.3.1.10	Waste Man	agement	
	4.3.2	Plutoniu	m Conversio	n Facility	
		4.3.2.1	Land Reso	urces	
		4.3.2.2	Site Infrast	ructure	
		4.3.2.3	Air Quality	and Noise	
		4.3.2.4	Water Reso	ources	
		4.3.2.5	Geology ar	nd Soils	
		4.3.2.6	Biological	Resources	
		4.3.2.7	Cultural an	d Paleontological Resources	
		4.3.2.8	Socioecono	omics	
		4.3.2.9	Public and	Occupational Health and Safety	
		4.3.2.10	Waste Man	lagement	
	4.3.3	Deep Bo	rehole Alter	native Category	
		4.3.3.1	Direct Disp	position Alternative	
			4.3.3.1.1	Land Resources	
			4.3.3.1.2	Site Infrastructure	
			4.3.3.1.3	Air Quality and Noise	
			4.3.3.1.4	Water Resources	
			4.3.3.1.5	Geology and Soils	
			4.3.3.1.6	Biological Resources	
			4.3.3.1.7	Cultural and Paleontological Resources	
			4.3.3.1.8	Socioeconomics	
			4.3.3.1.9	Public and Occupational Health and Safety	
			4.3.3.1.10	Waste Management	
				-	

1000

	4.3.3.2	Immobilized Disposition Alternative				
		4.3.3.2.1	Ceramic Imm	obilization Facility (for Borehole)	4-496	
			4.3.3.2.1.1	Land Resources	4-496	
	,		4.3.3.2.1.2	Site Infrastructure	4-499	
			4.3.3.2.1.3	Air Quality and Noise	4–504	
			4.3.3.2.1.4	Water Resources	4–508	
			4.3.3.2.1.5	Geology and Soils'	4-516	
			4.3.3.2.1.6	Biological Resources	4-517	
			4.3.3.2.1.7	Cultural and Paleontological Resources	4–522	
			4.3.3.2.1.8	Socioeconomics	4-525	
			4.3.3.2.1.9	Public and Occupational Health and Safety	4–527	
			4.3.3.2.1.10	Waste Management	4–540	
		4.3.3.2.2	Deep Borehol	e Complex	4–543	
			4.3.3.2.2.1	Land Resources	4–543	
			4.3.3.2.2.2	Site Infrastructure	4–544	
			4.3.3.2.2.3	Air Quality and Noise	4–546	
			4.3.3.2.2.4	Water Resources	4548	
			4.3.3.2.2.5	Geology and Soils	4–549	
			4.3.3.2.2.6	Biological Resources	4–550	
			4.3.3.2.2.7	Cultural and Paleontological Resources	4–551	
			4.3.3.2.2.8	Socioeconomics	4–552	
			4.3.3.2.2.9	Public and Occupational Health and Safety	4–553	
			4.3.3.2.2.10	Waste Management	4–559	
4.3.4	Immobili	ization Alter	native Categor	y	4-561	
	4.3.4.1	Vitrificatio	n Alternative	·····	4–561	
		4.3.4.1.1	Land Resourc	es	4-561	
		4.3.4.1.2	Site Infrastruc	ture	4-564	
		4.3.4.1.3	Air Quality ar	nd Noise	4-567	
		4.3.4.1.4	Water Resour	Ces	4-571	
		4.3.4.1.5	Geology and S	S011S	4-5/8	
		4.3.4.1.0	Biological Re	sources	4-579	
		4.3.4.1.7	Cultural and F		4-584	
		4.3.4.1.8	Socioeconomi Dublia and Oa	ICS	4-587	
		4.3.4.1.9	Public and Oc	cupational Health and Safety	4-389	
	1212	4.3.4.1.10	waste Manag	Cinent	4-002	
	4.J.4.2		I and Resource		4-005	
		43422	Site Infrastruc		4-608	
		43423	Air Quality ar	nd Noise	4-613	
		43474	Water Resource	na 140150	4-617	
		43425	Geology and S	Soile	4-674	
		43426	Biological Re		4-624	
		43427	Cultural and F	Paleontological Resources	4-630	
		4.3.4.2.8	Socioeconomi	ics	4-633	
		4.3.4.2.9	Public and Oc	cunational Health and Safety	4-635	
		4.3.4.2.10	Waste Manage	ement '	4-654	
	4.3.4.3	Electrometa	allurgical Treat	ment Alternative	4-657	
		4.3.4.3.1	Land Resource	es ''	4657	
		4.3.4.3.2	Site Infrastruc	turé	4658	
		4.3.4.3.3	Air Ouality an	d Noise	4-659	
		4.3.4.3.4	Water Resource	Ces	4-661	
		4.3.4.3.5	Geology and S	Soils	4662	

iv

		4.3.4.3.6	Biological Resources	4-663
		4.3.4.3.7	Cultural and Paleontological Resources	. 4–664
		4.3.4.3.8	Socioeconomics	.4-665
		4.3.4.3.9	Public and Occupational Health and Safety	4-666
		4.3.4.3.10	Waste Management	.4-670
4.3.5	Reactor	Category		.4-672
	4.3.5.1	Mixed Oxi	de Fuel Fabrication Facility	.4-673
		4.3.5.1.1	Land Resources	.4-673
		4.3.5.1.2	Site Infrastructure	.4–676
		4.3.5.1.3	Air Ouality and Noise	. 4680
		4.3.5.1.4	Water Resources	.4-682
		4.3.5.1.5	Geology and Soils	4-690
	-	43516	Biological Resources	4-691
		4.3.5.1.7	Cultural and Paleontological Resources	4-699
		43518	Socioeconomics	4_702
		43519	Public and Occupational Health and Safety	4_702
		435110	Waste Management	4_717
	4352	Fristing Li	aht Water Reactor Alternative	A_720
	4.3.3.2	43521	I and Resources	.4-720 A-720
		43522	Site Infrastructure	.4-720 A_721
		43523	Air Quality and Noise	A-721
		43524	Water Resources	Λ_723
		43525	Geology and Soils	Λ_72Λ
		43526	Biological Resources	A_725
		43527	Cultural and Paleontological Resources	Λ_726
		43528	Socioeconomics	A_727
		4.3.5.2.0	Public and Occupational Health and Safety	.4-121 1 778
		435210	Waste Management	.4-720 A_73A
	1353	Partially C	waste Management	A 725
	4.5.5.5	13531	Land Resources	A 735
		43532	Site Infractructure	A_736
		4.3.5.3.2	Air Ouality and Noise	A_737
		43534	Water Desources	A 730
		4.3.5.3.4	Geology and Soils	.4-739 A 740
		4.3.3.3.3	Piological Decourses	.4-740 A 741
		4.3.3.3.0	Cultural and Palaantalagical Resources	.4-741
		4.3.3.3.7	Cultural and Paleontological Resources	.4-142 1 712
		4.3.3.3.0	Dublic and Occupational Health and Safaty	.4-145
		4.3.3.3.9	Public and Occupational Health and Safety	.4-144
	1251	4.3.3.3.10 Evaluation	waste Management	.4-141
	4.3.3.4		I and Descurres	.4-749
		4.3.3.4.1	Land Resources	.4-749
		4.3.3.4.2	Site infrastructure	.4-152
		4.3.3.4.3	Air Quality and Noise	.4-151
		4.3.5.4.4	Water Resources	.4-/01
		4.3.3.4.3	Ucology and Solls	.4-113
		4.3.3.4.0	Biological Resources	.4-114
		4.3.3.4.1	Cultural and Paleontological Resources	.4-/84
		4.5.5.4.8	Socioeconomics	.4-187
		4.3.5.4.9	Public and Occupational Health and Safety	.4-789
		4.3.5.4.10	Waste Management	. 4–805

÷Ľ,

Stor	age and	Dispositio	n of Weapons-U	Isable	
Fiss	le Mate	rials Final	PEIS 12	The second providence	
		4.3.5.5	Canadian De	uterium Uranium Reactor Alternative (Retained Under	
			the Preferred	Alternative)	
	.		4.3.5.5.1	Effects Within the United States and Canada	
4.4	Inters	ite Transpo	ortation of Fissi	le Materials	
	4.4.1	Method	ology	· · · · · · · · · · · · · · · · · · ·	
-	4.4.2	Affecte	d Environment		
	,	4.4.2.1	Transportatio	on Procedures and Practices	
		4.4.2.2	Site Transpo	rtation Interfaces for Hazardous Materials	
		4.4.2.3	Packaging		
		4.4.2.4	Transportatio	on Routes and Emergency Preparedness Coordination	
			Among Fede	ral, State, and Local Agencies	
	4.4.3	Environ	mental Conseq	uences	
		4.4.3.1	No Action	4	
		4.4.3.2	Long-Term S	Storage Alternatives	
		4.4.3.3	Disposition A	Alternatives	
4.5	Enviro	onmental J	ustice in Minor	ity and Low-Income Populations	
	4.5.1	Method	ology		
	4.5.2	Environ	mental Conseq	uences	
4.6 °	Summ	ary of Imp	acts		
	4.6.1	Preferre	d Alternative I	mpacts	
	4.6.2	Long-To	erm Storage Al	ternatives	
		4.6.2.1	No Action	·····	4–896
		4.6.2.2	Upgrade		
		4.6.2.3	Consolidation	n	
		4.6.2.4	Collocation .	· · · · · · · · · · · · · · · · · · ·	
		4.6.2.5	Subalternativ	e Not Including Strategic Reserve and Weapons Resea	rch
			and Develop	ment Materials	
		4.6.2.6	Phaseout		
	4.6.3	Disposit	ion Alternative	3S	
		4.6.3.1	Activities Co	mmon to Disposition Alternatives	
		4.6.3.2	Deep Boreho	le Category	
			4.6.3.2.1 E	Direct Disposition	
			4.6.3.2.2 I	mmobilized Disposition	
		4.6.3.3	Immobilizatio	on Category	
			4.6.3.3.1 \	litrification	
			4.6.3.3.2 C	Ceramic Immobilization	
			4.6.3.3.3 E	ectrometallurgical Treatment	
		4.6.3.4	Reactor Cate	gory	
			4.6.3.4.1 N	Aixed Oxide Fuel Fabrication Facility	
-			4.6.3.4.2 E	xisting Light Water Reactor	
			4.6.3.4.3 P	artially Completed Light Water Reactor	
			4.6.3.4.4 E	volutionary Light Water Reactor	
			4.6.3.4.5 C	Canadian Deuterium Uranium Reactor	
4.7	Cumul	ative Impa	cts		
	4.7.1	Methodo	ology		
	4.7.2	Storage	Alternative Cu	mulative Impacts	
		4.7.2.1	Hanford Site		
			4.7.2.1.1 L	and Resources	
			4.7.2.1.2 S	ite Infrastructure	4-911
			4.7.2.1.3 A	Air Ouality and Noise	
			4.7.2.1.4 V	Vater Resources	
			4.7.2.1.5	eology and Soils	4-915

i

.....

	4.7.2.1.6	Biological Resources	
	4.7.2.1.7	Cultural and Paleontological Resources	
	4.7.2.1.8	Socioeconomics	
	4.7.2.1.9	Public and Occupational Health and Safety	
	4.7.2.1.10	Waste Management	
4722	Nevada Te	est Site	4-920
	4.7.2.2.1	Land Resources	4-920
	47222	Site Infrastructure	4-920
	47223	Air Quality and Noise	4_920
	47224	Water Resources	4_921
	1722	Geology and Soils	
	47226	Biological Descurges	1 023
	4.7.2.2.0	Cultural and Palaontological Pasources	4 923
	4.1.2.2.1	Cultural and Faleontological Resources	
	4.7.2.2.0	Dublic and Occurational Health and Safety	
	4.7.2.2.9	Public and Occupational Health and Salety	
4700	4.7.2.2.10	waste Management	
4.7.2.3	Idano Nati	Ional Engineering Laboratory	
	4.7.2.3.1	Land Resources	
	4.7.2.3.2	Site Infrastructure	
	4.7.2.3.3	Air Quality and Noise	
	4.7.2.3.4	Water Resources	
	4.7.2.3.5	Geology and Soils	
	4.7.2.3.6	Biological Resources	
	4.7.2.3.7	Cultural and Paleontological Resources	
	4.7.2.3.8	Socioeconomics	
	4.7.2.3.9	Public and Occupational Health and Safety	
	4.7.2.3.10	Waste Management	
4.7.2.4	Pantex Pla	int	
	4.7.2.4.1	Land Resources	
	4.7.2.4.2	Site Infrastructure	
	4.7.2.4.3	Air Quality and Noise	
	4.7.2.4.4	Water Resources	
	4.7.2.4.5	Geology and Soils	
	4.7.2.4.6	Biological Resources	
	4.7.2.4.7	Cultural and Paleontological Resources	
	4.7.2.4.8	Socioeconomics	
	4.7.2.4.9	Public and Occupational Health and Safety	
	4.7.2.4.10	Waste Management	
4.7.2.5	Oak Ridge	e Reservation	
	4.7.2.5.1	Land Resources	
	4.7.2.5.2	Site Infrastructure	
	4.7.2.5.3	Air Quality and Noise	
	4.7.2.5.4	Water Resources	
	4.7.2.5.5	Geology and Soils	
	4.7.2.5.6	Biological Resources	
	4.7.2.5.7	Cultural and Paleontological Resources	
	4.7.2.5.8	Socioeconomics	
	4.7.2.5.9	Public and Occupational Health and Safety	4–948
	4.7.2.5.10	Waste Management	4–949
4.7.2.6	Savannah	River Site	4–952
	4.7.2.6.1	Land Resources	4–952
	4.7.2.6.2	Site Infrastructure	4–952

-

į

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

				200 L 22	
~			4.7.2.6.3	Air Quality and Noise	
			4.7.2.6.4	Water Resources	
•			4.7.2.6.5	Geology and Soils	
			4.7.2.6.6	Biological Resources	4–956
			4.7.2.6.7	Cultural and Paleontological Resources	
			4.7.2.6.8	Socioeconomics	4–957
			4.7.2.6.9	Public and Occupational Health and Safety	
			4.7.2.6.10	Waste Management	4–958
		4.7.2.7	Rocky Flat	s Environmental Technology Site	
			4.7.2.7.1	Land Resources	
			4.7.2.7.2	Site Infrastructure	
			4.7.2.7.3	Air Quality and Noise	
			4.7.2.7.4	Water Resources	4–961
			4.7.2.7.5	Geology and Soils	4–961
			4.7.2.7.6	Biological Resources	4–961
			4.7.2.7.7	Cultural and Paleontological Resources	4–961
			4.7.2.7.8	Socioeconomics	4–962
			4.7.2.7.9	Public and Occupational Health and Safety	4–962
			4.7.2.7.10	Waste Management	4–962
		4.7.2.8	Los Alamo	s National Laboratory	4–963
			4.7.2.8.1	Land Resources	4–963
			4.7.2.8.2	Site Infrastructure	4–963
			4.7.2.8.3	Air Quality and Noise	4–963
			4.7.2.8.4	Water Resources	4–963
			4.7.2.8.5	Geology and Soils	4–963
			4.7.2.8.6	Biological Resources	4–963
			4.7.2.8.7	Cultural and Paleontological Resources	4–963
			4.7.2.8.8	Socioeconomics	4–964
			4.7.2.8.9	Public and Occupational Health and Safety	
		D	4.7.2.8.10	Waste Management	
	4.7.3	Dispositi	on Alternati	ves Cumulative Impacts	
		4.7.3.1	Land Resou	urces	
		4.7.3.2	Site Infrast		
		4.7.3.3	Air Quality		
		4.1.3.4	water Reso	JUICES	
		4.1.3.3	Geology an	IG SOIIS	
		4.1.3.0	Cultural an	d Dalaantalagigal Dagayraag	
		4.1.3.1	Cultural an	u Paleoniological Resources	
		4.1.3.0	Dublic and	Operational Upplith and Seferts	
		4.7.3.9	Public and	Occupational Health and Safety	
10	Lineve	4.7.3.10 Idabla Adu	waste wan	agement	
4.0	Augida	d Environ	erse Environ	ate and Impacts	4 074
4.9			lived Oxide	Evel Instead of Traditional Low Enriched Uranium Evel in	
	4.9.1	Nuclear l	Dower Diants	ruer instead of fradmonal Low-Enficied Oramum Fuer m	4 074
			Δ voided D	,	
		4017	Avoided A	in Onality Impacts	ر رح ب ۵۳۶_۸
		1012	Other Avei	n Quanty Inipacts	ر رجب مرحب ۸
		401 <i>1</i>	Impacts on	Ucu Environmental impacts Uranium Mining and Nuclear Fuel Cycle Industries	
	492	T.Z.1.4 Use of N	inclear Powe	r Plants Instead of Fossil Fuel Power Plants	<u>1_</u> 081
4,10	Relatio	nshin Betu	veen Short-T	erm Uses of Environment and the Maintenance and	
	Enhanc	ement of I	.ong-Term P	Productivity	4-983

4.11 4.12	Irreversible and Irretrievable Commitments of Resources Energy Requirements and Conservation Potential of the Various Alternatives and Mitigation Measures	4-984 4-985
Chap Refe	oter 5 rences	5–1
Chap Index	 x	6-1
Chap Gloss	oter 7 sary of Terms	7-1
Chap List (oter 8 of Preparers	8–1
Chap Fede	oter 9 ral, State, and Local Agencies and Organizations Contacted	9–1
Chap Distr	oter 10 ibution List	10–1

.

≂ it

ţ

LIST OF FIGURES

Volume II - Part B

Figure 4.5.1–1	Area	4-842
Figure 4.5.1–2	Low-Income Distribution by Poverty Status for Hanford Site and Surrounding Area.	4–843
Figure 4.5.1–3	Minority Population Distribution for Nevada Test Site and Surrounding Area.	4–844
Figure 4.5.1–4	Low-Income Distribution by Poverty Status for Nevada Test Site and Surrounding Area.	4–845
Figure 4.5.1–5	Minority Population Distribution for Idaho National Engineering Laboratory and Surrounding Area.	4–846
Figure 4.5.1–6	Low-Income Distribution by Poverty Status for Idaho National Engineering Laboratory and Surrounding Area.	4–847
Figure 4.5.1–7	Minority Population Distribution for Pantex Plant and Surrounding Area.	4–848
Figure 4.5.1–8	Low-Income Distribution by Poverty Status for Pantex Plant and Surrounding Area.	4–849
Figure 4.5.1–9	Minority Population Distribution for Oak Ridge Reservation and Surrounding Area.	4–850
Figure 4.5.1–10	Low-Income Distribution by Poverty Status for Oak Ridge Reservation and Surrounding Area.	4–851
Figure 4.5.1–11	Minority Population Distribution for Savannah River Site and Surrounding Area.	4–852
Figure 4.5.1–12	Low-Income Distribution by Poverty Status for Savannah River Site and Surrounding Area.	4–853
Figure 4.5.1–13	Minority Population Distribution for Rocky Flats Environmental Technology Site and Surrounding Area.	4–854
Figure 4.5.1–14	Low-Income Distribution by Poverty Status for Rocky Flats Environmental Technology Site and Surrounding Area.	
Figure 4.5.1–15	Minority Population Distribution for Los Alamos National Laboratory and Surrounding Area.	4856
Figure 4.5.1–16	Low-Income Distribution by Poverty Status for Los Alamos National Laboratory and Surrounding Area.	

۰<u>ء</u> ترت

1. S. S.

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS - - F - 1 5-5-1 , AND A THE THE A .; xii

LIST OF TABLES

Volume II - Part A

Table 4-	-1	Key to Locating Information on Environmental Consequences and Mitigation Measures for Storage Alternatives by Page Number	4–1
Table 4-	-2	Key to Locating Information on Environmental Consequences and Mitigation Measures for Disposition Alternatives by Page Number	4–2
Table 4.	1.8–1	Level of Service Letter Designations and Definitions	4–16
Table 4.	2.1.2-1	Site Infrastructure Changes Required for Operation at Hanford Site (Annual)—No Action (2005) and Storage Alternatives	430
Table 4.	2.1.3–1	Estimated Operational Concentrations of Pollutants at Hanford Site and Comparison With Most Stringent Regulations or Guidelines—No Action (2005) and Storage Alternatives	4–34
Table 4.	2.1.4–1	No Action and Potential Changes to Water Resources at Hanford Site—No Action (2005) and Storage Alternatives	4-40
Table 4.	2.1.9–1	Potential Radiological Impacts to the Public During Normal Operation at Hanford Site—No Action and Storage Alternatives	4–60
Table 4.	2.1.9–2	Potential Radiological Impacts to Workers During Normal Operation at Hanford Site—Storage Alternatives	461
Table 4.	2.1.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation at Hanford Site—No Action and Storage Alternatives	4–62
Table 4.	2.1.9–4	Upgrade Without Rocky Flats Environmental Technology Site or Los Alamos National Laboratory Material Alternative—Accident Impacts at Hanford Site	4–66
Table 4.	2.1.9–5	Upgrade With Rocky Flats Environmental Technology Site and Los Alamos National Laboratory Material Alternative—Accident Impacts at Hanford Site	4–68
Table 4.	.2.1.9–6	Consolidation Alternative Accident Impacts at Hanford Site	471
Table 4.	.2.1.9–7	Collocation Alternative Accident Impacts at Hanford Site	4–73
Table 4.	.2.1.10–1	Projected Spent Nuclear Fuel and Waste Management Under No Action (2005) at Hanford Site	4–76
Table 4.	.2.1.10–2	Estimated Annual Generated Waste Volumes at Hanford Site—No Action (2005) and Net Incremental for Storage Alternatives	4–78
Table 4	.2.2.2–1	Site Infrastructure Changes Required for Operation at Nevada Test Site (Annual)—No Action (2005) and Storage Alternatives	488
Table 4	.2.2.3–1	Estimated Operational Concentrations of Pollutants at Nevada Test Site and Comparison With Most Stringent Regulations or Guidelines—No Action (2005) and Storage Alternatives	4–91
Table 4	.2.2.4–1	No Action and Potential Changes to Water Resources at Nevada Test Site—No Action (2005) and Storage Alternatives	4–96

xiii

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Table 4.2.2.9–1	Potential Radiological Impacts to the Public During Normal Operation at Nevada Test Site—No Action and Storage Alternatives
Table 4.2.2.9–2	Potential Radiological Impacts to Workers During Normal Operation at Nevada Test Site—Storage Alternatives4–114
Table 4.2.2.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation at Nevada Test Site—No Action and Storage Alternatives
Table 4.2.2.9-4	Consolidation Alternative (P-Tunnel) Accident Impacts at Nevada Test Site
Table 4.2.2.9-5	Consolidation Alternative (New Storage Facility) Accident Impacts at Nevada Test Site
Table 4.2.2.9-6	Collocation Alternative (P-Tunnel) Accident Impacts at Nevada Test Site
Table 4.2.2.9–7	Collocation Alternative (New Pu and HEU Facilities) Accident Impacts at Nevada Test Site
Table 4.2.2.10–1	Projected Waste Management Under No Action (2005) at Nevada Test Site
Table 4.2.2.10-2	Estimated Annual Generated Waste Volumes at Nevada Test Site—No Action (2005) and Net Incremental for Storage Alternatives
Table 4.2.3.2-1	Site Infrastructure Changes Required for Operation at Idaho National Engineering Laboratory (Annual)—No Action (2005) and Storage Alternatives
Table 4.2.3.3–1	Estimated Operational Concentrations of Pollutants at Idaho National Engineering Laboratory and Comparison With Most Stringent Regulations or Guidelines—No Action (2005) and Storage Alternatives
Table 4.2.3.4-1	No Action and Potential Changes to Water Resources at Idaho National Engineering Laboratory—No Action (2005) and Storage Alternatives
Table 4.2.3.9–1	Potential Radiological Impacts to the Public During Normal Operation at Idaho National Engineering Laboratory—No Action and Storage Alternatives
Table 4.2.3.9–2	Potential Radiological Impacts to Workers During Normal Operation at Idaho National Engineering Laboratory—Storage Alternatives
Table 4.2.3.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation at Idaho National Engineering Laboratory—No Action (2005) and Storage Alternatives
Table 4.2.3.9-4	Upgrade Without Rocky Flats Environmental Technology Site or Los Alamos National Laboratory Material Alternative—Accident Impacts at Idaho National Engineering Laboratory
Table 4.2.3.9–5	Upgrade With Rocky Flats Environmental Technology Site and Los Alamos National Laboratory Material Alternative—Accident Impacts at Idaho National Engineering Laboratory
Table 4.2.3.9–6	Consolidation Alternative Accident Impacts at Idaho National Engineering Laboratory

.

.

Table 4.2.3.9–7	Collocation Alternative Accident Impacts at Idaho National Engineering Laboratory
Table 4.2.3.10-1	Projected Spent Nuclear Fuel and Waste Management Under No Action (2005) at Idaho National Engineering Laboratory
Table 4.2.3.10-2	Estimated Annual Generated Waste Volumes at Idaho National Engineering Laboratory—No Action (2005) and Net Incremental for Storage Alternatives
Table 4.2.4.2–1	Site Infrastructure Changes Required for Operation at Pantex Plant (Annual)—No Action (2005) and Storage Alternatives
Table 4.2.4.3–1	Estimated Operational Concentrations of Pollutants at Pantex Plant and Comparison With Most Stringent Regulations or Guidelines—No Action (2005) and Storage Alternatives
Table 4.2.4.4-1	No Action and Potential Changes to Water Resources at Pantex Plant—No Action (2005) and Storage Alternatives
Table 4.2.4.9–1	Potential Radiological Impacts to the Public During Normal Operation at Pantex Plant—No Action and Storage Alternatives
Table 4.2.4.9–2	Potential Radiological Impacts to Workers During Normal Operation at Pantex Plant—Storage Alternatives
Table 4.2.4.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation at Pantex Plant—No Action and Storage Alternatives
Table 4.2.4.9-4	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant
Table 4.2.4.9–4 Table 4.2.4.9–5	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant
Table 4.2.4.9–4 Table 4.2.4.9–5 Table 4.2.4.9–6	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant
Table 4.2.4.9–4 Table 4.2.4.9–5 Table 4.2.4.9–6 Table 4.2.4.9–7	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant
Table 4.2.4.9–4 Table 4.2.4.9–5 Table 4.2.4.9–6 Table 4.2.4.9–7 Table 4.2.4.9–8	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant
Table 4.2.4.9–4 Table 4.2.4.9–5 Table 4.2.4.9–6 Table 4.2.4.9–7 Table 4.2.4.9–8 Table 4.2.4.10–1	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant
Table 4.2.4.9-4 Table 4.2.4.9-5 Table 4.2.4.9-6 Table 4.2.4.9-7 Table 4.2.4.9-8 Table 4.2.4.10-1 Table 4.2.4.10-2	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant
Table 4.2.4.9-4 Table 4.2.4.9-5 Table 4.2.4.9-6 Table 4.2.4.9-7 Table 4.2.4.9-8 Table 4.2.4.10-1 Table 4.2.4.10-2 Table 4.2.5.2-1	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant
Table 4.2.4.9-4 Table 4.2.4.9-5 Table 4.2.4.9-6 Table 4.2.4.9-7 Table 4.2.4.9-8 Table 4.2.4.10-1 Table 4.2.4.10-2 Table 4.2.5.2-1 Table 4.2.5.3-1	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant
Table 4.2.4.9–4 Table 4.2.4.9–5 Table 4.2.4.9–6 Table 4.2.4.9–7 Table 4.2.4.9–8 Table 4.2.4.10–1 Table 4.2.5.2–1 Table 4.2.5.3–1 Table 4.2.5.4–1	Upgrade With Rocky Flats Environmental Technology Site Plutonium Pits Subalternative—Accident Impacts at Pantex Plant

そうないないない いたいない していたい しんない しょうしょう

....

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

~

CALC: N

Table 4.2.5.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation at Oak Ridge Reservation—No Action and Storage Alternatives
Table 4.2.5.9-4	Collocation Alternative Accident Impacts at Oak Ridge Reservation
Table 4.2.5.10–1	Projected Spent Nuclear Fuel and Waste Management Under No Action (2005) at Oak Ridge Reservation
Table 4.2.5.10-2	Estimated Annual Generated Waste Volumes at Oak Ridge Reservation
Table 4.2.6.2–1	Site Infrastructure Changes Required for Operation at Savannah River Site (Annual)—No Action (2005) and Storage Alternatives
Table 4.2.6.3–1	Estimated Operational Concentrations of Pollutants at Savannah River Site and Comparison With Most Stringent Regulations or Guidelines—No Action (2005) and Storage Alternatives
Table 4.2.6.4–1	No Action and Potential Changes to Water Resources at Savannah River Site—No Action (2005) and Storage Alternatives
Table 4.2.6.9–1	Potential Radiological Impacts to the Public During Normal Operation at Savannah River Site—No Action and Storage Alternatives
Table 4.2.6.9–2	Potential Radiological Impacts to Workers During Normal Operation at Savannah River Site—Storage Alternatives
Table 4.2.6.9-3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation at Savannah River Site—No Action and Storage Alternatives
Table 4.2.6.9-4	Upgrade With Rocky Flats Environmental Technology Site Non-Pit Plutonium Subalternative—Accident Impacts at Savannah River Site
Table 4.2.6.9–5	Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative—Accident Impacts at Savannah River Site
Table 4.2.6.9-6	Consolidation Alternative Accident Impacts at Savannah River Site
Table 4.2.6.9–7	Collocation Alternative Accident Impacts at Savannah River Site
Table 4.2.6.10–1	Projected Spent Nuclear Fuel and Waste Management Under No Action (2005) at Savannah River Site
Table 4.2.6.10-2	-Estimated Annual Generated Waste Volumes at Savannah River Site
Table 4.2.7.2–1	Site Infrastructure Changes Required for Operation at Rocky Flats Environmental Technology Site (Annual)—No Action (2005) and Storage Phaseout
Table 4.2.7.4–1	No Action and Potential Changes to Water Resources at Rocky Flats Environmental Technology Site—No Action (2005) and Storage Phaseout
Table 4.2.7.9–1	Potential Radiological Impacts to the Public During Normal Operation at Rocky Flats Environmental Technology Site—No Action
Table 4.2.7.9–2	Potential Radiological Impacts to Workers During Normal Operation at Rocky Flats Environmental Technology Site—No Action

xvi

List	of	Tab	les
------	----	-----	-----

Table 4.2.7.9-2	Potential Radiological Impacts to Workers During Normal Operation at Rocky Flats Environmental Technology Site—No Action
Table 4.2.7.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation at Rocky Flats Environmental Technology Site— No Action
Table 4.2.7.10–1	Projected Waste Management Under No Action (2005) at Rocky Flats Environmental Technology Site4-360
Table 4.2.8.2–1	Site Infrastructure Changes Required for Operation at Los Alamos National Laboratory (Annual)—No Action (2005) and Storage Phaseout4–365
Table 4.2.8.3–1	Estimated Operational Concentrations of Pollutants at Los Alamos National Laboratory and Comparison With Most Stringent Regulations or Guidelines—No Action (2005)4-367
Table 4.2.8.4–1	No Action and Potential Changes to Water Resources at Los Alamos National Laboratory—No Action (2005) and Storage Phaseout4–369
Table 4.2.8.9–1	Potential Radiological Impacts to the Public During Normal Operation at Los Alamos National Laboratory—No Action
Table 4.2.8.9–2	Potential Radiological Impacts to Workers During Normal Operation at Los Alamos National Laboratory—No Action
Table 4.2.8.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation at Los Alamos National Laboratory—No Action
Table 4.2.8.10-1	Projected Waste Management Under No Action (2005) at Los Alamos National Laboratory

Volume II - Part B

Table 4.3.1.2-1	Additional Site Infrastructure Needed for the Construction of the Pit Disassembly/Conversion Facility (Annual)4–388
Table 4.3.1.22	Additional Site Infrastructure Needed for the Operation of the Pit Disassembly/Conversion Facility (Annual)
Table 4.3.1.4–1	Potential Changes to Water Resources Resulting From Pit Disassembly/Conversion Facility4–394
Table 4.3.1.9–1	Potential Radiological Impacts to the Public During Normal Operation of the Pit Disassembly/Conversion Facility
Table 4.3.1.9–2	Potential Radiological Impacts to Workers During Normal Operation of the Pit Disassembly/Conversion Facility4-414
Table 4.3.1.9-3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation of the Pit Disassembly/Conversion Facility
Table 4.3.1.9-4	Pit Disassembly/Conversion Facility Accident Impacts at Hanford Site
Table 4.3.1.9-5	Pit Disassembly/Conversion Facility Accident Impacts at Nevada Test Site
Table 4.3.1.9-6	Pit Disassembly/Conversion Facility Accident Impacts at Idaho National Engineering Laboratory
Table 4.3.1.9–7	Pit Disassembly/Conversion Facility Accident Impacts at Pantex Plant

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

* 🗘

y (... j.k.

Table 4.3.1.9-8	Pit Disassembly/Conversion Facility Accident Impacts at Oak Ridge Reservation	4-421
Table 4.3.1.9–9	Pit Disassembly/Conversion Facility Accident Impacts at Savannah River Site	4–422
Table 4.3.1.10-1	Estimated Annual Generated Waste Volumes for the Pit Disassembly/Conversion Facility	4-425
Table 4.3.2.2–1	Additional Site Infrastructure Needed for the Construction of the Plutonium Conversion Facility (Annual)	4-430
Table 4.3.2.2–2	Additional Site Infrastructure Needed for the Operation of the Plutonium Conversion Facility (Annual)	4–'432
Table 4.3.2.31	Estimated Operational Concentrations of Pollutants and Comparison With Most Stringent Regulations or Guidelines—Plutonium Conversion Facility and No Action Alternative	4-436
Table 4.3.2.4–1	Potential Changes to Water Resources Resulting From the Plutonium Conversion Facility	4-442
Table 4.3.2.9–1	Potential Radiological Impacts to the Public During Normal Operation of the Plutonium Conversion Facility	4459
Table 4.3.2.9-2	Potential Radiological Impacts to Workers During Normal Operation of the Plutonium Conversion Facility	4-461
Table 4.3.2.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation of the Plutonium Conversion Facility	4-463
Table 4.3.2.9-4	Plutonium Conversion Facility Accident Impacts at Hanford Site	4–464
Table 4.3.2.9–5	Plutonium Conversion Facility Accident Impacts at Nevada Test Site	4–465
Table 4.3.2.9-6	Plutonium Conversion Facility Accident Impacts at Idaho National Engineering Laboratory	4–466
Table 4.3.2.9-7	Plutonium Conversion Facility Accident Impacts at Pantex Plant	4467
Table 4.3.2.9–8	Plutonium Conversion Facility Accident Impacts at Oak Ridge Reservation	4-468
Table 4.3.2.9-9	Plutonium Conversion Facility Accident Impacts at Savannah River Site	4-469
Table 4.3.2.10-1	Estimated Annual Generated Waste Volumes for the Plutonium Conversion Facility	4-471
Table 4.3.3.1.2-1	Additional Site Infrastructure Needed for the Construction of the Deep Borehole Complex—Direct Disposition Alternative (Annual)	4-476
Table 4.3.3.1.2–2	Additional Site Infrastructure Needed for the Operation of the Deep Borehole Complex—Direct Disposition Alternative (Annual)	4-477
Table 4.3.3:1.3–1	Estimated Operational Concentrations of Pollutants and Comparison With Most Stringent Regulations or Guidelines—Deep Borehole Complex and No Action Alternative—Direct Disposition Alternative	4–479
Table 4.3.3.1.4-1	Potential Changes to Water Resources Resulting From the Deep Borehole Complex—Direct Disposition Alternative	4-480
Table 4.3.3.1.9-1	Potential Radiological Impacts to the Public During Normal Operation of the Deep Borehole Complex—Direct Disposition Alternative	4-490

List	of	Tab	les
------	----	-----	-----

Table 4.3.3.1.9-2	Potential Radiological Impacts to Workers During Normal Operation of the Deep Borehole Complex—Direct Disposition Alternative
Table 4.3.3.1.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation of the Deep Borehole Complex—Direct Disposition Alternative
Table 4.3.3.1.9-4	Range of Accident Impacts for a Set of Accidents for the Deep Borehole Complex—Direct Disposition Alternative
Table 4.3.3.1.10-1	Estimated Annual Generated Waste Volumes for the Deep Borehole Complex—Direct Disposition Alternative
Table 4.3.3.2.1.2-1	Additional Site Infrastructure Needed for the Construction of the Ceramic Immobilization Facility (For Borehole)—Immobilized Disposition Alternative (Annual)
Table 4.3.3.2.1.2-2	Additional Site Infrastructure Needed for the Operation of the Ceramic Immobilization Facility (For Borehole)—Immobilized Disposition Alternative (Annual)
Table 4.3.3.2.1.3–1	Estimated Operational Concentrations of Pollutants and Comparison With Most Stringent Regulations or Guidelines—Ceramic Immobilization Facility and No Action Alternative (For Borehole)—Immobilized Disposition Alternative
Table 4.3.3.2.1.4–1	Potential Changes to Water Resources Resulting From Ceramic Immobilization Facility (For Borehole)—Immobilized Disposition Alternative
Table 4.3.3.2.1.9–1	Potential Radiological Impacts to the Public During Normal Operation of the Ceramic Immobilization Facility (For Borehole)—Immobilized Disposition Alternative
Table 4.3.3.2.1.9–2	Potential Radiological Impacts to Workers During Normal Operation of the Ceramic Immobilization Facility (For Borehole)—Immobilized Disposition Alternative
Table 4.3.3.2.1.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation of the Ceramic Immobilization Facility (For Borehole)—Immobilized Disposition Alternative
Table 4.3.3.2.1.9-4	Ceramic Immobilization Facility (For Borehole) for Immobilized Disposition Alternative—Accident Impacts at Hanford Site
Table 4.3.3.2.1.9-5	Ceramic Immobilization Facility (For Borehole) for Immobilized Disposition Alternative—Accident Impacts at Nevada Test Site
Table 4.3.3.2.1.9-6	Ceramic Immobilization Facility (For Borehole) for Immobilized Disposition Alternative—Accident Impacts at Idaho National Engineering Laboratory
Table 4.3.3.2.1.9-7	Ceramic Immobilization Facility (For Borehole) for Immobilized Disposition Alternative—Accident Impacts at Pantex Plant
Table 4.3.3.2.1.9–8	Ceramic Immobilization Facility (For Borehole) for Immobilized Disposition Alternative—Accident Impacts at Oak Ridge Reservation
Table 4.3.3.2.1.9-9	Ceramic Immobilization Facility (For Borehole) for Immobilized Disposition Alternative—Accident Impacts at Savannah River Site

j...'

时法

1,15

Ì

÷1

ì

~

Table 4.3.3.2.1.10-1	Estimated Annual Generated Waste Volumes for Ceramic Immobilization Facility (For Borehole)—Immobilized Disposition Alternative	541
Table 4.3.3.2.2.2–1	Additional Site Infrastructure Needed for the Construction of the Deep Borehole Complex—Immobilized Disposition Alternative (Annual)	544
Table 4.3.3.2.2.2-2	Additional Site Infrastructure Needed for the Operation of the Deep Borehole Complex—Immobilized Disposition Alternative (Annual)	545
Table 4.3.3.2.2.3-1	Estimated Operational Concentrations of Pollutants and Comparison With Most Stringent Regulations or Guidelines—Deep Borehole Complex and No Action Alternative—Immobilized Disposition Alternative	547
Table 4.3.3.2.2.9–1	Potential Radiological Impacts to the Public During Normal Operation of the Deep Borehole Complex—Immobilized Disposition Alternative4-4	554
Table 4.3.3.2.2.9-2	Potential Radiological Impacts to Workers During Normal Operation of the Deep Borehole Complex—Immobilized Disposition Alternative4-	555
Table 4.3.3.2.2.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation of the Deep Borehole Complex—Immobilized Disposition Alternative	556 [,]
Table 4.3.3.2.2.9-4	Range of Accident Impacts of the Deep Borehole Complex—Immobilized Disposition Alternative	557
Table 4.3.3.2.2.10-	1 Estimated Annual Generated Waste Volumes for the Deep Borehole Complex—Immobilized Disposition Alternative	560
Table 4.3.4.1.2–1	Additional Site Infrastructure Needed for the Construction of the Vitrification Facility Alternative (Annual)4–	565
Table 4.3.4.1.2-2	Additional Site Infrastructure Needed for the Operation of the Vitrification Facility Alternative (Annual)4-	-566
Table 4.3.4.1.3–1	Estimated Operational Concentrations of Pollutants and Comparison With Most Stringent Regulations or Guidelines—Vitrification Alternative and No Action Alternative	-568
Table 4.3.4.1.4-1	Potential Changes to Water Resources Resulting From the Vitrification Alternative4-	-572
Table 4.3.4.1.9–1	Potential Radiological Impacts to the Public During Normal Operation of the Vitrification Alternative4-	-590
Table 4.3.4.1.9–2	Potential Radiological Impacts to Workers During Normal Operation of the Vitrification Alternative4-	-593
Table 4.3.4.1.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation of the Vitrification Alternative4-	- 59 4
Table 4.3.4.1.9-4	Vitrification Alternative Accident Impacts at Hanford Site4-	-595
Table 4.3.4.1.9-5	Vitrification Alternative Accident Impacts at Nevada Test Site	-596
Table 4.3.4.1.9-6	Vitrification Alternative Accident Impacts at Idaho National Engineering Laboratory4	-597
Table 4.3.4.1.9–7	Vitrification Alternative Accident Impacts at Pantex Plant	-598
Table 4.3.4.1.9–8	Vitrification Alternative Accident Impacts at Oak Ridge Reservation4	-599
Table 4.3 4.1.9–9	Vitrification Alternative Accident Impacts at Savannah River Site4	-600

List	of	Tables
------	----	--------

Table 4.3.4.1.10-1	Estimated Annual Generated Waste Volumes for the Vitrification Alternative4-603
Table 4.3.4.2.2-1	Additional Site Infrastructure Needed for the Construction of the Ceramic Immobilization Alternative (Annual)
Table 4.3.4.2.2-2	Additional Site Infrastructure Needed for the Operation of the Ceramic Immobilization Alternative (Annual)
Table 4.3.4.2.3-1	Estimated Operational Concentrations of Pollutants and Comparison With Most Stringent Regulations or Guidelines—Ceramic Immobilization Alternative and No Action Alternative
Table 4.3.4.2.4–1	Potential Changes to Water Resources Resulting From the Ceramic Immobilization Alternative
Table 4.3.4.2.9–1	Potential Radiological Impacts to the Public During Normal Operation of the Ceramic Immobilization Alternative
Table 4.3.4.2.9–2	Potential Radiological Impacts to Workers During Normal Operation of the Ceramic Immobilization Alternative
Table 4.3.4.2.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation of the Ceramic Immobilization Alternative
Table 4.3.4.2.9–4	Ceramic Immobilization Alternative Accident Impacts at Hanford Site
Table 4.3.4.2.9–5	Ceramic Immobilization Alternative Accident Impacts at Nevada Test Site 4-644
Table 4.3.4.2.9-6	Ceramic Immobilization Alternative Accident Impacts at Idaho National Engineering Laboratory
Table 4.3.4.2.9–7	Ceramic Immobilization Alternative Accident Impacts at Pantex Plant
Table 4.3.4.2.9-8	Ceramic Immobilization Alternative Accident Impacts at Oak Ridge Reservation
Table 4.3.4.2.9–9	Ceramic Immobilization Alternative Accident Impacts at Savannah River Site 4-652
Table 4.3.4.2.10–1	Estimated Annual Generated Waste Volumes for the Ceramic Immobilization Alternative
Table 4.3.4.3.21	Additional Site Infrastructure Needed for the Operation of the Electrometallurgical Treatment Alternative at Idaho National Engineering Laboratory (Annual)
Table 4.3.4.3.3-1	Estimated Operational Concentrations of Pollutants and Comparison With Most Stringent Regulations or Guidelines—Electrometallurgical Treatment Alternative and No Action Alternative
Table 4.3.4.3.9-1	Potential Radiological Impacts to the Public During Normal Operation of the Electrometallurgical Treatment Alternative at Idaho National Engineering Laboratory
Table 4.3.4.3.9–2	Potential Radiological Impacts to Workers During Normal Operation of the Electrometallurgical Treatment at Idaho National Engineering Laboratory
Table 4.3.4.3.9–3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation of the Electrometallurgical Treatment Alternative at Idaho National Engineering Laboratory
Table 4.3.4.3.10-1	Estimated Annual Generated Waste Volumes for the Electrometallurgical Treatment Alternative

Storage and Dispos Fissile Materials F	sition of Weapons-Usable inal PEIS
Table 4.3.5.1.2–1	Additional Site Infrastructure Needed for the Construction of the Mixed Oxide Fuel Fabrication Facility (Annual)
Table 4.3.5.1.2–2	Additional Site Infrastructure Needed for the Operation of the Mixed Oxide Fuel Fabrication Facility (Annual)4–677
Table 4.3.5.1.4-1	Potential Changes to Water Resources Resulting From the Mixed Oxide Fuel Fabrication Facility
Table 4.3.5.1.9-1	Potential Radiological Impacts to the Public During Normal Operation of the Mixed Oxide Fuel Fabrication Facility4–705
Table 4.3.5.1.9-2 .	Potential Radiological Impacts to Workers During Normal Operation of the Mixed Oxide Fuel Fabrication Facility
Table 4.3.5.1.9-3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation of the Mixed Oxide Fuel Fabrication Facility
Table 4.3.5.1.9-4	Mixed Oxide Fuel Fabrication Facility Accident Impacts at Hanford Site
Table 4.3.5.1.9–5	Mixed Oxide Fuel Fabrication Facility Accident Impacts at Nevada Test Site 4–712
Table 4.3.5.1.9-6	Mixed Oxide Fuel Fabrication Facility Accident Impacts at Idaho National Engineering Laboratory4–713
Table 4.3.5.1.9-7	Mixed Oxide Fuel Fabrication Facility Accident Impacts at Pantex Plant
Table 4.3.5.1.9-8	Mixed Oxide Fuel Fabrication Facility Accident Impacts at Oak Ridge Reservation
Table 4.3.5.1.9–9	Mixed Oxide Fuel Fabrication Facility Accident Impacts at Savannah River Site
Table 4.3.5.1.10-1	Estimated Annual Generated Waste Volumes for the Mixed Oxide Fuel Fabrication Facility
Table 4.3.5.2.2–1	Additional Site Infrastructure Needed for the Operation of One Existing Light Water Reactor (Annual)4–721
Table 4.3.5.2.9-1	Potential Radiological Impacts to the Public During Normal Operation of the Existing Light Water Reactor
Table 4.3.5.2.9–2	Potential Radiological Impacts to Workers During Normal Operation of the Existing Light Water Reactor
Table 4.3.5.2.9–3	Accident Impacts for Existing Light Water Reactor With Mixed Oxide Fuels 4-732
Table 4.3.5.2.9-4	Ratio of Accident Impacts for Mixed Oxide Fueled and Uranium Fueled Reactors for Typical Severe Accidents (Mixed Oxide Impacts/Uranium Impacts)
Table 4.3.5.3.2–1	Additional Site Infrastructure Needed for the Operation of the Partially Completed Light Water Reactor (Annual)
Table 4.3.5.3.3–1	Estimated Incremental Operational Concentrations of Pollutants and Comparison With Most Stringent Regulations or Guidelines—Partially Completed Reactor
Table 4.3.5.3.9–1	Potential Radiological Impacts to the Public During Normal Operation of the Partially Completed Light Water Reactor
Table 4.3.5.3.9-2	Potential Radiological Impacts to Workers During Normal Operation of the Partially Completed Light Water Reactor

پو احد -

MA .

Table 4.3.5.3.10–1	Estimated Annual Waste Volumes Generated Per Reactor for Mixed Oxide Fuel in Partially Completed Light Water Reactors
Table 4.3.5.4.2–1	Additional Site Infrastructure Needed for the Construction of the Large or Small Evolutionary Light Water Reactor (Annual)
Table 4.3.5.4.2–2	Additional Site Infrastructure Needed for the Operation of the Large Evolutionary Light Water Reactor (Annual)
Table 4.3.5.4.2-3	Additional Site Infrastructure Needed for the Operation of the Small Evolutionary Light Water Reactor (Annual)
Table 4.3.5.4.3–1	Estimated Operational Concentrations of Pollutants and Comparison With Most Stringent Regulations or Guidelines—Evolutionary Light Water Reactor and No Action Alternative
Table 4.3.5.4.4–1	Potential Changes to Water Resources Resulting From the Large Evolutionary Light Water Reactor
Table 4.3.5.4.4-2	Potential Changes to Water Resources Resulting From the Small Evolutionary Light Water Reactor
Table 4.3.5.4.9–1	Potential Radiological Impacts to the Public During Normal Operation of the Large Evolutionary Light Water Reactor
Table 4.3.5.4.9-2	Potential Radiological Impacts to the Public During Normal Operation of the Small Evolutionary Light Water Reactor
Table 4.3.5.4.9-3	Potential Radiological Impacts to Workers During Normal Operation of the Large Evolutionary Light Water Reactor
Table 4.3.5.4.9-4	Potential Radiological Impacts to Workers During Normal Operation of the Small Evolutionary Light Water Reactor
Table 4.3.5.4.9–5	Potential Hazardous Chemical Impacts to the Public and Workers Resulting From Normal Operation of the Large or Small Evolutionary Light Water Reactor
Table 4.3.5.4.9–6	Evolutionary Light Water Reactor Accident Impacts at Hanford Site
Table 4.3.5.4.9-7	Evolutionary Light Water Reactor Accident Impacts at Nevada Test Site
Table 4.3.5.4.9–8	Evolutionary Light Water Reactor Accident Impacts at Idaho National Engineering Laboratory
Table 4.3.5.4.9-9	Evolutionary Light Water Reactor Accident Impacts at Pantex Plant
Table 4.3.5.4.9-10	Evolutionary Light Water Reactor Accident Impacts at Oak Ridge Reservation
Table 4.3.5.4.9-11	Evolutionary Light Water Reactor Accident Impacts at Savannah River Site 4–804
Table 4.3.5.4.10-1	Estimated Annual Generated Spent Nuclear Fuel and Waste Volumes for the Large Evolutionary Light Water Reactor
Table 4.3.5.4.10-2	Estimated Annual Generated Spent Nuclear Fuel and Waste Volumes for the Small Evolutionary Light Water Reactor
Table 4.4.2.2-1	Transportation Modes and Comparison Ratings by Site
Table 4.4.2.3–1	Transportation Summary of Radioactive Materials and Packagings for Alternatives

, Alter A

Storage and Dispo Fissile Materials I	sition of Weapons-Usable Final PEIS
Table 4.4.3.2-1	Total Potential Fatalities From Intersite Transportation Activities for the Preferred Alternative for Storage
Table 4.4.3.2-2	Total Potential Fatalities From the Transportation of Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium for the Upgrade Alternative
Table 4.4.3.2–3	Total Potential Fatalities From the Transportation of Plutonium for the Consolidation Alternative
Table 4.4.3.2-4	Total Potential Fatalities From the Transportation of Plutonium and Highly Enriched Uranium for the Collocation Alternative
Table 4.4.3.3-1	Total Potential Fatalities From the Transportation of Plutonium From Existing Storage Sites to a Pit Disassembly/Conversion or Plutonium Conversion Site
Table 4.4.3.3-2	Total Potential Fatalities From the Transportation of Plutonium and Immobilized Materials for the Deep Borehole Category of Alternatives4–825
Table 4.4.3.3–3	Total Potential Fatalities From the Transportation of Plutonium and Immobilized Materials for the Immobilization Category of Alternatives
Table 4.4.3.3-4	Total Potential Fatalities From the Transportation of Plutonium Oxide, Uranium Oxide, and Mixed Oxide Fuel for the Reactor Category of Alternatives
Table 4.4.3.3–5	Highest Number of Potential Fatalities From the Transportation of Materials for Each Disposition Alternative
Table 4.5.1–1	Selected Demographic Characteristics for the Hanford Site Region of Influence4-834
Table 4.5.1-2	Selected Demographic Characteristics for the Nevada Test Site Region of Influence
Table 4.5.1-3	Selected Demographic Characteristics for the Idaho National Engineering Laboratory Region of Influence
Table 4.5.1-4	Selected Demographic Characteristics for the Pantex Plant Region of Influence
Table 4.5.1–5	Selected Demographic Characteristics for the Oak Ridge Reservation Region of Influence
Table 4.5.1-6	Selected Demographic Characteristics for the Savannah River Site Region of Influence
Table 4.5.1–7	Selected Demographic Characteristics for the Rocky Flats Environmental Technology Site Region of Influence
Table 4.5.1-8	Selected Demographic Characteristics for the Los Alamos National Laboratory Region of Influence
Table 4.6.1–1	Storage and Disposition Actions at Department of Energy Sites Proposed by the Preferred Alternative
Table 4.6.1–2	Incremental Impact Indicators Over No Action From the Annual Operation of the Storage Actions Under the Preferred Alternative
Table 4.6.1–3	Land-Use Requirements From the Preferred Alternative

xxiv

List of Tables

Table 4.6.1-4	Site Infrastructure Requirements From the Preferred Alternative	4–865
Table 4.6.1–5	Estimated Operational Concentrations of Pollutants From the Preferred Alternative, Including No Action	4–866
Table 4.6.1–6	Potential Changes to Water Resources Resulting From the Preferred Alternative	4–870
Table 4.6.1–7	Changes to Economic and Demographic Indicators for the Preferred Alternative (Full Operation)	4–877
Table 4.6.1-8	Potential Human Health Impacts to the Public and Workers Under Normal Operation and Potential Accidents for the Preferred Alternative at Hanford Site	4–881
Table 4.6.1–9	Potential Human Health Impacts to the Public and Workers Under Normal Operation and Potential Accidents for the Preferred Alternative at Idaho National Engineering Laboratory	4–882
Table 4.6.1–10	Potential Human Health Impacts to the Public and Workers Under Normal Operation and Potential Accidents for the Preferred Alternative at Pantex Plant	4-883
Table 4.6.1–11	Potential Human Health Impacts to the Public and Workers Under Normal Operation and Potential Accidents for the Preferred Alternative at Savannah River Site	4–884
Table 4.6.1–12	Estimated Annual Generated Waste Volumes for the Preferred Alternative	4–887
Table 4.6.1–13	Total Potential Fatalities From the Transportation of Materials for the Preferred Alternative	4-893
Table 4.6.2–1	Maximum Incremental Direct Employment Over No Action Generated During Operation at Each Candidate Site	4–894
Table 4.6.2–2	Maximum Annual Net Incremental Water Usage Over No Action During Operation at Each Candidate Site	4–894
Table 4.6.2–3	Maximum Annual Net Incremental Volume of Solid Low-Level Waste Generated Over No Action During Operation at Each Candidate Site	4895
Table 4.6.2–4	Maximum Annual Net Incremental Volume of Solid Transuranic Waste Generated Over No Action During Operation at Each Candidate Site	4–895
Table 4.6.2–5	Maximum Annual Net Incremental Volume of Solid Hazardous Waste Generated Over No Action During Operation at Each Candidate Site	4–895
Table 4.6.2–6	Maximum Latent Cancer Fatalities Over No Action for Maximally Exposed Individual for 50 Years From Normal Operation	4–896
Table 4.6.3–1	Incremental Increase During Operation for Activities Common to Disposition Alternatives	4–900
Table 4.6.3–2	Incremental Net Increase During Operation by Disposition Alternative	4-901
Table 4.7.1–1	Reasonably Foreseeable Future Programs at Department of Energy Sites	4–910
Table 4.7.2.1.2-1	Site Infrastructure Cumulative Operation Impacts at Hanford Site	4-911
Table 4.7.2.1.3–1	Estimated Cumulative Operational Concentrations of Pollutants at Hanford Site and Comparison With Most Stringent Regulations or Guidelines—No Action or Storage Alternatives	4-912

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

L'issue maienais i a	A CALL AND AN AND A AND	
Table 4.7.2.1.4–1	Cumulative Annual Water Usage at Hanford Site	914
Table 4.7.2.1.4-2	Cumulative Annual Wastewater Discharge at Hanford Site	914
Table 4.7.2.1.8–1	Socioeconomic Cumulative Impacts at Hanford Site	915
Table 4.7.2.1.9–1	Estimated Average Annual Cumulative Radiological Doses and Resulting Health Effects to the Public and Workers From Normal Operation at Hanford Site4-	916 [°]
Table 4.7.2.1.10–1	Waste Management Cumulative Impacts at Hanford Site (2005)—Annual Volumes4–	917
Table 4.7.2.2.2–1	Site Infrastructure Cumulative Operation Impacts at Nevada Test Site4-	920
Table 4.7.2.2.4-1	Cumulative Annual Water Usage at Nevada Test Site4-	921
Table 4.7.2.2.4-2	Cumulative Annual Wastewater Discharge at Nevada Test Site4-	921
Table 4.7.2.2.3–1	Estimated Cumulative Operational Concentrations of Pollutants at Nevada Test Site and Comparison With Most Stringent Regulations or Guidelines—No Action and Storage Alternatives4–	922
Table 4.7.2.2.8-1	Socioeconomic Cumulative Impacts at Nevada Test Site4-	923
Table 4.7.2.2.9-1	Estimated Average Annual Cumulative Radiological Doses and Resulting Health Effects to the Public and Workers From Normal Operation at Nevada Test Site4-	-924 _.
Table 4.7.2.2.10-1	Waste Management Cumulative Impacts at Nevada Test Site (2005)—Annual Volumes4-	-925
Table 4.7.2.3.2-1	Site Infrastructure Cumulative Impacts at Idaho National Engineering Laboratory4-	-927
Table 4.7.2.3.3-1	Estimated Cumulative Operational Concentrations of Pollutants at Idaho National Engineering Laboratory and Comparison With Most Stringent Regulations or Guidelines—No Action and Storage Alternatives	-928
Table 4.7.2.3.4-1	Cumulative Annual Water Usage at Idaho National Engineering Laboratory4-	-930
Table 4.7.2.3.4-2	Cumulative Annual Wastewater Discharge at Idaho National Engineering Laboratory4-	-930
Table 4.7.2.3.8-1	Socioeconomic Cumulative Impacts at Idaho National Engineering Laboratory 4-	-931
Table 4.7.2.3.9–1	Estimated Average Annual Cumulative Radiological Doses and Resulting Health Effects to the Public and Workers From Normal Operation at Idaho National Engineering Laboratory4-	-932
Table 4.7.2.3.10-1	Waste Management Cumulative Impacts at Idaho National Engineering Laboratory (2005)—Annual Volumes4-	-93 ['] 3
Table 4.7.2.4.2-1	Site Infrastructure Cumulative Operation Impacts at Pantex Plant4-	-935
Table 4.7.2.4.3–1	Estimated Cumulative Operational Concentrations of Pollutants at Pantex Plant and Comparison With Most Stringent Regulations or Guidelines—No Action and Storage Alternatives4	-936
Table 4.7.2.4.4–1	Cumulative Annual Water Usage at Pantex Plant4	-939
Table 4.7.2.4.4–2	Cumulative Annual Wastewater Discharge at Pantex Plant	-939
Table 4.7.2.4.8-1	Socioeconomic Cumulative Impacts at Pantex Plant4	-939

i

Table 4.7.2.4.9–1	Estimated Average Annual Cumulative Radiological Doses and Resulting Health Effects to the Public and Workers From Normal Operation at Pantex Plant
Table 4.7.2.4.10-1	Waste Management Cumulative Impacts at Pantex Plant (2005)—Annual Volumes4–941
Table 4.7.2.5.2-1	Site Infrastructure Cumulative Operation Impacts at Oak Ridge Reservation 4–943
Table 4.7.2.5.3–1	Estimated Cumulative Operational Concentrations of Pollutants at Oak Ridge Reservation and Comparison With Most Stringent Regulations or Guidelines—No Action and Storage Alternatives
Table 4.7.2.5.4–1	Cumulative Annual Water Usage at Oak Ridge Reservation
Table 4.7.2.5.4-2	Cumulative Annual Wastewater Discharge at Oak Ridge Reservation
Table 4.7.2.5.8–1	Socioeconomic Cumulative Impacts at Oak Ridge Reservation
Table 4.7.2.5.9–1	Estimated Average Annual Cumulative Radiological Doses and Resulting Health Effects to the Public and Workers From Normal Operation at Oak Ridge Reservation
Table 4.7.2.5.10–1	Waste Management Cumulative Impacts at Oak Ridge Reservation (2005)—Annual Volumes
Table 4.7.2.6.2–1	Site Infrastructure Cumulative Operation Impacts at Savannah River Site
Table 4.7.2.6.3–1	Estimated Cumulative Operational Concentrations of Pollutants at Savannah River Site and Comparison With Most Stringent Regulations or Guidelines—No Action and Storage Alternatives
Table 4.7.2.6.4–1	Cumulative Annual Water Usage at Savannah River Site
Table 4.7.2.6.4-2	Cumulative Annual Wastewater Discharge at Savannah River Site
Table 4.7.2.6.8–1	Socioeconomic Cumulative Impacts at Savannah River Site 4–957
Table 4.7.2.6.9–1	Estimated Average Annual Cumulative Radiological Doses and Resulting Health Effects to the Public and Workers From Normal Operation at Savannah River Site4–958
Table 4.7.2.6.10-1	Waste Management Cumulative Impacts at Savannah River Site (2005)—Annual Volumes
Table 4.7.2.7.8-1	Socioeconomic Cumulative Impacts at Rocky Flats Environmental Technology Site
Table 4.7.3.1-1	Contribution to Land-Use Cumulative Impacts From the Disposition Scenario
Table 4.7.3.2-1	Contribution to Site Infrastructure Cumulative Impacts From the Disposition Scenario
Table 4.7.3.4–1	Contribution to Water Resource Cumulative Impacts From the Disposition Scenario
Table 4.7.3.8–1	Contribution to Socioeconomic Cumulative Impacts From the Disposition Scenario
Table 4.7.3.9-1	Contribution to Public and Occupational Health and Safety Cumulative Impacts From the Disposition Scenario4–968

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

11.

Table 4.7.3.10–1	Contribution to Waste Management Cumulative Impacts From the Disposition Scenario
Table 4.9.1-1	Comparison of Uranium Fuel and Mixed Oxide Fuel Cycles
Table 4.9.1.1-1	Comparison of Radionuclide Atmosphere Emissions
Table 4.9.1.1–2	Comparison of Potential Radiological Human Health Impacts to the General Public
Table 4.9.1.1–3	Comparison of Potential Radiological Human Health Impacts to Workers4-978
Table 4.9.1.2-1	Comparison of Potential Emission Rates of Criteria Pollutants
Table 4.9.2–1	Comparison of Potential Emission Rates of Criteria Pollutants Between the Mixed Oxide Fuel Cycle Using Partially Completed Light Water Reactors and Conventional Power Plants
Table 4.9.2–2	Comparison of Potential Emission Rates of Criteria Pollutants Between the Mixed Oxide Fuel Cycle Using Evolutionary Light Water Reactors and Conventional Power Plants
Table 10-1	Representatives From Affected Areas by State (Colorado, Georgia, Idaho, Nevada, New Mexico, South Carolina)10-4
Table 10-2	Representatives From Affected Areas by State (Tennessee, Texas, Washington)10–5
Table 10-3	Individuals Who Provided Comments on the Draft Programmatic Environmental Impact Statement or Have Requested Copies

. <u>.</u> . .

LIST OF ACRONYMS AND ABBREVIATIONS

AADT	Average Annual Daily Traffic
ACEC	Area of Critical Environmental Concern
ACGIH	American Conference of Governmental Industrial Hygienists
AEA	Atomic Energy Act
AEC	Atomic Energy Commission
AGV	automated guided vehicle
ALARA	as low as reasonably achievable
ALE	Arid Lands Ecology Reserve
ANL-W	Argonne National Laboratory-West
APSF	Actinide Packaging and Storage Facility
AQCR	Air Quality Control Region
ARA	Auxiliary Reactor Area
ARIES	Advanced Recovery and Integrated Extraction System
BEA	Bureau of Economic Analysis
BEIR	biological effects of ionizing radiation
BLM	Bureau of Land Management
BOP	balance-of-plant
BPA	Bonneville Power Administration
BWR	boiling water reactor
CAA	Clean Air Act
CANDU	Canadian deuterium uranium
CAS	Chemical Abstracts Service
CCDF	complimentary cumulative distribution function
CEQ	Council on Environmental Quality
CERCLA	Comprehensive Environmental Response, Compensation, and Liability Act
CFA	Central Facilities Area
CFR	Code of Federal Regulations
CGTO	Consolidated Group of Tribes and Organizations
CI	confidence interval
CIC	can-in-canister
CLUP	Comprehensive Land-Use Plan
CMR	Chemistry and Metallurgy Research

South States in the second

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

ι. Έ. Έ.

COE	Corps of Engineers	5.7 15	
Complex	Nuclear Weapons Complex	···· · · ·	
CRD	Comment Response Document	,	
CRT	Cargo Restraint Transporters	ţ	
CWA	Clean Water Act		
D&D	decontamination and decommissioning	યું અ કે અંગ જ	
DAF	Device Assembly Facility	-	
DCG	derived concentration guide	· · · · ·	
DHLW	defense high-level waste		
DNB	departure of nucleate boiling		
DNFSB	Defense Nuclear Facilities Safety Board		
DNL	day and night average sound levels	,	•
DNWR	Desert National Wildlife Range		,
DoD	Department of Defense		,
DOE	Department of Energy	· * ~	
DOT	Department of Transportation	, -	
DP	Office of Defense Programs		
DRCOG	Denver Regional Council of Governments		
DWPF	Defense Waste Processing Facility		*
EA	environmental assessment	r 	
EBR	Experimental Breeder Reactor		v
EDNA	Environmental Design for Noise Abatement	-	
EIA	Energy Information Administration		•
EIS	environmental impact statement		
EM	Office of Environmental Management		
EPA	Environmental Protection Agency	~~ 4	
ERR	excess relative risk		
ES&H	Office of Environment, Safety, and Health	, [*] - ,	
ESA	Endangered Species Act		,
ETF	effluent treatment facility	,	
FAIR	Forest, Agriculture, Industry, and Research	· · · · · ·	,
FCF	Fuel Cycle Facility	• • • • • •	
FEMA	Federal Emergency Management Agency	**	ĩ
FFCA	Federal Facility Compliance Agreement	· · · · · · · · ·	,
FFTF	Fast Flux Test Facility		

XXX
1 _

FLPMA	Federal Land Planning Management Act
FMEF	Fuels and Materials Examination Facility
FMF	Fuel Manufacturing Facility
FONSI	Finding of No Significant Impact
FR	Federal Register
FSAR	Final Safety Analysis Report
GBZ	Glass-bonded zeolite
GESMO	Generic Environmental Statement on Mixed Oxide
GIS	Geographical Information System
GMA	Growth Management Act
GMODS	Glass Material Oxidation Dissolution System
HAD	hazard analysis document
Hanford	Hanford Site
HE	high explosives
HEAST	Health Effects Summary Table
HEPA	high-efficiency particulate air
HEU	highly enriched uranium
HEU EIS	Disposition of Surplus Highly Enriched Uranium Environmental Impact Statement
HFEF	Hot Fuel Examination Facility
HI	Hazard Index
HLW	high-level waste
HQ	Hazard Quotient
HRA EIS	Hanford Remedial Action Environmental Impact Statement and Comprehensive Land Use Plan
HVAC	Heating Ventilation and Air Conditioning
HWR	Heavy Water Reactor
IAEA	International Atomic Energy Agency
ICPP	Idaho Chemical Processing Plant
ICRP	International Commission of Radiological Protection
INEL	Idaho National Engineering Laboratory
IRIS	Integrated Risk Information System
ISCST2	Industrial Source Complex Short-Term Model Version 2
ISO	International Standards Organization
IWG	Interagency Working Group
K-25	K-25 Site

Storage and Dis Fissile Materials	position of Weapons-Usable Final PEIS			·
L/ER	Energy Research Program Office	1. 5 5 1. 5 5 5	х, к. с	
LA	Limited Area		e f Is en	,
LAA	Limited Access Area			
LANL	Los Alamos National Laboratory	•	,	
LANSCE	Los Alamos Neutron Scattering Center		те ,	
LCF	latent cancer fatalities		£	,
LDR	Land Disposal Restriction	,	- '	
LEU	low-enriched uranium	ʻ.	· -	
LIGO	Laser Interferometer Gravitational-Wave O	bservatory		
LLNL	Lawrence Livermore National Laboratory			
LLW	low-level waste		. ,	,
LOB	Laboratory Office Building	· .		
LWR	Light Water Reactor	·		
MAA	Material Access Area			
MACCS	Melcor Accident Consequence Code System	m		
MC&A	Material Control and Accountability		e e e e e e e e e e e e e e e e e e e	
MD	Office of Fissile Materials Disposition			~
MEI	maximally exposed individual		,	
MHR	Modular Helium Reactor			-
MMI	Modified Mercalli Intensity			
MOX	mixed oxide			,
MSL	mean sea level			
NAAQS	National Ambient Air Quality Standards		t	
NAGPRA	Native American Graves Protection and Re	epatriation Act	r t	
NAS	National Academy of Sciences	t -	۰. <u>,</u>	
NCDC	National Climatic Data Center	٠, ٨		
NCRP	National Commission of Radiological Prote	ection ,		
NEIC	National Earthquake Information Center	, - <i>*</i>		
NEPA	National Environmental Policy Act	· · · · ·		
NERP	National Environmental Research Park			
NESHAP	National Emission Standards for Hazardou	s Air Pollutants 👘		
NFS	Nuclear Fuel Services Fuel Fabrication Pla	nt	A y	
NHPA	National Historic Preservation Act	- *s.,	• .	
NIOSH	National Institute of Occupational Safety as	nd Health		
NMSF	Nuclear Material Storage Facility			

xxxii

List of Acronyms and Abbreviations

NMSM	Nuclear Materials and Stockpile Management
NOAA	National Oceanic and Atmospheric Administration
NOI	Notice of Intent
NPDES	National Pollutant Discharge Elimination System
NPL	National Priorities List
NRC	Nuclear Regulatory Commission
NRF	Naval Reactors Facility
NRHP	National Register of Historic Places
NTS	Nevada Test Site
NTS EIS	Environmental Impact Statement for the Nevada Test Site and Off-Site Locations in the State of Nevada
NWI	National Wetlands Inventory
NWPA	Nuclear Waste Policy Act
NWS	National Weather Service
OCRWM	Office of Civilian Radioactive Waste Management
ORISE	Oak Ridge Institute for Science and Education
ORNL	Oak Ridge National Laboratory
ORR	Oak Ridge Reservation
OSHA	Occupational Safety and Health Administration
PA	Protected Area
Pantex	Pantex Plant
Pantex EIS	Environmental Impact Statement for the Continued Operation of the Pantex Plant and Associated Storage of Nuclear Weapon Components
PBF	Power Burst Facility
PCV	Primary Containment Vessel
PEIS	programmatic environmental impact statement
PEL	Permissible Exposure Level
PFP	Plutonium Finishing Plant
PFP EIS	Plutonium Finishing Plant Stabilization Environmental Impact Statement
PIDAS	Perimeter Intrusion Detection and Alarm System
PNNL	Pacific Northwest National Laboratory
PPA	Property Protection Area
PRA	probabilistic risk assessment
PSAR	Preliminary Safety Analysis Report
PSD	Prevention of Significant Deterioration

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

PUREX	Plutonium-Uranium Extraction Plant	\$ s.			
PWR	pressurized water reactor	, '			
R&D	Research and Development			,	
RCRA	Resource Conservation and Recovery Act				
REA	regional economic area		1		
RIA	reactivity insertion accident				
RFETS	Rocky Flats Environmental Technology Site				
	Regional Input-Output Modeling System				
RL	Richland Operations Office				,
ROD	Record of Decision				-
ROD	region of influence				
DOWE	Padioactive Scrap and Waste Facility				
NSWI ^C	Radioactive Scrap and waste Facility				\$
RWMC	Radioactive Waste Management Site				
RWWS SAD	Safaty Analysis Deport				
SAR	Superfund Americand Production Act				
SAKA	Superjuna Amenamenis and Keaumonzanon Act				
SU	Standard deviation				
SDWA	Saje Drinking water Act				
SEB	Security Equipment Building				
SHPO	State Historic Preservation Officer				
SIP	State Implementation Plan		1		
SISMP	Site Integrated Stabilization and Management Plan				
SMR	Standardized Mortality Ratio				
SNF	spent nuclear fuel				
SNL	Sandia National Laboratories	v	*	,	
SRR	standardize rate ratio		· .1		' `
SRS	Savannah River Site	_ 11	٢	,	: •
Stockpile Stewardship and Management PEIS	Programmatic Environmental Impact Statement for S	Stockpile	Stewards	'iip and Ma	nagement
Storage and Disposition PEIS	Storage and Disposition of Weapons-Usable Fissile Impact Statement	Materials	rogram	matic Envi	ronmental
	· · ·	~	~		
SST	safe secure trailer				
START	Strategic Arms Reduction Talks				
xxxiv					

List of Acronyms and Abbreviations

TA	Technical Area
TAN	Test Area North
TCLP	toxicity characteristic leaching procedure
TDEC	Tennessee Department of Environmental Conservation
TDS	total dissolved solids
TI	transport index
TLV	Threshold Limit Values
TNRCC	Texas Natural Resources Conservation Commission
TRA	Test Reactor Area
TRU	transuranic
TSCA	Toxic Substance Control Act
TSD	Transportation Safeguards Division
TSP	total suspended particulates
TSR PEIS	Tritium Supply and Recycling Programmatic Environmental Impact Statement
TVA	Tennessee Valley Authority
USFWS	United States Fish and Wildlife Services
USGS	United States Geological Survey
VOC	volatile organic compound
VRM	Visual Resource Management
WAC	Waste Acceptance Criteria
Waste Management PEIS	Waste Management Programmatic Environmental Impact Statement for Managing Treatment, Storage, and Disposal of Radioactive and Hazardous Waste
WIPP	Waste Isolation Pilot Plant
WMIS	Waste Management Information System
WNP	Washington Nuclear Power
WPPSS	Washington Public Power Supply System
WSA	Weapons Storage Area
WSCC	Western Systems Coordinating Council
WSCF	Waste Sampling and Characterization Facility
Y-12	Y-12 Plant
Y-12 EA	Environmental Assessment for the Proposed Interim Storage of Enriched Uranium Above the Maximum Historical Level at the Y–12 Plant, Oak Ridge, Tennessee
YMSCO	Yucca Mountain Site Characterization Office
ZPPR	Zero Power Physics Reactor

. .

1.

CHEMICALS AND UNITS OF MEASURE

		1. N	
°C	degrees Celsius		
Ci	curie		
cm	centimeter		1
CO	carbon monoxide	1. 1	
CO ₂	carbon dioxide		
Co-60	cobalt-60		
Cs	cesium		
Cs-137	cesium-137		
CsCl	cesium chloride	,	-
Cu	copper		
dB	decibel	ı.	Ē. ¹ .
dBA	decibel A-weighted		ſ '
°F	degrees Fahrenheit 💴		,
ft	feet	1	- *
ft ²	square feet		
ft ³	cubic feet		
g	gram		
G	gravitational acceleration		
gal	gallon		
Gd	gadolinium		
GWd	gigawatt-days		
ha	hectare		
H ₂	hydrogen	۰ ۲	
HF	hydrogen fluoride	,1 i i	
HNO ₃	nitric acid		
hr	hour	ł , ,	
I-129	iodine-129		
in	inch		
k _{eff}	effective neutron multiplication factor		
kg	kilogram	1	
km	kilometer		<i>د</i> غ
km ²	square kilometer		٤

List of Acronyms and Abbreviations

	1
Kr	krypton
kV	kilovolt
I	liter
lb	pound
m	meter
m ²	square meter
m ³	cubic meter
mCi	millicurie
mg	milligram
mi	mile
mi ²	square miles
min	minute
mph	miles per hour
mrem	millirem (one thousandth of a rem)
MW	megawatt
MWe	megawatt electric
N ₂	nitrogen
nCi	nanocurie (one-billionth of a Curie)
Ni	nickel
NO ₂	nitrogen dioxide
NO _x	nitrogen oxides
O ₃	ozone
oz	ounce
Рb	lead
PCB	polychlorinated biphenyl
pCi	picocurie (one-trillionth of a Curie)
- PM ₁₀	particulate matter less than or equal to 10 microns
ppm	parts per million
Pu	plutonium
PuCl	plutonium chloride
PuO ₂	plutonium dioxide
rad	radiation absorbed dose
rem	roentgen equivalent man
RfC	Reference Concentration

ţ

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

S	second
SO ₂	sulfur dioxide
Sr-90	strontium-90
t	metric ton
Tc-99	technetium-99
ton	short ton
U .	uranium
U-233	uranium-233
U-234	uranium-234
U-235	uranium-235
U-236	uranium-236
U-238	uranium-238
UF ₆	uranium hexafluoride
UNH	uranyl nitrate hexahydrate
UO ₂	uranium dioxide
U ₃ O ₈	triuranium octaoxide
VOC	volatile organic compound
yd	yard
yr	year
μg	microgram (one-millionth of a gram)

J ...

Metric Conversion Chart

To Convert Into Metric			To Convert Out of Metric		
If You Know	Multiply By	To Get	If You Know	Multiply By	To Get
Length					
inches	2.54	centimeters	centimeters	0.3937	inches
feet	30.48	centimeters	centimeters	0.0328	feet
feet	0.3048	meters	meters	3.281	feet
yards	0.9144	meters	meters	1.0936	yards
miles	1.60934	kilometers	kilometers	0.6214	miles
Area					
sq. inches	6.4516	sq. centimeters	sq. centimeters	0.155	sq. inches
sq feet	0.092903	sq. meters	sq. meters	10.7639	sq. feet
sq. yards	0.8361	sq. meters	sq. meters	1.196	sq. yards
acres	0.40469	hectares	hectares	2.471	acres
sq miles	2.58999	sq. kilometers	sq kilometers	0.3861	sq. miles
Volume					
fluid ounces	29.574	milliliters	milliliters	0.0338	fluid ounces
gallons	3.7854	liters	liters	0.26417	gallons
cubic feet	0.028317	cubic meters	cubic meters	35.315	cubic feet
cubic yards	0.76455	cubic meters	cubic meters	1.308	cubic yards
Weight					
ounces	28.3495	grams	grams	0.03527	ounces
pounds	0.45360	kılograms	kılograms	2.2046	pounds
short tons	0.90718	metric tons	metric tons	1.1023	short tons
Temperature					
Fahrenheit	Subtract 32 then multiply by 5/9ths	Celsius	Celsius	Multiply by 9/5ths, then add 32	Fahrenheit

METRIC CONVERSION CHART

METRIC PREFIXES

Prefix	Symbol	Multiplication Factor
exa-	Е	$1\ 000\ 000\ 000\ 000\ 000\ 000\ = 10^{18}$
peta-	Р	$1\ 000\ 000\ 000\ 000\ 000 = 10^{15}$
tera-	Т	$1\ 000\ 000\ 000\ 000 = 10^{12}$
giga-	G	$1\ 000\ 000\ 000 = 10^9$
mega-	М	$1\ 000\ 000 = 10^6$
kilo-	k	$1\ 000 = 10^3$
hecto-	h	$100 = 10^2$
deka-	da	$10 = 10^1$
deci-	d	$0.1 = 10^{-1}$
centi-	с	$0.01 = 10^{-2}$
milli-	m	$0\ 001 = 10^{-3}$
micro-	μ	$0\ 000\ 001 = 10^{-6}$
nano-	n	$0\ 000\ 000\ 001 = 10^{-9}$
pico-	р	$0.000\ 000\ 000\ 001 = 10^{-12}$
femto-	f	$0.000\ 000\ 000\ 000\ 001 = 10^{-15}$
atto-	a	$0.000\ 000\ 000\ 000\ 001 = 10^{-18}$

xxxix

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

1 155110 1110/01/01/01 11/01/1 12:0	-74 15 14	-	22 x 23x	
<u></u>	11.00 %		ng d th	
	۰۵ ت	35 ×		
		•		
	,	. (,	

·

xl

i

Chapter 4 Environmental Consequences

The potential environmental consequences resulting from the proposed weapons-usable fissile materials storage alternatives, the Pu disposition alternatives, and the No Action Alternative are described in this chapter for each of the sites, environmental settings, and resource/issue areas described in Section 3.0. The impact methodology applied for each of these resource/issue areas is described in Section 4.1. In addition, descriptions of methodologies applied and the resulting environmental consequences for intersite transportation and environmental justice are found in Sections 4.4 and 4.5, respectively.

Section 4.2 describes the environmental consequences associated with the No Action Alternative and the longter'm storage alternatives. Table 4–1 identifies the pages within the document where environmental consequences of the alternatives are analyzed for each of the sites. [Text deleted.] Environmental impacts of each long-term storage alternative and the No Action Alternative are analyzed for each of the six candidate storage sites to allow (1) the comparison of impacts by sites for each alternative and (2) the comparison of impacts by alternatives for each site. As a result, decisions can be made to select a single storage alternative for all sites, or a combination of different alternatives for different sites.

Table 4–1.	Key to Locating Information on Environmental Consequences and Mitigation Measures for
	Storage Alternatives by Page Number ^a

	Sites								
Alternatives	Hanford	NTS	INEL	Pantex	ORR	SRS	RFETS	LANL	
No Action Alternative	4-23	4-83	4-131	4-182	4–240	4-291	4-343	4-363	
Preferred Alternative	4-859	4-859	4-859	4-859	4-859	4-859	4-859	4-859	
Upgrade at Multiple Sites Alternative	4-23 .	NA	4–131	4–182	4–240	4–291	NA	NA	
Consolidation of Pu Alternative	4–23	4-83	4-131	4–182	NA ^b	4–291	NA	NA	
Collocation of Pu and HEU Alternative	4–23	4-83	4-131	4–182	4–240	4–291	NA	NA	

^a See also Sections 4 4 (Intersite Transportation) and 4.5 (Environmental Justice).

I

^b Consolidated Pu storage at ORR would create an HEU collocation condition and is analyzed as part of the Collocation Alternative. Note: NA=alternative does not apply at the site.

For the Pu disposition alternatives analyzed in Section 4.3, specific site characteristics, as well as generic representative environmental settings, have been used in this PEIS for analyzing the potential environmental impacts of disposition technology alternatives. Six DOE sites and other generic and specific sites were used for assessing the environmental impacts of various disposition technologies and strategies. The locations of the new facilities considered for the various disposition technologies are representative and for analysis purposes only. Table 4–2 identifies the locations in this document where the disposition alternatives and common activities, and sites and environmental settings, have been analyzed. Since the disposition decision to be made encompasses the programmatic strategy and technology or technology mix that would be employed for disposition of the various forms of surplus weapons-usable Pu materials, generic cumulative environmental impact analysis of locating more than one disposition facility at a single site is described in Sections 4.6 and 4.7. Site-specific analysis would be made in follow-on tiered NEPA documentation.

Although analyzed individually, both pit disassembly/conversion and Pu conversion are front-end processes common to all of the disposition alternatives (except No Action). MOX fuel fabrication is a common activity to all of the reactor alternatives. In addition to a potential new MOX fuel fabrication facility that would be built in the United States, the Existing LWR Alternative initially could include MOX fuel fabrication abroad on an interim basis.

	Analysis Sites and Environmental Settings						
Alternatives and Common Activities	Hanford	NTS	INEL	Pantex -	ORR	SRS	Generic
Preferred Alternative	4-859	NA	4-859	4-859	NA	4-859	NA
Common Activities					-		
Pit Disassembly/Conversion Facility	4-383	4-383	4-383	4–383	4–383	4–383	NA
Pu Conversion Facility	4-427	4–427	4-427	4-427	4-427	4-427	NA
Deep Borehole Category							-
Direct Disposition Alternative	NA	NA	NA	NA	NA	NA	4-474
Immobilized Disposition Alternative						-	-
Ceramic Immobilization Facility (for Borehole)	4–496	4-496	4-496	4-496	4-496	4-496	NA -
Deep Borehole Complex	⁻ NA	NA	NA	NA	NA	NA	4543
Immobilization Category							
Vitrification Alternative	4-561	4-561	4561	4561	4561	4561	NA
Ceramic Immobilization Alternative	4-605	4-605	4-605	4-605	4-605	4-605	NA
Electrometallurgical Treatment Alternative ^a	NA	NA -	4-657 -	NA	NA -	NA	NA
Reactor Category and Common Activities			* 1	e ,	-	· · · · ·	
MOX Fuel Fabrication		** *	19. 19. 19.	. s		<i>.</i>	-
New Domestic Facility	4-673	4-673	4-673	4-673	4-673	4-673	4-673
Existing European Facility	NA	ŇA	NA	NA	NA	NA ~	NA
Existing LWR Alternative	NA -	NA 🖃	NA	NA	NA	NA	4–720
Partially Completed LWR Alternative	NA	NA	[•] NA	NĂ	NA	NA	4 735
Evolutionary LWR Alternative	4-749	4-749	4–749	4-749	4–749	4-749 [•]	NA
CANDU Reactor Alternative ^b	NA	NA	NA	NA	NĂ	NA	4-809

Table 4–2. Key to Locating Information on Environmental Consequences and Mitigation Measures for Disposition Alternatives by Page Number

^a INEL is a representative site for the Electrometallurgical Treatment Alternative. If this alternative is selected, any of the six DOE sites could be used and subsequent NEPA documents would be prepared, as appropriate.
^b Activities in Canada are not fully analyzed in this PEIS but are discussed in Section 4.3.5 5 and Appendix I.

Note: NA=not applicable. 4 Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

For the ocean transportation of fissile materials associated with the European MOX fuel fabrication option, the intersite transportation section (Section 4 4) analyzes potential impacts to the global commons, pursuant to Executive Order 12114, *Environmental Effects Abroad of Major Federal Actions*. Although actions proposed in other countries are not analyzed in this PEIS, changes in CANDU reactor operations due to the use of MOX fuel are described in Appendix I.

As noted in Section 2.4, the final disposal of immobilized forms and domestically produced spent fuel for some disposition alternatives would be accomplished by sending the materials to a geologic repository pursuant to NWPA.¹ Since a repository site has not yet been recommended by the President and approved by Congress, this PEIS does not analyze impacts to a repository. Such a geologic repository, if approved under the provisions of the NWPA, would serve primarily as the disposal site for commercial and DOE-owned spent nuclear fuel and HLW. This impact analysis would be done in a separate EIS for the repository. Operation issues regarding the repository, however, are described in Appendix H.

A summary of the environmental consequences for all the proposed storage and disposition alternatives and the preferred alternative is presented in Section 4.6. Cumulative impacts associated with the alternatives are described in Section 4.7. Sections 4.8 through 4.11 provide descriptions of the following with regard to the alternatives: unavoidable adverse environmental impacts, avoided environmental impacts, relationship between short-term uses of the environment and the maintenance and enhancement of long-term productivity, and irreversible and irretrievable commitment of resources. Section 4.12 provides energy requirements and conservation potential of the various alternatives and mitigation measures.

4.1 METHODS FOR ASSESSING ENVIRONMENTAL IMPACTS

The environmental impact assessment methodologies discussed in this section address the full range of natural and human resource and issue areas pertinent to the sites considered for constructing and operating weaponsusable fissile materials storage and disposition alternatives. These resource/issue areas are land resources, site infrastructure, air quality and noise, water resources, geology and soils, biological resources, cultural and paleontological resources, socioeconomics, public and occupational health and safety, and waste management. A summary of major Federal and State environmental, safety, and health statutes, regulations, and orders applicable to the various resource/issue areas is provided in Appendix J.

As part of the impact assessment process, mitigation measures are identified that could be used to reduce and minimize potential impacts. Because these mitigation measures vary from site to site, some mitigation strategies may need to be addressed in subsequent site-specific environmental documents and/or through agency coordination, as required.

4.1.1 LAND RESOURCES

I

The land resources analysis assesses land use and visual resources for the defined ROI. The ROI for land use includes land within 3.2 km (2 mi) of a given candidate site; the ROI for visual resources includes those lands within the affected viewshed of the proposed action. Land resources impacts are analyzed within the context of NEPA and other related Federal legislation and executive orders. Land-use impacts associated with site-specific alternatives are compared to existing land-use patterns and densities, and to land-use plans, policies, and controls. Visual resource impacts are associated with changes in the existing landscape character that could result from the proposed action. Land resources impacts associated with alternatives for which site-specific locations have not been identified are assessed generically. For example, based on technology descriptions and site infrastructure requirements, likely land resources impacts are characterized for a range of land-use and

¹ Under the NWPA, NRC may determine by rule that highly radioactive material requires permanent isolation as HLW in a geologic repository. Such a determination or legislative clarification may be required in order for the immobilized forms generated by the Immobilization Alternatives to be disposed of in a repository pursuant to NWPA.

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

.

11 34

27.

visual resources conditions that could support the specific technology. The No Action Alternative is based on the most current site development and facility utilization plans.

بر في

• • •

121

10 .

Land Use. The land-use analysis estimates the magnitude and extent of potential impacts to current land-use patterns and densities due to the implementation of the alternative under consideration. The impact analysis identifies the amount of land disturbed during construction and used during operation; the total amount of land required; and potential changes in land use and conflicts with land-use plans, policies, and controls. Potential impacts to special status lands, including prime farmland, are emphasized. Possible conflict with any related legislation or regulation is assessed since major regulatory conflicts could conceivably impede or prevent implementation. Conflicts with State and local land-use plans, policies, and controls could result in potential impacts. Conflicts with site development and facility utilization plans could result in potential impacts.

For storage alternatives, the Storage and Disposition PEIS identifies the land area required for operation as the area occupied by the footprint of each building and nonbuilding support areas. In conjunction with paved road, parking areas, graveled areas, and any land graded and cleared of vegetation, they constitute the total land area requirements. Of the land required during operation, the "disturbed area" is the land that was not in use and was previously undisturbed natural habitat. Previously disturbed areas, include graveled areas and any land previously graded and cleared of vegetation. All areas within the existing perimeter intrusion detection boundaries are classified as already disturbed areas. The construction laydown area accommodates material and equipment storage, and provides limited onsite fabrication areas and temporary construction buildings to support construction activities. The criteria for determining disturbed and undisturbed land area during construction is the same as for operation. Therefore, facilities or construction laydown areas proposed to be situated entirely on previously disturbed land area would not create any newly disturbed area.

Visual Resources. The visual resources analysis identifies and evaluates the magnitude and extent of potential changes to the visual environment that could result from implementation of the proposed alternative. Visual resources assessments are based on the BLM VRM methodology. The existing landscape at each analysis site was assigned a VRM classification as defined in Section 3.1.1, ranging from 1 (pristine areas including designated wilderness and wild and scenic rivers) to 5 (areas where the natural character of the landscape no longer exists). Important objectives of visual resources analysis are the identification of the degree of contrast between the proposed action and the existing landscape, the location and sensitivity levels of viewpoints accessible to the public, and the visibility of the proposed action from the viewpoints. Sensitivity levels of viewpoints are based on the number and concerns of the observers. The distance from a viewpoint to the affected area and atmospheric conditions are also taken into consideration because distance and haze can diminish the degree of contrast and visibility. a fair a the star

The Storage and Disposition PEIS does not include a complete and detailed VRM assessment for each disposition alternative because design plans are not available. An assessment of visual impacts is performed using the degree of visual contrast between the proposed facilities or activities and the existing landscape. character as seen from viewpoints accessible to the public, such as public roadways, recreation areas, special status lands, and residential areas. Sensitivity levels of the viewpoints and visibility of the affected area from those viewpoints are also taken into consideration. The reduction of assigned VRM classification could result if the affected area could be seen from a viewpoint with a high sensitivity level.

4.1.2 SITE INFRASTRUCTURE

ないのできょうである

1

Site infrastructure impacts are assessed by overlaying the support requirements of the various alternatives on the projected site infrastructure capacities. These impact assessments focus on electrical power requirements, road networks, rail interfaces, and fuel requirements. Projection of electricity availability, site development plans, and other DOE mid- and long-range planning documents are utilized to describe existing site infrastructure conditions. Tables are presented depicting additional infrastructure requirements estimated for each alternative. Data for these comparison tables were obtained from detailed data input reports describing each potential facility that would be required under the various alternative actions. Design mitigation considerations that might reduce infrastructure demand are identified, where possible, on a site-by-site basis.

Under some Pu disposition alternatives, specific analysis sites are not identified. As a result, no planning documents are available to provide descriptions of the site infrastructure or to establish a detailed baseline from which environmental consequences can be estimated. For these cases, generic environmental baselines were developed to define conditions for analysis purposes. Site infrastructure impacts are assessed by analyzing the specific facilities and activities associated with the alternative that are necessary to locate new missions at the unspecified sites. As stated above, tables are presented to depict the additional infrastructure requirements for each alternative as a result of the new missions. Impact assessments focus on the additional electrical power, road networks, rail interfaces, and fuel required to perform the new missions located at the unspecified sites.

Detailed assessments of the availability of electric power for the various storage and disposition facilities' electrical power requirements in the year 2005 timeframe are not considered practical. Electric utilities would not be expected to reliably project how they would meet the needs of these facilities (that is, whether by new power generation, power imports, or demand-side management). Therefore, the basis for this PEIS assessment is the supply and demand projections of the U.S. electric utilities published annually by the North American Electric Reliability Council.

For purposes of analysis, electricity generation is based on the assumption that electricity would be supplied by the power pool currently supplying the facility in question by using a mix of fuels and generating sources representative of those projected. These data are used to determine whether there would be enough reserve margin within a particular power pool to accommodate electrical requirements or if additional power is required. A detailed analysis, based on the proportional contributions from each fuel source, would be conducted in sitespecific NEPA documents, as appropriate.

Four of the Pu disposition alternative technologies, specifically the existing LWR, evolutionary LWR, partially completed LWR, and CANDU reactor, have the ability to produce electricity while using Pu as MOX fuel. The environmental effects of this steam production are included in the analysis of the basic technologies since design of these reactors includes steam turbines to generate electricity. Selection of either of two alternatives, the partially completed LWR or the evolutionary LWR, would result in additional power line construction to distribute electricity to the power grid. The environmental impacts of this additional power line construction for these two alternative actions are not addressed in this PEIS but would be assessed in tiered, site-specific NEPA documents should either be chosen for implementation.

4.1.3 AIR QUALITY AND NOISE

Air Quality. Potential effects on the environment associated with air pollutant emissions from normal operations for each alternative are evaluated, and effects of construction emissions are discussed. The assessment of air quality impacts requires identification of applicable criteria for assessing impacts, the development of emission inventories, and the estimation of air pollutant concentrations. The assessment of impacts is based on the estimated concentrations, data on the existing environment, and the assessment criteria. Human health effects due to air pollutant emissions are discussed in the Public and Occupational Health and Safety sections and include consideration of airborne radioactive chemical releases and from accidents initiated ---- by natural phenomenon (for example, tornadoes).

13-1-1-

The assessment of potential impacts to air quality is based upon comparison of effects of each alternative with applicable State, local, or NAAQS. Assessment criteria for air pollutants include the EPA primary and secondary ambient air quality standards for criteria pollutants, specified in 40 CFR 50, and ambient standards established by each State. The more stringent of either the EPA or State standards serve as the assessment criteria. The assessment criteria for hazardous or toxic pollutants include guidelines or standards adopted or proposed by each State. Hazardous and toxic air pollutants include those listed in Title III of the 1990 CAA Amendments, in the NESHAP, in 40 CFR 61, and in standards or guidelines proposed or adopted by the respective states.

Certain hazardous pollutants for specific industrial processes are regulated by NESHAPs. No sources to which NESHAPs apply have been identified for the alternatives being evaluated except for sources of beryllium which are discussed in the health effects sections. Some of the sites have existing NESHAPs sources for which emissions are quantified under the No Action alternative.

No Action concentrations of pollutants were determined by modeling site emissions. Site-specific emissions are modeled in accordance with the guidelines presented in the EPA's Guidelines on Air Quality Models. The EPA recommended Industrial Source Complex Short Term Modél, Version 2 (ISCST2) was selected as the most appropriate model to perform the air dispersion modeling analysis because it is designed to support the EPA regulatory modeling program and is capable of handling multiple sources, including different source types. The air quality modeling analysis performed for the sites is a "screening level" analysis incorporating conservative assumptions applied to each of the sites so that the impacts associated with the respective alternatives could be compared among the sites. The "highest-high" concentration for each pollutant and averaging time is selected for comparison to the applicable assessment criteria, instead of the less conservative EPA-recommended "highest-high" and "highest second-highest" concentration for longer-term and short-term averaging times, respectively. The concentrations evaluated are the maximum occurring at or beyond the site boundary or public access road and include the contribution from the alternative and the baseline concentration. Concentrations presented for each alternative include the criteria pollutants CO, Pb, NO2, PM10, SO2; pollutants for which there are State-mandated standards; and hazardous and other toxic pollutants that are emitted for the specific alternative. Appendix F contains additional information on the ISCST2 model. 1. 11 . (n. 11 T, * 11.

Ozone is a secondary pollutant formed in the atmosphere and is not directly emitted as a pollutant from the candidate sites. The primary pollutants contributing to ozone formation are NO_x and VOCs. The formation of ozone in the atmosphere is a regional issue and is contributed to by vehicular sources (mobile sources), natural sources, and stationary sources in a region. Ozone, as a criteria pollutant, was not evaluated for this programmatic evaluation since it is not directly emitted or monitored by the candidate sites, but NO_2 and VOC emissions were analyzed where applicable.

Additional assumptions incorporated into the air quality modeling at each site include: major source criteria pollutant emissions were modeled using actual source locations and stack parameters to determine environmental baseline and No Action criteria pollutant concentrations; toxic/hazardous pollutant emissions

and sources with incomplete source characteristic information were modeled from a single source centrally located within the complex of facilities on each site assuming a 10-m (33-ft) stack height, a stack diameter of 0.3 m (1 ft), stack exit temperature equal to ambient temperature, and a stack exit velocity equal to 0.03 m/s (0.1 ft/s).

Input data for the model representing site emission rates were provided by the individual sites. Air emissions data for each alternative, other than No Action, are based on preconceptual design reports. Onsite or representative NWS meteorological data are used to define the dispersion characteristics of the site. Actual terrain heights are used for those sites not considered "flat."

The environmental consequences of construction activities on air quality are also described. Source control measures are described that are commonly used to control air pollutant emissions from construction activities. The concentrations presented in this document are the highest concentrations predicted by the model to present conservative estimates of pollutant concentrations.

The CAA, as amended in 1990 (U.S.C. 7506(c)), requires Federal actions to conform with the host State's "State Implementation Plan" (SIP). A SIP provides for the implementation, maintenance, and enforcement of NAAQS for the six criteria pollutants (that is, SO₂, PM₁₀, CO, O₃, NO₂ and Pb). The SIP's purpose is to eliminate or reduce the severity and number of violations of NAAQS and to expedite the attainment of these standards. No department, agency, or instrumentality of the Federal Government shall engage in, support in any way, or provide financial assistance for, license or permit, or approve any activity that does not conform to an applicable implementation plan.

The final rule for *Determining Conformity of General Federal Actions to State or Federal Implementation Plans* was established by EPA in November 30, 1993 (58 FR 63214), and took effect on January 31, 1994 (40 CFR Parts 6, 51, and 93). This rule establishes the conformity criteria and procedures necessary to meet the CAA until the required conformity SIP revision by each State is approved by EPA. Criteria for determining conformity are specified in some detail in the final rule, but basically ensure that emissions of all criteria air pollutants and VOCs from the action are specifically identified and accounted for in the SIP's attainment or maintenance demonstration.

The EPA has striven to ensure that the new conformity procedures are consistent with NEPA. This way, Federal agencies can incorporate the new conformity procedures into their existing NEPA procedures.

I

The EPA has limited the Conformity Rule to "nonattainment" areas or those areas classified after November 15, 1990, as "maintenance" areas. An area is designated as nonattainment for a criteria pollutant if it does not meet national primary or secondary ambient air quality standards for the pollutant (or if the area contributes to the ambient air quality of a nearby area that does not meet national primary or secondary ambient air area is an area a State has redesignated from nonattainment to attainment; accordingly, the State must submit to EPA a plan for maintaining NAAQS as a revision to the SIP.

The candidate sites at Hanford, NTS, INEL, Pantex, ORR, and SRS are within existing attainment areas. Therefore, in regard to these criteria pollutants, the proposed storage and disposition action is not affected by the provisions of this rule.

Toxic air pollutants are addressed in both the Air Quality and Noise sections and the Public and Occupational Health and Safety sections for each of the potential sites. In the Air Quality sections, the maximum concentrations of toxic air pollutants at or beyond the site boundary are compared with the most stringent standard or guidance. VOCs are not addressed because there are no regulatory standards and because they are a secondary source of ozone. In addition, there is a wide variety of VOCs that could be emitted.

KARASER I

沙亚中国和爱

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

4-8

For the disposition facilities, the maximum concentrations of hazardous pollutants, including cleaning solvents and hydrocarbons, are also not addressed because there are no regulatory standards. There are a wide variety of cleaning solvents and hydrocarbons available. However, DOE would use cleaning solvents that would reduce the chemical exposure to the public and to workers to the maximum extent possible. Specific cleaning solvents would be identified and analyzed in detail in tiered NEPA documentation. In the Public and Occupational Health and Safety sections, a health risk is calculated based upon chemical concentration, toxicity, and PEL and RfC for non-cancer effects and a cancer risk based on the slope factor (cancer potency) of those compounds that are regulated or carcinogenic.

These differences in analytical method result in the presentation of pollutants that differ between the air quality analysis and public and occupational health analysis. In the air quality analysis, toxic pollutants with low emission rates and high toxicity will, in most cases, result in extremely low concentrations at or beyond the site boundary and are therefore not presented in the air quality analysis. On the other hand, these chemical pollutants may expose an onsite worker located 100 m (328 ft) from the emission source to an unacceptable risk and therefore are presented in the analysis, hence the type and form of data used to evaluate these two areas can and will be different and may require different data sources. For example, compliance to standards neither considers what health effects are expected nor the interaction between several chemicals that may cumulatively cause health responses even if they separately are at acceptable concentrations.

Noise. The acoustical environment both onsite and offsite may be affected during construction and operations of the proposed facilities. Construction and operation noise sources and the potential for onsite and offsite impacts are discussed. The discussion is based on information available on the potential types of noise sources and the location of the proposed facilities relative to the site boundary and noise-sensitive locations. Acoustics impacts are assessed on the basis of the potential degree of change in noise levels at sensitive receptors (for example, residences near the site boundary) with respect to ambient conditions. Most nontraffic noise sources associated with construction and operation of the proposed storage and disposition facilities are located at sufficient distances from offsite noise-sensitive receptors that the contribution to offsite noise levels is expected to be small. The analysis uses available information on the potential types of noise sources and the location of proposed alternative facilities relative to the site boundary and noise-sensitive locations.

1 t

4.1.4 WATER RESOURCES

I

The availability and quality of water resources (surface water and groundwater) and the presence of floodplains may play a significant role in the acceptability of sites for this program. Impacts to water resources at a site could affect surrounding communities as well as the proposed project. Important legislation that protects water resources includes the CWA, especially Sections 402 (NPDES) and 307(b) (Pretreatment Standards), and the SDWA. In addition, DOE regulations 10 CFR 1022, *Compliance with Floodplains/Wetlands Environmental Review Requirements*, which implement Executive Orders 11988 and 11990, require evaluation of the potential effects of their actions in floodplains and wetlands.

Water resources include both surface water and groundwater. The main issues related to water resources include the following: (1) whether there is sufficient water available for both the proposed alternative and local domestic consumption, (2) whether water quality will be degraded or further degraded, (3) whether proposed actions challenge legislative or regulatory compliance, and (4) whether actions are threatened by flooding.

Surface Water Availability. Impact assessment on water availability was analyzed by comparing the rates of water consumption and wastewater discharge during construction and operation of proposed long-term storage and disposition facilities to volumes or use rates for No Action water sources. For sites that obtain water from surface water sources, a comparison was performed for both No Action water use and stream flow. For each storage and disposition alternative, water-use rates for construction and operation activities at each site or for each technology under each option or alternative were taken from technical reports prepared for specialized alternatives. The water use at a site was the sum of water use for waste treatment, storage, or disposal operations. The existing water supply is evaluated to determine its capacity to support an increased demand by comparing a projected increase with the capacity of the supplier and with existing water rights, agreements, and allocations.

The analysis assumed that water for the proposed long-term storage and disposition activities would be withdrawn from the current water source at each site. The surface water, groundwater, or municipal water source is part of the water resources affected and described in Chapter 3. However, at Pantex, the city of Amarillo has offered the use of reclaimed wastewater from the Hollywood Road Waste Treatment Plant as a potential source of water when large water requirements are needed. Where surface water is the current source, surface flow data were also evaluated.

Domestic consumption of regional water supplies by the anticipated workforce is addressed to determine if the projected supply of public water might result in or intensify potential water shortages. The impact assessment methodology presented is applied for all alternatives, including No Action.

Surface Water Quality. The assessment of potential water quality impacts include the type (for example, wastewater effluent, or stormwater runoff, or cooling tower blowdown [rapid discharge of water from the cooling tower to the receiving water body]), rate, and concentrations of potential discharge constituents, including sediment loading. Water quality management practices employed at each site also are reviewed. Parameters with the potential to further degrade existing water quality, along with parameters that could violate existing NPDES permit limits, are identified.

During the construction period, impacts to surface water resources could occur from runoff and sedimentation as a result of site clearing. During operations, impacts to water resources could occur through increased runoff from buildings, parking lots, and cleared areas. These impacts would be proportional to the amount of land disturbed and would be minimized by application of standard management practices for stormwater runoff and erosion control.

During operations, stormwater runoff could be contaminated with materials deposited from airborne pollutants. Impacts from stormwater runoff are expected to be minor, but they are highly site-specific and would depend on the design, precipitation, topography, and surface water body.

机构研究

Floodplains. Floodplains are identified from maps and existing environmental documents. Any facility operating within a 100-year floodplain, or a critical action within a 500-year floodplain, is analyzed for the environmental consequence to the floodplain. [Text deleted.]

4+ *R

If possible, no new storage or disposition facilities would be located in 100-year or 500-year floodplain areas identified from maps and existing environmental documents. The impacts of disposition activities on floodplains cannot be determined at this time because the specific locations of the disposition facilities for generic sites have not been selected. Therefore, the requirements of Executive Order 11988 (*Floodplain Management*) and 10 CFR 1022 (*Compliance with Floodplain/Wetland Environmental Review Requirements*) would be examined in follow-on NEPA documentation, as appropriate.

Groundwater Availability. Environmental impacts to groundwater availability and usage are determined by addressing the potential change in demand as a result of program activities during the construction and operation phases. Estimates of drawdowns in the affected aquifers, both onsite and offsite, are calculated. Instances are identified where groundwater use may exceed locally developed groundwater supplies. Existing water rights for the major water users, as well as contractual agreements for water supply to the sites from support communities, are summarized. Impacts associated with construction and operation withdrawals are described, including potential effects on existing areas of contaminated groundwater. Withdrawals of groundwater to supply water for the storage and disposition facilities could cause movement of existing groundwater contamination plumes. This could occur where water levels are lowered by withdrawals. If necessary, potential impacts on existing areas of contaminated in detail in site-specific environmental documents.

Groundwater Quality. Potential groundwater quality impacts associated with pollutant discharges during construction and operation phases (for example, sanitary wastes) are examined. The results of the groundwater quality projections then are compared to Federal and State groundwater quality standards, effluent limitations, and safe drinking water standards to assess the impacts of each alternative.

4.1.5 GEOLOGY AND SOILS

The geology and soil resource impact analysis consists of an evaluation of the effects generated by the construction and operation of a proposed alternative on specific geologic and soil resource attributes. Construction activities represent the principal means by which an effect to the geologic resource (limiting access to mineral or energy resources) and the soil resource (increasing the potential for soil erosion) would occur. The impact analysis also considers the effects on these attributes resulting from other past, present, and reasonably foreseeable future site improvements within the defined ROI. The principal element in assessing the effect on the geologic and soil resource is the amount and location of land disturbed during construction of the alternative.

Key factors in the impact analysis of geologic and soil resources for the disposition alternatives focus on the amount of land disturbed during construction and operations. For the soil resource impact analysis, the climatic conditions at the sites are considered. Since these key factors varied only slightly for the disposition alternatives, the impact analysis is similar for all alternatives.

Geology. Impact analysis on the geologic resource by a proposed alternative involves the evaluation of potential effects to critical geologic resource attributes such as access to mineral and energy resources, destruction of unique geologic features, vibratory ground motion induced by seismic activity, subsidence induced by groundwater, and mass movement or ground shifting induced by the construction of facilities associated with an alternative. The impact analysis includes the analysis of large-scale geological conditions such as earthquakes, volcanism, geological resources, and sinkhole development. Human health effects from accidents initiated by natural phenomenon (for example, earthquakes) are discussed in the Public and Occupational Health and Safety sections. These conditions tend to affect broad expanses of land and are not typically restricted to smaller discrete areas of land. The analysis of these conditions is provided for the storage alternatives and also applies to the disposition alternatives as appropriate.

Soil. Impact analysis on the soil resource by a proposed alternative involves the evaluation of potential effects to specific soil attributes such as soil erosion and shrink-swell potential by construction activities. [Text deleted.] Affects to the soil resource occur to small, discrete areas of land. The impact analysis for the soil resource is provided for each alternative.

4.1.6 BIOLOGICAL RESOURCES

During construction of storage and disposition facilities, impacts to biological resources may result from landclearing activities, erosion and sedimentation, human disturbance and noise, and dewatering of foundations. Operations may affect biological resources as a result of changes in land use, salt drift from cooling systems, emissions of radionuclides, water withdrawal, wastewater discharge, and human disturbance and noise. In general, potential impacts are assessed based on the degree to which various habitats or species could be affected by an alternative. Where appropriate, impacts are evaluated with respect to Federal and State protection regulations and standards. The ROI for site-specific alternatives includes the entire DOE site under consideration, while the ROI for non-site-specific alternatives includes conditions representing various regions of the United States within which the alternative could be located.

37.

Terrestrial Resources. Impacts of project activities on terrestrial plant communities are evaluated by comparing data on site or regional vegetation communities to proposed land requirements for both construction and operation. The analysis of impacts to wildlife is based to a large extent on plant community loss or modification, which directly affects animal habitat. The loss of important or sensitive habitats and species is considered more important than the loss of a regionally abundant type. Where alternatives are associated with specific sites, the disturbance, displacement, or loss of wildlife is evaluated. Migratory birds are discussed in a broader sense where alternatives are not site-specific, since they encompass a wide geographic area. Alternatives that may cause salt deposition are identified. [Text deleted.] Impacts on biotic resources from the release of radionuclides would be expected to be less than that on the human population. Humans have generally been shown to be the most sensitive organism to radiation release (AEC 1968a:220; NAS 1972a:34); radiological effects on humans are addressed in the human health sections.

Wetlands. The potential direct loss of wetlands resulting from construction and operation of storage and disposition facilities is addressed in a similar fashion as for terrestrial plant communities, that is, by comparing data on site or regional wetlands to proposed land requirements. Sedimentation impacts are evaluated based on the proximity of wetlands to project areas and the knowledge that erosion and sedimentation controls will be required. Where alternatives are site-specific, impacts resulting from increased flows are evaluated based on a relative comparison of expected discharge rates with stream flow rates. Impacts resulting from the introduction of thermal and chemical pollution into a wetland system are addressed recognizing that effluents are required to meet Federal and State standards. DOE regulations concerning proposed actions in wetlands (10 CFR 1022) were also considered.

Aquatic Resources. Impacts to aquatic resources resulting from sedimentation, increased flows, and the introduction of waste heat and chemicals are evaluated as described for wetlands. Potential impacts from radionuclides are not addressed for the same reasons described for terrestrial resources. Where alternatives are site-specific, impingement and entrainment impacts are evaluated based on a relative comparison of stream flow and intake volumes, recognizing that when intake volumes represent a large percentage of stream flow, the possibility of impingement and entrainment impacts exists. Compliance with protective measures, such as the Anadromous Fish Conservation Act (Appendix J), are addressed, as appropriate.

Threatened and Endangered Species. Impacts on threatened and endangered species are determined in a manner similar to that described for terrestrial and aquatic resources because the sources of potential impacts are similar. A list of species potentially present on each site or in each region was developed using information obtained from the USFWS, National Marine Fisheries Service, and appropriate State agencies and site-specific reports. This list, along with consideration of site environmental and engineering data, is used to evaluate

whether the various alternatives could impact any threatened or endangered plant or animal (or its habitat). Consultations with USFWS and State agencies would be conducted at the site-specific level, as appropriate.

£

4.1.7 CULTURAL AND PALEONTOLOGICAL RESOURCES

The cultural and paleontological resources impact analysis evaluates the potential effects of the construction and operation of a proposed alternative on prehistoric, historic, Native American, and paleontological resources. Appropriate information regarding cultural resources includes the possibility that NRHP-eligible sites or important Native American resources may be affected, as well as the procedures for successful completion of the NHPA Section 106 identification or mitigation process. Section 106 requires Federal agencies to take into account the effects of their activities and programs on NRHP-eligible properties. Section 106 procedures include identification of resources, evaluation for NRHP eligibility, assessment of project effects, preservation or mitigation of affected eligible resources, and coordination and consultation with the appropriate SHPO and the Advisory Council on Historic Preservation.

Compliance with the American Indian Religious Freedom Act and the Native American Graves Protection and Repatriation Act is also part of the regulatory process. These acts require that consultation with interested Native American groups be initiated as part of the compliance process.

Information collected to identify relevant cultural and paleontological resources includes the general prehistory and history of a region, results of previous cultural or paleontological resources surveys or studies within or near the affected project area, and identification of any resources eligible for or currently on the NRHP, or part of the National Natural Landmarks Program.

Impacts to prehistoric and historic resources are postulated if alternatives could substantially add to existing disturbance of resources in the project areas, could affect NRHP-eligible resources, or could cause loss or destruction of important scientific, cultural, or historic resources. Impacts to Native American resources are postulated if alternatives have the potential to affect sites important for their position in the Native American physical universe or religion, or to reduce access to traditional use areas, plant or animal communities, or sacred sites. Impacts to paleontological resources are postulated if alternatives could affect deposits with high research potential.

[Text deleted.] On a programmatic level specific detail on the types and locations of resources within the proposed project areas is not available for all locations, and the level of detail varies by facility or alternative. General information concerning site types and NRHP eligibility of those types in the region, however, is appropriate for the PEIS. At the site-specific level and depending on the alternative, more detailed information may be required including file investigations, Native American consultation, implementation of the DOE American Indian Policy, or predictive modeling to determine the types, numbers, locations, and NRHP eligibility or importance of cultural resources in the proposed project area.

4–14

4.1.8 SOCIOECONOMICS

The socioeconomic impact analysis assesses the environmental consequences of demographic and economic changes resulting from implementing a proposed alternative. Developing a new facility or adding new operations to an existing DOE or commercial facility may create new jobs. The creation of new jobs may affect the economy of the regional economic area and may place an increased demand on housing, community services, and local transportation within the ROI, if the local labor market is unable to meet the labor requirements. Therefore, this study assesses the potential impacts of each proposed alternative on these characteristics. The socioeconomic environment is defined for two geographic regions: the REA and the ROI. REAs are used to assess potential effects on the regional economy, and ROIs are used to assess effects that are more localized in political jurisdictions surrounding the sites.

The REA for each site encompasses a broad market that involves trade among and between regional industrial and service sectors and is characterized by strong economic linkages between the communities in the region. These linkages determine the nature and magnitude of multiplier effects of economic activity (purchases, earnings, and employment) at each site. REAs are defined by BEA and consist of an economic node that serves as the center of economic activity and the surrounding counties that are economically related and include the places of work and residences of the labor force.

Potential demographic impacts were assessed for the ROI, a smaller geographic area where the housing market and local community services would be the most affected. Site-specific ROIs were identified as those counties where approximately 90 percent of the current DOE and/or contractor employees reside. This residential distribution reflects existing commuting patterns and attractiveness of area communities for people employed at each site and is used to estimate the future distribution of in-migrating workers.

The following sections summarize the methodology used for assessing the potential socioeconomic impacts from the alternatives.

Baseline Comparisons. Data on the current socioeconomic conditions were compiled for each ROI and REA. The data were then projected out to the period of full operation for the alternatives based on population forecasts developed by BEA. These projections were compared with the estimates of project-related impacts.

Proposed project alternatives may require additional workers during the construction and operation phases. An analysis of the projected labor availability was performed to determine the number of workers that may inmigrate to the region to meet employment needs. In addition to jobs and income directly created by the proposed project alternatives, other jobs and income may be created indirectly within the REA. These indirect jobs and resulting income are measured by employing the most recent version of the Regional Input-Output Modeling System developed by the BEA.

Population increases due to in-migration of new workers and their families are estimated by the number of new workers and the national average household size. The national average is used because the new population would come from unknown locations outside the region.

The impact on community services is assessed by determining the increase in each community service required to maintain the level of service that currently exists within the ROI, such as student-to-teacher ratio, police officer-to-population ratio, and hospital capacity.

Local Transportation. Traffic impacts are modeled using a proprietary transportation model developed in accordance with the *Highway Capacity Manual* methodology for freeways, multilane roads, and two-lane roads. Average Annual Daily Traffic (AADT) volumes are projected for 1995 and the impact year using No Action population projections.

and strates a

The AADT for impacts beyond the No Action alternative are calculated from the direct employment and direct in-migrating population associated with each alternative. Only roads in the vicinity of the site on which at least 5 percent of workers travel were modeled. It was assumed that any additional workers required for a proposed alternative would use the same roads as the current workforce.

* *

Service flow, the maximum hourly rate at which vehicles can reasonably be expected to cross a point or uniform section of a lane during a given time period under current roadway and traffic conditions, is calculated for the road segments. Capacity of the road segments is calculated from the number of lanes and capacity per lane. The service flow equation is solved for the volume-to-capacity ratio.

Level of service is a measure describing operational conditions within a traffic stream. A level of service describes these conditions in terms of factors such as speed and travel time, freedom to maneuver, traffic interruptions, comfort and convenience, and safety. The level of service designations range from A to F, with each level defined by a range of volume to capacity ratios. The level of service designations, given in Table 4.1.8–1, are based primarily on the *Highway Capacity Manual* Special Report 209.

Letter Designation	Operating Conditions	Level of Service Definition
A	Good	This is a condition of free flow, accompanied by low volumes and high speeds.
В	Good	This occurs in the zone of stable flow, with operating speeds beginning to be restricted somewhat by traffic conditions.
С	Good	This is still in the zone of stable flow, but speeds and maneuverability are more closely controlled by the higher volumes.
D ;	Below average	This level of service approaches unstable flow, with tolerable operating speeds maintained, though considerably affected by changes in operating conditions.
E	Maximum capacity	This cannot be described by speed alone, but represents operations at lower operating speeds, typically, but not always, in the neighborhood of 30 miles per hour, with volumes at or near the capacity of the highway.
F	Traffic jam	This describes a forced-flow operation at low speeds, where volumes are above

Table 4.1.8–1. Level of Service Letter Designations and Definiti	el of Service Letter Designations and Definitions
--	---

Source: National 1985a.

4.1.9 PUBLIC AND OCCUPATIONAL HEALTH AND SAFETY

の記述です。

The scope of the public and occupational health and safety analysis includes a determination of effects on human health that result from exposures to ionizing radiation and hazardous chemicals. Storage and disposition alternatives addressed in this PEIS use a broad variety of processes involving both radioactive and chemical materials that can be hazardous to human health. The degree of hazard is directly related to the type and quantity of the particular radioactive or chemical material to which a person is exposed. The health effects that may be associated with radiation and chemical induction of cancer are determined by estimating the radiological and chemical doses to workers and the general public that result from these exposures. The risk from these doses is then converted to potential cancers. This is done for normal operations and postulated accident scenarios. Additionally, for hazardous chemicals, a Hazard Quotient (HQ) is calculated based on the exposure concentration of each chemical; the HQs are calculated for all known noncancer adverse effects. The HQs are then summed to obtain the Hazard Index (HI) for each operation at a given site. The HI is used as an indicator of safe exposure levels to avoid noncancer effects of an acute or chronic nature, and is independent of the cancer risk. More detailed information about the public and occupational health safety analysis for this PEIS is found in Appendix M.

Computer models and operational histories are used to predict the impacts on the health of workers and the public due to both normal operation and postulated accidents. The computer models include GENII and Melcor Accident Consequence Code System (MACCS) for airborne and liquid radioactive releases during normal operation and accident conditions, respectively; and ISCST2 for hazardous chemical releases.

Since direct chemical monitoring data on worker exposure is not available for specific operations, the onsite worker is assumed to receive the maximum exposure that any involved (the workers directly involved in the proposed facility) or noninvolved (the workers onsite but not directly involved in the proposed facility) onsite person will receive. OSHA-regulated levels (that is, the PEL) are applied to all hazardous chemicals that are released at the site. This includes both the process-specific releases as well as those that are a result of other site operations. All onsite exposures are assumed to occur at a distance of 100 m (330 ft) from a centralized point of release, which will yield a conservative concentration level for each chemical. The concentrations are derived through the ISCST 2 model recommended by EPA. The noncancer risks to the MEI of the public consist of HQs that compare chemical exposure levels to the RfC values published by EPA in the Integrated Risk Information System (IRIS). For the onsite worker, the HQ values are calculated by comparing the calculated chemical exposure levels to the PEL established by OSHA. The HI for either the MEI or the onsite worker is the sum of the HQs for every hazardous chemical analyzed. If the HIs are 1.0 or less, then the noncancer regulated levels for all hazardous chemicals have been met and no health effects are expected.

The lifetime cancer risk to the MEI and the onsite worker from exposure to a carcinogenic chemical can be derived from either of the two following ways. The calculated exposure dose in mg/kg of body weight is multiplied by that individual chemical's slope factor published in EPA's IRIS or the EPA's yearly published Health Effects Summary Tables (HEAST), or the exposure concentration in mg/m³ is multiplied by unit risk (found in IRIS or HEAST) to give the lifetime cancer risk. Only carcinogenic chemicals will have slope factors and/or unit risk values. The cancer risk coming from all carcinogenic chemicals is derived by summing up the risks from all individual carcinogenic chemicals.

The chemical pollutant exposures to the public are considered to be for 24 hours a day, 52 weeks per year, and 70 years for a lifetime. The worker exposure is based on an 8-hour day and 52 weeks of 40 hours each (0.237 fractional year). The HI values and cancer risks are conservative because all emissions are assume to come from a centralized source term. The cancer risks to the noninvolved worker for each chemical are computed from the dose (converted from air concentrations) and the unit risk or slope factors to yield a probable risk. The risks are also conservative because a single point at or near the maximum onsite concentration is selected for exposure of the noninvolved facility worker. Actual risks are likely lower than estimated risks.

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

1.× + x

The HI is the sum of the HQs for every hazardous chemical analyzed, the HQ being the ratio of the actual (or calculated) concentration to the regulated concentration value. Even if OSHA PELs and EPA RfCs for several hazardous chemicals are met, but they affect the same target organ or tissue, they may be expected to collectively cause injury or ill health because their damage will be cumulative. Thus, an HI of much greater than 1.0 is meaningful because the sum of the effects has the same effect as that of a single hazardous chemical exceeding the OSHA PEL or the RfC. The HI and HQ are not regulated values, but serve to assess whether regulated values have been met and whether there is a potential for a detrimental health effect to occur. ٠,

Toxic air pollutants are addressed in the Air Quality and Noise sections and the Public and Occupational Health sections for each of the potential sites. In the Air Quality sections, the maximum concentration of toxic air. pollutants at or beyond the site boundary are compared with a standard to determine compliance. In the Public and Occupational Health sections, a health risk is calculated based upon chemical concentration, toxicity, and PEL or RfC for noncancer effects and a cancer risk based on the slope factor (cancer potency) of those compounds which are regulated or carcinogens.

11 "

These differences in analytical method result in the presentation of pollutants that differ between the air quality analysis and public and occupational health analysis. In the air quality analysis, toxic pollutants with low emission rates and high toxicity will, in most cases, result in extremely low concentrations at or beyond the site boundary and are therefore not presented in the air quality analysis. On the other hand, these chemical pollutants may expose an onsite worker located 100 m (330 ft) from the emission source to an unacceptable risk, and therefore are presented in the analysis. Hence, the type and form of data used to evaluate these two areas can and will be different and may require different data sources.

The assessments use technology-specific releases of radioactivity and hazardous chemicals and site-specific factors such as meteorology, population distribution, agricultural production, and an assumed facility location on the site. Atmospheric dispersion modeling performed for the Air Quality section is also utilized to develop exposure concentrations in the evaluation of impacts to workers from hazardous chemicals.

A discussion of calculational uncertainties is included to provide an understanding of how the various steps of an impact analysis can lead to predictions different from those presented in this PEIS. However, efforts have been made in all assessments to choose conservative assumptions and parameters so that impacts are not underestimated. Further discussions of assessment methodologies, including values of parameters used in the various computer models, are presented in Appendix M.

Health Impacts on Plant Workers During Normal Operation. Because radiation workers are individually monitored, experiences from past and current operations that are similar to future operations are used to estimate the radiological doses to workers directly involved with fissile material storage and disposition activities. Doses are also determined for noninvolved workers (that is, workers who are onsite but not involved with the proposed storage and disposition activities).

Radiological doses are converted to health effects using appropriate health risk estimators. Health effects from chemicals are predicted by applying appropriate risk estimators to modeled concentrations. Radiological and chemical impacts are also compared with applicable regulatory compliance and guidance requirements. The health risk estimators and applicable standards are presented and discussed in detail in Appendix M. , , , , , , . .

. .

34

. .

Health Impacts on the Public During Normal Operation. Public health impacts could result from exposure to radioactive or hazardous chemical materials released during operation. The effect is the sum of internal exposure resulting from breathing air, eating food, and drinking water; and of external exposure from standing on contaminated ground, being exposed to the air, and being submerged in water.

4 - 18

Impacts to the MEI and to the population around each site are calculated. The differences in impacts among the various sites are the result of many site-specific factors including the distance to the site boundary, the population distribution, and meteorological dispersion.

Modeling is used to estimate the type and amount of material released and the associated radiological doses and chemical concentrations and doses. The predictive models used to calculate radiological doses have been found to give results that tend to be higher than if calculated by converting actual measured concentrations to doses. The radiological doses and the chemical doses and concentrations are converted to health effects using appropriate health risk estimators. Radiological doses, chemical concentrations, and health risks are compared with regulatory limits and, for perspective, with background levels in the vicinity of the site. The health risk estimators used in the assessments and the applicable regulatory standards are presented and discussed in detail in Appendix M.

Accident Analysis for Postulated Accident Scenarios. The relative consequences of postulated accidents are considered in the evaluation of each alternative. In evaluating the magnitude and consequences of each alternative, accident analysis is performed. The concepts used are analogous to a formal Probabilities Risk Assessment, which would be appropriate for a project-level analysis, although the accident analysis involves considerably less detail for programmatic decision-making. The accident analysis addresses only a spectrum of bounding accidents (high consequence, low probability) and a representative spectrum of possible operational accidents (low consequence but high probability of occurrence). The technical approach for the selection of accidents is consistent with the DOE Office of NEPA Oversight *Recommendations for the Preparation of Environmental Assessments and Environmental Impact Statements* (May 1993) guidance that recommends consideration of two major categories of accidents: within design basis accidents (also referred to as evaluation basis accidents).

I

1

For the purpose of this assessment, risk is defined as the mathematical product of the probability and consequence of an accident. The risk-contributing scenarios consider both evaluation basis and beyond evaluation basis accidents. The specific accidents consider the types of facilities. Examples of accidents include those resulting from operator errors, spills, criticalities, fire, explosions, airplane crashes, common-cause failures, severe weather, earthquakes, and transportation. Information on potential accidents includes those that have been postulated and analyzed for similar facilities. The risks of the various storage and disposition facilities are evaluated in terms of the incremental increase in risk from postulated accidents with respect to normal day-to-day risks to which the general population is exposed

For each alternative, a number of evaluation basis and beyond evaluation basis accidents have been identified and are analyzed to define the accident consequences and risks. Evaluation basis accidents are generally in a frequency range of greater than 10^{-6} per year, while beyond evaluation basis accidents are generally in a frequency range of 10^{-7} to 10^{-6} per year. In some cases, accidents less than 10^{-7} are included in the set of beyond evaluation basis accidents when their impacts are relevant to decisionmaking.

Uncertainties. The sequence of analyses performed to generate the radiological and hazardous chemicals impacts estimates from normal operation and facility accidents include (1) selection of normal operational modes and accident scenarios and their probabilities, (2) estimation of source terms, (3) estimation of environmental transport and uptake of radionuclides and hazardous chemicals, (4) calculation of radiation and chemical doses to exposed individuals, and (5) estimation of health effects. There are uncertainties associated with each of these steps. Uncertainties exist in the way the physical systems being analyzed are represented by the computational models and in the data required to exercise the models (due to measurement errors, sampling errors, or natural variability).

Of particular interest are the uncertainties in the estimate of cancer deaths from exposure to radioactive materials. The numerical values of the health risk estimates used in this PEIS (refer to Appendix M) were

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

- × . . .

obtained by the practice of linear extrapolation from the nominal risk estimate for lifetime total cancer mortality at 10 rad. Other methods of extrapolation to the low-dose region could yield higher or lower estimates of cancer deaths. Studies of human populations exposed at low doses are inadequate to demonstrate the actual level of risk. There is scientific uncertainty about cancer risk in the low-dose region below the range of epidemiological observation, and the possibility of no risk or even health benefits (hormesis effects) cannot be excluded. Because the health risk estimators are multiplied by conservatively calculated radiological doses to predict fatal cancer risks, the fatal cancer values presented in this PEIS are expected to be overestimates.

The widely used algorithms for estimating the risk of latent cancers from radiation are based on high dose rates, and the impacts are then extrapolated to low rates by presumed linear response models. These models are known to overestimate the risk for low dose rates, and the actual risk may be zero.

For the purposes of presentation in this PEIS, the impacts calculated from the linear model are treated as an upper bound case, consistent with the widely used methodologies for quantifying radiogenic health impacts. This does not imply that health effects are expected. Moreover, in cases where the upper bound estimators predict a number of latent cancer deaths that is greater that 1, this does not imply that the latent cancer death(s)

Uncertainties are also introduced when accident analyses performed for similar existing facilities have been used as a major source of data. Although the radionuclide composition of source terms are reasonable estimates, there are uncertainties in the radionuclide inventory and release fractions that affect the estimated consequences. Accident frequencies for low probability sequences of events are always difficult to estimate, even for operating facilities, because there is little or no record of historical occurrences. For a new facility, such as a Pu storage facilities would tend to further compound the effects of uncertainties. There are also uncertainties attributed to information for designs that are in the conceptual stage.

In summary, the radiological and hazardous chemical impact estimates presented in this document were obtained by:

- Using the latest available data
- Considering the processes, events, and accidents reasonably foreseeable for the storage and disposition facilities described in this PEIS
- Making conservative assumptions when there is doubt about the exact nature of the processes and events taking place, such that the chance of underestimating health impacts is small.

222

(c) a subject of the second s second s second se

4-20

4.1.10 WASTE MANAGEMENT

Waste management facilities that support the long-term storage or disposition of weapons-usable fissile materials would treat and package all waste generated in support of this activity into forms that enable long-term storage and/or disposal in accordance with the *Atomic Energy Act* (AEA), RCRA, and other applicable statutes as outlined in Section E.1.2. The preconceptual designs of the new facilities provide their own waste treatment capability and incorporate waste minimization and pollution prevention practices.

Since any changes in wastes volumes could have an impact on the existing and projected waste activities at DOE sites, the PEIS assesses the impacts on waste management from the waste streams generated by all alternatives, including the No Action Alternative. All alternatives would be required to fully comply with all current applicable regulations. The waste management assessment is very important to the overall DOE decisionmaking process and is being coordinated with EM, which evaluates alternatives for implementing an integrated waste management program. The Draft Waste Management PEIS (DOE/EIS-0200-D) focuses on waste management strategies required to treat, store, or dispose of existing wastes and wastes that will be generated in the future as a result of DOE operations. The Draft Waste Management PEIS also evaluates the environmental impacts of transporting wastes. [Text deleted.]

Waste management activities that would support the long-term storage or disposition of weapons-usable fissile materials program are assumed to be per current site practices and are contingent upon decisions to be made through the Waste Management PEIS. Any future waste management facilities that may be required to support the long-term storage or disposition of weapons-usable fissile materials program would be coordinated with any decisions resulting from the Waste Management PEIS and any respective site-specific NEPA documentation. For example, depending on decisions in the waste-type-specific RODs for the Waste Management PEIS and in subsequent waste-type-specific RODs and NEPA documents, wastes could be treated and disposed of onsite or at regionalized or centralized DOE sites.

The construction and operation of facilities to support the storage and disposition of weapons-usable fissile material program would generate spent nuclear fuel and several types of wastes, depending on the alternative. [Text deleted.] Construction wastes would be similar to those generated by any construction project of comparable scale. Spent nuclear fuel and the following waste categories are analyzed: TRU, mixed TRU, low-level, mixed low-level, hazardous, and nonhazardous. This PEIS also analyzes the management and onsite storage of wastes and spent nuclear fuel, until DOE either disposes of the wastes or places them in long-term retrievable storage. To provide a framework for addressing the impacts of waste management, descriptive information is presented on waste management activities anticipated for each DOE site and for each storage or disposition technology and site. These estimates have included waste minimization provisions. The impact assessment compares the projected waste type and waste volume generation from the various disposition facilities at each site with that of the No Action Alternative. Impacts are assessed in the context of existing site practices for treatment, storage, and disposal, including current Federal, State, and local regulations and agreements.

The number of TRU waste shipments to WIPP (depending on decisions from the Supplemental EIS for the disposal phase of WIPP) was estimated by using data in the *Comparative Study of Waste Isolation Pilot Plant Transportation Alternatives* (DOE/WIPP 93-058). From this report, the PEIS uses 8.7 m³ (11.4 yd³) per truck shipment, 17.5 m³ (22.9 yd³) per regular train shipment, and 52.4 m³ (68.6 yd³) per dedicated train shipment (rail shipment may not be applicable at all sites). DOE sites used in the analysis (with the exception of Pantex) either have, or have planned, an onsite LLW disposal facility. As discussed in Section E.1.4, land-usage factors for the disposal of LLW were developed from data in the DOE Integrated Data Base Program for each existing DOE LLW disposal facility (Hanford, INEL, NTS, and SRS). For the proposed Class II LLW disposal facility at ORR, a 3,300 m³/ha (1,700 yd³/acre) usage factor was assumed (OR DOE 1995e:1). For the purposes of this

I

I

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

PEIS, it was assumed that all LLW generated at Pantex would be shipped to NTS per current practice. The number of additional shipments required to transport LLW from the site to a DOE LLW disposal facility is estimated. A typical shipment was assumed to consist of eighty 208-1 (55-gal) drums, which results in a total shipment volume of approximately 16.6 m³ (21.7 yd³). The risks associated with additional shipments are addressed as part of the intersite transportation assessment.

Under some PEIS alternatives, specific candidate sites are not identified. As a result, no planning or environmental documents are available to provide descriptions of the waste management activities or to establish a detailed baseline from which environmental consequences can be estimated. For these cases, generic environmental baselines are presented to define conditions. Waste management impacts are assessed by analyzing the specific waste management facilities and activities associated with the alternative that are necessary to locate new missions at the unspecified sites.

4.2 NO ACTION ALTERNATIVE AND LONG-TERM STORAGE ALTERNATIVES

This section describes the potential environmental consequences of the No Action Alternative and of constructing and operating various long-term storage alternatives. A key for locating information on the storage alternatives and candidate sites analyzed is shown in Table 4–1. Included in the analysis of each site is storage phaseout. The phaseout of Pu storage at Hanford, INEL, Pantex, SRS, RFETS, and LANL, and HEU storage at ORR, is a common activity associated with both storage and disposition when the storage requirements for these fissile materials are eliminated at a particular site. The impact analysis in this section is organized by site.

4.2.1 HANFORD SITE

A listing of the proposed long-term storage alternatives, subalternatives, and related actions, including the No Action Alternative, at Hanford is provided below. The potential impacts of implementing these alternatives and related actions at Hanford are described in the following sections: land resources, site infrastructure, air quality and noise, water resources, geology and soils, biological resources, cultural and paleontological resources, socioeconomics, public and occupational health and safety, and waste management. The specific longterm storage alternatives proposed for Hanford are the Upgrade Alternative, the Consolidation Alternative, and the Collocation Alternative.



Proposed Storage Activities at Hanford Site

- No Action Alternative (Preferred Alternative): Continue to store Hanford Pu material within the scope of this PEIS in the PFP in stabilized form pursuant to DNFSB Recommendation 94-1.
- Upgrade Alternative: There are two subalternatives under this storage alternative.
 - Upgrade Without RFETS Pu or LANL Pu Subalternative: Two options to accommodate Hanford Pu material: Modify selected areas of FMEF in 400 Area; or construct a new facility in 200 West Area.
 - Upgrade With All or Some RFETS Pu and LANL Pu Subalternative: Two options to accommodate Hanford, RFETS and LANL Pu material: Increase modification of selected areas of the FMEF in 400 Area; or construct a larger new facility in 200 West Area.

11.

I

I



• Phaseout: Hanford Pu material within the scope of this PEIS would be moved out of Hanford to the Consolidation or Collocation site (located at another DOE site) or to disposition (for surplus Pu).

4.2.1.1 Land Resources

Preferred Alternative: No Action Alternative

Under this alternative, Pu storage would continue at the current interim storage location in stabilized form at the PFP pursuant to DNFSB Recommendation 94-1 and the Record of Decision for the *Plutonium Finishing Plant Stabilization Environmental Impact Statement* (PFP EIS). The ongoing (no new action) activities conform with present and future land-use plans, policies, and controls. Therefore, no impacts on land resources would be anticipated at Hanford beyond the effects of existing activities and future activities that are independent of the proposed action.

Upgrade Alternative

I

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Part of the existing FMEF, located in the 400 Area of Hanford, would be converted into a Pu storage facility. The modified FMEF would house the Pu storage vault and all the functions needed to support storage. No new land disturbance would occur during construction nor would additional land not previously disturbed be used during operations.

Land Use. Modifying the FMEF for Pu storage would conform with existing and future land use as described in the current *Hanford Site Development Plan* and with ongoing discussions in the CLUP Process. According to the *Hanford Site Development Plan*, 400 Area land use is identified as reactor operations, which can include other operational uses such as material storage (HF DOE 1993c:13,14). Since this option would involve the modification of an existing facility, construction would occur completely within an existing, but not operating, protected area and would use the same areas previously disturbed during original construction (for example, laydown areas). During operation, the entire previously disturbed protected area (6.25 ha [15 45 acres]) would be required. In addition, there could be dedicated areas outside the fence (for example, parking lot). Other Hanford land uses or special status lands will not be affected. As discussed in Section 4.2.1.8, no in-migration of workers would be required during the construction and only a small increase in population would occur during operation. There would be no change to the region's housing market during construction and an insignificant effect during operation. Therefore, no indirect impacts to the offsite land use would be anticipated.

Construction and operation would be compatible with State and local (Benton, Franklin, and Grant Counties and the city of Richland) land-use plans, policies, and controls since Hanford provides information to these jurisdictions for use in their efforts to comply with the GMA (HF DOE 1993c:17).

Visual Resources. This alternative would not result in new land disturbance or new structures, so the existing landscape would remain virtually unchanged. No impacts to visual resources caused by construction and operation would occur. The appearance of the upgraded facilities would remain consistent with the industrialized landscape character and the current VRM Class 5 designation of the developed areas of Hanford.

Construct New 200 West Area Facility for Plutonium Storage

The new Pu storage facility would be a two-level facility that would be located on vacant land within the 200 West Area northwest of the 234-5Z building. No new land disturbance would occur during construction nor would additional land not previously disturbed be used during operations.

ان مربع مربعة

۰,

it , , , \$ Land Use. Construction of a new facility for Pu storage would conform with existing and future land use as described in the current Hanford Site Development Plan and with ongoing discussions in the CLUP Process. According to the Hanford Site Development Plan, 200 Area land use is identified as waste operations, which includes radioactive material management and storage (HF DOE 1993c:13,14). Although existing land use would change from unused to developed land, the proposal would be consistent with land-use plans and construction and operation would occur completely within an existing protected area and would not involve new disturbed land. During operation, the entire 10.5 ha (26 acres) protected area would be required. In addition, there could be dedicated area outside the fence (for example, parking lot). Other Hanford land uses or special status lands would not be affected. As discussed in Section 4.2.1.8, workforce requirements for construction and operation would be the same as modification of the FMEF. Therefore, no offsite land use would be indirectly affected. [Text deleted.]

ч.:;,.

Construction and operation would be compatible with State and local (Benton, Franklin, and Grant Counties and the city of Richland) land-use plans, policies, and controls since Hanford provides information to these jurisdictions for use in their efforts to comply with the GMA (HF DOE 1993c:17).

Visual Resources. [Text deleted.] The appearance of the proposed facility would be consistent with the existing industrialized landscape character, and the current VRM Class 5 designation of the 200 West Area would remain. Because of the existing industrial character of the 200 West Area and the distance to any sensitive viewpoints, no visual impacts would occur.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

14.

Die ,

This option would modify selected areas of the existing FMEF for Pu storage. Although the additional interior area needed to accommodate the RFETS and LANL material would slightly increase footprint, the facility would still be contained within the existing protected area. Therefore, land area requirements during modification and operations would be equal to the Upgrade Without RFETS or LANL Pu'Subalternative. Potential impacts to land resources would be the same as the Upgrade Without RFETS Pu or LANL Pu Subalternative.

Construct New 200 West Area Facility for Plutonium Storage

CIC.

A new facility to accommodate Hanford, RFETS, and LANL Pu material would be constructed in the 200 West Area. Land area requirements during construction and operations would be equal to that of the Upgrade Without RFETS Pu or LANL Pu Subalternative. Therefore, potential impacts to land resources would be similar to the Upgrade Without RFETS Pu or LANL Pu Subalternative. .. -

63 ...

· • •

Consolidation Alternative

Construct New Plutonium Storage Facility

able is the the only of Under this alternative, all Pu within the scope of this PEIS would be stored at a new storage facility to be constructed at Hanford west of the 200 East Area. The potential facility location has been improved with a railroad spur and utility line (HF 1996a:1). The alternative would disturb 58.5 ha (144 acres) of land area during construction of which 56 ha (138 acres) would be used during operations. A buffer zone would be provided between operations and the Hanford site boundary. Pu storage in existing DOE storage facilities would be phased out.
Land Use. Construction and operation of a new consolidated facility for Pu storage would conform with existing and future land use as described in the current *Hanford Site Development Plan* and in the CLUP Process. The *Hanford Site Development Plan*, identifies land use in the 200 Area as waste operations, which includes radioactive material management and storage (HF DOE 1993c:13,14). Although existing land use would change from undeveloped to industrial, this alternative conforms with land-use plans. As discussed in Section 4.2.1.8, no in-migration of workers would be required during the construction phase, and only a small in-migration would occur during operations, with no effect on the housing sector. Therefore, no indirect offsite land-use effects would occur. Compatibility with the use of special status lands, and consistency with State and local land-use plans, policies, and controls would be similar to the Construct New 200 West Area Facility for Pu Storage option of the Upgrade Alternative.

Visual Resources. The appearance of the proposed action would be consistent with the industrial landscape character and current VRM Class 5 designation of the 200 Area. Potential impacts to visual resources would be similar to the Construct New 200 West Area Facility for Pu Storage option of the Upgrade Alternative.

Collocation Alternative

1

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

All HEU within the scope of the PEIS would be stored at a new storage facility at Hanford collocated with the Pu storage facility. The facilities would be located west of the 200 East Area. The potential facility location has been improved with a railroad spur and utility line (HF 1996a:1). The facilities would disturb 89.5 ha (221 acres) of land area during construction of which 87 ha (215 acres) would be used during operations. A buffer zone would be provided between the facilities and the Hanford site boundary. Pu storage in existing facilities at Hanford would be phased out.

Land Use. Construction and operation of new collocated facilities would conform with existing and future land-use plans as described in the current *Hanford Site Development Plan* and in the CLUP Process. The *Hanford Site Development Plan* identifies land use in the 200 Area as waste operations, which includes radioactive material management and storage (HF DOE 1993c:13,14) Vacant land would be used. As discussed in Section 4.2 1.8, no in-migration would occur during construction. A small number of workers are projected to in-migrate during operation. Projected housing vacancies would be sufficient to accommodate this growth. Therefore, no indirect impacts to offsite land uses would be anticipated. Compatibility with the use of special status lands, and consistency with land-use plans, policies, and controls, would be similar to the Construct New 200 West Area Facility for Pu Storage option of the Upgrade Alternative.

Visual Resources. The appearance of the proposed action would be consistent with the industrial landscape character and current VRM Class 5 designation of the 200 Area. Potential impacts to visual resources would be similar to the Construct New 200 West Area Facility for Pu Storage option.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Under this subalternative, land effects during construction and operation would be almost the same in extent and magnitude to the No Action Alternative, Upgrade with All or Some RFETS Pu and LANL Pu Subalternative, Consolidation Alternative, and Collocation Alternative because the facility would be almost the same. However, because the smaller quantity of material would require smaller facilities, it is likely that less land area would be disturbed during construction and used during, operations. [Text deleted.]

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

> ----

1 4.6

Phaseout

.

ļ

I

No new construction or upgrade of existing facilities would occur under the phaseout of the Pu mission. Hanford Pu would be moved out of Hanford to the Consolidation or Collocation site or to disposition. Therefore, no impacts to land resources would be anticipated at Hanford beyond the effects of existing activities and future activities that are independent of the proposed action. . ! • • ŧ.

۰.

[Text deleted.]

1.4

: ::

. چې مې چې

71.1

. 7

ţĭ 1

1 12:

2. 17 ;,

4-28

4.2.1.2 Site Infrastructure

For selected site infrastructure parameters, Table 4.2.1.2–1 shows the site availability, projected usage under No Action, and projected usage associated with each storage alternative at Hanford, including phaseout of the Pu storage mission. Adequate infrastructure is available to accommodate all long-term storage alternatives. In addition, there is adequate space for any new facilities required to support the various storage alternatives.

Preferred Alternative: No Action Alternative

The infrastructure currently in place at Hanford is capable of handling all anticipated missions and functions associated with the No Action Alternative. However, certain actions could result in changes to the site infrastructure under No Action. Some of these actions are described below. Since the PFP processing facilities (234-5Z) were designed and constructed in 1947, it is reasonably foreseeable that upgrades of the PFP and support services/utilities could be required to complete stabilization and packaging activities for the current inventory of weapons-usable Pu. These activities are described in the *Hanford Site Integrated Stabilization Management Plan* (SISMP). In addition, glovebox-scale processing and a new packaging capability are required to implement the DOE-STD-3013-94 storage standard in SISMP timeframes.

In the PFP-EIS, DOE has assessed impacts of activities required at Hanford to remove and stabilize readily retrievable Pu residues in the nonstorage portions of the PFP (that is, 234-5Z processing facility) while its current material storage capability is maintained. The actions resulting from the PFP-EIS could affect certain site infrastructure parameters. Hanford will continue to evaluate low-assay, non-weapons usable, Pu materials currently in storage for potential treatment, packaging, and disposal as TRU waste to WIPP to avoid unnecessary costs, handling, and personnel exposure. [Text deleted.]

There could be additional changes to Hanford's site infrastructure under No Action resulting from construction of safe, environmentally sound, and economic, dry interim storage of the K-East Basin (105-KE Basin) spent fuel. The Hanford SISMP outlines plans to remove the spent nuclear fuel from the 105-KE Basin by December 1999. Such removal will resolve the safety and environmental concerns identified in DNFSB Recommendation 94-1 that are associated with the deteriorating spent fuel in the 105-KE Basin and would occur under No Action.

Upgrade Alternative

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium subalternatives

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Modifying selected areas of the FMEF to accommodate long-term storage of existing quantities of Pu at Hanford would not affect the site infrastructure. Data for construction are presented in Appendix C. Operations impacts to the Hanford infrastructure under this option would be minimal. This is because the FMEF is a fully capable nuclear materials facility with the required infrastructure, including access roads, in place. All infrastructure requirements for this subalternatives are within site capacities.

Construct New 200 West Area Facility for Plutonium Storage

Constructing a new storage facility to accommodate long-term storage of existing quantities of Pu at Hanford would not impact the site infrastructure. Data for construction are presented in Appendix C. Operations impacts to the Hanford infrastructure under this subalternatives would be minimal. [Text deleted.] As shown in Table 4.2.1.2–1, less than 5 km (3 mi) of roads would need to be constructed. All other infrastructure requirements are within site capacities.

 Table 4.2.1.2–1.
 Site Infrastructure Changes Required for Operation at Hanford Site (Annual)—No Action (2005) and

 Storage Alternatives

	Transportation		Electrical					
Alternative	Roads (km)	Railroads (km)	Energy (MWh/yr)	Peak Load (MWe)	Oil (l/yr)	Natural Gas (m ³ /yr)	Coal (t/yr)	_
No Action			•.					<u> </u>
Site availability	420	204	1,678,700	281	14,775,000	21,039,531	91,708	
Projected usage	420	204	345,500	58	9,334,800	21,039,531	0	
Upgrade (without RFETS or LANL Pu material)				v		-		
Modify FMEF			-			ه سر چ سر ³		~
Projected usage with upgrade facility	<i>.</i> 420 .	204	365,500	63	9,338,740	21,039,531	0	
Amount required in excess to site availability	0 🐔	0	, 0	0	0	0	0	
New Pu Storage Facility			r I					۲۳- م مربع
Projected usage with upgrade facility	425	204	365,500	63	9,338,740	21,039,531	0	· ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~
Amount required in excess to site availability	. <5	0	0	0	0	0.	0	
Upgrade (with RFETS and LANL Pu material)			-					
Modify FMEF			h .		-			•
Projected usage with upgrade facility	420	204	<u>366,650</u>	63	9,338, 740	21,039,531	0	
Amount required in excess to site availability	0	0	. 0	, O	- 0	0	0	, ·
New Pu Storage Facility	,		· .			T ,	•	د.
Projected usage with upgrade facility	425	204	366,650	63	9,338,740 -	21,039,531	<i>,</i> 0	
Amount required in excess to site availability	ব	0	⁻ 0	´0	- 0	0	0	
Consolidation						•		_
Projected usage with consolidated facility	425	204	416,500	67	9,372,800	21.039.531	- 0	1.
Amount required in excess to site availability	<5	0	0	0	0	. 0	`	
Collocation					-	, –	•	
Projected usage with collocated facilities	425	204	437,500	76	9,372,800	21.039.531	0	
Amount required in excess to site availability	<5.	0	 	· · · · ·	ř , 0	0	÷ 0.	
Phaseout				N 1 1		- ,	•	1
Projected usage with storage phaseout	[•] 420	204	345.500	58	9,334.800	21.039.531	0	÷
Amount required in excess to site availability	0	0	0	- 0	² ² ² ² ²		ů N	

, ,

4 30

ANT IS ANT ANT ANT STORE AND

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium subalternatives

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Construction to modify selected areas of the FMEF to accommodate existing quantities of Pu plus material relocated from RFETS and LANL would not impact the site infrastructure. Data for construction are presented in Appendix C. Operations impacts under this option would also be minimal. Requirements are within site capacities and would not impact the site infrastructure.

Since impacts associated with this option for storing all of the RFETS Pu and LANL Pu material at Hanford are minimal, storing only a portion of the RFETS Pu and LANL Pu material at Hanford would result in minimal impacts to the site infrastructures as well. Additional annual electrical energy requirements would be proportionately less than that required for storage of the full amount of RFETS Pu and LANL Pu material.

Construct New 200 West Area Facility for Plutonium Storage

Constructing a new storage facility to accommodate existing quantities of Pu plus material relocated from RFETS and LANL would not impact the site infrastructure. Data for construction are presented in Appendix C. Operations impacts under this option would also be minimal. Requirements are within site capacities and would not impact the site infrastructure. Less than 5 km (3 mi) of access road would need to be constructed.

Since impacts associated with this option for storing all of the RFETS Pu and LANL Pu material at Hanford are minimal, storing only a portion of the RFETS Pu and LANL Pu material at Hanford would result in minimal impacts to the site infrastructures as well. Additional annual electrical energy requirements would be proportionately less than that required for storage of the full amount of RFETS Pu and LANL Pu material.

Consolidation Alternative

Construct New Plutonium Storage Facility

The site infrastructure impacts of constructing a consolidated Pu storage facility at Hanford would be minimal. Data for this construction are located in Appendix C. The site infrastructure impacts of operating a consolidated Pu storage facility at Hanford are shown in Table 4.2.1.2–1. The Hanford site infrastructure would be fully capable of supporting the operations of such a new facility without major modifications to the existing site infrastructure and utility resource requirements. Less than 5 km (3 mi) of access road would need to be constructed.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

The site infrastructure impacts of constructing consolidated Pu and collocated HEU storage facilities at Hanford would be minimal. Data for this construction are located in Appendix C. The site infrastructure impacts of operating consolidated Pu and collocated HEU storage facilities at Hanford are shown in Table 4.2.1.2–1. Hanford site infrastructure would be fully capable of supporting the operations of such new facilities without major modifications to the existing site infrastructure and utility resource requirements. Less than 5 km (3 mi) of access road would need to be constructed.

Subalternatives Not Including Strategic Reserve and Weapons Research and Development Materials

Since the existing Hanford site infrastructure would be fully capable of supporting construction/modification and operation of facilities for the Upgrade Alternative (both subalternatives), the Consolidation Alternative, and the Collocation Alternative, constructing and operating such facilities without including provisions for storage of strategic reserve and weapons R&D materials could be accommodated as well. Expected reductions in amounts of annual electrical energy requirements from those of the various storage alternatives for all the strategic reserve and weapons R&D materials are the only site infrastructure changes expected if this subalternatives is chosen because electric usage is dependent on the amount of material. [Text deleted.]

, 11,

.

1.1

٦۲

Phaseout

Phaseout of the Pu storage mission at Hanford would have no impact on the facilities and site infrastructure. While Pu storage operations would cease, the storage facilities would remain and utility service would continue at about the same usage level until D&D is accomplished.

4.2.1.3 Air Quality and Noise

Construction and operation activities associated with the No Action Alternative and the proposed storage alternatives would generate criteria and toxic/hazardous pollutants. To evaluate the air quality impacts at Hanford, criteria and toxic/hazardous concentrations from the No Action Alternative and the proposed storage alternatives are compared with Federal and State standards and guidelines. Impacts to human health from radiological and chemical airborne emissions are described in Section 4.2.1.9.

In general, all of the proposed storage facilities would emit the same types of air pollutants during construction. It is expected emissions would not exceed Federal, State, or local air quality regulations. PM_{10} and TSP concentrations will be increased, especially during peak construction periods.

The principal sources of emissions during construction include the following:

- Fugitive dust from land clearing, site preparation, excavation, and wind erosion of exposed ground surfaces
- Exhaust and road dust generated by construction equipment, vehicles delivering construction materials, and vehicles carrying construction workers

During operation, impacts from each of the individual storage facilities with respect to the concentrations of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Table 4.2.1.3–1 presents the estimated pollutant concentrations for each of the proposed storage alternatives, indicating little difference between alternatives with respect to impacts to air quality.

Emission rates attributed to operation of the proposed storage facilities are presented in Tables F 1.3–1 to F.1.3–3. [Text deleted.] Air pollutant emission sources associated with operations include the following:

- Operation of boilers for space heating
- Operation of diesel generators and periodic testing of emergency diesel generators
- Exhaust and road dust generated by vehicles delivering supplies and bringing employees to work
- Toxic/hazardous pollutant emissions from facility processes

Noise impacts during either construction or operation are expected to be low. Air quality and noise impacts for each storage alternative are described separately. Supporting data for the air quality and noise analyses are presented in Appendix F.

AIR QUALITY

An analysis was conducted of the potential impacts on air quality from emissions from each of the storage alternatives described in Section 2.2.1.

Section 176 (c) of the 1990 CAA Amendments requires that all Federal actions conform with the applicable SIP. The EPA has implemented rules that establish the criteria and procedures governing the determination of conformity for all Federal actions in nonattainment and maintenance areas. These are discussed in Section 4.1.3. The attainment status of the area in which Hanford is located is discussed in Section 3.2.3. Since the area is considered to be an attainment area for the criteria pollutants, the proposed actions at this site do not require that a conformity analysis be performed.

Pollutant	Averaging Time	Most Stringent g Regulations or No Guidelines ^a Action (μg/m ³) (μg/m ³)		Upgrade (μg/m ³)	Consolidation (µg/m ³)	Collocation (µg/m ³)	n	
Criteria Pollutants								
Carbon monoxide	8-hour	10,000 ^b	0.08		0.09	0.17	0.17	
	1-hour	40,000 ^b	0.3		0.37	1.04	1.04	
Lead	Calendar Quarter	1.5 ^b	<0.01		<0.01	<0.01	<0.01	
	24-hour	0.5 ^c	<0.01	1	<0.01	<0.01	<0.01	ĩ
Nitrogen dioxide	Annual	100 ^b	0.03	ł	0.03	0.04	0.04	1
Ozone	1-hour	235 ^b	đ		đ	d	đ	
Particulate matter less than or equal to 10 microns in	Annual	50 ^b	<0.01		<0.01	<0.01	<0.01	
diameter	Of hour	trob	0.02		0.02	0.00	0.02	
9	24-nour	150°	0.02	,-	0.02	0.02	0.02	
Sulfur dioxide	Annual	, 52 ⁻	<0.01		<0.01	<0.01	<0.01	
	24-nour	200 ⁻	<0.01		<0.01	<0.01	0.01	
	3-nour	1,300°	0.01	,	0.01	0.01	0.11	
م بر ۲۰	I-nour	1,018	0.02		0.02	0.22	0.22	
	I-nour	022,0	0.02		0.02	0.22	0.22	*
Mandated by Washington	ن د. د						0.04	÷.
Total suspended	Annual	60°	<0.01		<0.01	<0.01	<0.01	2
	24-hour	150°	0.02 f		0.02 f	0.02	0.02 f	
Gaseous fluorides	30-day	0.80	f		f	f	ŕ	
	7-day	1.7	r f		f	£ ,	ŕ	
the state of the s	24-hour	2.9°	1 5		i F	r -	r E	
, , , , , , , , , , , , , , , , , ,	12-hour	3.7°	1		I	I	1. 1	
Hazardous and Other Toxic Compounds			ب ۵ ۶		х 7 °, х	s ~ , 12	۴ ,	
Ammonia	24-hour	100 ^c	<0.01		<0.01	<0.01	<0.01	
Chlorine	24-hour	5 ^c	f	-	f	<0.01 ^g	<0.01 ^g	

Table 4.2.1.3–1.Estimated Operational Concentrations of Pollutants at Hanford Site and
Comparison With Most Stringent Regulations or Guidelines—No Action (2005)
and Storage Alternatives

4-34

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

950 .

•

		3				
Pollutant	Averaging Time	Most Stringent Regulations or Guidelines ^a (μg/m ³)	No Action (μg/m ³)	Upgrade (µg/m ³)	Consolidation (µg/m ³)	Collocation (µg/m ³)
Hydrogen chloride	24-hour	7°	f	f	<0.01 ^g	<0.01 ^g
Hydrazine	Annual	0.0002 ^c	f	f	<0.0001 ^g	<0.00001 ^g
Nitric acid	24-hour	17°	f	f	<0.01 ^g	<0.01 ^g
Phosphoric acid	24-hour	3.3°	f	f	< 0.01 ^g	<0.01 ^g
Sulfuric acid	24-hour	3.3°	f	f	<0 01 ^g	<0.01 ^g

Table 4.2.1.3–1.Estimated Operational Concentrations of Pollutants at Hanford Site and
Comparison With Most Stringent Regulations or Guidelines—No Action (2005)
and Storage Alternatives—Continued

^a The more stringent of the Federal and State standard is presented if both exist for the averaging time.

^b Federal and State standard

^c State standard or guideline.

^d Ozone, as a criteria pollutant, is not directly emitted or monitored by the candidate site See Section 4.1.3 for a discussion of ozone-related issues.

^e The standard is not to be exceeded more than twice in any seven consecutive days

f No sources of this pollutant have been identified

^g The concentration represents the alternative contribution only.

Note Concentrations are based on site contribution, including concentrations from ongoing activities (No Action), and do not include the contribution from non-facility sources (for example, traffic).

Source. 40 CFR 50; DOE 1996e, DOE 1996f; HF 1995a 1, HF DOE 1996a; WA Ecology 1994a.

Preferred Alternative: No Action Alternative

This alternative utilizes estimated air emissions data from total site operations at Hanford assuming continuation of site missions as described in Section 3.2. These data reflect conservative estimates of criteria and toxic/hazardous emissions at Hanford. The emission rates for the criteria and toxic/hazardous pollutants for No Action are presented in Table F.1.2.2–1. Table 4.2.1.3–1 presents the No Action concentrations. Increased PM₁₀ and TSP concentrations may occur due to ongoing construction associated with other activities (that are outside of the scope of the PEIS) under the No Action Alternative. Concentrations of all other criteria and toxic/hazardous air pollutants at the site boundary or public access highways are expected to remain within applicable Federal, State, and local ambient air quality standards.

Upgrade Alternative

Upgrade Without Rocky Flats Environmental Technology Site or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

It is expected that pollutant concentrations will remain within applicable Federal and State ambient air quality standards during modification of the FMEF.

During operation, concentrations of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Estimated pollutant concentrations attributable to increased operations associated with this storage alternative, plus the No Action concentrations, are presented in Table 4.2.1.3-1. Concentrations of air pollutants are expected to be the same with or without the RFETS or LANL material.

Construct New 200 West Area Facility for Plutonium Storage

In addition to the types of sources of emissions during construction associated with the No Action Alternative and the storage upgrade alternatives, fugitive dust resulting from the operation of a concrete batch plant would be an additional emission source associated with construction of a new facility.

Increased PM_{10} and TSP concentrations may occur during the peak construction period, and during dry and windy conditions. Appropriate control measures would be followed to minimize pollutant concentrations during construction. Concentrations of all pollutants at the site boundary or public access highways would remain within applicable Federal and State ambient air quality standards during construction.

During operation, concentrations of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Estimated pollutant concentrations attributable to increased operations associated with this storage alternative, plus the No Action concentrations, are presented in Table 4.2.1.3–1. Concentrations of air pollutants are expected to be similar with or without the RFETS or LANL material since pollutant emissions are the result of combustion of fossil fuels for heating the facility.

Upgrade With All or Some Rocky Flats Environmental Technology Site and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Air quality impacts for construction and operation for this option are expected to be similar to those previously described for modifying the FMEF.

Construct New 200 West Area Facility for Plutonium Storage

Air quality impacts for construction and operation for this option are expected to be similar to those previously described for the new Pu storage facility.

Consolidation Alternative

بۇ ي

Construct New Plutonium Storage Facility

Air quality impacts for construction and operation for this option are expected to be similar to those for the options previously discussed for the Upgrade Alternative.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

The collocation of Pu and HEU materials facilities would be located in the same area as the consolidated Pu materials facilities and would have similar air quality impacts, with the following exceptions.

During operation, emissions would be slightly higher, as shown in Appendix F. Concentrations of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Estimated pollutant concentrations attributable to increased operations associated with this storage alternative and No Action are presented in Table 4.2.1.3–1.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Air quality impacts for construction and operation for this subalternative are expected to be similar to those previously described for the Upgrade With All or Some RFETS and LANL Pu Subalternative, the Consolidation Alternative, and the Collocation Alternative. Since the same facility or a slightly smaller facility would be constructed or upgraded, the construction activities would be expected to have the same or less emissions. Storage of a smaller quantity of material would be expected to result in less emissions during operation of the facility as a result of reduced combustion emissions, laboratory operations, and other activities. [Text deleted.]

Phaseout

Phaseout of existing Pu inventories as a result of consolidating Pu at another site is expected to result in a small reduction in air pollutant concentrations from the No Action concentrations and would be in compliance with Federal and State standards. Some emissions may occur as a result of transporting materials from Hanford. Quantity of emissions is dependent on transportation mode.

NOISE

The location of the storage facilities relative to the site boundary and sensitive receptors was examined to evaluate the potential for onsite and offsite noise impacts. Noise sources during construction may include heavy construction equipment and increased traffic. Increased traffic would occur onsite and along offsite local and regional transportation routes used to bring construction material and workers to the site.

Preferred Alternative: No Action Alternative

Nontraffic noise sources associated with continued interim storage (No Action) and other ongoing missions are the same as described in Chapter 3. The continuation of operations at Hanford would result in no appreciable change in traffic noise and onsite operational noise sources from current levels. Nontraffic noise sources are

the transf

5

located at sufficient distance from offsite areas that the contribution to offsite noise levels would continue to be small. Due to the size of the site, noise emissions from construction equipment and operations activities would not be expected to cause annoyance to the public. Some noise sources may be located close enough to onsite noise sensitive areas to result in impacts, such as disturbances of wildlife.

Upgrade, Consolidation, and Collocation Alternatives

Nontraffic noise sources associated with the storage alternatives would be similar to those for existing facilities as discussed in Chapter 3. Nontraffic, operational noise sources associated with the alternatives include new or existing equipment and machines (cooling systems, vents, motors, and material handling equipment). These noise sources would be located at sufficient distance from offsite areas that the contribution to offsite noise levels would be small. Due to the size of the site, noise emissions from construction equipment and operations activities would not be expected to cause annoyance to the public. Some noise sources may result in impacts, such as disturbance of wildlife.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Noise impacts for construction and operations for this option are expected to be almost the same to those previously described for the Upgrade With All or Some RFETS and LANL Pu Subalternative, the Consolidation Alternative, and the Collocation Alternative because noise impacts are based on the use of the facility and not the size. [Text deleted.]

Phaseout

A reduction in noise levels associated with facility operations may result from the phaseout of storage facilities.

1 -

4.2.1.4 Water Resources

Construction and operation of the potential long-term storage facilities at Hanford would affect water resources. All facility options (in either the 400 or 200 Areas) are above the 100-year, 500-year, probable maximum flood (40,000 m³/s [1.4 million ft³/s]), flooding from dam failures, and flooding from a landslide resulting in river blockage. At Hanford, surface water resources, primarily the Columbia River, would be used to meet all construction and operation water requirements for facilities located in the vicinity of the 200 Area. The Columbia River has sufficient flow to support any of the alternatives. No construction- or operation-related impacts would exceed 1.1 percent of the Columbia River's average flow. Groundwater would be used to meet water requirements for facilities located in the 400 Area. During construction and operation of the facilities, treated wastewater would continue to be discharged in compliance with NPDES permit requirements, to infiltration ponds in the 200 Area, or nearby streams, or would be recycled at newly constructed wastewater treatment facilities. Stormwater runoff would be collected and treated, if necessary, before discharge to natural drainage channels in accordance with permit requirements. [Text deleted.]

Minimal impacts to groundwater are anticipated because no direct discharges would occur during construction and operation. Table 4.2.1.4–1 presents No Action water resources uses and discharges and the potential changes to water resources at Hanford resulting from the long-term storage alternatives.

Preferred Alternative: No Action Alternative

Surface Water. [Text deleted.] A description of the activities that would continue at Hanford is provided in Section 3.2. Under this alternative, surface water withdrawals from the Columbia River are not expected to increase from the current usage of 13,511 million l/yr (3,569 million gal/yr) by 2005. Treated wastewater discharged to infiltration/evaporation ponds is expected to remain at 246 million l/yr (65 million gal/yr). Under this alternative, current restoration programs would continue, and water quality is anticipated to improve.

Groundwater. Under this alternative, no additional impacts to groundwater resources are anticipated. Withdrawals from current operations in the 400 Area (195 million l/yr [51.6 million gal/yr]) are not anticipated to increase by 2005.

Upgrade Alternative

山へい

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Surface Water. There are no unique construction characteristics associated with water requirements and discharges from the modify FMEF option. Since the facilities are located in the 400 Area, no surface water would be withdrawn for any modification or operation activities. Groundwater from the unconfined aquifer would be used to meet water requirements. Since upgrades will take place in an existing facility, no impact to surface water would result from soil erosion of disturbed land and siltation of surface drainage channels during modifications. During operation, stormwater runoff would be collected and treated, if necessary, before discharge to natural drainage channels.

During modification of selected areas of the FMEF, sanitary wastewater (approximately 3.9 million l/yr [1.0 million gal/yr]) would be generated and discharged to the existing wastewater treatment systems at the 400 Area. This would cause a 1.6-percent increase in the effluent discharged at Hanford. During operation, wastewater would be discharged to infiltration/evaporation ponds. [Text deleted.]

			Upg	rade				
		Without RFETS or LANL Pu Material		With RFETS and LANL Pu Material				
Affected Resource Indicator	No Action	Modify FMEF	New Storage Facility	Modify FMEF	New Storage Facility	Consolidation	Collocation	Phaseout
Water Source	Surface/ Ground	Ground	Surface	Ground	Surface	Surface	Surface	Surface
Construction								
Water Availability and Use								
Total water requirement (million l/yr)	NA ^a	5.0	5.0	7.8	7.8	85	105	0
Percent increase in projected water use ^b	NA ^{a'}	2.6	0.04	4.0	0.06	0.6	0.8	0
W-4- 0	L							
water Quality		,		·		7 f f *24 (*	4 j 2 2 2	\$
(million l/yr)	24	3.9	3.9	5.9 <i>"</i>	5.9	7.7	12.5	· · · 0
Percent change in wastewater discharge ^c	NA ^a	1.6	1.6	, 3.0	0.04	ar 3.1 3.1	· . 5.1. · ·	, 0
many marks in a second s	4 × 1	7	т. (з	i v 1	, en 🧳 👻	et s in the to	•	t in t
Water Availability and Use	· · · ·		·· · · · · · · · · · · · · · · · · · ·		. 1 4		٣	
Total water requirement	13,511/195	8.4	8.4	8.9	8.9	Í10	150	0
Dement increases in projected			, , , , , , , , , , , , , , , , , , ,	, , <u>,</u> ,	, ,			
water use ^d a strong to t	V/U	• 4.3 • • • • • • • • • •	0.06	4.6	0.07	0.8	1.1 ^{***} -**	···· · O **
Water Quality		•						,
Total wastewater discharge (million l/yr)	246	~ 0	Ō	0	0	0	0	0
Percent change in wastewater discharge ^e	· 0	0	0 *	0	0	0 ,	, O	, 0
loodplain		- 1 7	، ۲. ۲.۴۰۰		بىر ج			
Is action in 100-year floodplain?	NA	No	No	No	No	No	No	No

Table 4.2.1.4–1. No Action and Potential Changes to Water Resources at Hanford Site—No Action (2005) and Storage Alternatives

4 1 , • t, 7 4.4 4 113

is denom sin

a the first and the main and the main and the second state of the second state of the second state and the second state

			Upg	rade				
		Without RFETS or LANL Pu Material		With RFETS and LANL Pu Material				
Affected Resource Indicator	No Action	Modify FMEF	New Storage Facility	Modify FMEF	New Storage Facility	Consolidation	Collocation	Phaseout
Is critical action in 500-year floodplain?	NA	No	No	No	No	No	No	No

 Table 4.2.1.4–1.
 No Action and Potential Changes to Water Resources at Hanford Site—No Action (2005) and

 Storage Alternatives—Continued

^a See operations section of table for No Action water data.

^b Percent increases in water requirements for construction at Hanford are calculated by dividing No Action water requirements (13,511 million l/yr for surface water and 195 million l/yr for groundwater) with that for each storage alternative: Modify FMEF without RFETS or LANL Pu material (5.0 million l/yr), new Pu storage facility without RFETS or LANL Pu material (5.0 million l/yr), Modify FMEF with all RFETS and LANL Pu material (7.8 million l/yr), new Pu storage facility with all RFETS and LANL Pu material (7.8 million l/yr), new Pu storage facility (85 million l/yr), new Pu and HEU storage facility (105 million l/yr), and storage phaseout (0 l/yr).

^c Percent changes in wastewater discharged during construction at Hanford are calculated by dividing No Action wastewater discharges (246 million l/yr) with that for each storage alternative: modify FMEF without RFETS or LANL Pu material (3.9 million l/yr), new Pu storage facility without RFETS or LANL Pu material (3.9 million l/yr), new Pu storage facility with all RFETS and LANL Pu material (5.9 million l/yr), new Pu storage facility with all RFETS and LANL Pu material (5.9 million l/yr), new Pu storage facility with all RFETS and LANL Pu material (5.9 million l/yr), new Pu storage facility (7.7 million l/yr), new Pu and HEU storage facility (12.5 million l/yr), and storage phaseout (0 l/yr [0 gal/yr]).

^d Percent increases in water requirements for operation at Hanford are calculated by dividing No Action water requirements (13,511 million l/yr for surface water and 195 million l/yr for groundwater) with that for each storage alternative: modify FMEF without RFETS or LANL material (8.4 million l/yr), new Pu storage facility without RFETS or LANL material (8.4 million l/yr), modify FMEF with all RFETS and LANL Pu material (8.8 million l/yr), new Pu storage facility with all RFETS and LANL Pu material (8.9 million l/yr), new Pu storage facility (110 million l/yr), new Pu and HEU storage facility (146 million l/yr), and storage phaseout (0 l/yr).

^c Percent changes in wastewater discharged during operation at Hanford are calculated by dividing No Action wastewater discharges (246 million l/yr) with that for each storage alternative: modify FMEF without RFETS or LANL Pu material (0 million l/yr), new Pu storage facility without RFETS or LANL Pu material (0 million l/yr), modify FMEF with all RFETS and LANL Pu material (0 million l/yr), new Pu storage facility with all RFETS and LANL Pu material (0 million l/yr), new Pu and HEU storage facility (0 l/yr), and storage phaseout (0 l/yr).

Note: NA=not applicable. During operation of the new facilities, all wastewater will be recycled; construction impacts are considered to be temporary, lasting only throughout the construction period. Impacts from operations occur continuously.

Source: DOE 1996e; DOE 1996f; HF 1995a:1; HF DOE 1995e:1; HF DOE 1996a.

1

al makes -

Fire sprinkler water and truck hose-down water would be collected in tanks, monitored for radioactivity, and if uncontaminated, discharged to storm drains that discharge to local drainage channels. If contaminated, this water would be treated as required.

The FMEF is located in the 400 Area above the floodplain from the probable maximum flood of 40,000 m³/s (1.4 million ft³/s), which is greater than the 500-year flood. The possibility of flooding from dam failures with a flood wave of 600,000 m³/s (21 million ft³/s) has been studied by the COE. In addition to the areas inundated by the probable maximum flood, the remainder of the 100 Area, the 300 Area, and nearly all of Richland, but not the 400 Area, would be flooded. A landslide resulting in river blockage downstream of the 400 Area, and flooding along the Columbia River during a river flood flow of 17,000 m³/s (600,000 ft³/s), would not inundate the FMEF. Additionally, it is unlikely that the landslide would be downstream.

Groundwater. During modification activities, the quantity of water required would be approximately 5.0 million l/yr (1.3 million gal/yr), which would represent a 2.6-percent increase over the projected No Action groundwater withdrawal (195 million l/yr [52 million gal/yr]). During operation, groundwater would be obtained from existing supply systems in the 400 Area. The total annual requirement for the modified FMEF would be 8.4 million l/yr (2.2 million gal/yr), which would represent a 4.3-percent increase over the projected groundwater withdrawal (195 million l/yr [52 million gal/yr]). It is not expected that these small increases would impact regional groundwater levels.

No wastewater would be discharged directly to groundwater, so groundwater quality would not be affected. However, some of the treated wastewater discharged to evaporation/percolation ponds would percolate downward into the groundwater. The water discharged to and from the ponds would be monitored and would not be discharged until contaminant levels were within the limits specified. Impacts to groundwater quality are therefore not expected. In addition, other factors contributing to a lessening of potential impacts to groundwater are the combined effects of a deep water table, low discharge volumes, and high evaporation rates.

Similarly, some stormwater runoff and other discharges routed to storm drains could percolate into the subsurface. Storm sewer and storm drain discharges would be monitored under the NPDES stormwater regulations. No impacts to groundwater quality are expected.

Construct New 200 West Area Facility for Plutonium Storage

Surface Water. Because the new Hanford Pu storage facility would be located in the 200 West Area, surface water would be used to meet water requirements. During construction, approximately 5 million l/yr (1.3 million gal/yr) of water would be required. This represents a much less than 1-percent increase in the projected No Action surface water withdrawal. This additional withdrawal would not cause any impacts. During operation, approximately 8.4 million l/yr (2.2 million gal/yr) of water would be required. This represents a much less than 1-percent increase in the projected annual surface water withdrawal, and it would increase Hanford's total withdrawal from the Columbia River to less than 1.0x10⁻⁶ of the river's average minimum flow. This would not cause any impacts to surface water availability.

During construction of the new Hanford Pu storage facility, sanitary wastewater (approximately 3.9 million l/yr [1.0 million gal/yr]) would be generated and discharged to the existing wastewater treatment systems at the 200 West Area. This would cause a 1.6-percent increase in the effluent discharged at Hanford. During operation, treated wastewater would be discharged to evaporation/infiltration ponds. [Text deleted.] All discharges would be monitored to comply with discharge requirements. Makeup water for the closed-cycle cooling system would be recycled.

The new facility would be located in the 200 Area, which is above the 100-year, 500-year, probable maximum floods, flooding from dam failures, and flooding from a landslide resulting from river blockage.

Groundwater. Because surface water would be used during construction and operation, no impact on groundwater availability is anticipated. No wastewater would be discharged from the ponds directly to groundwater, so groundwater quality would not be affected. However, some of the treated wastewater discharged to evaporation/percolation ponds could percolate downward into the groundwater. The water would be monitored and would not be discharged until contaminant levels were within the limits specified in the NPDES permit. Impacts to groundwater quality are therefore not expected. In addition, other factors contributing to a lessening of potential impacts to groundwater are the combined effects of a deep water table, low discharge volumes, and high evaporation rates. Similarly, some stormwater runoff routed to storm drains could percolate into the subsurface. These discharges would be monitored under the NPDES stormwater regulations. No impacts to groundwater quality are expected.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Modification activities would require 7.8 million l/yr (2.1 million gal/yr) of water, a 4.0-percent increase over the projected No Action water use. This is approximately 2.8 million l/yr (0.74 million gal/yr) of water more than that required for the Pu storage upgrade without RFETS Pu or LANL Pu material water requirements. During operations, 8.9 million l/yr (2.4 million gal/yr) of water would be required, a 4.6-percent increase over projected No Action water use. All other water requirements of the Pu storage upgrade with RFETS Pu and LANL Pu material are identical to the modified FMEF without RFETS Pu or LANL material.

Modifying FMEF to store RFETS Pu and LANL Pu material would increase water discharges by 5.9 million l/yr (1.6 million gal/yr) or 1.9 percent during construction activities over the projected No Action discharge. During operations, wastewater would be discharged to infiltration/evaporation ponds. All other wastewater requirements of the upgrade with RFETS Pu and LANL Pu material are similar to the modified FMEF without RFETS Pu or LANL Pu material.

Construct New 200 West Area Facility for Plutonium Storage

同時になるが、「日本の日本の日本」

During construction, the facility would require 7.8 million l/yr (2.1 million gal/yr), a much less than 1-percent increase over projected No Action water use. All other water requirements of the new Pu storage upgrade with RFETS Pu and LANL Pu material are identical to the new Hanford Pu facility without RFETS Pu or LANL Pu material. During operations, 8.9 million l/yr (2.4 million gal/yr) of water would be required. This represents a less than 1-percent increase in surface water withdrawal.

Water resources impacts during construction and operation with RFETS Pu and LANL Pu material would increase water discharges by 5.9 million l/yr (1.6 million gal/yr) or 1.9 percent of the projected No Action discharge. During operations, wastewater would be discharged to infiltration/evaporation ponds. All other wastewater discharges of the upgrade with RFETS Pu and LANL Pu material are the same as previously discussed for the new Hanford Pu storage facility without RFETS Pu or LANL Pu material.

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Consolidation Alternative

Construct New Plutonium Storage Facility

The new consolidated Pu storage facility would be located west of the 200 East Area of Hanford. Impacts associated with it are the same as those discussed above for the upgrade of the existing Pu storage area, with the following exceptions. The water requirements for construction and operation of this option are approximately 85 million l/yr (22.5 million gal/yr) and 110 million l/yr (29 million gal/yr), respectively. These additional requirements represent 0.6- and 0.8-percent increases, respectively, in the projected annual surface water withdrawals from the Columbia River and should not cause any impacts.

4.6 × P

1. 1

The quantity of sanitary wastewater generated during construction of this option would be approximately 7.7 million l/yr (2 million gal/yr). This represents a 3.1-percent increase in the projected annual wastewater effluent that would be discharged to evaporation/infiltration ponds. During operations, sanitary, utility, and process wastewaters would be recycled. No impacts to groundwater are expected from discharges. Groundwater would not be used for this alternative, so no impacts to groundwater availability or quality would be expected.

Collocation Alternative

- 1 m at 11

1.-

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

المرجع الم

÷ ,

1 1 1

F ()

These storage facilities would be located west of the 200 East Area of Hanford, and the impacts associated with them are the same as those discussed above, with the following exceptions. The water requirements for construction and operation of this option are greater, approximately 105 million l/yr (27.7 million gal/yr) and 150 million l/yr (39.6 million gal/yr), respectively. These additional requirements represent 0.8- and 1.1-percent increases, respectively, in the projected annual surface water withdrawals from the Columbia River and should not cause any impacts.

20. 1

The quantity of sanitary wastewater generated during construction of this option would be approximately 12.5 million l/yr (3.3 million gal/yr). This represents a 5.1-percent increase in the projected annual wastewater effluent that would be discharged to evaporation/infiltration ponds. During operations, sanitary, utility, and process wastewater would be recycled at newly constructed wastewater treatment systems. No impacts are expected. Groundwater would not be used for this alternative. Therefore, no impacts to groundwater availability or quality would be expected.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Water resource impacts for construction and operation of this subalternative are expected to be slightly less than those for the Upgrade With All or Some RFETS Pu and LANL Pu, the Pu Consolidated, and the Pu and HEU Collocation Storage Alternatives at Hanford described previously because of the reduction in the amount of material. [Text deleted.]

Phaseout

Should the current Pu storage mission at Hanford be phased out, surface water withdrawals from the Columbia River and nonhazardous wastewater discharge to evaporation/percolation ponds would decrease by negligible quantities. No noticeable impacts would occur or be alleviated due to these decreases.

[Text deleted.]

4.2.1.5 Geology and Soils

1 いっとってんかがしたい

Construction and operation of the alternatives at Hanford would have no effect on the geologic resources. A low seismic risk exists, but would be considered in the design of the proposed alternatives. The existing seismic risk does not preclude the safe construction and operation of the proposed alternative facilities. The facilities would be designed for earthquake-generated ground accelerations in accordance with DOE O 420.1, *Facility Safety*. Intensities of approximately V to VII on the MMI scale are possible in the general region. This could affect the integrity of poorly designed or nonreinforced structures, but should not affect newly designed facilities. Human health effects from accidents initiated by natural phenomenon (for example, earthquakes) are discussed in Section 4.2.1.9. A volcanic event of the Mt. St. Helens type could occur in the Cascade Mountain region to the west, resulting in a possible ashfall at Hanford. A recurrence of a similar event would not have an effect on the construction and operation of any of the proposed storage alternatives. It is highly unlikely that landslides, sinkhole development, or other nontectonic events would affect project activities. Slopes and underlying foundation materials are generally considered stable. Geologic resources at Hanford consist of crushed rock, sand, and gravel that have low economic value. New construction may increase the use of these materials; however, because large volumes of these materials are present, the impact is anticipated to be negligible.

Impacts to the geologic and soil resource occur during, or as a result of, ground-disturbing construction activities. Construction of the alternatives may involve ground disturbing activities that could affect the soil resources. The amount of land disturbed is specified below for each alternative. Impacts would depend on the specific soil units in the disturbed area, the extent of land disturbing activities, and the amount of soil disturbed. Control measures would be employed during construction activities to minimize soil erosion. [Text deleted.]

[Text deleted.]

Preferred Alternative: No Action Alternative

Under the No Action Alternative, DOE would continue current and ongoing activities at Hanford. There would be no ground-disturbing activities beyond those associated with existing and future site improvements. Because neither new construction nor the associated ground disturbance for potential soil erosion would occur, the No Action Alternative would have no effect on the geologic or soil resources at the site.

Upgrade Alternative

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

No apparent direct or indirect effects on the geologic resources are anticipated because neither facility construction and operational activities nor site infrastructure improvements will limit access to potential geologic resources. Design of the facilities would preclude potential impacts by any potential hazardous geological conditions.

Construction activities will occur completely within FMEF protected area, using existing gravel areas for construction laydown. Modification of the existing FMEF for Pu storage may disturb as much as 6.3 ha (15.5 acres) from construction activities, which would affect the soil profile and lead to a possible increase in soil erosion as s result of stormwater runoff and wind action. Soil impacts during operation are expected to be minimal.

.

Construct New 200 West Area Facility for Plutonium Storage وديد الله الم

No apparent direct or indirect effects on the geologic resource are anticipated because neither facility construction and operational activities nor site infrastructure improvements will limit access to potential geologic resources. Design of the facilities would preclude potential impacts by any potential hazardous geological conditions. Ŧ · L .

,

٦, · · · · · · · 5 A. C. A. S. C. 27 Construction activities will occur completely within PFP protected area and use existing gravel areas for construction laydown. Construction of the new Pu storage facility will involve ground-disturbing construction activities (approximately 10.5 ha [26 acres]) that will affect the soil profile and potentially cause a temporary increase in soil erosion. Construction activities (foundation preparation) and associated building construction laydown areas can expose the soil profile and lead to a possible increase in soil erosion as a result of wind and water action. Soil loss would depend on the frequency and severity of rain, wind velocities (increases in wind velocities and durations increase potential soil erosion), and the size, location, and duration of grounddisturbing activities.

, Net soil disturbance during operations would be considerably less than during construction because areas temporarily used for construction laydown would be restored. Although stormwater runoff and wind action could occur during operation, they are anticipated to be minimal. [Text deleted.]

....

- . . - ·

ta e see the e

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

۰ I

The construction and operation land-use requirements to modify the FMEF are not affected by the inclusion of a RFETS and LANL Pu material. Therefore, the impacts would be similar to those discussed for the modify FMEF without RFETS or LANL Pu material. ·

Construct New 200 West Area Facility for Plutonium Storage

The construction and operation land use requirements for the new Hanford Pu storage facility are not affected by the inclusion or exclusion of RFETS and LANL Pu material. Therefore, the impacts would be similar to those discussed for the new 200 West Area Pu storage facility without RFETS or LANL Pu material.

Consolidation Alternative

Construct New Plutonium Storage Facility

Construction of the new Pu storage facility will occur completely on previously undisturbed land as defined by Section 4.1.1 in a location west of the 200 East Area. Implementation of the Consolidation Alternative will involve ground-disturbing activities (58.5 ha [144 acres]) that will affect the soil profile and potentially cause a temporary increase in soil erosion. No apparent direct or indirect effects on the geologic resource are anticipated because neither facility construction and operational activities nor site infrastructure improvements will limit access to potential geologic resources. Analysis in this section is the same as that provided for the new 200 West Area Pu storage facility without RFETS or LANL Pu material.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Construction and operation effects on geologic and soil resources for the Collocation Alternative would be similar to those described for the Consolidation Alternative. However, additional soil impacts would be expected. Construction activities would occur completely on undisturbed land (west of the 200 East Area) as defined by Section 4.1.1 and involve approximately 89.5 ha (221 acres) of land disturbance for the new facilities, affecting the soil profile and leading to a possible temporary increase in erosion as a result of stormwater runoff and wind action. Soil impacts during operation are expected to be minimal.

No apparent direct or indirect effects on the geologic resource are anticipated, because neither facility construction and operational activities nor site infrastructure improvements will restrict access to potential geologic resources.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Exclusion of strategic reserve and weapons R&D materials would give almost the same effects to the geologic and soil resources for the Pu Subalternative, the Consolidation Alternative, and the Collocation Alternative. By excluding these materials, the size of a facility would be similar, thus not changing the amount of land disturbed by construction activities. No effect to the geologic resource is anticipated as a result of this subalternative. [Text deleted.]

Phaseout

時間で設定の設すが

The phaseout of storage capacity would have no apparent effects on the geologic resources. However, phaseout could result in beneficial effects on the soils of the area. Hazardous, radioactive, and mixed waste sources would be eliminated from the area, thus decreasing the potential for future soil contamination.

[Text deleted.]

4.2.1.6 Biological Resources

Preferred Alternative: No Action Alternative

Under No Action, the Pu storage mission described in Section 2.2.1 would continue at Hanford. These activities would result in no appreciable change to current conditions of biological resources at Hanford as described in Section 3.2.6. · · · ار · · · ·

- . . I

Upgrade Alternative

1

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Modifying selected areas of the FMEF within the 400 Area at Hanford would cause minimal disturbance to biological resources. This is because all activities would involve existing structures and would take place within an area that is currently disturbed. Noise associated with modifying the FMEF could cause some temporary disturbance to wildlife, but this impact would be minimal because animals living adjacent to the current facility have already adapted to its presence. Water withdrawal would be through wells and would involve relatively minor volumes, so wetlands and aquatic resources would not be affected. Wastewater would be discharged to evaporation/infiltration ponds. Since the upgrade would take place within a developed area, impacts to threatened and endangered species would not be expected.

Construct New 200 West Area Facility for Plutonium Storage

A new Pu storage facility would be constructed within the protected area of the PFP in the 200 West Area of Hanford. Although new construction would be involved in this option, it will take place within an area of the 200 West Area that is currently disturbed. Impacts to biological resources would be expected to be minimal and similar to those described above for the modification of the FMEF.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

the state of the second Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Upgrading with all or some RFETS and LANL materials would not be expected to change impacts to biological resources from those described above for the modification of the FMEF.

Construct New 200 West Area Facility for Plutonium Storage

· ·

Upgrading with all or some RFETS and LANL materials would not be expected to change impacts to biological I resources from those described above for the new Pu storage facility.

- Contraction of the

Consolidation Alternative

· · · · · · · · Construct New Plutonium Storage Facility

- 11 . . Under this alternative, Pu would be consolidated in a new storage facility located adjacent to the 200 East Area. Impacts to terrestrial resources, wetlands, aquatic resources, and threatened and endangered species are discussed below.

4-48

Terrestrial Resources. Construction and operation of the consolidated Pu storage facility would disturb 58 5 ha (144 acres) of terrestrial habitat, or about 0.04 percent of Hanford. This includes areas on which plant facilities would be constructed, as well as areas revegetated following construction. Vegetation within the proposed location would be destroyed during land-clearing operations. The project site falls within the sagebrush/ cheatgrass or Sandberg's bluegrass community. Sagebrush communities are well represented on Hanford, but they are relatively uncommon regionally because of widespread conversion of shrub-steppe habitats to agriculture. Disturbed areas are generally recolonized by cheatgrass, a nonnative species, at the expense of native plants.

「「「「「「「」」」」

Construction of the Pu storage facility would affect animal populations. Less mobile animals within the project area, such as reptiles and small mammals, would not be expected to survive Construction activities and noise would cause larger mammals and birds in the construction and adjacent areas to move to similar habitat nearby If the area to which they moved was below its carrying capacity, these animals would be expected to survive. However, if the area was already supporting the maximum number of individuals, the additional animals would compete for limited resources, which could lead to habitat degradation and eventual loss of the excess population. Nests and young animals living within the proposed location may not survive. The location would be surveyed as necessary for the nests of migrating birds prior to construction. Areas disturbed by construction, but not occupied by facility structures, would be of minimal value to wildlife because they would be maintained as landscaped areas.

Activities associated with facility operations, such as noise and human presence, could affect wildlife living immediately adjacent to the Pu storage facility. These disturbances may cause some species to move from the area. Disturbance to wildlife living adjacent to the facility would be minimized by preventing workers from entering undisturbed areas. Salt drift generated by mechanical draft cooling systems would be minimal, so impacts are not expected.

Wetlands. Construction and operation of the Pu storage facility would not affect wetlands since no wetlands exist near the proposed location. Due to the relatively small amount of water required during both construction and operation, existing intake structures would be used. It would not be necessary to disturb wetlands along the Columbia River. Construction- and operation-related discharges would be directed to evaporation ponds and, thus, would not impact wetlands. All wastewater discharges would be treated, as necessary, to meet NPDES permit requirements.

Aquatic Resources. Construction of a Pu storage facility at Hanford would not impact aquatic resources since there are no surface water bodies near the proposed location. Water requirements during both construction and operation would be met by existing site sources. Since new intake structures would not be required, direct disturbance of aquatic resources in the Columbia River would not occur. Water withdrawal during both construction and operation would represent a small percentage of the Columbia River's average flow and would have little effect on the flow of the river. Flow-related impacts to aquatic resources from impingement and entrainment impacts would be minimal and unlikely to affect fish populations in the Columbia River. In compliance with the Anadromous Fish Conservation Act (16 USC 757a et seq.), populations of anadromous fish species would be to evaporation ponds, which would provide temporary aquatic habitat.

Threatened and Endangered Species. It is unlikely that federally listed threatened and endangered species would be affected by construction and operation of the Pu storage facility. However, this alternative would disturb 58.5 ha (144 acres) of sagebrush habitat. Sagebrush habitat is important nesting/breeding and foraging habitat for several State-listed and candidate species, such as the ferruginous hawk, loggerhead shrike, western burrowing owl, pygmy rabbit, western sage grouse, sage sparrow, and sage thrasher. Preactivity surveys would

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

16.51

be conducted as appropriate prior to construction to determine the presence of plant species or animal species in the area to be disturbed. Consultation with the USFWS and State agencies would be conducted at the sitespecific level, as appropriate.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Under this alternative, Pu would be stored with HEU inventories in a new collocated storage facility located in the 200 East Area. Construction and operation of collocated storage facilities at Hanford would have similar, but somewhat greater, effects on biological resources as those described for the consolidated storage facility alone. Construction of the collocated storage alternative would disturb 89.5 ha (221 acres) of habitat.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

The exclusion of strategic reserve and weapons R&D materials would have almost the same effects to the Upgrade With All or Some RFETS and LANL Pu Subalternative, the Consolidation Alternative, and the Collocation Alternative. The size of facility would be similar and would not result in the reduction of disturbed habitat and fewer facility modifications and the potential impacts to biological resources would be similar. [Text deleted.]

Phaseout

The phaseout of Pu storage facilities at Hanford is not expected to affect biological resources, although increased human activity could temporarily disturb some wildlife species in the vicinity of the site.

1.

n and set the first set of the se

4.2.1.7 Cultural and Paleontological Resources

Preferred Alternative: No Action Alternative

Under this alternative, DOE would continue the existing and planned missions at Hanford, which include continued storage of Pu material in the PFP in stabilized forms pursuant to DNFSB Recommendation 94-1. All inventory and evaluation of cultural resources at Hanford is conducted within the framework of the Hanford Cultural Resources Management Plan (PNL-6942 UC-600, June 1989), which has been in place since 1989. Any impacts to cultural or paleontological resources from these missions would be independent of the proposed action and would be addressed through separate NHPA, American Indian Religious Freedom Act, and Native American Graves Protection and Repatriation Act regulatory compliance procedures.

Upgrade Alternative

ないないで、ことになったいで、ことのできたのであった。

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

The FMEF is located in the 400 Area of Hanford in an existing protected area. This option involves only modification to selected areas of the existing facility. No new construction would be necessary. Because it was recently built, the FMEF itself is not NRHP eligible. The 400 Area has been surveyed, and no archaeological or historic resources were identified. Consequently, any land-disturbing activity associated with building modification (such as equipment staging areas and temporary roads) should not affect cultural resources. Similarly, operation does not involve increased activity or ground disturbance, so it would not result in impact.

Construct New 200 West Area Facility for Plutonium Storage

The new Pu storage facility would be constructed in the 200 West Area northwest of the PFP facility within the existing protected area. The total land required during construction and operation is 10.5 ha (26 acres). All land was previously disturbed, and there will be no construction on undisturbed land.

[Text deleted.]

ł

A non-systematic archaeological survey was conducted across 50 percent of the 200 West Area. No prehistoric or historic resources were identified, except for the White Bluffs Freight Road, which was used during both prehistoric and historic times. This portion of the road is a noncontributing element of the NRHP-listed resource, and has been given a buffer zone to protect it from development. The road is outside the proposed construction area and will not be affected. Depending on siting, construction, and operation, the new facility may affect the functional and historic setting of the PFP, which is an NRHP-eligible property. The PFP was constructed between 1947 and 1949 and was used to produce Pu metal during the Cold War Era. Some scientifically valuable Pliocene and Pleistocene paleontological deposits may also exist in the areas to be excavated during construction; although this is unlikely as previous construction activities did not reveal these kinds of resources. Archaeological and paleontological resources would not be affected by facility operation because operation does not involve additional ground disturbance or increased activity.

To date, no Native American groups have identified any areas of special concern in proximity to the 200 Areas. [Text deleted.] Operation may result in reduced access to traditional hunting, fishing, and gathering areas, or visual and auditory intrusion into sacred spaces. Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

and ded

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

<u>بر این ا</u>

The inclusion of RFETS and LANL material would not increase the total area requirement of the Upgrade Without RFETS or LANL Subalternative, therefore, no ground-breaking construction would be necessary. All materials could be accommodated within existing facilities. Consequently, construction and operation are not expected to affect cultural or paleontological resources.

Construct New 200 West Area Facility for Plutonium Storage

Total land disturbed by construction and operation would not increase with the inclusion of RFETS and LANL materials. Construction of a slightly larger Pu storage facility is not expected to have more of an effect on cultural or paleontological resources, as discussed under the new Pu storage facility option.

Consolidation Alternative

I

Construct New Plutonium Storage Facility

This alternative would involve the construction of a new facility west of the 200 East Area. During construction, 58.5 ha (144 acres) would be disturbed. The total land required for operation is 56 ha (138 acres). A 1.6-km (1 mi) reduced-access buffer zone-exists and would be maintained around the facility. Pu storage in existing DOE storage facilities would be phased out.

The 200 areas have been surveyed, and no prehistoric or historic resources were identified. The area is previously disturbed. Some significant paleontological materials may occur within this acreage. The potential for impacts to paleontological resources is greatest during construction. Operation would not have an additional impact on resources, should any be identified during construction. As discussed under the new Hanford Pu storage facility upgrade option, Native American groups have not identified any resources in proximity to the 200 Areas. Additional consultation may be necessary for resource identification.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

This alternative would involve the construction of a new HEU storage facility to be collocated with a consolidated special nuclear material plant adjacent to the 200 East Area. Land required during operation would be 87 ha (215 acres). Construction of this facility is expected to disturb 89.5 ha (221 acres). A 1.6-km (1-mi) reduced-access buffer zone exists and would be maintained around the facility. Pu and HEU storage in existing facilities would be phased out. Potential for impacts to these resources would be similar to that discussed under the previous Consolidation Alternative.

î

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Under this subalternative, facility and other resource requirements will be almost the same as the Upgrade With All or Some RFETS Pu and LANL Pu Subalternative, the Consolidation Alternative, and the Collocation Alternative. Therefore, impacts to cultural and paleontological resources would be equal to those previously discussed. [Text deleted.]

Phaseout

Impacts to archaeological resources are not anticipated because phaseout is not expected to result in groundbreaking activity. Likewise, no impacts to paleontological remains are expected. It may affect, through alteration, if subsequently proposed, some NRHP-eligible historic structures at Hanford. Impacts to Native American resources are not expected.

[Text deleted.]

4.2.1.8 Socioeconomics

Preferred Alternative: No Action Alternative

Under this alternative, the existing storage facility would remain operational. No new employment or inmigration of workers would be required.

Regional Economy Characteristics. Total employment in the REA is projected to increase by about 1.3 percent annually between 1995 and 2000, reaching 322,000 in the latter year. Long-range projections indicate slower growth after the year 2000, when employment will increase by less than 1 percent annually and reach 446,300 in 2040. Unemployment in the REA was 9.1 percent in 1994 and is expected to remain at this level into the near future. Per capita income is projected to increase from approximately \$18,996 in 1995 to \$28,079 in 2040. Projections for the No Action Alternative are presented in Table L.1–10.

Population and Housing. Population in the ROI is projected to increase from approximately 384,700 in 1995 to 568,600 by 2040. The total number of housing units in the ROI is projected to increase from approximately 140,900 to 208,200 during the same period. Population and housing projections for the No Action Alternative are presented in Tables L.1–11 and L.1–12, respectively.

Community Services. Education, public safety, and health care characteristics are used to assess the level of community services in the Hanford ROI. School enrollments are projected to increase from 76,891 students in 1995 to 113,659 students by 2040. To maintain the current student-to-teacher ratio of 18.9:1, the number of teachers in the ROI would need to increase from 4,077 in 1995 to 6,023 in 2040. These projections are presented in Tables L.1–13 and L.1–14, respectively.

The projected numbers of sworn police officers and firefighters serving in ROI communities over the period 1995 to 2000 are shown in Tables L.1–15 and L.1–16, respectively. Under No Action, the number of sworn police officers is projected to increase from 503 in 1995 to 742 in 2040 to maintain the current service level of 1.6 officers per 1,000 persons. The number of firefighters in the ROI would need to increase from 1,544 in 1995 to 2,281 in 2040 to maintain the current level of service of 4.0 firefighters per 1,000 persons.

Hospital occupancy rates are based on current capacity. Hospital occupancy rates and the estimated number of practicing physicians serving the ROI population between 1995 and 2040 are presented in Tables L.1–17 and L.1–18, respectively. Hospital occupancy rates for the ROI are projected to increase from 51 percent in 1995 to 75 percent in 2040. To maintain the current physician-to-population ratio of 1.2 physicians per 1,000 persons, the total number of physicians in the ROI would need to increase from 472 in 1995 to 696 in 2040.

Local Transportation. The worker population at Hanford would not increase. Therefore, any increases in traffic ... or air traffic would be due to the projected growth in the area unrelated to DOE activities. [Text deleted.]

Upgrade Alternative

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

;

[Text deleted.] A total of 54 workers would be employed during peak construction of the modified facility. During the operational phase, 225 workers would be required. Projections indicate that there would be sufficient available labor in the REA to fill both direct and indirect jobs generated as a result of construction and all indirect jobs generated by operation of the modified facility. Some workers would in-migrate to fill a portion of the direct jobs generated during operations. **Regional Economy Characteristics.** During peak construction, the project would add up to 108 (54 direct and 54 indirect) jobs to the regional economy. All of these new jobs would be filled by available labor force in the REA and unemployment would fall from the No Action level of 9.1 percent to 9.0 percent (Socio 1996a). Per capita income would remain virtually unchanged, increasing by much less than 1 percent over the No Action Alternative.

Operation of the facility without storage of RFETS or LANL material would generate a total of 759 jobs (225 direct and 534 indirect) in the REA. These additional jobs would reduce regional unemployment by much less than 1 percent from the No Action level. Per capita income would increase by much less than 1 percent in the year 2005, when the facility would become fully operational (Socio 1996a).

Population, Housing, and Community Services. During construction, all newly created jobs would be filled by the resident labor force. Therefore, there would be no change to the region's population, housing market, or demand for community services beyond the No Action projections. A small increase in population would occur during operation of the facility due to the in-migration of five workers. Accordingly, there would be an insignificant effect on the housing market and the demand for community services (Socio 1996a).

Local Transportation. During the peak construction period, 104 vehicle trips per day would be generated by workers involved in facility modification. This increase would not affect level of service on the road segments analyzed. During operations, the workers would generate 432 vehicles trips per day. This increase over the No Action level would not affect the level of service on the local road segments analyzed (Socio 1996a).

Construct New 200 West Area Facility for Plutonium Storage

2.

Work force requirements for the construction and operation of the new Pu storage facility are the same as for modification of the FMEF. Therefore, the magnitude of socioeconomic impacts for this option would be the same as those discussed above for the FMEF.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

A peak of 77 workers would be employed to modify the facility to store all of the RFETS and LANL material. During the operational phase, 252 workers would be required. Projections indicate that there would be sufficient labor available in the REA to fill all direct and indirect jobs generated by construction and all indirect jobs generated by operation of the modified facility. Some workers would in-migrate to fill a portion of the direct jobs generated by operation.

Regional Economy Characteristics. During peak construction, the project would generate 154 jobs (77 direct and 77 indirect) in the regional economy. All of these new jobs would be filled by available labor within the REA. Unemployment would fall from the No Action level of 9.1 percent to 9.0 percent (Socio 1996a). Per capita income would remain virtually unchanged, increasing by much less than one percent over the No Action Alternative.

Operation of the facility would generate a total of 850 (252 direct and 598 indirect) jobs. Regional unemployment would be reduced slightly from the No Action projection of 9.1 percent to 8.9 percent. Per capita income would increase by much less than 1 percent (Socio 1996a).

Population, Housing, and Community Services. All jobs generated by construction would be filled by the resident labor force. Therefore, there would be no change to the region's population from the No Action projections. Accordingly, there would be minimal effects on the housing market or demand for community

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

services. A small increase in population would occur during operation of the facility due to the in-migration of eight workers. Such an increase would have an insignificant effect on the housing market and the demand for community services (Socio 1996a).

a 14

If only a portion of the RFETS or LANL materials were transferred to Hanford, between 225 and 252 workers would be required to operate the facility. The exact number of workers would depend on the amount of material that would actually reside at Hanford. The size of the construction workforce would be between 54 and 77 workers in the peak year of construction. Between 108 and 154 jobs (direct and indirect) would be generated during construction while between 759 and 850 jobs (direct and indirect) would be generated during operations. There would be no changes to the ROI population over the No Action projections during construction, but could be some in-migration during operations. In all cases, the socioeconomic impacts to the region would be slight.

Local Transportation. During the peak construction period, 148 vehicle trips per day would be generated by workers involved in facility modification. During operations, workers would generate 484 vehicle trips per day. These increases would not affect the level of service on the local road segments analyzed. (Socio 1996a).

Construct New 200 West Area Facility for Plutonium Storage

Workforce requirements for the construction and operation of the new Pu storage facility are the same as for the modification of the FMEF. Therefore, the magnitude of socioeconomic impacts for this option would be the same as those discussed above for the FMEF.

Consolidation Alternative

Construct New Plutonium Storage Facility

To consolidate storage of Pu that is currently stored at multiple DOE sites, a new storage facility would need to be constructed at Hanford. A few workers would in-migrate to fill a portion of the direct jobs created during the operation of the new facility.

Regional Economy Characteristics. Construction would involve over 1,000 workers on site and add a total of 2,129 jobs (1,064 direct and 1,065 indirect) to the REA during the peak period of activity, an increase of less than 1 percent over the No Action level. All of these jobs would be filled by available labor in the REA. Unemployment would drop from 9.1 percent to about 8.5 percent. Per capita income would increase by much less than 1 percent in the peak year of construction (Socio 1996a).

a 1 a - 12

The operation of the facility would add a total of 1,495 jobs (443 direct and 1,052 indirect) to the regional economy, an increase of less than 1 percent over the No Action level. A small percentage of the direct workers would in-migrate to fill some specialized employment requirements. Operation workers would begin phasing in as construction nears completion. Unemployment would rise from 8.5 percent during peak construction to 8.7 percent during operation, but would remain lower than the No Action level of 9.1 percent. Per capita income would increase by much less than 1 percent over No Action (Socio 1996a).

Population, Housing, and Community Services. A small increase in population would occur during the operation phase due to the in-migration of 27 workers. Such an increase would have no effect on the housing sector and would have an insignificant effect on the demand for community services (Socio 1996a). 1 1 1 the also

Local Transportation. During the peak construction period, workers would generate 2,043 vehicle trips per day. This increase would not affect the level of service on the local road segments analyzed. During operations, workers would generate 851 vehicles trips per day, and the increase to roadway traffic would be less than during 4 2 construction (Socio 1996a). . · · · 1.00

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Construction of new storage facilities would be required in order to store Pu and HEU at Hanford. Workers would in-migrate to fill some of the direct jobs created during operation of the new storage facility. Construction employment would reach 1,076 during the peak period of activity. Operations would require 572 workers.

Regional Economy Characteristics. Construction of the new facility would generate a total of 2,153 jobs (1,076 direct and 1,077 indirect) during the peak construction year. The resident available labor force would be sufficient to fill all of the direct and indirect jobs created during the construction phase. Total employment in the REA would increase by less than 1 percent. Unemployment would decrease from 9.1 percent to 8.5 percent. Per capita income would increase by much less than 1 percent in the peak construction year (Socio 1996a).

Operation of the facility would produce a total of over 1,930 new jobs (572 direct and 1,358 indirect) within the REA. A majority of direct jobs and all of the indirect jobs generated would be filled by the resident labor force. Total employment in the REA would increase by less than 1 percent in the year 2005, when the facility would become fully operational. Operation workers would begin phasing in as construction nears completion. Unemployment would rise from 8.5 percent during peak construction to 8.6 percent during operation, but would remain lower than the No Action level of 9.1 percent. Per capita income is projected to increase by less than 1 percent (Socio 1996a).

Population and Housing. A small number of workers are projected to in-migrate during the operation period. The population increase would be negligible, and projected housing vacancies would be sufficient to accommodate the incremental population increase (Socio 1996a).

Community Services. The additional population would slightly increase demand for some community services. Worker in-migration would lead to an increase in ROI school enrollment by approximately 35 students during operation. In order to maintain the No Action student-to-teacher ratio, the number of teachers would have to increase by one (Socio 1996a). [Text deleted.]

In order to maintain the No Action level of service, two firefighters would need to be hired during the operational period. No additional police would be required to maintain the No Action level of service during the operational period (Socio 1996a).

The small population change would have a negligible effect on health services, increasing hospital occupancy by much less than 1 percent. The number of physicians in the ROI would be sufficient to maintain the No Action level of service.

Local Transportation. During the peak construction period, workers would generate 2,066 vehicle trips per day. This increase would not affect the level of service on the road segments analyzed. During operations, workers would generate 1,098 vehicles trips per day and the increase in roadway traffic would be less than during construction (Socio 1996a).

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

If strategic reserve and weapons R&D materials are not included in the storage requirements for Hanford, there would be a small reduction in worker requirements for construction and operation of the facility due to fewer, workers being needed. Therefore, the socioeconomic effects would be less than under those alternatives that include storage of nonsurplus RFETS and LANL material for the Upgrade With All or Some RFETS and LANL Pu Subalternative, the Consolidation Alternative, and the Collocation Alternative. [Text deleted.]

4–57

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Phaseout

Phaseout of the existing Pu storage facility at Hanford would result in the loss of 675 total jobs (200 direct and 475 indirect) in the REA. The total direct and indirect employment loss would be much less than a 1 percent reduction in the projected regional employment levels for the year 2005, when the phaseout would be implemented.

4 \$ 7.7-

٢

The follow

s

In the longer term, some unemployed workers may migrate out of the REA to seek new employment opportunities. Even if all of these workers were to leave the REA with their families, population would decrease by much less than 1 percent compared to No Action. The impact on housing and community services, including health care, education, and public safety would not be substantial. For example, there could be a slight increase in housing vacancies or a decline in new housing construction and also a small decrease in demand for educational and health services (for example, teachers and physicians). These minor impacts would be further reduced if the storage mission is phased out over more than 1 year.

Phaseout of the existing Pu storage facilities at Hanford would reduce the number of vehicle trips per day by 384. There would be no significant traffic impact to the local road network.

4.2.1.9 Public and Occupational Health and Safety

The assessments of potential radiological and chemical impacts associated with the storage alternatives at Hanford are presented in this section. Summaries of radiological impacts from normal operations are presented in Tables 4.2.1.9–1 and 4.2.1.9–2 for the public and workers, respectively. Impacts from hazardous chemicals are presented in Table 4.2.1.9–3. Summaries of impacts associated with postulated accidents are given in Tables 4.2.1.9–4 through 4.2.1.9–7. Detailed results are presented in Appendix M.

Preferred Alternative: No Action Alternative

This section describes the radiological and hazardous chemical releases and their associated impacts resulting from normal operations involving the Hanford sitewide missions, including interim storage of Pu. The impacts would be within applicable regulatory limits. For facility accidents, the risks and consequences are described in site safety documentation.

Normal Operation. The current mission at Hanford, where Pu is in interim storage, is described in Section 3.2. The site has identified those facilities that will continue to operate under the No Action Alternative, including interim Pu storage facilities and others, if any, that will become operational by 2005. Based on that information, the radiological and chemical releases to the environment in 2005 and beyond (future operation) were developed and used in the impact assessments. The resulting doses and potential health effects on the public and workers at Hanford are described below.

Under the No Action Alternative, Hanford would continue to store Pu-bearing materials in the storage vaults and approved vault-type rooms of the PFP. All Pu materials would be stabilized and repackaged, as necessary, to ensure safe storage. Activities supporting stabilization, repackaging, and storage of the Pu materials are identified and discussed in the DNFSB Recommendation 94-1 Hanford Site Integrated Stabilization Management Plan (VHC-EP-0853). This plan calls for transforming the Pu-bearing materials to a stable form that meets the DOE standard Criteria for Safe Storage of Pu Metals and Oxides (DOE-STD-3013-94) by 2002 for materials with greater than 50 percent Pu. Some PFP plant systems that are required to provide basic facility services would be upgraded for storage facility operations for the No Action Alternative.

Radiological Impacts. Under this alternative at Hanford, Table 4.2.1.9–1 presents the calculated annual dose to the average and MEI of the public, from total site operation, the projected fatal cancer risks to these individuals from 50 years of operation, the dose to the population within 80 km (50 mi) due to total site operation in the year 2030, and the projected number of fatal cancers in this population from 50 years of operation. The annual dose of 5.3×10^{-3} mrem to the MEI is within the radiological limits specified in NESHAPS (40 CFR 61, Subpart H) and DOE Order 5400.5. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 1.3×10^{-7} . The annual dose of 1.6 person-rem to the population of 621,000 would be within the limit in proposed 10 CFR 834. The corresponding number of fatal cancers in this population from 50 years of operation would be 0.039. To put operational doses into perspective, comparisons with natural background radiation doses are included in the table. The doses and projected fatal cancers associated with the storage component of the No Action Alternative are included in Table 4.2.1.9–1. These are seen to be much lower than those from total site operations.

Under the No Action Alternative shown in Table 4.2.1.9–2, the annual average dose to a noninvolved (No Action) site worker and the annual dose to the noninvolved (No Action) total site workforce would be 31 mrem and 296 person-rem, respectively. The associated risk of fatal cancer to the average worker from 50 years of total site operations would be 6.0×10^{-4} and the projected number of fatal cancers from 50 years of total site operations would be 5.1.

The annual average dose to a worker involved in No Action storage operations would be 250 mrem/year, with a total involved No Action workforce dose of 49 person-rem. The increased risk of latent cancer fatality to the

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

1

ALL ALL DATE OF THE PARTY AND A PARTY AND

				_				
,	No Action		Up	Upgrade		Consolidation		ocation
Receptor	Storage Facility	Total Site	Storage Facility	Total Site ^b	Storage Facility	e Total Site ^b	Storage Facility	e Total
Exposed Individual Member of the Public ^c								
Atmospheric release pathway (mrem)	4.1x10 ⁻⁴	4.4x10 ⁻³	1.8x10 ⁻⁶	4.1x10 ⁻	³ 2.5x10 ⁻⁶	4.1x10 ⁻²	³ 2.5x10 ⁻⁶	⁵ 4.1x10 ⁻³
Drinking water pathway (mrem) 0	0	0	0	0	0	0	
Total liquid release pathway (mrem)	0	9.5x10 ⁻⁴	0	9.5x10-4	0	9.5x10 ⁻⁴	0 1 0	0 9.5x10 ⁻⁴
Atmospheric and liquid release pathways combined (mrem)	4.1x10 ⁻⁴	5.3x10 ⁻³	1.8x10 ⁻⁶	5.1x10 ⁻³	2.5x10 ⁻⁶	5.1x10 ⁻³	2.5x10 ⁻⁶	5.1x10 ⁻³
Percent of natural background ^d	1.4x10 ⁻⁴	1.8x10 ⁻³	6.0x10 ⁻⁷	-1.7x10-3	8 3 + 10-7	1 7-10-3	0.0.107	
50-year fatal cancer risk	1.0x10 ⁻⁸	1.3x10 ⁻⁷	4.5×10-11	1 3×10-7	6 2 10-11	1,7,10 -7	8.3X10 ·	1.7×10^{-5}
Population Dose Within 80 Kilometers for Year 2030 ^e				5	0.2210	1.5X10 ·	6.2x10***	1.3x10-'
Atmospheric release pathway (person-rem)	0.047	0.46	4.7x10 ⁻⁵	0.41	1.1x10 ⁻⁴	0.41	1.1x10 ⁻⁴	0.41
Liquid release pathway (person-rem)	0	1.1	0	1.1	0	1.1	0	1.1
Atmospheric and liquid release pathways combined (person-rem)	0.047	1.6	4.7x10 ⁻⁵	1.5	1.1x10 ⁻⁴	1.5	1.1x10 ⁻⁴	1.5
Percent of natural background ^d	2.5x10 ⁻⁵	8.4x10 ⁻⁴	2 5x10-8	8 1~10-4	5 010-8			
50-year fatal cancers	1.2×10^{-3}	0.039	1.2 × 10-6	0.1710	5.9X10 °	8.1x10-	5.9x10°°	8.1x10 ⁻⁴
Annual Dose to the Average Individual Within 80 Kilometers ^f					2.8810	0.038	2.8x10 ⁻⁰	0.038
Atmospheric and liquid release pathways combined (mrem)	7.6x10 ⁻⁵	2.6x10 ⁻³	7.6x10 ⁻⁸ 2	2.4x10 ⁻³	1.8x10 ⁻⁷	2.4x10 ⁻³	1.8x10 ⁻⁷	2.4x10 ⁻³
50-year fatal cancer risk	1.9x10 ⁻⁹ é	5.0x10 ⁻⁸	l.9x10 ⁻¹² (5.0x10 ⁻⁸	4.4x10 ⁻¹²	6.0x10 ⁻⁸	4 4x 10-12	6 0~ 10-8
		_					TITAIU !	0.0710 -

Potential Radiological Impacts to the Public During Martin 10
Number of the stability
No Action and Storage Alternatives

The radiological impacts associated with the new Hanford Pu storage facility are smaller than for the modified FMEF (refer to the text). The impacts from total site operations are virtually the same with operations of either of the two storage alternatives. The radiological impacts are calculated based on measured releases from facilities at Hanford, RFETS, and LANL.

^b Includes impacts from No Action facilities. The location of the MEI may be different under No Action than for the other alternatives. Therefore, the impacts may not be directly additive.

^c The applicable radiological limits for an individual member of the public from total site operations are 10 mrem per year from the air pathways as required by the NESHAPS (40 CFR 61, Subpart H) under the CAA; 4 mrem per year from the drinking water pathway as required by the SDWA, and 100 mrem per year from all pathways combined. Refer to DOE Order 5400.5. [Text deleted.]

a

^d The annual natural background radiation level at Hanford is 300 mrem for the average individual; the population within 80 km in the year 2030 receives 186,400 person-rem.

^e For DOE activities, proposed 10 CFR 834 (see 58 FR 16268) would generally limit the potential annual population dose to 100 person-rem from all pathways combined, and would require an ALARA program. [Text deleted.]

Obtained by dividing the population dose by the number of people projected to live within 80 km of Hanford in 2030 (621,000). f Source: Section M.2.

	•			
Receptor	No Action ^a	Upgrade ^b	Consolidation ^b	Collocation ^b
Involved Workforce ^c				
Average worker dose (mrem/yr) ^d	250	250	258	264
50-year risk of fatal cancer	5.0x10 ⁻³	5.0x10 ⁻³ **	5.2x10 ⁻³	5.3x10 ⁻³
Total dose (person-rem/yr)	46	52	24	25
50-year fatal cancers	0.92	1.0	0.48	0.50
Noninvolved Workforce ^e				
Average worker dose (mrem/yr) ^d	27	27	27	27
50-year risk of fatal cancer	5.5x10 ⁻⁴	5.5x10 ⁻⁴	5.5x10 ⁻⁴	5.5x10 ⁻⁴
Total dose (person-rem/yr)	250	250	250	250
50-year fatal cancers	5.1	5.1	5.1	5.1
Total Site Workforce ^f				
Dose (person-rem/yr)	296	302	274	275
50-year fatal cancers	5.9	6.0	5.5	5.5
	•			

 Table 4.2.1.9–2.
 Potential Radiological Impacts to Workers During Normal Operation at Hanford Site—

 Storage Alternatives

^a It is expected that for the No Action Alternative the number of involved workers and average dose would be the same as for the Upgrade Without RFETS Pu or LANL Pu Subalternative. Additional workers would be required for the Upgrade With RFETS Pu and LANL Pu Subalternative. Therefore, the total worker dose would increase accordingly.

^b Under the Upgrade Alternative, 225 in-plant workers (of which 185 are badged with dosimeters to monitor radiation exposure) would be required to operate the storage facility, with an estimated additional 27 in-plant workers (22 badged) needed if Pu is transferred from the RFETS and LANL. The impacts given in the upgrade column include those associated with these additional workers. The number of involved badged workers for the Consolidation and Collocation Alternatives would be 92 and 95, respectively.

^c The involved worker is associated with operations of the proposed action. The maximum dose to an involved worker will be kept below 500 mrem per year. Based on a review of worker doses associated with similar operations (Section M 2.3.2), an average worker dose of 250 mrem per year was conservatively assumed. However, an effective ALARA program will ensure that exposure will be reduced to that level which is as low as reasonably achievable.

^d The radiological limit for an individual worker is 5,000 mrem/year (10 CFR 835). However, DOE has also established an administrative control level of 2,000 mrem per year (DOE 1992t); the site must make reasonable attempts to maintain worker doses below this level.

^e The noninvolved worker is onsite but not associated with operations of the proposed action. The projected number of noninvolved badged workers in 2005 and beyond is 9,300. The noninvolved workforce is equivalent to the No Action workforce.

^f The impact to the total site workforce is the summation of the involved worker impact and the noninvolved worker impact.

[Text deleted.]

「「「「「「「」」」」

Source: Section M.2.

	1				ł.	-	~
,	No Action	Upgrade		Consoli	dation	Collocation	
Receptor	Total Site ^a	Facility ^b .	Total Site ^a	, Facility ^b	Total Site ^a	Facility ^b	Total Site ^a
Maximally Exposed Individual (Public)		· ' ·	يڭ ۽	· , , , ,	-	3 y	• •
Hazard Index ^c	6.2x10 ⁻⁵	9.4x10 ⁻⁷	6.3x10 ⁻⁵	4.0x10 ⁻⁶	6.6x10 ⁻⁵	1.6x10 ⁻⁵	7.8x10 ⁻⁵
Cancer risk ^d	0	0.	0	· 2.7x10 ⁻⁸	,2.7x10 ⁻⁸	2.7x10 ⁻⁸	2.7x10 ⁻⁸
Worker Onsite		1	•	۶		τ.	
Hazard Index ^e	4.0x10 ⁻³	1.9x10 ⁻⁵	4.0x10 ⁻³	2.8x10 ⁻⁴	4.3x10 ⁻³	7.1x10⁻⁴	4.7x10 ⁻³
Cancer risk ^f	0.	0	· 0·	¹ 1.2x10 ⁻⁵	1.2x10 ⁻⁵	1.2x10 ⁻⁵	1.2x10 ⁻⁵

Table 4.2.1.9–3.	Potential Hazardous Chemical Impacts to the Public and Workers During Normal
	Operation at Hanford Site—No Action and Storage Alternatives

Sincl?

^a Total=Sum of the No Action plus the contributions of the above activity.

12.12

^b Facility=Contribution from the above activity only (for example, the amount of increase over the existing, No Action level at the site).

^c Hazard index for MEI=Sum of Individual Hazard Quotients (Noncancerous health effects) for MEI.

^d Cancer risk for MEI=(Emissions for 8-hr) x (0.286 [Converts concentrations to doses]) x (Slope Factor [SF]).

^e Hazard index for workers=Sum of Individual Hazard Quotients (Noncancerous health effects) for workers.

^f Cancer risk for workers=(Emissions for 8-hr) x (0.286 [Converts concentrations to doses]) x (0.237 [Fraction of year exposed]) x (0.571 [Fraction of lifetime working]) x (SF).

Note: Where there are no known carcinogens among the hazardous chemicals emitted, there are no slope factors, therefore the calculated risk value is 0.

Source: Section M.3, Tables M.3.4-1 through M.3.4-4.

average No Action worker from 50 years of operation would be 5.0x10⁻³, and the projected number of latent fatal cancers to the No Action workforce from 50 years of operation would be 0.92.

Hazardous Chemical Impacts. Hazardous chemical impacts to the public resulting from the normal operation under No Action at Hanford are presented in Table 4.2.1.9–3. The hazardous chemical impacts from current site operations represent the baseline site impacts for the various storage alternatives. The noncancerous health effects and the risk of cancer due to the total chemical exposures were estimated. Since the major releases due to normal operation at Hanford are expected to make up nearly all of the exposures to onsite workers and to the public in adjacent communities, contributions to the hazardous chemical concentrations from all other sources (for example, industrial operations) are considered negligible for purposes of risk calculations.

The HI to the MEI of the public at Hanford resulting from normal operation under the No Action Alternative is 6.2×10^{-5} , and the cancer risk from hazardous chemicals is zero (because no carcinogens are released from the hazardous chemicals used). The HI to the onsite worker is 4.0×10^{-3} , and the cancer risk is zero (because no carcinogens are released from the hazardous chemicals used).

Facility Accidents. Under the No Action Alternative, Pu would continue to be stored at Hanford in existing facilities. These facilities currently operate in accordance with DOE Orders which ensure that the risk to the public of prompt fatalities due to accidents or cancer fatalities due to operations will be minimized. The safety to workers and the public from accidents at existing facilities is also controlled by Technical Safety Requirements specified in detail in a Safety Analysis Report (SAR) or a Basis for Interim Operations document prepared and maintained specifically for a facility or process within a facility. Under these controls, any change in approved operations or to facilities would cause a halt in operations until it can be established that worker and public safety has not been compromised.

The *Plutonium Finishing Plant Safely Analysis Report* (WHC-SD-CP-SAR-021) analyzes a wide spectrum of accidents that are primarily associated with processing rather than vault storage. This is because a release from a vault would require more severe accident conditions than are normally analyzed in an SAR. The accidents in
the SAR consist of potential process accidents such as fires, explosions, and criticality as well as an externally initiated aircraft crash and earthquake. An estimate of the effects of potential accidents in the existing storage vault at Hanford can be derived from similar storage accidents that have been postulated for an upgraded storage facility. A severe consequence, low frequency accident for storage under the No Action Alternative would be a beyond design basis earthquake. If this accident were to occur, there would be an estimated 0.12 latent cancer fatalities in the offsite population within 80 km (50 mi). The estimated frequency of the earthquake with sufficient damage to cause a release is approximately 1.0×10^{-7} per year, which corresponds to a risk of 1.2×10^{-8} latent cancer fatalities, respectively, if the accident occurred. The risks would be 1.7×10^{-5} and 2.2×10^{-10} latent cancer fatalities per year. A potentially more frequent accident is penetration of the PCV caused by corrosion. If this accident were to occur, the estimated number of cancer fatalities in the offsite population would be 1.3×10^{-3} . The estimated frequency of this accident is 6.4×10^{-3} per year, which corresponds to a risk of 8.3×10^{-5} cancer fatalities per year. For the MEI and noninvolved worker the corresponds to a risk of 8.3×10^{-5} cancer fatalities per year. For the MEI and noninvolved worker the corresponds to a risk of 8.3×10^{-5} cancer fatalities per year. For the MEI and noninvolved worker the corresponding impacts are 1.8×10^{-7} and 1.8×10^{-5} latent cancer fatalities, respectively, if the accident occurred. The risks would be 1.2×10^{-9} and 1.2×10^{-7} latent cancer fatalities per year. For the MEI and noninvolved worker the corresponding impacts are 1.8×10^{-7} and 1.8×10^{-5} latent cancer fatalities, respectively, if the accident occurred. The risks would be 1.2×10^{-9} and 1.2×10^{-7} latent cancer fatalities per year.

Upgrade Alternative

This section describes the radiological and hazardous chemical releases and their associated impacts resulting either from normal operation or from accidents involved with the modified FMEF or a new storage facility at Hanford. The section describes the impacts from normal facility operations at Hanford, followed by a description of impacts from facility accidents.

During normal operation at Hanford, the operation of any of these Pu storage facilities would result in impacts that are within applicable regulatory limits.

[Text deleted.]

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Normal Operation. There would be no radiological releases during the modification of the FMEF at Hanford. Construction worker exposures to material potentially contaminated with radioactivity (for example, from construction activities involved with existing contaminated soil) would be limited to assure that doses are maintained as low as reasonably achievable (ALARA). Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of the construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be radiological and hazardous chemical releases to the environment as well as direct exposures. The resulting doses and potential health effects on the public and workers at Hanford are described below.

Radiological Impacts. Doses to the public from storage would be expected to decrease from No Action for the Upgrade Alternative, as shown in Table 4.2.1.9–1. This is because the storage facility safety and design features would improve. The dose to the MEI of the public due to annual storage facility operation would be 1.8×10^{-6} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 4.5×10^{-11} . The impacts to the average individual would be less. As a result of storage facility operation in the year 2030, the population dose would be 4.7×10^{-5} person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 1.2×10^{-6} .

The dose to the MEI due to annual total site operations is within the radiological limits specified in NESHAPS (40 CFR 61, Subpart H) and DOE Order 5400.5, and would be 5.1×10^{-3} mrem. From 50 years of operations, the

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

194.4

·F. (4)

corresponding risk of fatal cancer to this individual would be 1.3×10^{-7} . These values are presented in Table 4.2.1.9-1. The impacts to the average individual would be less. This activity would be included in a program to ensure that doses to the public are ALARA. As a result of total site operations in the year 2030, the population dose would be within the limit in proposed 10 CFR 834 and would be 1.5 person-rem. The corresponding number of fatal cancers in this population from 50 years of operation would be 0.038.

Doses to onsite workers from normal operations are given in Table 4.2.1.9–2. Included are involved workers ' directly associated with the modified facility for Pu storage, workers who are not involved with the modified facility, and the entire workforce at Hanford. All doses fall within regulatory limits and administrative control levels. The associated risks and numbers of fatal cancers among the different workers from 50 years of operation are included in the table. Dose to individual workers would be kept low by instituting badged monitoring and ALARA programs and also workers rotations. As a result of the implementation of these mitigation measures, the actual number of fatal cancers calculated would be lower for the operation of this facility

Hazardous Chemical Impacts. Hazardous chemical impacts to the public and to the onsite worker resulting from the normal operations of the upgraded storage facilities at Hanford are presented in Table 4.2.1.9–3. The impacts from all site operations, including the upgraded storage facilities are also included in this table. Total site impacts, which include the No Action impact plus the facility are provided. All analyses to support the values presented in this table are provided in Section M.3.

The HI from the facility to the MEI of the public is 9.4×10^{-7} , and the cancer risk from hazardous chemicals from the facility is zero (because no carcinogens are released from the hazardous chemicals used) as a result of operation of the upgraded storage facilities in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, because exposures would be expected to remain the same. The total site operation, including the upgrade facility, would result in an HI of 6.3×10^{-5} and a cancer risk of zero (because no carcinogens are released) for the MEI in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

The HI from the facility to the onsite worker would be 1.9×10^{-5} and the cancer risk from the facility is zero (because no carcinogens are released from the hazardous chemicals used) as a result of operation of the upgraded storage facilities in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, provided exposures remain the same. The total site operation including the upgrade facility would result in an HI of 4.0×10^{-3} and a cancer risk of zero (because no carcinogens are released) for the worker in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

Facility Accidents. Modification of the existing Pu storage facilities at the Hanford site may change the existing risks of accidents to workers and the public. Under this action, the FMEF would be modified and would be in compliance with applicable DOE Orders and other regulations and standards. This may result in a reduction of risk compared to No Action.

A set of potential accidents have been postulated for upgraded storage at FMEF of existing Pu without LANL or RFETS Pu for which there may be releases of Pu that may impact onsite workers and the offsite population. The accident consequences and risks to a worker located 1,000 m (3,280 ft) from the accident release point, the maximum offsite individual located at the site boundary, and the population located within 80 km (50 mi) of the accident release point are summarized in Table 4.2.1.9-4. For the set of accidents analyzed, the maximum number of cancer fatalities in the population within 80 km (50 mi) would be 0.12 at Hanford for the beyond design basis earthquake accident scenario with an estimated probability of $1.0x10^{-7}$ per year (for example, probability of severe earthquake occurring is estimated to be about $1.0x10^{-5}$, once in 100,000 years, multiplied by a damage and release probability of 0.01). The corresponding 50-year facility lifetime risk from the same accident scenario for the population, maximum offsite individual, and worker at 1,000 m (3,280 ft), would be $6.1x10^{-7}$, $8.3x10^{-11}$, and $1.1x10^{-8}$, respectively. The maximum population 50-year facility lifetime risk would

L

be 4.2×10^{-4} (for example, one fatality in over 100,000 years) at Hanford for the PCV penetration by corrosion accident scenario with a probability of 6.4×10^{-3} per year. The corresponding maximum offsite individual and worker 50-year facility lifetime risks would be 5.7×10^{-8} and 5.7×10^{-6} , respectively. Section M.5 presents additional facility accident data and summary descriptions of the accident scenarios identified in Table 4.2.1.9-4.

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions, and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

Construct New 200 West Area Facility for Plutonium Storage

一般所にたちと

Normal Operation. There would be no radiological releases during the construction of new storage facilities at Hanford. Construction worker exposures to material potentially contaminated with radioactivity (for example, from construction activities involved with existing contaminated soil) would be limited to assure that doses are maintained ALARA. Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of the construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be radiological and hazardous chemical releases to the environment as well as direct exposures. The resulting doses and potential health effects to the public and workers at Hanford are described below.

Radiological Impacts. The doses and associated health risks to the public associated with this new storage facility are expected to be even smaller than those for the modified FMEF. Total site doses and resulting health risks would be virtually the same for both storage facilities. The doses and associated health risks to workers are assumed to be the same as for the modified FMEF (Table 4.2.1.9–2). This is because the operations would be similar and the amount of material handled would be the same.

Hazardous Chemical Impacts. Hazardous chemical emissions from the new Pu storage facility would be less than the emissions from the modified FMEF. The resultant health risks to the public and workers from hazardous chemical emissions associated with this new storage facility would be even smaller than those given in Table 4.2.1.9–3 for the modified FMEF.

1.1

Facility Accidents. A new Pu storage facility for continued storage of Pu would incorporate new safety features that should reduce the consequences and risks of accidents compared with No Action. The consequences and risks of accidents for this new facility would be bounded by the consequences and risks presented in Table 4.2.1.9-4 for the upgraded FMEF at Hanford.

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions, and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

					_		
	Wor 1,00	ker at)0 m	Maximu Indiv	m Offsite idual	Popula 80	ation to km	· · · · · · · · · · · · · · · · · · ·
Accident Description	Risk of Cancer Fatality (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatality (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatalities (per 50 yr) ^a	Number of Cancer Fatalities ^c	Accident Frequency (per yr)
PCV puncture by forklift	1.3x10 ⁻⁷	4.4×10^{-6}	1.3x10 ⁻⁹	4.4x10 ⁻⁸	9.6x10 ⁻⁶	3.2×10^{-4}	6.0×10^{-4}
PCV breach by firearms discharge	7.7x10 ⁻⁹	4.4x10 ⁻⁷	7.7x10 ⁻¹¹	4.4x10 ⁻⁹	5.6x10 ⁻⁷	3.2x10 ⁻⁵	3.5x10 ⁻⁴
PCV penetration by corrosion	5.7x10 ⁻⁶	1.8x10 ⁻⁵	5.7x10 ⁻⁸	1.8x10 ⁻⁷	4.2x10 ⁻⁴	1.3x10 ⁻³	6.4x10 ⁻³
Vault fire	5.8x10 ⁻⁹	1.2x10 ⁻³	4.6x10 ⁻¹¹	9.2x10 ⁻⁶	3.4×10^{-7}	0.067	1 0x10-7
Truck bay fire	3.1x10 ⁻⁹	6.1x10 ⁻⁴	3.1x10 ⁻¹¹	6.1x10 ⁻⁶	2.2×10^{-7}	0.045	1.0×10-7
Spontaneous combustion	3.1x10 ⁻¹¹	8.8x10 ⁻⁷	3.1×10^{-13}	8.8x10 ⁻⁹	2.2x10 ⁻⁹	64×10^{-5}	7.0×10-7
Explosion in the vault	7.2x10 ⁻¹⁰	1.4x10 ⁻⁴	7.2×10^{-12}	1.4x10 ⁻⁶	5.3x10 ⁻⁸	0.011	1 0v10-7
Explosion outside of vault	3.3x10 ⁻¹¹	6.6x10 ⁻⁶	3.3x10 ⁻¹³	6.6x10 ⁻⁸	2.4x10 ⁻⁹	4.8x10 ⁻⁴	1.0×10^{-7}
Nuclear criticality	2.1x10 ⁻¹¹	4.2x10 ⁻⁶	1.6x10 ⁻¹³	3.3x10 ⁻⁸	1.8x10 ⁻¹⁰	3.5x10 ⁻⁵	1.0x10 ⁻⁷
Beyond design basis earthquake	. 1.1x10 ⁻⁸	2.2x10 ⁻³	8.3x10 ⁻¹¹	1.7x10 ⁻⁵	6.1x10 ⁻⁷	0.12	1.0x10 ⁻⁷
Expected risk ^d	5.8x10-6		5.8x10 ⁻⁸	_	4.3x10 ⁻⁴	· · ·	,* i · ·

Table 4.2.1.9–4.	Upgrade Without Rocky Flats Environmental Technology Site or Los Alamos National
	Laboratory Material Alternative—Accident Impacts at Hanford Site

^a The risk values are calculated by multiplying the probability of cancer fatality (for the worker at 1,000 m or the maximum offsite individual) or the number of cancer fatalities (for the population to 80 km) by the accident frequency and the number of years in operation.

^b Increased likelihood (or probability) of cancer fatality to a hypothetical individual (a single onsite worker at a distance of 1,000 m or the site boundary, whichever is smaller or to a hypothetical individual in the offsite population located at the site boundary) if exposed to the indicated dose. The value assumes the accident has occurred

^c Estimated number of cancer fatalities in the entire offsite population out to a distance of 80 km if exposed to the indicated dose. The t-value assumes the accident has occurred.

^d Expected risk is the sum of the risks for each accident over the 50-year lifetime of the facility.

Note: All values are mean values

Source: Calculated using Table 4.2 1.9-6 data adjusted for existing inventory of Pu at Hanford.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

Normal Operation. As described for the Upgrade Without RFETS or LANL Pu, there would be no radiological releases during the modification of the FMEF at Hanford. Construction worker exposures to material potentially contaminated with radioactivity would be limited to assure that doses are maintained ALARA. Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of the construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be radiological and hazardous chemical releases to the environment as well as direct exposures. The resulting doses and potential health effects on the public and workers at Hanford are described below.

Radiological Impacts. During normal operations, there would be only a negligible difference in radiological impacts if Pu from the RFETS and LANL is included in the upgrade storage alternative. Therefore, the impacts are essentially the same as presented in the previous section, which discusses the upgrade without RFETS or LANL Pu.

Hazardous Chemical Impacts. Hazardous chemicals associated with storage of Pu from RFETS and LANL does not measurably contribute to hazardous chemical emissions from the facility for this subalternative. Therefore resultant hazardous chemical impacts to the public and worker are essentially the same as presented in the previous section, which discusses the upgrade without RFETS or LANL Pu.

Facility Accidents. Upgrade of the existing Pu storage facilities at the Hanford site may change the existing risks of accidents to workers and the public. Under upgrade, all Pu storage facilities would be brought into compliance with applicable DOE Orders and other regulations and standards. This may result in a reduction of risk compared to No Action.

A set of potential accidents have been postulated for the RFETS and LANL Pu storage increment for which there may be releases of Pu that may impact onsite workers and the offsite population. The accident consequences and risks to a worker located 1,000 m (3,280 ft) from the accident release point, the maximum offsite individual located at the site boundary, and the population located within 80 km (50 mi) of the accident release point are summarized in Table 4.2.1.9-5. For the set of accidents analyzed, the maximum number of cancer fatalities in the population within 80 km (50 mi) would be 0.12 at Hanford for the beyond design basis earthquake accident scenario with an estimated probability of 1.0x10⁻⁷ per year (for example, probability of severe earthquake occurring is estimated to be about 1.0x10⁻⁵, once in 100,000 years, multiplied by a damage and release probability of 0.01). The corresponding 50-year facility lifetime risk from the same accident scenario for the population, maximum offsite individual, and worker at 1,000 m (3,280 ft), would be 6.2x10⁻⁷, 8.5×10^{-11} , and 1.1×10^{-8} , respectively. The maximum population 50-year facility lifetime risk would be 4.3×10^{-4} (for example, on fatality in over 100,000 years) at Hanford for the PCV penetration by corrosion accident scenario with a probability of 6.6x10⁻³ per year. The corresponding maximum offsite individual and worker 50-year facility lifetime risks would be 5.9×10^{-8} and 5.9×10^{-6} , respectively. Table 4.2.1.9–5 also shows the Combined Expected Risk for the upgraded storage of existing Pu, the RFETS Pu and the LANL Pu increment. Section M.5 presents additional facility accident data and summary descriptions of the accident scenarios identified in Table 4.2.1.9-5.

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions, and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

Construct New 200 West Area Facility for Plutonium Storage

State Liter 1.

Normal Operation. As described for the Upgrade Without RFETS or LANL Pu, there would be no radiological releases during the construction of the new 200 West Area Facility. Construction worker exposures to material potentially contaminated with radioactivity (for example, from construction activities involved with existing contaminated soil) would be limited to assure that doses are maintained ALARA. Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of the construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be radiological and hazardous chemical releases to the environment as well as direct exposures. The resulting doses and potential health effects to the public and workers at Hanford are described below.

Radiological Impacts. During normal operations, there would be only a negligible difference in radiological impacts if Pu from the RFETS and LANL is included in the upgrade storage alternative. Therefore, the impacts are essentially the same as presented in the previous section, which discusses the Upgrade Without RFETS or LANL Pu.

200001						# \ <u>\</u>	
	Worl	ker at	Maximu	m Offsite	Popul	ation to	
	- 1,00	0 m 🔐 😁	👝 💤 Indi	vidual 🗤 🔄	<u> </u>	km	· .
-	Risk of	Probability	Risk of	Probability	Risk of	Number of	t > ~
	Cancer	of Cancer	Cancer	of Cancer	Cancer	Cancer	Accident
	Fatality	Fatality ^b	Fatality	Fatality ^b	Fatalities	Fatalities ^c	Frequency
Accident Description	(per 50 yr) ^a		(per 50 yr) ⁸	L	(per 50 yr) ⁸	l	(per yr)
PCV puncture by forklift	1.3x10 ⁻⁷	4.4x10 ⁻⁶	1.3x10 ⁻⁹	4.4x10 ⁻⁸	9.6x10 ⁻⁶	3.2×10^{-4}	6.0x10 ⁻⁴
PCV breach by	7.7x10 ⁻⁹	4.4x10 ⁻⁷	7.7x10 ⁻¹¹	4.4x10 ⁻⁹	5.6x10 ⁻⁷	3.2x10 ⁻⁵	3.5x10 ⁻⁴
firearms discharge	•		,		•	· ·	
PCV penetration	5.9x10 ⁻⁶	1.8x10 ⁻⁵	΄ 5.9 x10 ⁻⁸	1.8x10 ⁻⁷	4.3x10 ⁻⁴	- 1.3x10 ⁻³	6.6x10 ⁻³
by corrosion	•	2	11	· · · · · · · · · · · · · · · · · · ·	7	2 15E	
Vault fire	5.9x10 ⁻⁹	1.2x10 ⁻³	4.7x10 ⁻¹¹	9.4x10 ⁻⁰	3.5x10'	0,069	1.0x10 ⁻⁷
Truck bay fire	3.1x10 ⁻⁹	6.1×10^{-4}	3.1x10 ⁻¹¹	. 6.1x10 ⁻⁰	2.2×10^{-7}	0.045	1.0x10 ⁻⁷
Spontaneous combustion	3.1x10 ⁻¹¹	8.8x10 ⁻⁷	3.1x10 ⁻¹³	8.8x10 ⁻⁹	2.2x10 ⁻⁹	, 6.4x10 ⁻³	7.0x10 ⁻⁷
Explosion in the vault '	7.4x10 ⁻¹⁰	1.4×10^{-4}	7.4x10 ⁻¹²	1.4x10 ⁻⁰	5.4x10 ⁻⁸	. 0.011	1.0×10^{-7}
Explosion outside of vault	3.3x10 ⁻¹¹	6.6x10 ⁻⁶	3.3×10^{-13}	6.6x10 ⁻⁸	2.4×10^{-9}	4.8x10 ⁻⁴	1.0×10^{-7}
Nuclear criticality	2.1x10 ⁻¹¹	4.2x10 ⁻⁶	1.6x10 ⁻¹³	3.3x10 ⁻⁸	1.8x10 ⁻¹⁰	3.5x10 ⁻⁵	1.0x10 ⁻
Beyond evaluation basis	1.1x10 ⁻⁸	2.3x10 ⁻³	8.5x10 ⁻¹¹	1.7x10 ⁻⁵	6.2x10 ⁻⁷	0.12	1.0x10 ⁻⁷
earthquake		· •			· ·		
Expected risk ^d	6.0x10 ⁻⁶	- 1	6.0x10 ⁻⁸	· –	4.3x10 ⁻⁴	-	J
Combined expected risk ^e	1.2×10^{-5}	<u> </u>	1.2x10 ⁻⁷	<u>ر ۲٬۰٬۰</u>	8.6x10 ⁻⁴	-	· <u> </u>

 Table 4.2.1.9–5.
 Upgrade With Rocky Flats Environmental Technology Site and Los Alamos National

 Laboratory Material Alternative—Accident Impacts at Hanford Site

^a The risk values are calculated by multiplying the probability of cancer fatality (for the worker at 1,000 m or the maximum offsite individual) or the number of cancer fatalities (for the population to 80 km) by the accident frequency and the number of years in operation.

^b Increased likelihood (or probability) of cancer fatality to a hypothetical individual (a single onsite worker at a distance of 1,000 m or the site boundary, whichever is smaller or to a hypothetical individual in the offsite population located at the site boundary) if exposed to the indicated dose. The value assumes the accident has occurred.

^c Estimated number of cancer fatalities in the entire offsite population out to a distance of 80 km if exposed to the indicated dose. The value assumes the accident has occurred.

^d Expected risk is the incremental risk for storing the additional RFETS and LANL material for each accident over the 50-year lifetime of the facility.

^e Combined expected risk for base case without RFETS or LANL Pu material plus RFETS and LANL Pu material increment. Note: All values are mean values.

Source: Calculated using in Tables 4.2.1.9-6. Data adjusted for additional RFETS and LANL Pu.

Hazardous Chemical Impacts. Hazardous chemicals associated with storage of Pu from RFETS and LANL associated with building a new Pu facility are essentially the same as presented in the previous section, which discusses the upgrade without RFETS or LANL Pu. The resultant hazardous chemical impacts to the public and worker are essentially the same as presented in the previous section.

Facility Accidents. The new 200 West Area Facility constructed for continued storage of Pu would incorporate new safety features that should reduce the consequences and risks of accidents compared with No Action. The consequences and risks of accidents for this facility would be bounded by the consequences and risks presented in Table 4.2.1.9–5 for the Modify Existing FMEF for Pu Storage Subalternative at Hanford.

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions, and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

Consolidation Alternative

Construct New Plutonium Storage Facility

This section includes a description of radiological and hazardous chemical releases and their associated impacts resulting from either normal operation or accidents involving the new consolidated Pu storage facility at Hanford.

[Text deleted.]

Normal Operation. There would be no radiological releases during the construction of a new consolidated Pu storage facility at Hanford. Construction worker exposures to material potentially contaminated with radioactivity (for example, from construction activities involved with existing contaminated soil) would be limited to assure that doses are maintained ALARA. Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be radiological and hazardous chemical releases to the environment as well as direct in-plant exposures. The resulting doses and potential health effects to the public and workers at Hanford are described below.

Radiological Impacts. Radiological impacts to the public resulting from the normal operation of the new consolidated Pu storage facility are presented in Table 4.2.1.9-1. The impacts from all site operations, including the new consolidated Pu storage facility, are also given in the table. To put operational doses into perspective, comparisons of operational doses with natural background radiation doses are included in the table.

The dose to the MEI due to annual storage facility operation would be 2.5×10^{-6} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 6.2×10^{-11} . As a result of storage plant operation in the year 2030, the population dose would be 1.1×10^{-4} person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 2.8×10^{-6} .

The dose to the MEI due to annual total site operations is within the radiological limits specified in NESHAPS (40 CFR 61, Subpart H) and DOE Order 5400.5 and would be 5.1×10^{-3} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 1.3×10^{-7} . The impacts to the average individual would be less. This activity would be included in a program to ensure that doses to the public are ALARA. As a result of total site operation in the year 2030, the population dose would be within the limit in proposed 10 CFR 834 and would be 1.5 person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 0.038.

Doses to onsite workers from normal operations are given in Table 4.2.1.9-2. Included are involved workers directly associated with the new consolidated Pu storage facility, workers who are not involved with the new storage facility, and the entire workforce at Hanford. All doses fall within regulatory limits and administrative control levels. The associated risks and numbers of fatal cancers among the different workers from 50 years of operation are included in the table. Dose to individual workers would be kept low by instituting badged monitoring and ALARA programs and also workers rotations. As a result of the implementation of these mitigation measures, the actual number of fatal cancers calculated would be lower for the operation of this facility.

Hazardous Chemical Impacts. Hazardous chemical impacts to the public and to the onsite worker resulting from the normal operations of the new consolidated Pu storage facility at Hanford are presented in Table 4.2.1.9–3. The impacts from all site operations, including the consolidated storage facility, are included in this table. Total site impacts, which include the No Action impact plus the added facility impact, are provided. All analyses to support the values presented in this table are provided in Section M.3.

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

The HI to the MEI of the public is 4.0×10^{-6} , and the cancer risk is 2.7×10^{-8} as a result of operation of the new consolidated Pu storage facility in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, provided exposures remain the same. The total site operation including the upgrade facility would result in an HI of 6.6×10^{-5} and a cancer risk of 2.7×10^{-8} for the MEI in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

35000

The HI to the onsite worker would be 2.8×10^{-4} , and the cancer risk is 1.2×10^{-5} as a result of operation of the new consolidated Pu storage facility in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, provided exposures remain the same. The total site operation including the upgrade facility would result in an HI of 4.3×10^{-3} and a cancer risk of 1.2×10^{-5} for the MEI in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

Facility Accidents. A set of potential accidents for the consolidation of Pu alternative at Hanford for which there may be releases of Pu that may impact onsite workers and the offsite population has been postulated. The accident consequences and risks to a worker located 1,000 m (3,280 feet) from the accident release point, the maximum offsite individual located at the site boundary, and the population located within 80 km (50 mi) of the accident release point are summarized in Table 4.2.1.9–6. For the set of accidents analyzed, the maximum number of cancer fatalities in the population within 80 km (50 mi) would be 1.2 at Hanford for the beyond design basis earthquake accident scenario with an estimated probability of 1.0×10^{-7} per year (that is, probability of severe earthquake occurring is estimated to be about 1.0×10^{-5} , once in 100,000 years, multiplied by a damage and release probability of 0.01). The corresponding 50-year facility lifetime risk from the same accident scenario for the population, maximum offsite individual, and worker at 1,000 m (3,280 ft), would be 6.1×10^{-6} , 8.3×10^{-10} , and 1.1×10^{-7} , respectively. The maximum population 50-year facility lifetime risk would be 4.2×10^{-3} (that is, one fatality in about 12,000 years) at Hanford for the PCV penetration by corrosion accident scenario with a probability of 0.064 per year. The corresponding maximum offsite individual and worker 50-year facility lifetime risks would be 5.7×10^{-7} and 5.7×10^{-5} , respectively. Section M.5 presents additional facility accident data and summary descriptions of the accident scenarios identified in Table 4.2.1.9-6.

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions, and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

This section includes a description of radiological and hazardous chemical releases and the associated impacts resulting from either normal operation or accidents involved with the consolidation of Pu storage and collocation with HEU storage facilities at Hanford. This storage would take place in a new Pu and HEU storage facility.

Normal operation of the new collocated storage facility at Hanford would result in impacts that are within applicable regulatory limits.

[Text deleted.]

Normal Operation. There would be no radiological releases during the construction of a new collocated storage facility at Hanford. Construction worker exposures to materials potentially contaminated with radioactivity (for

	Worl 1,00	ker at 0 m	Maximu Indiv	m Offsite ` idual	Popula 80	ation to km	
	Risk of Cancer Fatality	Probability of Cancer Fatality ^b	Risk of Cancer Fatality	Probability of Cancer Fatality ^b	Risk of Cancer Fatalities	Number of Cancer Fatalities ^c	Accident Frequency
Accident Description	(per 50 yr) ^a		(per 50 yr) ^a		(per 50 yr)*		(per yr)
PCV puncture by forklift	1.3x10 ⁻⁷	4.4x10 ⁻⁶	1.3x10 ⁻⁹	4.4x10 ⁻⁸	9.6x10 ⁻⁰	3.2x10 ⁻⁴	6.0x10 ⁻⁴
PCV breach by firearms discharge	7.7x10 ⁻⁹	4.4x10 ⁻⁷	7.7x10 ⁻¹¹	4.4x10 ⁻⁹	5.6x10 ⁻⁷	3.2x10 ⁻³	3.5x10 ⁻⁴
PCV penetration	5.7x10 ⁻⁵	1.8x10 ⁻⁵	5.7x10 ⁻⁷	1.8x10 ⁻⁷	4.2x10 ⁻³	1.3x10 ⁻³	0.064
Voult fire	5.8×10^{-8}	0.012	4.6×10^{-10}	9.2x10 ⁻⁵	3.4x10 ⁻⁶	0.67	1.0x10 ⁻⁷
Truck hav fire	3 1x10 ⁻⁹	6.1×10^{-4}	3.1x10 ⁻¹¹	6.1×10^{-6}	2.2x10 ⁻⁷	0.045	1.0x10 ⁻⁷
Spontaneous combustion	3.1×10^{-11}	8.8×10 ⁻⁷	3.1×10^{-13}	8.8x10 ⁻⁹	2.2x10 ⁻⁹	6.4x10 ⁻⁵	7.0x10 ⁻⁷
Spontaneous compustion	7.2×10 ⁻¹¹	1.4×10^{-3}	7.2x10 ⁻¹¹	1.4x10 ⁻⁵	5.3x10 ⁻⁷	0.11	1.0x10 ⁻⁷
Explosion outside of vault	3.3×10 ⁻¹¹	6.6x10 ⁻⁶	3.3x10 ⁻¹³	6.6x10 ⁻⁸	2.4x10 ⁻⁹	4.8x10 ⁻⁴	1.0x10 ⁻⁷
Nuclear criticality	2.1×10^{-11}	4.2×10^{-6}	1.6x10 ⁻¹³	3.3x10 ⁻⁸	1.8x10 ⁻¹⁰	3.5x10 ⁻⁵	1.0x10 ⁻⁷
Beyond evaluation basis earthquake	1.1x10 ⁻⁷	2.2x10 ⁻²	8.3x10 ⁻¹⁰	1.7x10 ⁻⁴	6.1x10 ⁻⁶	1.2	1.0x10 ⁻⁷
Expected risk ^d	5.8x10 ⁻⁵	· _	5.8x10 ⁻⁷		4.3x10 ⁻³	. –	

Table 4.2.1.9–6. Consolidation Alternative Accident Impacts at Hanford Site

The risk values are calculated by multiplying the probability of cancer fatality (for the worker at 1,000 m or the maximum offsite individual) or the number of cancer fatalities (for the population to 80 km) by the accident frequency and the number of years of operation.

^b Increased likelihood (or probability) of cancer fatality to a hypothetical individual (a single onsite worker at a distance of 1,000 m or the site boundary, whichever is smaller or to a hypothetical individual in the offsite population located at the site boundary) if exposed to the indicated dose. The value assumes the accident has occurred.

^c Estimated number of cancer fatalities in the entire offsite population out to a distance of 80 km if exposed to the indicated dose. The value assumes the accident has occurred.

^d Expected risk is the sum of the risks for each accident over the 50-year lifetime of the facility. Note: All values are mean values.

ないないというというというという

ちんち しんしょうちょうせいしい

i,

Source: Calculated using the source terms in Tables M.5.2.1.1-5 and M.5.2.1.1-6 and the MACCS computer code.

example, from construction activities involved with existing contaminated soil) would be limited to assure that doses are maintained ALARA. Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be radiological and hazardous chemical releases to the environment as well as direct in-plant exposures. The resulting doses and potential health effects to the public and workers are described below.

Radiological Impacts. Radiological impacts to the public resulting from the normal operation of the new collocated storage facility at Hanford are presented in Table 4.2.1.9–1. The impacts from all site operations, including the new storage facility, are also given in the table. To put operational doses into perspective, comparisons of operational doses with natural background radiation doses are included in the table.

The dose to the MEI of the public due to annual storage facility operation would be 2.5×10^{-6} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 6.2×10^{-11} . The impacts to the average individual would be less. As a result of storage facility operation in the year 2030, the population

dose would be 1.1×10^{-4} person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 2.8×10^{-6} .

The dose to the MEI of the public due to annual total site operations is within radiological limits and would be 5.1×10^{-3} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 1.3×10^{-7} . The impacts to the average individual would be less. This activity would be included in a program to ensure that doses to the public are ALARA. As a result of total site operation in the year 2030, the population dose would be within the limit in proposed 10 CFR 834 and would be 1.5 person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 0.038.

Doses to onsite workers due to normal operations are given in Table 4.2.1.9–2. Included are involved workers directly associated with the new storage facility, workers who are not involved with the new storage facility, and the entire workforce at Hanford. All doses fall within regulatory limits and administrative control levels. The associated risks and numbers of fatal cancers among the different workers from 50 years of operation are included in the table. Dose to individual workers would be kept low by instituting badged monitoring and ALARA programs and also worker rotations. As a result of the implementation of these mitigation measures, the actual number of fatal cancers calculated would be lower for the operation of this facility.

Hazardous Chemical Impacts. Hazardous chemical impacts to the public and to the onsite worker resulting from the normal operations of the new consolidation of Pu storage and collocation with HEU storage facilities at Hanford are presented in Table 4.2.1.9–3. The impacts from all site operations, including the consolidation of Pu storage and collocation with HEU storage facilities, are also included in this table. Total site impacts, which include the No Action impact plus the added facility impacts, are provided. All analyses to support the values presented in this table are provided in Section M.3.

The HI to the MEI of the public is 1.6×10^{-5} , and the cancer risk is 2.7×10^{-8} as a result of operation of the new consolidation of Pu storage and collocation with HEU storage facilities in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, because exposures would be expected to remain the same. The total site operation, including the new facility, would result in an HI of 7.8×10^{-5} and a cancer risk of 2.7×10^{-8} for the MEI in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

いたちままでのになけたいでは、「ないない」

The HI to the onsite worker is 7.1×10^{-4} , and the cancer risk is 1.2×10^{-5} as a result of operation of the new consolidation of Pu storage and collocation with HEU storage facilities in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, because exposures would be expected to remain the same. The total site operation including the new facility would result in an HI of 4.7×10^{-3} and a cancer risk of 1.2×10^{-5} for the onsite worker in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

Facility Accidents. A set of potential accidents for collocation of Pu and HEU at Hanford for which there may be releases of Pu or HEU that may impact onsite workers and the offsite population has been postulated. The consequences and risks of potential accidents for Pu and HEU storage would be bounded by the impacts associated with the release of Pu. The accident consequences and risks to a worker located 1,000 m (3,280 feet) from the accident release point, the maximum offsite individual located at the site boundary, and the population located within 80 km (50 mi) of the accident release point are summarized in Table 4.2.1.9–7. For the set of accidents analyzed, the maximum number of cancer fatalities in the population within 80 km (50 mi) would be 1.2 at Hanford for the beyond design basis earthquake accident scenario with an estimated probability of 1.0x10⁻⁷per year (that is, probability of severe earthquake occurring is estimated to be about 1.0x10⁻⁵, once in 100,000 years, multiplied by a damage and release probability of 0.01). The corresponding 50-year facility lifetime risk from the same accident scenario for the population, maximum offsite individual, and worker at 1,000 m (3,280 ft), would be 6.1x10⁻⁶, 8.3x10⁻¹⁰, and 1.1x10⁻⁷, respectively. The maximum population 50-year facility lifetime risk would be 4.2x10⁻³ (that is, one fatality in about 12,000 years) at Hanford for the PCV penetration by corrosion accident scenario with a probability of 0.064 per year. The corresponding maximum

4–72

I

I

offsite individual and worker 50-year facility lifetime risks would be 5.7×10^{-7} and 5.7×10^{-5} , respectively. Section M.5 presents additional facility accident data and summary descriptions of the accident scenarios identified in Table 4.2.1.9–7.

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions, and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

If the strategic reserve and weapon's R&D materials are not included, the incremental impacts to the public and to workers from the accident-free storage activities would be reduced in proportion to the decrease in the amount of material stored. The impacts from total site operations would decrease slightly. This subalternative applies to the Upgrade With All or Some RFETS and LANL Pu Subalternative, the Consolidation Alternative, and the Collocation Alternative. The risks due to accidents would also tend to be lower.

	Worl 1,00	ker at 10 m	Maximu Indiv	m Offsite vidual	Popula 80	ation to km	_
Accident Description	Risk of Cancer Fatality (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatality (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatalities (per 50 yr) ^a	Number of Cancer Fatalities ^c	Accident Frequency (per year)
PCV puncture by forklift	1.3x10 ⁻⁷	4.4x10 ⁻⁶	1.3x10 ⁻⁹	4.4x10 ⁻⁸	9.6x10 ⁻⁶	3.2x10 ⁻⁴	6.0x10 ⁻⁴
PCV breach by firearms discharge	7.7x10 ⁻⁹	4.4x10 ⁻⁷	7.7x10 ⁻¹¹	4.4x10 ^{.9}	5.6x10 ⁻⁷	3.2x10 ⁻⁵	3.5x10 ⁻⁴
PCV penetration by corrosion	5.7x10 ⁻⁵	1.8x10 ⁻⁵	5.7x10 ⁻⁷	1.8x10 ⁻⁷	4.2x10 ⁻³	1.3x10 ⁻³	0.064
Vault fire	5.8x10 ⁻⁸	0.012	4.6x10 ⁻¹⁰	9.2x10 ⁻⁵	3.4x10 ⁻⁶	0.67	1.0x10 ⁻⁷
Truck bay fire	3.1x10 ⁻⁹	6.1x10 ⁻⁴	3.1x10 ⁻¹¹	6.1x10 ⁻⁶	2.2x10 ⁻⁷	0.045	1.0x10 ⁻⁷
Spontaneous combustion	3.1x10 ⁻¹¹	8.8x10 ⁻⁷	3.1x10 ⁻¹³	8.8x10 ⁻⁹	2.2x10 ⁻⁹	6.4x10 ⁻⁵	7.0x10 ⁻⁷
Explosion in the vault	7.2x10 ⁻⁹	1.4x10 ⁻³	7.2x10 ⁻¹¹	1.4x10 ⁻⁵	5.3x10 ⁻⁷	0.11	1.0x10 ⁻⁷
Explosion outside of vault	3.3x10 ⁻¹¹	6.6x10 ⁻⁶	3.3x10 ⁻¹³	6.6x10 ⁻⁸	2.4x10 ⁻⁹	4.8x10 ⁻⁴	1.0x10 ⁻⁷
Nuclear criticality	· 2.1x10 ⁻¹¹	4.2x10 ⁻⁶	1.6x10 ⁻¹³	3.3x10 ⁻⁸	1.8x10 ⁻¹⁰	3.5x10 ⁻⁵	1.0x10 ⁻⁷
Beyond evaluation basis earthquake	1.1x10 ⁻⁷	0.022	8.3x10 ⁻¹⁰	1.7x10 ⁻⁴	6.1x10 ⁻⁶	1.2	1.0x10 ⁻⁷
Expected risk ^d	5.8x10 ⁻⁵	c	5.8x10 ⁻⁷	-	4.3x10 ⁻³	-	-

Table 4.2.1.9–7. Collocation Alternative Accident Impacts at Hanford Site

^a The risk values are calculated by multiplying the probability of cancer fatality (for the worker at 1,000 m or the maximum offsite individual) or the number of cancer fatalities (for the population to 80 km) by the accident frequency and the number of years of operation.

^b Increased likelihood (or probability) of cancer fatality to a hypothetical individual (a single onsite worker at a distance of 1,000 m or the site boundary, whichever is smaller or to a hypothetical individual in the offsite population located at the site boundary) if exposed to the indicated dose. The value assumes the accident has occurred.

Estimated number of cancer fatalities in the entire offsite population out to a distance of 80 km if exposed to the indicated dose. The value assumes the accident has occurred.

^d Expected risk is the sum of the risks for each accident over the 50-year lifetime of the facility. 1.3

Note: All values are mean values.

「「ない」というという。

Source: Calculated using the source terms in Tables M.5.2.2.1-3 and M.5.2.2.1-4 and the MACCS computer code.

Starage and Disposition of Weapons-Usable Fissile Materials Final PEIS

动的

120

Phaseout

Normal Operation. A phaseout of existing Pu storage facilities at Hanford would reduce the impacts from radiological and chemical releases and exposures to levels slightly less than the No Action exposures. As shown in 'lable 4.2.1.9–1, the dose to the MEI from annual operation would be reduced by 4.1×10^{-4} mrem; the dose to the population would be reduced by 0.047 person-rem. The associated reductions in fatal cancer are included in the table. All workers involved in the transfer of the Pu would be monitored to assure that their doses remain willhin regulatory limits and ALARA.

かった

~~ + ~

Facility Accidents. The phaseout operation will be conducted in accordance with DOE Orders to ensure that the tisk to the public of prompt fatalities due to accidents or of cancer fatalities due to operations will be minimized. For current operations in the facility that would be phased out, the safety of workers and the public from accidents is controlled by Technical Safety Requirements that are specified in SARs or Basis for Interim Operations documents that have been prepared for the facility. Prior to initiating phaseout, the potential for accidents that could affect workers and the public will be assessed and, if necessary, applicable existing safety documentation will be modified to ensure safety for workers and the public.

4-74:

11

4.2.1.10 Waste Management

10000

This section summarizes the impacts on waste management at Hanford under No Action, each of the long-term storage alternatives, and the phaseout of Pu storage. There is no spent nuclear fuel or HLW associated with Pu or HEU storage. Table 4.2.1.10-1 lists the projected sitewide waste generation rates and treatment, storage, disposal capacities under No Action for 2005. Projections for No Action were derived from the most recent available environmental data, with the assumption that operational requirements for waste generation in 2005 would be approximately equal to the 1993 generation volume. The projection does not include wastes from future, yet uncharacterized environmental restoration activities, such as content characterization and decommissioning of 149 single shell tanks, treating 28 double shell tanks, and removing over 500 buildings. The projections for No Action could change significantly depending on the decisions resulting from the PEIS on waste management being prepared by DOE. Table 4.2.1.10-2 provides the estimated incremental operational waste volumes projected to be generated at Hanford as a result of the various storage alternatives prior to treatment. Some of the waste values described in this section are different than the waste values in the table. For those values that differ (for example LLW), the table gives waste generated pre-treatment values and the text discusses post-treatment values (indicated as after treatment and volume reduction). The waste volumes generated from the various storage alternatives and the resultant waste effluent used for the waste impacts analysis can be found in Section E.3.1. Facilities that would support the storage of Pu and/or HEU would treat and package all waste generated into forms that would enable staging and/or disposal in accordance with RCRA and other applicable statutes. Depending in part on decisions in waste-type-specific RODs for the Waste Management PEIS, wastes could be treated and disposed of onsite or at regionalized or centralized DOE sites. For the purposes of analyses only, this PEIS assumes that TRU and mixed TRU waste would be treated onsite to the current planning-basis WIPP WAC, and shipped to WIPP for disposal. This PEIS also assumes that LLW, mixed LLW, hazardous, and nonhazardous wastes would be treated and disposed of in accordance with current site practice.

Preferred Alternative: No Action Alternative

Under this alternative, high-level, TRU, low-level, mixed, hazardous, and nonhazardous wastes, and spent nuclear fuel would continue to be managed from the missions outlined in Section 3.2. Hanford no longer has a weapons production mission. Its focus is to decommission the reactors and site facilities, as well as cleanup approximately 1,450 km² (560 mi²) of land. The impacts of the wastes generated as part of environmental restoration and D&D activities are addressed in the *Final Environmental Impact Statement: Decommissioning of Eight Surplus Production Reactors at the Hanford Site, Richland, Washington* (DOE/EIS–0119F) and the *Draft Hanford Remedial Action Environmental Impact Statement and Comprehensive Land Use Plan* (DOE/EIS–0222D). Under No Action, Hanford would continue to store its inventory of Pu, and treat, store, and dispose of its legacy and newly generated wastes in current and planned facilities.

The Pu addressed in this PEIS is limited to materials currently stored within protected vaults and gloveboxes, and additional materials within process lines and process equipment within the PFP Complex in the 200 West Area. The PFP had been used to conduct Pu processing operations such as Pu purification, Pu recovery, oxide production, metal production, and parts fabrication. The PFP has also been used for receipt and large-scale storage of onsite and offsite Pu scrap and product materials. [Text deleted.] Modifications to the facilities will proceed following the ROD resulting from the PFP EIS (DOE/EIS-0244F) to meet current regulations and provide for interim storage. Maintenance, assay, packaging, and monitoring of the inventory would produce TRU, low-level, hazardous, and nonhazardous wastes. These wastes would be treated, stored, and disposed of in compliance with existing regulations.

Under No Action, the processing of legacy wastes would require new facilities, since the necessary treatment, storage, and disposal facilities either do not exist or are nearing capacity. Spent nuclear fuel would be managed in accordance with the amended ROD (61 FR 9441) from the Department of Energy Programmatic Spent Nuclear Fuel Management and Idaho National Engineering Laboratory Environmental Restoration and Waste

Category	Annual Generation (m ³)	Treatment Method	Treatment Capacity (m ³ /yr)	Storage Method	Storage Capacity (m ³)	Disposal Method	Disposal Capacity (m ³)	
Spent Nuclear Fuel	None ^a	Encapsulation	Planned	Reactor basins Non-Hanford production reactor spent fuel to be sent to INEL	2,133 t	To HLW Program	NA	
High-Level								
Liquid	None ^a	Evaporation ^{b,c}	50,000	Tank farm	146,000 ^d	NA	NA	
Solid	None	NĂ	NA	NA	NA	To HLW Program,	NA	
Transuranic				·				Lance 4
Liquid	None ^a	Included in HLW	Included in HLW	Tank farm	Included in HLW	NA	NA	
Solid	271	None	NA	Containers on asphalt pads	15,370	WIPP or alternate	None	
) — J	بر به ا	,	<u>ي</u> د			facility-	* + <u>+</u>	
Mixed Transuranic						planieu		
Liquid	None ^a	Included in HLW	Included in HLW	Tank farm	Included in HLW	NA	NA	
Solid	98 ्	Included in TRU	Included in TRU	Containers on asphalt pads	15,370	WIPP or alternate	None	
$\frac{1}{2} = \frac{1}{2}$	•,	•				facility-	,	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1
Low-Level						planned		
Liquid	None	Evaporation, separation, solidification (vitrification)	Evaporator in 'service, new facilities planned	None	NA	NA	NA	
Solid	3,390	Compaction	4,000 ^f	Not stored	NA	Burial	902,900 ^g	
	· ,	: :,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			,		^	
• •		••• ••		بر ب	-	~		
1					•		·	
		•			- /*		.,	

4 . •

_ ` ·

Table 4.2.1.10–1. Projected Spent Nuclear Fuel and Waste Management Under No Action (2005) at Hanford Site

- ---

* . ** 1

·1 ·

.

۰. ۳

1 ...

1. A. 1. 2. . . .

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

~

4-76

1

. ..

8

. .

Category	Annual Generation (m ³)	Treatment Method	Treatment Capacity (m ³ /yr)	Storage Method	Storage Capacity (m ³)	Disposal Method	Disposal Capacity (m ³)
Mixed Low-Level			<u></u>	· · · · · · · · · · · · · · · · · · ·			
Liquid	3,760	Evaporation, ion exchange ^c	50,000	Storage tanks, basins planned	446,500 ^h	None	NA
Solid	1,505	None	NA	RCRA facility, retrievable	1,218,700	Landfill, LLW burial grounds 218-E-NN	Included in LLW
Hazardous			,				
Liquid	Included in solid	None	NA ′	RCRA building	Included in solid	Commercial ⁱ	NA
Solid	560	None	NA	RCRA building	127	Commercial ⁱ	NA
Nonhazardous (Sanitary)							
Liquid	414,000 ^j	None	NA	None	NA	Septic tanks, french drains	Expandable
Solid	5,107	None	NA	None	NA	Richland Sanitary Landfill	Expandable
Nonhazardous (Other)							
Liquid	Included in sanitary	None	NA	None	NA	Percolation ponds, leachfields	Expandable
Solid	Included in sanitary	None	NA	None	NA	Landfill	Expandable

Table 4.2.1.10–1. Projected Spent Nuclear Fuel and Waste Management Under No Action (2005) at Hanford Site—Continued

^b Vitrification planned.

^c Assumes 242-A Evaporator as treatment method for liquid HLW, liquid TRU, and liquid mixed TRU.

^d Consists of HLW and liquid TRU wastes in Double-Shell Tanks; Pu recovery and extraction aging waste. Includes 241-AN, 241-AP, 241-AY, 241-AZ, and 241-SY Tank Farms.

^e Disposal at WIPP would depend in part on decisions pursuant to the Waste Isolation Pilot Plant Disposal Phase Supplemental Environmental Impact Statement.

^f Compaction by LLW Compactor (213-W).

⁸ Includes the LLW Burial Grounds (unit 218-E-NN) and Low-Level Mixed Waste Disposal Facility (Project-W-025).

^h Assumes storage of liquid mixed LLW in tanks and planned basins.

ⁱ Offsite at RCRA facility.

^j Estimate based on 14,586 employees, 30 gal per day per employee, and 250 days per year of operation.

Note: NA=not applicable.

<u>1</u>

Source: 61 FR 9441; DOE 1992f; DOE 1993a; DOE 1993h; HF 1993a:1; HF 1995a:1.

-		Upg	rade		-	
Category	No Action ^a (m ³)	Without RFETS or LANL Material ^b (m ³) ⁵	With RFETS and LANL Material ^b (m ³)	Consolidation ^b (m ³)	Collocation ^b (m ³)	Phaseout (m ³)
Transuranic			5			0
Liquid	None	0	0	0.02	0.02	0.
Solid	271	20	21	10	10	0
Mixed Transuranic		-	j na		_	0
Liquid	None	0	- 0 `	0	0	U
Solid	· · · 98	0	0	4	4	0
Low-Level			-	ł	(0
Liquid	None	0.08°	0.08 ^c	2 ^c	2.10	0
Solid	- 3,390	85	89	1,260	1,300	0
Mixed Low-Level						<u> </u>
Liquid	3,760	× 0	, 0	0.2	0.2	0 Ò
Solid	_ 1,505	j 5 -	- 5	65	66	0 -
	n ta di Contra n n		•			1
Hazardous		21 20 , 1 1	· · ·	•	•	,
Liquid	- Included in solid	. 0.57	0.57	- 2 -	2	
Solid	560 - 5	- 4	4	2	2	0
Nonhazardous (Sanitary)		•	, -			0
Liquid	414,000	8,330	8,780	110,000	146,000	- U
Solid	5,107	917	967	1,140	1,760	0 °
Nonhazardous (Oth	ier)				•••••••	
Liquid	Included in sanitary	/ Included in sanitary	/ Included in sanitary	Included in sanitary	Included in sanitary	~ <u>~</u> U ,
Solid	Included in sanitary	y 0	- 0 - E	1,400 ^a	2,200	U,

Table 4.2.1.10–2. Estimated Annual Generated Waste Volumes at Hanford Site—No Action (2005) and Net Incremental for Storage Alternatives

^a The No Action waste volumes are from Table 4.2.1.10–1. $\epsilon_{\rm c}$

^b Generated waste volumes for storage alternatives shown in this table are found in Section E.3.1 (Tables E 3.1.1-1, E.3.1.1-5, E.3.1.2-1, and E.3.1.3-1). Waste effluents (that is, after treatment and volume reduction) which are used in the narrative description of the impacts are also provided in these tables.

-

.....

. 1

^c Liquid TRU and LLW would be treated and solidified prior to disposal.

^d Recyclable wastes.

5

4-78

扪

Management Programs Final Environmental Impact Statement (DOE/EIS-0203-F) and the ROD (61 FR 10736) from the follow-on tiered site-specific NEPA analysis, Final Environmental Impact Statement on the Management of Spent Nuclear Fuel from the K Basins at the Hanford Site, Richland, Washington (DOE/EIS-0245). TRU waste already packaged to current planning-basis WIPP WAC would either be stored or have been shipped. In compliance with the Federal Facility Compliance Act of 1992, mixed waste would have been treated and disposed of according to the Hanford Tri-Party Agreement. Solid LLW would continue to be buried at the onsite low-level disposal facility.

Upgrade Alternative

Upgrade Without Rocky Flats Environmental Technology Site or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

The modification of the FMEF or construction of a new storage facility for the continued storage of Pu would have a small impact on existing Hanford waste management activities. Construction waste volumes as presented in Table E.3.1.1–1 would have minimal impact on Hanford waste management activities. Waste generated during construction would consist of wastewater and solid nonhazardous and hazardous wastes. Nonhazardous waste would be disposed of as part of the construction project by the contractor, and the hazardous waste would be shipped offsite to commercial RCRA-permitted treatment and disposal facilities. Operational waste volumes as shown in Table 4.2.1.10–2 would increase slightly due to increased surveillance activities over No Action.

Approximately 20 m³ (26 yd³) of TRU waste from damaged PCVs and contaminated glovebox panels, windows, and gaskets would need to be treated and packaged to meet the current planning-basis WIPP WAC or alternative treatment level. While awaiting shipment to WIPP (depending on decisions made in the ROD associated with the supplemental EIS for the proposed continued phased development of WIPP for disposal of TRU waste), the TRU waste would be stored in above-grade storage facilities at the Hanford Central Waste Complex and the Transuranic Waste Storage and Assay Facility. Three additional truck shipments per year or, if applicable, two regular train shipments per year or one dedicated train shipment every 3 years, would be required to transport this waste to WIPP.

After treatment and volume reduction, approximately 42 m^3 (56 yd³) of LLW from solidified liquid LLW (such as decontamination solutions), protective clothing, HEPA filters, glovebox gloves, and decontamination equipment and materials would require disposal in the 200 Area LLW Burial Grounds. Assuming a land usage of 3,400 m³/ha (1,800 yd³/acre), this would require 0.01 ha/yr (0.03 acre/yr) of LLW disposal area.

Contaminated shielding and cleaning materials would be the major contributors to the 5 m³ (7 yd³) of mixed LLW. This small amount of mixed LLW could be treated and disposed of in accordance with the Hanford Tri-Party Agreement through the use of existing and planned facilities.

The 0.57 m³ (150 gal) of liquid hazardous waste such as lubricants, cleaning solvents, paint, and lube oil and 4 m^3 (5 yd³) of solid hazardous waste such as lead packing, wipes, and solid materials contaminated with oils, lubricants, and cleaning solvents would have minimal impact on waste management activities at Hanford. The hazardous wastes would be packaged in DOT-approved containers and shipped offsite to commercial RCRA-permitted treatment and disposal facilities.

Approximately 8,330 m³ (2,200,000 gal) of liquid nonhazardous waste to include sanitary, utility and process wastewaters, and cooling system blowdown would be processed using the 200 West Area Treatment Facility or one of the numerous septic tanks/subsurface disposal systems. Existing and planned liquid nonhazardous waste facilities are adequate. After volume reduction, approximately 459 m³ (600 yd³) of solid nonhazardous waste

such as clean non-Pu metals, packing materials, office trash, defective and damaged equipment, and industrial waste from utility and maintenance operations would be shipped to one of the onsite landfills.

Construct New 200 West Area Facility for Plutonium Storage

The construction and operation of a new storage facility for the continued storage of the current inventory of Pu would have a small impact on existing Hanford waste management activities. The impacts are identical to those identified in the preceding option of modifying the FMEF.

Upgrade With All or Some Rocky Flats Environmental Technology Site and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing Fuels and Materials Examination Facility for Plutonium Storage

As shown in Table E.3.1.1-5 construction waste volumes would increase for the additional required construction. The types of operational waste are identical to those discussed earlier, but there would be a small increase in volume. Approximately 21 m^3 (27 yd^3) of TRU waste would be treated and packaged to meet the current planning-basis WIPP WAC or alternative treatment level. While awaiting shipment to WIPP (depending on decisions resulting from the supplemental PEIS noted earlier), the TRU and mixed TRU waste would be stored in above-grade storage facilities in the Hanford Central Waste Complex and the Transuranic Waste Storage and Assay Facility. Three additional truck shipments per year or, if applicable, two regular train shipments per year or one dedicated train shipment every 3 years, would be required to transport these wastes to WIPP.

After treatment and volume reduction, approximately 45 m^3 (59 yd³) of LLW would require disposal in the 200 Area LLW Burial Grounds. Assuming a land usage of 3,400 m³/ha (1,800 yd³/acre), this would require 0.01 ha/yr (0.03 acre/yr) of LLW disposal area. The 5 m³ (7 yd³) of solid mixed LLW would be treated and disposed of in accordance with the Hanford Tri-Party Agreement through the use of existing and planned facilities. The 0.57 m³ (150 gal) of liquid hazardous wastes and 4 m³ (5 yd³) of solid hazardous wastes would have minimal impact on waste management activities at Hanford. The hazardous wastes would be packaged in DOT-approved containers and shipped offsite to commercial RCRA-permitted treatment and disposal facilities. Approximately 8,780 m³ (2,320,000 gal) of liquid nonhazardous wastes to include sanitary, utility and process wastewaters, and cooling system blowdown would be processed using the 200 West Area Treatment Facility or one of the numerous septic tanks/subsurface disposal systems. Existing and planned liquid nonhazardous waste facilities are adequate. After volume reduction, approximately 483 m³ (632 yd³) of solid nonhazardous waste would require disposal at one of the onsite landfills.

Distributing the RFETS and LANL material to more than one site would reduce the operational waste volumes. The decrease would be proportional to the amount of material.

Construct New 200 West Area Facility for Plutonium Storage

The impacts of constructing and operating a new storage facility to include RFETS and LANL Pu would be identical to those identified in the preceding option of modifying the FMEF to include RFETS and LANL Pu.

Consolidation Alternative

Construct New Plutonium Storage Facility

Construction and operation of a consolidated Pu storage facility would have an impact on existing Hanford waste management activities, increasing the generation of TRU, low-level, mixed, hazardous, and nonhazardous wastes. Waste generated during construction would consist of wastewater and solid nonhazardous

; [

and hazardous wastes. The solid nonhazardous waste would be disposed of as part of the construction project by the contractor, and the hazardous waste would be shipped to commercial RCRA-permitted treatment and disposal facilities. No soil contaminated with hazardous or radioactive constituents is expected to be generated during construction. However, if any was generated it would be managed in accordance with site practice and all applicable Federal and State regulations. The types of operational wastes from the consolidated Pu storage facility would be the same as those from the Upgrade Alternative, but the quantity would change.

After treatment and volume reduction of TRU waste, approximately 5 m^3 (7 yd³) of TRU waste and 4 m^3 (5 yd³) of mixed TRU waste from leaded gloves and windows and contaminated lead shielding would be treated and packaged to meet the current planning-basis WIPP WAC or alternative treatment level. While awaiting shipment to WIPP (depending on decisions resulting from the supplemental EIS noted earlier), the TRU and mixed TRU wastes would be stored in above-grade storage facilities in the Hanford Central Waste Complex and the TRU Waste Storage and Assay Facility. One additional truck shipments per year or, if applicable, one regular train shipment every 2 years or one dedicated train shipment every 6 years, would be required to transport these wastes to WIPP.

Following treatment and volume reduction, approximately 630 m³ (824 yd³) of LLW would require disposal in the 200 Area LLW Burial Grounds. Assuming a land usage of 3,400 m³/ha (1,800 yd³/acre), this would require approximately 0.2 ha/yr (0.5 acre/yr) of LLW disposal area. The 0.2 m³ (50 gal) of liquid mixed LLW and 65 m³ (85 yd³) of solid mixed LLW would be treated and disposed of in accordance with the Hanford Tri-Party Agreement through the use of existing and planned facilities. The 2 m³ (476 gal) of liquid hazardous waste and 2 m³ (3 yd³) of solid hazardous waste would have minimal impact on waste management activities at Hanford. The hazardous wastes would be packaged in DOT-approved containers and shipped offsite to commercial RCRA-permitted treatment and disposal facilities. Approximately 110,000 m³ (29,000,000 gal) of liquid nonhazardous waste would be treated and recycled by the consolidated facility. After volume reduction, 570 m³ (746 yd³) of solid nonhazardous waste would require disposal at one of the onsite landfills.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Construction and operation of a consolidated Pu storage facility collocated with HEU storage would have an impact on existing Hanford waste management activities, increasing the generation of TRU, low-level, mixed, hazardous, and nonhazardous wastes. Waste generated during construction would consist of wastewater and solid nonhazardous and hazardous wastes. The solid nonhazardous waste would be disposed of as part of the construction project by the contractor, and the hazardous waste would be shipped to commercial RCRA-permitted treatment and disposal facilities. No soil contaminated with hazardous or radioactive constituents is expected to be generated during construction. However, if any was generated it would be managed in accordance with site practice and all applicable Federal and State regulations. Since there is no TRU or mixed TRU wastes associated with HEU storage, the impacts from TRU and mixed TRU wastes are identical to those identified in the consolidated Pu storage alternative. The sources of waste are similar to those of the Pu storage facilities except the source of radioactive contamination from HEU storage is uranium.

Following treatment and volume reduction, approximately 630 m³ (824 yd³) of LLW contaminated with Pu and 20 m³ (26 yd³) of LLW contaminated with uranium would require disposal in the 200 Area LLW Burial Grounds. Assuming a land usage of 3,400 m³/ha (1,800 yd³/acre), this would require approximately 0.2 ha/yr (0.5 acre/yr) of LLW disposal area. The 0.2 m³ (55 gal) of liquid mixed LLW and 66 m³ (86 yd³) of solid mixed LLW would be treated and disposed of in accordance with the Hanford Tri-Party Agreement through the use of existing and planned facilities. The 2 m³ (530 gal) of liquid hazardous waste and 2 m³ (3 yd³) of solid hazardous waste would have minimal impact on waste management activities at Hanford. The hazardous wastes would be packaged in DOT-approved containers and shipped offsite to commercial RCRA-permitted treatment and disposal facilities. The 146,000 m³ (39,000,000 gal) of liquid nonhazardous waste would require construction

of sanitary, utility, and process wastewater treatment systems. After volume reduction 880 m³ (1,150 yd³) of solid nonhazardous wastes would require disposal at one of the onsite landfills.

કેટ ક

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials - 1. 法书书: まんちょう いい

The exclusion of strategic reserve and weapons R&D materials would reduce the amount of operational waste volumes shown in Table 4.2.1.10-2 for the Upgrade With All or Some RFETS and LANL Pu Subalternative, the Consolidation Alternative, and the Collocation Alternative. The decrease would be proportional to the amount of material excluded. [Text deleted.]

ž

Phaseout

The phaseout of Pu storage would have no impact on Hanford waste management activities. The volume of waste would not decrease until the facilities in which Pu is stored were D&D. ۰. :. -t,

÷,

やう事に相比較にないたりの、利用での記録は目出していた日本

4.2.2 NEVADA TEST SITE

同時にいたい

A listing of the proposed long-term storage alternatives, subalternatives, and related actions, including the No Action Alternative, at NTS is provided below. The potential impacts of implementing these alternatives and related actions at NTS are described in the following sections: land resources, site infrastructure, air quality and noise, water resources, geology and soils, biological resources, cultural and paleontological resources, socioeconomics, public and occupational health and safety, and waste management. The specific longterm storage alternatives proposed for NTS are the Consolidation Alternative and the Collocation Alternative.

11' F ...



Proposed Storage Activities at Nevada Test Site No Action Alternative (Preferred Alternative): There is no Pu or HEU storage mission currently at NTS; does not add Pu or HEU storage at NTS. Upgrade Alternative: This storage alternative does not apply to NTS. Consolidation Alternative: Two options to accommodate all Pu material within the scope of this PEIS: Modify the existing network of tunnel drifts and construct a new material handling building at the P-Tunnel; or construct a new facility near DAF. Collocation Alternative: Two options to accommodate all Pu and HEU material within the scope of this PEIS: Modify the existing network of tunnel drifts and construct a new facility near DAF.

• Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials: Facility and other resource requirements would be smaller than the Consolidation Alternative and the Collocation Alternative.

• **Phaseout:** This storage activity does not apply to NTS.

4.2.2.1 Land Resources

Preferred Alternative: No Action Alternative

Neither Pu nor HEU is presently stored at NTS. Under the No Action Alternative, existing and planned missions at NTS would continue. The ongoing (no new action) activities would conform with present and future land-use plans, policies, and controls. No effects to land resources would be anticipated at NTS beyond those of existing and future activities that are independent of this action.

Consolidation Alternative

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

This option would modify the existing P-Tunnel and construct a new material handling building in Area 12 to accommodate all Pu material within the scope of this PEIS. During construction, 29 ha (72 acres) of land area would be required of which 27 ha (68 acres) would be used during operation. Construction laydown area and the operating facility would be situated entirely on previously disturbed land and would not create any newly, disturbed area. As a modification of an existing facility, the 1.6-km (1-mi) buffer zone is established.

10 - E - A

「「「「「「「「」」」」」

Land Use. Utilization of the P-Tunnel in Area 12 would not conform with the master plan of the Nevada Test Site Development Plan, which designates the North area of NTS as an underground nuclear weapons test area (NT DOE 1995d:7). However, the P-Tunnel is a potential site for long-term storage and disposition of weaponsusable fissile materials as part of the NTS defense program materials disposition activities considered under the Expanded Use Alternative (part of the Preferred Alternative) of the Final Environmental Impact Statement for the Nevada Test Site and Off-Site Locations in the State of Nevada (NTS EIS) (NT DOE 1996c:3-8,3-9; NT DOE 1996e:A-18). Should the Consolidation Alternative P-Tunnel option be selected, the Expanded Use Alternative of the NTS EIS could be used to revise the current Nevada Test Site Development Plan. With these changes, any required construction would be consistent with the land use plan. As discussed in Section 4.2.2.8, sufficient available labor exists within the region to fill the jobs created during construction and operations. There would be no increased demand for housing. Therefore, no indirect impacts on offsite land use would be anticipated.

Use of the P-Tunnel would not affect special status lands as shown in Figure 3.3.1–1. The buffer zone and security area associated with the long-term storage alternative would preclude development within the immediate area. However, the NTS EIS Expanded Use Alternative indicates that adequate land area is available at NTS for facility siting (NT DOE 1996c:3-14, 3-15). The proposal would not affect offsite grazing allotments. No prime farmlands exist onsite. The alternative would not be in conflict with land-use plans, policies, or controls of adjacent jurisdictions since none of these counties or municipalities currently undertake land-use planning. Storage of Pu in the P-Tunnel could impact weapons effects testing ability. It is likely that the P-Tunnel be closed during testing, although it could be kept manned if appropriate safety considerations were met (NT DOE 1995e:1). However, the potential for impacts could be eliminated by test or tunnel design (NT DOE 1996f:1).

Visual Resources. [Text deleted.] Construction and operation of the facility would be compatible with the existing industrialized landscape character of Area 12 and the current VRM Class 5 designation. Although U.S. Route 95 is a heavily traveled public roadway, travelers are unable to view Area 12 facilities because of mountainous terrain and distance.

Construct New Plutonium Storage Facility

All Pu within the scope of this PEIS would be stored at a new storage facility to be constructed at NTS Area 6 near the DAF. The consolidated Pu storage plant at NTS would disturb 58.5 ha (144 acres) of land area during

ţ

3

;

construction of which 56 ha (138 acres) would be used during operation. A buffer zone would be provided between the facility and the NTS site boundary. Pu storage in existing storage facilities at other DOE sites would be phased out.

Land Use. Construction and operation of the Pu facility would convert undeveloped land in Area 6. The proposed action would not conform with the current *Nevada Test Site Development Plan*, which designates the southeast area of NTS as a nonnuclear test area (NT DOE 1995d:7). However, Area 6 is a potential site for long-term storage and disposition of weapons-usable fissile materials as part of the NTS defense program materials disposition activities considered under the Expanded Use Alternative (part of the Preferred Alternative) of the NTS EIS (NT DOE 1996c:3-8,3-9; NT DOE 1996e:A-18). As discussed in Section 4.2.2.8, no in-migration of workers would be required during construction and operations. No increase in housing demand would be anticipated, with offsite land use not subject to indirect land-use impacts.

Construction and operation would not affect other land uses at NTS or special status lands. The buffer zone and security area associated with the long-term storage alternative would preclude development within the immediate area. However, the Expanded Use Alternative of the NTS EIS indicates that adequate land area is available at NTS for facility siting (NT DOE 1996c:3-14,3-15). The alternative would not affect offsite grazing allotments. No prime farmlands exist onsite. The alternative would not be in conflict with land-use plans, policies, or controls of adjacent jurisdictions since none of these counties or municipalities currently undertake land-use planning.

Visual Resources. [Text deleted.] Construction and operation of the facility would be compatible with the industrial landscape character of the adjacent DAF and the current VRM Class 5 designation of Area 6. Views of the alternative would be blocked from sensitive viewpoints accessible to the public by mountainous terrain.

Collocation Alternative

「「「「「「「」」」」」

1

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Under this action, the existing P-Tunnel located in Area 12 is proposed to be utilized, and a new material handling building would be constructed at the P-Tunnel. Land disturbance would be 29 ha (72 acres) during construction of which 27 ha (68 acres) would be used during operations. Construction laydown area and the operating facility would be situated entirely on previously disturbed land and would not create any newly disturbed area. Effects to land resources during construction and operation would be similar to those of the new and modified P-Tunnel for the Consolidation Alternative. As discussed in Section 4.2.2.8, in-migration would occur only during the operation phase. Projected vacancies within the housing stock would be sufficient to accommodate the slight increase in demand. Therefore, no indirect effects to offsite land use would be anticipated.

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

The new storage facility would be located on undisturbed land in Area 6 near the DAF and would disturb a land area of 89.5 ha (221 acres) during construction of which 87 ha (215 acres) would be used during operations. A buffer zone would be provided between operations and the NTS site boundary. Direct and indirect land resources effects would be similar to the Consolidation Alternative, new storage facility.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Under this subalternative, land effects during construction and operation would be almost the same in extent and magnitude to the Consolidation Alternative and Collocation Alternative because the facility would be almost the

same. However, because the smaller quantity of material would require smaller facilities, it is likely that less land area would be disturbed during construction and used during operations. [Text deleted.]

[Text deleted.] '

a - - t

A STATE AND A STATE AN

うたちからいのかった

「「「「」」のできてい

11

4.2.2.2 Site Infrastructure

Nevada Test Site had an extensive infrastructure to handle the underground test program. With the cessation of nuclear testing, many of the operations around the site have been terminated. However, the facilities remain in place and are considered to be available.

Preferred Alternative: No Action Alternative

The infrastructure currently in place at NTS is capable of handling all anticipated missions and functions associated with the No Action Alternative.

Consolidation Alternative

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Construction to modify the P-Tunnel and construct a new material handling building to accommodate long-term storage of Pu at NTS would not impact the site infrastructure. Data for construction are presented in Appendix C. Operations impacts to NTS infrastructure under this option are in the areas of electrical energy and fuel requirements for the site. As shown in Table 4.2.2.2–1, additional electrical energy would be required to operate the facility. A small amount of oil would be required to operate the modified P-Tunnel for storage of Pu. Since oil availability is governed by usage and not by storage capacity onsite, the additional oil could be procured through normal contractual means. The preconceptual facility design uses natural gas as a fuel source. The final facility design for NTS would be converted to an energy source already available at NTS. With this conversion from natural gas to oil, site infrastructure requirements are within site capacities.

Construct New Plutonium Storage Facility

Constructing a new storage facility to accommodate long-term storage of Pu at NTS would not affect the site infrastructure. Data for construction are presented in Appendix C. Operations impacts to NTS infrastructure under this option are in the area of fuel requirements. As shown in Table 4.2.2.2–1, a small amount of oil would be required to operate the new facility for storage of Pu. Since oil availability is governed by usage and not by storage capacity on site, the additional oil could be procured through normal contractual means. Adequate electrical energy is available from the regional power grid. The preconceptual facility design uses natural gas as a fuel source. The final facility design for NTS would be converted to an energy source already available at NTS. With this conversion from natural gas to oil, site infrastructure requirements are within site capacities.

Collocation Alternative

I

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Construction to modify the P-Tunnel and construct a new material handling building to accommodate long-term storage of Pu and HEU at NTS would not affect site infrastructure. Data for construction are presented in Appendix C. Operations impacts to NTS infrastructure under this option are in the areas of electrical energy and fuel requirements for the site. As shown in Table 4.2.2.2–1, additional electrical energy would be required to operate the facility. A small amount of oil would be required to operate the modified P-Tunnel for storage of Pu and HEU. Since oil availability is governed by usage and not by storage capacity on site, the additional oil could be procured through normal contractual means. Adequate electrical energy is available from the regional power grid. The preconceptual facility design uses natural gas as a fuel source. The final facility design for NTS would be converted to an energy source already available at NTS. With this conversion from natural gas to another energy source, site infrastructure requirements are within site capacities.

-	Trans	sportation	Elec	trical	- -	Fuel	,	
Alternative	Roads (km)	Railroads (km)	Energy (MWh/yr)	Peak Load (MWe)	Oil (l/yr)	Natural Gas	· Coal	-
No Action						(iii / j1)	(uji)	-
Site availability	1100 ^a	0	176,844	45	5.716.000	0	0	
Projected usage	645	0	124,940	25	5,716,000	0	ົ່ ເ	
Consolidation			·		-	Ū	U	
Modify Existing Tunnel Drifts and Construct New Material Handling Building				*		r.		
Projected usage with consolidated facility	650	0	196,940	35	5.754.000	3 200 000	0	
Amount required in excess to site availability	0	0	20,096	0	38,000	3,200,000	° 0	* * *
New Pu Storage Facility				-	•			
Projected usage with consolidated facility	650	0	173,940	33	, 5.754.000	2,800,000	. 0	
Amount required in excess to site availability	0	0	0	0.	38,000	2,800,000	0	, , , ,
Collocation				14 %				,
Modify Existing Tunnel Drifts and Construct New Material Handling Building				2° 4 4		· · ·		
Projected usage with consolidated and collocated upgrade facilities	650	0	213,940	38	5,754,000	3,600,000	0	`
Amount required in excess to site availability	0	0	37,096	0 .	38,000 ^b	3,600,000 ^c	· · 0	هم د کې د کې
New Pu and HEU Storage Facilities								- *
Projected usage with and new collocated facilities	650	0	189,940	36	5,754,000	3,200,000	. 0	-
Amount required in excess to site availability	0	0	13,096	0	38,000 ^b	3,200,000 ^c	0	

Table 4.2.2.2–1. Site Infrastructure Changes Required for Operation at Nevada Test Site (Annual)—No Action (2005) and Storage Alternatives

⁹ Fuel oil requirements in excess to site availability could be procured through normal contractual means.

^c Facility would be adapted to use fuel oil instead of natural gas. Note: Modified from NTS 1993a:4.

4188

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

1

ことういうののないないないのです。

Constructing a new storage facility to accommodate long-term storage of Pu and HEU at NTS would not affect the site infrastructure. Data for construction are presented in Appendix C. Operations impacts to NTS infrastructure under this option are in the areas of electrical energy and fuel requirements. As shown in Table 4.2.2.2–1, additional electrical energy would be required to operate the facility. A small amount of oil would be required to operate the new facility for storage of Pu and HEU. Since oil availability is governed by usage and not by storage capacity on site, the additional oil could be procured through normal contractual means. Adequate electrical energy is available from the regional power grid. The preconceptual facility design uses natural gas as a fuel source. The final facility design for NTS would be converted to an energy source already available at NTS. With this conversion from natural gas to oil, site infrastructure requirements are within site capacities.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

With a change to the preconceptual facility designs that would allow use of a fuel source already in place at NTS, the existing site infrastructure would be fully capable of supporting construction/modification and operation of facilities for the Consolidation of Pu and Collocation of Pu and HEU Alternatives. With this change, constructing and operating such alternatives, without including provisions for storage of strategic reserve and weapons R&D materials could be accommodated as well. Expected reductions in annual electrical energy requirements from that of the various storage alternatives for all the nonsurplus materials are the only site infrastructure changes expected if this subalternative is chosen because electric usage is dependent on the amount of material. [Text deleted.]

4.2.2.3 Air Quality and Noise

Construction and operation activities associated with the No Action Alternative and the proposed storage alternatives would generate criteria and toxic/hazardous pollutants. To evaluate the air quality impacts at NTS, criteria and toxic/hazardous concentrations from the No Action Alternative and the proposed storage alternatives are compared with Federal and State standards and guidelines. Impacts from radiological airborne emissions are described in Section 4.2.2.9.

ه و

27.24.2

A STATE OF THE PARTY AND A STATE OF

In general, all of the proposed storage facilities would emit the same types of air pollutants during construction. It is expected emissions would not exceed Federal, State, or local air quality regulations. PM_{10} concentrations will be increased especially during peak construction periods.

The principal sources of emissions during construction include the following:

- Fugitive dust from land clearing, site preparation, excavation, and wind erosion of exposed ground surfaces
- Exhaust and road dust generated by construction equipment, vehicles delivering construction materials, and vehicles carrying construction workers

During operation, impacts from each of the individual storage facilities with respect to the concentrations of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Table 4.2.2.3–1 presents the estimated pollutant concentrations for each of the storage alternatives, indicating little difference between alternatives with respect to impacts to air quality.

Emission rates attributed to operation of the proposed storage facilities are presented in Tables F.1.3–2 and F.1.3–3. [Text deleted.] Air pollutant emission sources associated with operations include the following:

- · Operation of boilers for space heating
- Operation of diesel generators and periodic testing of emergency diesel generators
- Exhaust and road dust generated by vehicles delivering supplies and bringing employees to work
- Toxic/hazardous pollutant emissions from facility processes

Noise impacts during either construction or operation are expected to be low. Air quality and noise impacts for each storage alternative are described separately. Supporting data for the air quality and noise analyses are presented in Appendix F.

AIR QUALITY

I

An analysis was conducted of the potential air quality impacts of emissions from each of the storage alternatives as described in Section 4.1.3.

Section 176 (c) of the 1990 CAA Amendments requires that all Federal actions conform with the applicable SIP. EPA has implemented rules that establish the criteria and procedures governing the determination of conformity for all Federal actions in nonattainment and maintenance areas. These are discussed in Section 4.1.3. The attainment status of the area in which NTS is located is discussed in Section 3.3.3. Since the area is considered to be an attainment area for the criteria pollutants, the proposed actions at this site do not require that a conformity analysis be performed.

4-90 ·

•

			<u>.</u>	Conse	olidation	Coll	ocation
Pollutant	Averaging Time	Most Stringent Regulations or Guidelines ^a (µg/m ³)	- No Action (µg/m ³)	New Storage Facility (µg/m ³)	Modify P-Tunnel (µg/m ³)	New Storage Facilities (µg/m ³)	Modify P-Tunnel (µg/m ³)
Criteria Pollutants		······································					
Carbon monoxide	8-hour	10,000 ^b ~~	2,290	2,290.29	2,290.54	2,290.33	2,290.60
,	1-hour	40,000 ^b	2,748	2,750.02	2,751.77	2,750.30	2,752.22
Lead ^	Calendar	1.5 ^b	c	c	c	c	C
	Quarter						· ·
Nitrogen dioxide	Annual	100 ^b	c	<0.01 ^d	0.01 ^d	<0.01 ^d	0.01 ^a
Ozone	1-hour	235 ^b	e	c	c	c	c
Particulate matter less than or equal to 10 microns in diameter	Annual	50 ^b	9.4	9.4	9.4	9.4	9.4
	24-hour	150 ^b	106	106	106	106	106
Sulfur dioxide	Annual	80 ^b	8.4	8.4	8.4	8.4	8.4
	24-hour	365 ^b	94.6	94.6	94.6	94.6	94.6
	3-hour	1,300 ^b	725	725	725	725	725
Mandated by Nevada Hydrogen sulfide	1-hour	112 ^f	с	c	c	с	÷ c

Table 4.2.2.3–1. Estimated Operational Concentrations of Pollutants at Nevada Test Site and
Comparison With Most Stringent Regulations or Guidelines—No Action (2005)
and Storage Alternatives

.

	, ,		1	-	Conse	lidation '	Col	location
Pollutant	, , , ,	Averaging Time	Most Stringent Regulations or Guidelines ^a (µg/m ³)	No Action (µg/m ³)	New Storage Facility (µg/m ³)	Modify P-Tunnel (µg/m ³)	New Storage Facilities (μg/m ³)	Modify P-Tunn (ue/m ³)
azardous and Other 7 Compounds	Foxic		4	-	, 3 · ·			<u>, , , , , , , , , , , , , , , , , , , </u>
hlorine		8-hour	35.7 ^f	d	< 0.01 ^d	<0.01 ^d	<0.01d	~0.01d :
ydrogen chloride	•	8-hour	g	đ	<0.01 ^d 1	< 0.01 ^d	<0.01 ^d	<0.01 ^d
ydrazine		8-hour	3.1 ^f	đ	<0.01 ^d	<0.01 ^d	<0.01 ^d	<0.01 ^d
itric acid		{ 8-hour	123.8 ^f	ď	<0.01 ^d	<0.01 ^d	(0.01 ^d	<0.01 <0.01
hosphoric acid	· ·	8-hour	23.8 ^f	🔨 d	<0.01 ^d	<0.01 ^d	<0.01 ^d	<0.01 ^d
		·	22 of	đ	<0.01 ^d	~0.01d	<0.01 ^d	<0.01 J
ulfuric acid The more stringent of the l Federal and State stands To sources of this pollutar The concentration represent	Federa ard. nt have	8-hour l and State stan been identified alternative con	dard is presented if bo	th exist for the a	veraging time.	et '	<u><u> </u></u>	
Ilfuric acid he more stringent of the l dederal and State stands to sources of this pollutar he concentration represen ozone, as a criteria polluta tate standard or guideline	Federa ard. nt have nts the ant is n	8-hour l and State stan been identified alternative com ot directly emit	dard is presented if bo tribution only. ted or monitored by th	oth exist for the a	veraging time. See Section 4.1.3 for	r a discussion of ozone-re	elated issues.	~~~~~
Ilfuric acid he more stringent of the l ederal and State standa o sources of this pollutar he concentration represen zone, as a criteria polluta tate standard or guideline ot Applicable.	Federa ard. nt have nts the ant is n	8-hour l and State stan been identified alternative cont ot directly emit	dard is presented if bo tribution only. ted or monitored by th	oth exist for the a	veraging time. See Section 4.1.3 for	r ^t	elated issues.	
Ilfuric acid the more stringent of the l ederal and State stands to sources of this pollutar the concentration represent zone, as a criteria pollutar tate standard or guideline tot Applicable. e: Concentrations are base concentrations concentrations are base concentrations concentrations concentrations are base concentrations concentrations	Federa ard. Int have nts the ant is n sed on	8-hour l and State stan been identified alternative com ot directly emit site contributio	dard is presented if bo tribution only. ted or monitored by th	oth exist for the a ne candidate site. ations from ongoi	veraging time. See Section 4.1.3 for ing activities (No Act	r a discussion of ozone-re	elated issues.	ion-facility sources (fo
Ilfuric acid the more stringent of the l dederal and State stands to sources of this pollutar the concentration represen zone, as a criteria polluta tate standard or guideline tot Applicable. te: Concentrations are base example, traffic). arce: 40 CFR 50: DOE 19	Federa ard. In thave not the ant is n sed on	8-hour l and State stand been identified alternative cont ot directly emit site contributio	dard is presented if bo tribution only. ted or monitored by th n, including concentra	th exist for the and the candidate site.	veraging time. See Section 4.1.3 for ing activities (No Act	r a discussion of ozone-re	elated issues.	ion-facility sources (fo
alfuric acid the more stringent of the l rederal and State stands to sources of this pollutar the concentration represen- zone, as a criteria polluta- tate standard or guideline tot Applicable. te: Concentrations are bas- example, traffic). arce: 40 CFR 50; DOE 19	Federa ard. Int have nts the ant is n sed on 996e; D	8-hour l and State stan been identified alternative com ot directly emit site contributio POE 1996f; NT	dard is presented if bo tribution only. ted or monitored by th n, including concentra DOE 1996a; NV DCI	oth exist for the a ne candidate site. ations from ongoi NR 1992a; NV D	veraging time. See Section 4.1.3 for ing activities (No Act CNR 1995a.	t a discussion of ozone-re	elated issues.	con-facility sources (fo
Ilfuric acid the more stringent of the l rederal and State standa to sources of this pollutar the concentration represen- zone, as a criteria pollutar tate standard or guideline tot Applicable. te: Concentrations are bas- example, traffic). arce: 40 CFR 50; DOE 19	Federal ard	8-hour l and State stand been identified alternative cont ot directly emit site contribution POE 1996f; NT	dard is presented if bo tribution only. ted or monitored by th n, including concentra DOE 1996a; NV DCI	oth exist for the a ne candidate site. ations from ongoi NR 1992a; NV D	veraging time. See Section 4.1.3 for ing activities (No Act CNR 1995a.	a discussion of ozone-re	elated issues.	con-facility sources (fo
Ifuric acid the more stringent of the lederal and State standard o sources of this pollutar the concentration represent zone, as a criteria pollutar ate standard or guideline of Applicable. e: Concentrations are base example, traffic). rce: 40 CFR 50; DOE 19	Federa ard. nt have nts the ant is n sed on 096e; D	8-hour l and State stan been identified alternative cont ot directly emit site contributio OE 1996f; NT	dard is presented if bo tribution only. ted or monitored by th n, including concentra DOE 1996a; NV DCI	oth exist for the a ne candidate site. ations from ongoi NR 1992a; NV D	veraging time. See Section 4.1.3 for ing activities (No Act CNR 1995a.	a discussion of ozone-re	elated issues.	ion-facility sources (fc
Ilfuric acid he more stringent of the l ederal and State standa o sources of this pollutar he concentration represen zone, as a criteria polluta tate standard or guideline ot Applicable. e: Concentrations are bas example, traffic). rrce: 40 CFR 50; DOE 19	Federa ard. it have nts the ant is n sed on 996e; D	8-hour l and State stan been identified alternative com ot directly emit site contribution POE 1996f; NT	dard is presented if bo tribution only. ted or monitored by th n, including concentra DOE 1996a; NV DC	oth exist for the a ne candidate site. ations from ongo NR 1992a; NV D	veraging time. See Section 4.1.3 for ing activities (No Act CNR 1995a.	t a discussion of ozone-re	elated issues.	ion-facility sources (fo
Infuric acid the more stringent of the l rederal and State stands to sources of this pollutar the concentration represen zone, as a criteria polluta tate standard or guideline tot Applicable. te: Concentrations are bas example, traffic). arce: 40 CFR 50; DOE 19	Federa ard. In thave ant is n ant is n sed on 296e; E	8-hour l and State stand been identified alternative cont ot directly emit site contribution POE 1996f; NT	dard is presented if bo tribution only. ted or monitored by th n, including concentra DOE 1996a; NV DCI	oth exist for the a ne candidate site. ations from ongoi NR 1992a; NV D	veraging time. See Section 4.1.3 for ing activities (No Act CNR 1995a.	a discussion of ozone-re	elated issues.	ton-facility sources (fo
Ilfuric acid he more stringent of the l ederal and State standa to sources of this pollutar he concentration represen zone, as a criteria polluta tate standard or guideline tot Applicable. e: Concentrations are bas example, traffic). arce: 40 CFR 50; DOE 19	Federa ard. In thave nuts the ant is n sed on 096e; D	8-hour l and State stan been identified alternative com ot directly emit site contributio	dard is presented if bo tribution only. ted or monitored by th n, including concentra DOE 1996a; NV DCI	oth exist for the a ne candidate site. ations from ongoi NR 1992a; NV D	veraging time. See Section 4.1.3 for ing activities (No Act CNR 1995a.	t a discussion of ozone-re	elated issues.	ion-facility sources (fo

Table 4.2.2.3–1. Estimated Operational Concentrations of Pollutants at Nevada Test Site and Comparison With Most Stringent Regulations or Guidelines—No Action (2005) and Storage Alternatives—Continued

ı.

4-92

.

- 32 A

Preferred Alternative: No Action Alternative

This alternative utilizes estimated air emissions data from operations at NTS assuming continuation of site missions as described in Section 3.3. The emission rates for the criteria and toxic/hazardous pollutants for No Action are presented in Table F.1.2.3-1. Table 4.2.2.3-1 presents the No Action concentrations for the total site. Concentrations of all criteria and toxic/hazardous air pollutants at the site boundary are expected to remain within applicable Federal, State, and local ambient air quality standards.

Consolidation Alternative

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

In addition to the sources of emissions during construction associated with the No Action Alternative, fugitive dust resulting from the operation of a concrete batch plant may be an additional emission source associated with a new facility.

Increases in PM_{10} concentrations may occur during the peak construction period for a new facility and during dry and windy conditions. Appropriate control measures would be followed to minimize pollutant concentrations during construction. Concentrations of all pollutants at the site boundary would remain within applicable Federal and State ambient air quality standards during construction.

During operation of the modified P-Tunnel, impacts with respect to the concentrations of criteria and toxic/ hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Estimated pollutant concentrations attributable to increased operations associated with this storage alternative, plus the No Action concentrations, are presented in Table 4.2.2.3–1.

Construct New Plutonium Storage Facility

The new storage facility option would have air quality impacts similar to those of the modified P-Tunnel, with the following exceptions. During operation, emissions would be slightly lower, as shown in Appendix F. Impacts for the new storage facility option with respect to the concentrations of criteria pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Estimated pollutant concentrations attributable to increased operations associated with this option for the storage alternative, plus the No Action concentrations, are presented in Table 4.2.2.3-1.

Collocation Alternative

State of the state of the state

いいかん あまんがい

Atrices & A

This at margine with

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

The P-Tunnel option would have slightly higher emissions than for the consolidation of Pu modified P-Tunnel, as shown in Appendix F. Impacts for this alternative are also expected to be in compliance with Federal, State, and local air quality regulations and guidelines. Estimated pollutant concentrations attributable to increased operations associated with this option for the storage alternative, plus the No Action concentrations, are presented in Table 4.2.2.3–1.

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

The new storage facility option would be located in the same area as the consolidation of Pu new storage facility and would have similar air quality impacts with the following exceptions.

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

During operation, emissions would be slightly higher than for consolidation of Pu new storage facility option, as shown in Appendix F. Impacts for the new storage facilities option with respect to the concentrations of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Estimated pollutant concentrations attributable to increased operations associated with this option for the storage alternative, plus the No Action concentrations, are presented in Table 4.2.2.3–1.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Air quality impacts for construction and operations for this subalternative are expected to be similar to those previously described for the Consolidation Alternative and the Collocation Alternative. [Text deleted.]

NOISE

The location of the storage facilities relative to the site boundary and sensitive receptors was examined to evaluate the potential for onsite and offsite noise impacts. Noise sources during construction may include heavy construction equipment and increased traffic. Increased traffic would occur onsite and along offsite local and regional transportation routes used to bring construction material and workers to the site.

Preferred Alternative: No Action Alternative

Nontraffic noise sources associated with continued interim storage and other ongoing missions would be the same as described in Chapter 3. The continuation of operations at NTS would result in no appreciable change in traffic noise and onsite operational noise sources from current levels. Nontraffic noise sources are located at sufficient distance from offsite areas that the contribution to offsite noise levels would continue to be small. Due to the size of the site, noise emissions from construction equipment and operations activities would not be expected to cause annoyance to the public. Some noise sources may be located close enough to onsite noise sensitive areas to result in impacts, such as disturbance of wildlife.

Consolidation and Collocation Alternatives

Nontraffic, operational noise sources associated with the storage alternatives include existing or additional equipment and machines (cooling systems, vents, motors, and material handling equipment). These noise sources would be located at sufficient distance from offsite areas that the contribution to offsite noise levels would be small. Due to the size of the site, noise emissions from construction equipment and operations activities would not be expected to cause annoyance to the public. Some noise sources may result in impacts, such as disturbance of wildlife. e

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Noise impacts for construction and operations for this option are expected to be almost the same as those previously described for the Consolidation Alternative and the Collocation Alternative because noise impacts are based on the use of the facility and not the size. [Text deleted.] 7 2 F.A.T c tree

1 .

_ 1

The A C. A.L.

. 5

11.1

4.2.2.4 Water Resources

のないないであるという

ţ

ł

Construction and operation of the potential long-term storage facilities at NTS could affect water resources. All water required for construction and operation would be supplied from groundwater. The proposed facilities do not lie within areas historically prone to flooding. During construction, treated sanitary wastewater would be discharged to containment and sewage ponds, which would be built in accordance with applicable regulations to avoid impacts on groundwater. While the potential impacts to surface waters during the construction phase would be erosion and sedimentation, the relatively dry climate, along with the implementation of best management practices for stormwater runoff and erosion control, should preclude these potential impacts. No excess wastewater would be discharged to surface waters during the operation of the facilities, so no impacts to surface water quality are expected. Stormwater runoff would be collected and treated, if necessary, before discharge to natural drainage channels. [Text deleted.] Table 4.2.2.4–1 presents No Action water resources uses and discharges and the potential changes to water resources at NTS resulting from the long-term storage alternatives.

Preferred Alternative: No Action Alternative

Surface Water. A description of the activities that would continue at NTS is provided in Section 3.3.4. Under this alternative, no impacts to surface water resources are anticipated because there are no surface water withdrawals, offsite surface drainage system, or publicly owned treatment works. Treated wastewater discharged to evaporation/infiltration ponds is expected to continue at a rate of 82 million l/yr (21.7 million gal/yr). Contamination that has occurred from past practices and is limited to onsite areas would continue to be characterized and remediated.

Groundwater. Under this alternative, no additional impacts to groundwater availability or quality are anticipated. Baseline conditions and operations, described in Section 3.3 would continue at NTS. Current groundwater usage of 2,400 million l/yr (634 million gal/yr) is not anticipated to increase by the year 2005. Groundwater would continue to be withdrawn from local groundwater sources. No additional impacts to groundwater quality are anticipated since there are no direct discharges to groundwater.

Consolidation Alternative

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Surface Water. No surface water would be withdrawn during construction or operation of the proposed facilities. Consequently, impacts to surface water availability and quality are not expected. Approximately 7.8 million l/yr (2.1 million gal/yr) of nonhazardous wastewater would be generated during construction and subsequently treated, and discharged to evaporation/infiltration ponds that would be designed to minimize seepage. No impacts to surface water from these discharges are expected, because wastewaters are not discharged to natural flowing surface water bodies.

[Text deleted.]

ł

During operation, utility, process, and sanitary wastewater from the proposed facilities would be treated and recycled. Treated effluent would be monitored to comply with discharge permit limits. The extent to which treated effluent or stormwater would be recycled for reuse within the plant would be determined during site-specific studies.

[Text deleted.]

There have been no studies conducted to assess the 500-year floodplain at NTS. However, information on the location of the 500-year floodplain could be developed as part of the siting process. Studies of the 100-year

Affected Resource Indicator	No Action	Consolidation		Collocation	
		Modify P-Tunnel	New Storage Facility	Modify -	New Storage
Water Source	Ground	Ground	Ground	<u>r-tuimer</u>	Facilities
Construction		oround,	Olonia	Ground	Ground
Water Availability and Use	•			1	
Total water requirement (million l/yr)	NA ^a	35	85	35	104.7
Water Quality	NA ^a	1.5	3.5	1.5	4.4
Wastewater discharge (million l/yr)	NAª	7.8	7.8	8.7	11.8
Operation	NAª	9.5	9.5	10.6	14.4
Water Availability and Use					
Total water requirement (million l/yr) Percent increase in projected water use ^d	2,400 0	130 5.4	110 4.6	190 7 9	150
Water Quality	•	. *	3 6	· · · · · · · · · · · · · · · · · · ·	0.5
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^e	82 0,	0,		· · 0 · ·	0
Is action in 100-year floodplain? Is critical action in 500-year floodplain?	NA NA	No Unlikely	No Unlikely	No	No

Table 4.2.2.4–1. No Action and Potential Changes to Water Resources at Nevada Test Site— No Action (2005) and Storage Alternatives

Storage and Disposition **Fissile Materials Final PEIS**

of Weapons-Usable

^b Percent increases in water requirements during construction at NTS are calculated by dividing No Action water requirements (2,400 million l/yr) with that for each storage option: . modified P-Tunnel drift (35 million l/yr), new Pu storage facility (85 million l/yr), modify P-Tunnel drifts (35 million l/yr), and new Pu and HEU storage facility (104.7 million l/yr). ^c Percent changes in wastewater discharged during construction at NTS are calculated by dividing No Action wastewater discharges (82 million l/yr) with that for each storage option: modified P-Tunnel drift (7.8 million l/yr), new Pu storage facility (7.8 million l/yr), modify P-Tunnel drifts (8.7 million l/yr), and new Pu and HEU storage facility (11.8 million l/yr). ^d Percent increases in water requirements during operation at NTS are calculated by dividing No Action water requirements (2,400 million l/yr) with that for each storage option: modified P-Tunnel drift (130 million l/yr), New Pu storage facility (110 million l/yr), modify P-Tunnel drifts (190 million l/yr), and new Pu and HEU storage facility (150 million l/yr). * Percent changes in wastewater discharged during operation at NTS are calculated by dividing No Action wastewater discharges (82 million l/yr) with that for each storage option: modified P-Tunnel drift (0 l/yr), new Pu storage facility (0 l/yr), modify P-Tunnel drifts (0 l/yr), and new Pu and HEU storage facility (0 l/yr). Note: NA=not applicable.

States and the state of the sta

Source: DOE 1996e; DOE 1996f; NT DOE 1996a; NTS 1993a 4.

<u>1</u>6

floodplain showed it to be confined to the Jackass Flats and Frenchman Lake areas. The proposed site for the new Pu storage facilities is outside these areas. However, since NTS is in a region where most flooding occurs by locally intense thunderstorms that can create brief (less than 6 hours) flash floods, the facilities would be designed to withstand such flooding.

Groundwater. All water required for construction and operation would be supplied from groundwater via the existing supply system. The Lower and Upper Carbonate, the Volcanic, and the Valley-Fill Aquifers are the source of water for operations at NTS. Groundwater required for construction (35 million l/yr [9 million gal/yr]) and operation (130 million l/yr [34.3 million gal/yr]) would represent a 1.5- and 5.4-percent maximum increase, respectively, over the projected No Action 2005 groundwater withdrawal, and 0.09- and 0.3-percent, respectively, of the minimum estimated annual recharge to the regional aquifer under the entire NTS. This is based on several studies conducted in recent years, which estimated recharge to be 38 to 57 billion 1 (10 to 15 billion gal). These amounts would also be within NTS's allotment and would not be expected to cause depletion of the aquifer. Groundwater required for both construction and operation and the percent increase in projected water use is shown in Table 4.2.2.4–1.

Construction and operation of the potential modified P-Tunnel drifts would not result in direct discharges to groundwater. Recycling of all treated wastewater is expected. However, if treated wastewater generated during construction was discharged to disposal ponds it could percolate downward into the groundwater of the Valley-Fill Aquifer. This water would be monitored and would not be discharged until contaminant levels are within the limits specified. In addition, other factors contributing to a lessening of potential impacts to groundwater are the combined effects of a deep water table, low discharge volumes, and high evaporation rates. Similarly, some stormwater runoff and other discharges routed to storm drains could percolate into the subsurface These discharges would be monitored and no impacts to groundwater quality are expected.

Construct New Plutonium Storage Facility

Surface Water. There are no unique construction characteristics associated with water requirements and discharges from a new Pu storage facility. No surface water would be withdrawn for any construction or operation activities associated with any of the proposed facilities. Therefore, there would be no impacts to surface water availability. During the construction phase, approximately 7.8 million l/yr (2.1 million gal/yr) of nonhazardous wastewater would be generated. This treated wastewater would be discharged to evaporation/infiltration ponds. No impacts to surface water from these discharges are expected, because wastewaters are not discharged to natural flowing surface waterbodies.

[Text deleted.]

そうない いいちょう うくうたい うくてい ちょうり 御子 かいかい ちょうり

During operation, utility, process, and sanitary wastewater from the proposed facilities would be treated and recycled to the cooling tower and/or boiler. Cooling system blowdown is directed to the utility wastewater treatment facility and is also recycled for use as cooling tower makeup. Treated effluent would be monitored to comply with discharge permit limits. Floodplain issues are the same as described for the previous alternative.

[Text deleted.]

Groundwater. All water required for construction and operation would be supplied from groundwater via the existing supply system. Groundwater required for both construction and operation and the percent increase in projected water use are shown in Table 4.2.2.4–1 for operations at NTS.

Construction and operation water requirements for the proposed facilities (85 million l/yr [22.5 million gal/yr], and 110 million l/yr [30 million gal/yr]) represent approximately 0.2 and 0.3 percent of the minimum estimated annual recharge (38 billion 1 [10 billion gal]) to the regional aquifer under the entire NTS. As shown in Table 4.2.2.4–1, the quantities of water required for construction and operation of the proposed facilities

煤气温料...

represent approximately 3.5- and 4.6-percent increases over the projected No Action groundwater usage. These small increases boost the total projected groundwater withdrawal to less than 7 percent of the estimated annual recharge; there should be no impact on groundwater availability.

*r

. try 1. 18 %

.

Construction and operation of the proposed consolidated Pu storage facilities would not result in direct discharges to groundwater. Recycling of all treated wastewater is expected. However, if treated wastewater generated during construction was discharged to disposal ponds it could percolate downward into the groundwater of the Valley-Fill Aquifer. This water would be monitored and would not be discharged until contaminant levels are within the limits specified. In addition, other factors contributing to a lessening of potential impacts to groundwater are the combined effects of a deep water table, low discharge volumes, and high evaporation rates. Impacts to groundwater quality are therefore not expected. · · · 12

Similarly, some stormwater runoff and other discharges routed to storm drains could percolate into the subsurface. These discharges would be monitored under the State of Nevada stormwater regulations and therefore no impacts to groundwater quality are expected.

ì

• * f _ -

'a _

Collocation Alternative Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Surface Water. Under this alternative, modifying the P-Tunnel would increase water discharges by 10.6 percent over the projected No Action discharge during construction. During operations, wastewater would be recycled. All other wastewater requirements of the option would be similar to modifying P-Tunnel under the Consolidation Alternative.

.

Groundwater. During construction, the Pu and HEU storage upgrade using the P-Tunnel would require 35 million l/yr (9.2 million gal/yr) of groundwater, or a 1.5-percent increase over projected No Action water use. During operations, 190 million l/yr (50.2 million gal/yr) of water would be required, a 7.9-percent increase over the projected No Action water use, representing 0.5 percent of the minimum estimated recharge. All other water requirements of the option are identical to those discussed previously for modifying the P-Tunnel.

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Since the new Pu and HEU collocated storage facilities would be located in the same area as the new Pu storage facility, the impacts associated with them are similar to those previously discussed for the Consolidation New Storage Facility Alternative, with the following exceptions. Sanitary wastewater quantities generated during construction would be greater than for the new storage facility option and are approximately 11.8 million l/yr (3.1 million gal/yr). These effluents would be discharged to evaporation/infiltration ponds under a State of Nevada permit. No impacts are expected. During operations, wastewater would be recycled. The groundwater requirements for construction of this option are greater than those for the previous option. This option would require approximately 104.7 million l/yr (27.7 million gal/yr) and 150 million l/yr (39.6 million gal/yr) for construction and operation, respectively. These additional requirements represent 4.4- and 6.3-percent increases, respectively, in the projected No Action groundwater withdrawals. These amounts increase the total projected site groundwater withdrawal to less than 7 percent of the estimated annual recharge; there should be no impact on groundwater availability.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Water resource impacts for construction and operation of this option are expected to be slightly less than those described for the Consolidation Alternative and the Collocation Alternative because of the reduction in the amount of material. [Text deleted.]
4.2.2.5 Geology and Soils

Construction and operation of the alternatives at NTS would have no effect on the geologic resources. A moderate seismic risk exists, but would be considered in the design of the proposed alternatives. The existing seismic risk does not preclude the safe construction and operation of the proposed alternative facilities. The facilities would be designed for earthquake-generated ground accelerations, in accordance with DOE O 420.1, Facility Safety. The Yucca and Carpet Bag faults are considered to be capable faults and represent a potential for ground rupture as a result of an earthquake. However, ground shaking is more likely in the areas of the proposed alternatives. Intensities of approximately VII on the MMI scale are possible at NTS. A peak ground acceleration of 0.67 g with a Richter magnitude of 6.7, has been estimated for the Cane Spring fault, with a recurrence interval of 10,000 to 30,000 years. This could affect the integrity of inadequately designed or nonreinforced structures but should not affect newly designed facilities. Human health effects form accidents initiated by natural phenomenon (for example, earthquakes) are discussed in Section 4.2.2.9. Volcanic activity is improbable during the life of the alternatives and is not anticipated to affect the construction and operation of the alternatives. The most likely risk to NTS is possible ash fall from the Long Valley, California area located approximately 214 km (150 mi) to the west-northwest. Lava extrusions from sources at NTS could recur but are unlikely. Precursors, such as shallow earthquakes, gas venting activity, and an increase in groundwater temperatures provide advance warning of most eruptions of this type; no such activity is currently indicated at NTS. It is unlikely that landslides, sinkhole development, or other nontectonic events would affect project activities. Slopes and underlying foundation materials are generally considered stable. Properties and conditions of the soils typical of NTS have no limitation on construction. Soils would be affected by construction and operations of a proposed alternative.

None of the sites has known economically viable geologic resources that would be affected by the construction and operations of an alternative. Except for the potential existence of gold, tungsten, and molybdenum at NTS, geologic resources consist of surficial sand, gravel, or clay deposits that have low economic value. New construction may increase the use of the materials, but because large volumes of these materials are present, consequently the impact is anticipated to be negligible.

Construction of the proposed alternative facilities may involve ground-disturbing activities that could affect the soil resource. The amount of land disturbed is specified below for each alternative. Impacts would depend on the specific soil units in the disturbed area, the extent of the land disturbing activities, and the amount of soil disturbed. Within NTS, the soil erosion potential is directly related to the amount of land disturbed because soil and climatic conditions are similar throughout the site. Control measures would be employed during construction to minimize soil erosion.

Preferred Alternative: No Action Alternative

[Text deleted.]

「「「「「「「」」」」」

なみ、かんちゃうちゃうちゃうない、 しちないち

and the second sec

I

Impacts to geologic and soil resources occur during, or as a result of, ground-disturbing construction activities. Under the No Action Alternative, DOE would continue current and ongoing activities at NTS. There would be no ground-disturbing activities beyond those associated with existing and future site improvements. Because new construction and the associated ground disturbance for potential soil erosion would not occur, the No Action Alternative would have no effect on the geologic or soil resources at the site.

Consolidation Alternative

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

1 THE

· • • • • • • •

No apparent direct or indirect effects on the geologic resource are anticipated, because neither facility construction and operational activities nor site infrastructure improvements will restrict access to potential geologic resources. Design of the facilities would ensure that they would not be affected by potentially hazardous geological conditions.

、肥果生

[Text deleted.] Construction activities will occur completely on previously disturbed land, as described in Section 4.2.2.1 and involve land disturbance of approximately 29 ha (72 acres) that will affect the soil profile and potentially cause a temporary increase in soil erosion. Soil disturbance would occur primarily from ground-disturbing construction activities (foundation preparation) and activities associated with building construction laydown areas that can expose the soil profile and lead to a possible increase in soil erosion as a result of wind and water action. Soil loss would depend on the frequency and severity of rainstorm wind velocities (increased velocities and durations increase potential erosion); the frequency and severity of storms; and the size, location, and duration of ground-disturbing activities.

Net soil disturbance during operations would be considerably less than during construction because areas temporarily used for construction laydown would be restored. Although stormwater runoff and wind action could occur during operation, it is anticipated to be minimal. [Text deleted.]

Construct New Plutonium Storage Facility

Construction and operation effects on geologic resources for the storage facility are the same as those described for the modified P-Tunnel. However, additional soil impacts would be expected from the construction of the storage facility which will occur completely on previously disturbed land as described in Section 4.2.2.1. Approximately 58.5 ha (144 acres) would be disturbed for construction of the new facility, affecting the soil profile and leading to a possible temporary increase in soil erosion as a result of stormwater runoff and wind action. Soil losses would depend on frequency of storms, wind velocity, and location of the facility with respect to drainage and wind patterns; slope, shape, and area of the tracts of ground disturbed; and the duration of time the soil is bare. Soil impacts during operation are expected to be minimal.

Collocation Alternative

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Construction and operation effects on geological and soil resources for this alternative are the same as those discussed previously for the Consolidation Alternative using the modified P-Tunnel. Construction will occur completely on previously disturbed land, as described in Section 4.2.2.1, and involve land disturbance of approximately 29 ha (72 acres).

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

[Text deleted.] Additional soil impacts would be anticipated from the construction of the storage facilities which will occur completely on previously undisturbed land, as described in Section 4.2.2.1. Approximately 89.5 ha (221 acres) would be disturbed for construction of the Collocation Alternative, affecting the soil profile and leading to a possible temporary increase in soil erosion as a result of storm water runoff and wind action. Soil losses would depend on frequency of storms, wind velocity, and location of the facility with respect to drainage and wind patterns; slope, shape, and area of the tracts of ground disturbed; and the duration of time the soil is exposed. Soil impacts during operation are expected to be minimal.

I

i

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Exclusion of strategic reserve and weapons R&D materials would give almost the same effects to the geologic and soil resources for the Consolidation Alternative and Collocation Alternative. By excluding these materials, the size of a facility would be similar, thus not changing the amount of land disturbed by construction activities. No effect to the geologic resource is anticipated as a result of this option. [Text deleted.]

[Text deleted.]

4.2.2.6 Biological Resources

Preferred Alternative: No Action Alternative

The missions described in Section 2.2.2 would continue at NTS. This would result in no changes to current conditions of biological resources at NTS as described in Section 3.3.6.

Consolidation Alternative

Consolidated storage facilities would be located in modified P-Tunnel drifts or in new facilities in the Frenchman Flat area.

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Terrestrial Resources. If consolidated storage facilities were located within the modified P-Tunnel area, existing and new facilities would be required at both the modified P-Tunnel site itself and the Area 12 Camp. Modification of existing structures would have minimal impact on terrestrial resources. New construction would also have minimal impact, because construction would take place within presently developed areas. The material handling facility, which is the principal structure to be built, would be located in an area on which excavated material from the modified P-Tunnel was deposited. Foundation preparation would require some additional soil, which would be imported from other as-yet-unspecified areas of NTS. If this material was excavated from a new borrow pit, terrestrial resources at the location could be affected. Construction and operation could result in disturbance to wildlife by noise and human activity, but impacts to wildlife would be minimal because animals would have already adjusted to ongoing activities.

Wetlands. Construction and operation of consolidated storage facilities would not affect wetlands since there are no wetlands in the vicinity of the modified P-Tunnel.

Aquatic Resources. Construction and operation of consolidated storage facilities would not affect aquatic resources since there are no permanent surface water bodies in the vicinity of the modified P-Tunnel.

Threatened and Endangered Species. Construction and operation of consolidated storage facilities would have minimal effect on threatened and endangered species in the modified P-Tunnel area since the habitat is already disturbed. The range of the federally listed threatened desert tortoise does not extend to the modified P-Tunnel area. Site surveys would be performed as necessary to determine the presence of special status species. Consultation with USFWS and State agencies would be conducted at the sit-specific levels, as appropriate.

Construct New Plutonium Storage Facility

Under this alternative, Pu would be consolidated in a new storage facility located in the Frenchman Flat area of NTS. Impacts to terrestrial resources, wetlands, aquatic resources, and threatened and endangered species are discussed below.

2

Terrestrial Resources. Construction of the consolidated Pu storage facility at NTS would result in the disturbance of 58.5 ha (144 acres) of terrestrial resources, or less than 0.02 percent of NTS. This includes areas on which facilities would be constructed, as well as areas used for construction laydown. Vegetative cover within the proposed project area, which is primarily creosote bush (Figure 3.3.6–1), would be destroyed during land-clearing operations. Creosote bush communities are well represented on NTS.

Construction of the Pu storage facility would affect animal populations. Less-mobile animals, such as reptiles and small mammals, within the project area would not be expected to survive. Construction activities and noise would cause larger mammals and birds in the construction and adjacent areas to move to similar habitat nearby.

I

If the area to which they moved was below its carrying capacity, these animals would be expected to survive. However, if the area was already supporting the maximum number of individuals, the additional animals would compete for limited resources, which could lead to habitat degradation and eventual loss of the excess population. Nests and young animals living within the proposed site may not survive. The site would be surveyed as necessary for the nests of migrating birds prior to construction. Areas disturbed by construction, but not occupied by facility structures, would be of minimal value to wildlife because of the difficulty in establishing vegetative cover in a desert environment.

Activities associated with operation, such as noise and human presence, could affect wildlife living immediately adjacent to the facility. These disturbances may cause some species to move from the area. Disturbance to wildlife living adjacent to the facility would be minimized by preventing workers from entering undisturbed areas. Impacts to vegetation from salt drift would not occur since dry cooling systems would be used.

Wetlands. Construction and operation of the Pu storage facility would not affect wetlands because there are no wetlands near the assumed facility location.

Aquatic Resources. Construction and operation of the Pu storage facility would not affect aquatic resources because there are no permanent surface water bodies near the assumed facility location.

Threatened and Endangered Species. The desert tortoise is a federally listed threatened species that could be affected by construction of the Pu storage facility at NTS. Construction activities such as land-clearing operations, trenches, and excavation could pose a threat to any tortoises residing within the disturbed area. An increase in vehicular traffic is an additional hazard to the tortoise. Measures from previous projects at NTS designed to avoid impacts to the desert tortoise have been implemented as a result of a Biological Opinion issued by the USFWS (NT DOI 1992b:8-15). Recommended mitigation measures included providing worker training; putting restrictions on vehicle speeds and off-road movement; conducting clearance surveys prior to surface disturbance; approving stop work authority if tortoises are found within work areas; removing tortoises from roadways and work areas; placing permanent and temporary tortoise-proof fencing around trenches, landfills, and treatment ponds; inspecting trenches; and having biologists present when heavy equipment is in use. The USFWS would be consulted, and similar USFWS recommendations would be implemented should NTS be selected as the location for the Pu storage facility.

[Text deleted.] Any listed plant species located within the construction area could be lost or affected during landclearing activities. Preactivity surveys would be conducted as appropriate prior to construction to determine the presence of these species in the area to be disturbed. Consultation with USFWS and State agencies would be conducted at the site-specific levels, as appropriate.

During facility operation, vehicular traffic would pose a hazard to the desert tortoise similar to the hazard caused by current traffic. Extensive measures, including personnel training, are presently being taken to ensure that drivers on NTS avoid the tortoise. [Text deleted.] Groundwater levels in Devils Hole are not expected to change due to operation of the Pu storage facility (Section 4.2.2.4), so impacts to the Devils Hole pupfish are not expected. Similarly, other rare endemic aquatic species found in the Ash Meadows area would not be affected.

Collocation Alternative

Under this alternative, consolidated Pu would be stored with HEU inventories in existing and new facilities in the modified P-Tunnel area or in a new collocated Pu storage facility sited at the same location as the consolidated storage facility.

· 1

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Impacts to biological resources from placing collocated storage facilities in modified P-Tunnel drifts would be similar to those described previously for consolidated storage facilities. This is because both facilities are of a similar size and both would be placed within developed portions of the P-Tunnel site and Area 12 Camp.

· ·

*** ***

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Construction and operation of a collocated storage facility at the Frenchman Flat area of NTS would have similar, but somewhat greater effects on biological resources as those described for the consolidated storage facility. Construction of the collocated storage alternative would disturb 89.5 ha (221 acres) of habitat.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

5 15 1 The exclusion of strategic reserve and weapons R&D materials would have almost the same effects to the Consolidation Alternative and the Collocation Alternative. The size of facility would be similar and would not result in the reduction of disturbed habitat and/or fewer facility modifications and thus lessen the potential... - 1., t e 1.11 impacts to biological resources would be similar. [Text deleted.]

١.

2 d ĥ.

ale : Clark e ezt a part of the second ر معيدوه معرور فارار x- 1+,+ x

1. N. 2. C. la - at a friet a second the states I • • · , · , to the she ^{*}۱' . and the second states of the second and the state of the second 2.10

· .

4.2.2.7 Cultural and Paleontological Resources

Preferred Alternative: No Action Alternative

Under this alternative, DOE would continue the existing and planned missions at NTS. Management of NTS's cultural resources follows the Programmatic Agreement between DOE, the Nevada SHPO, and the Advisory Council on Historic Preservation. Any impacts to cultural or paleontological resources from these missions would be independent of the proposed action and would be addressed through separate NHPA, American Indian Religious Freedom Act, and Native American Graves Protection and Repatriation Act regulatory compliance procedures.

Consolidation Alternative

ľ

b

I

I

the social lines of a

1

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

The existing P-Tunnel is located in Area 12. Modification would result in ground disturbance of 29 ha (72 acres) on previously disturbed land. No new, undisturbed land would be used. The P-Tunnel is on Rainier Mesa, in Area 12. In 1990, DOE entered into a Programmatic Agreement with the Nevada SHPO and the Advisory Council on Historic Preservation that has resulted in an in-depth cultural resources study of 11 percent of Pahute and Rainier Mesas. Additional surveys would be completed prior to construction on any currently unexamined tracts. This area is rich in cultural resources, including many archaeological sites. Construction of the proposed facility may affect some potentially NRHP-eligible resources. Recorded prehistoric sites in the area include quarries, lithic workshops, campsites, and rock shelters. Historic site types include remains of ranches and mines. Operation would not have an additional impact on prehistoric or historic resources. Some paleontological resources may also be affected by land disturbance during construction. Operation would not result in additional impacts to those resources.

The CGTO is an alliance of Native American groups that have ties to the land at NTS. This alliance has in the past requested and been denied access to Rainier Mesa. Consequently, no important Native American resources have been identified there to date. Oral histories indicate that some resources, such as ancestral campsites, are located within Rainier Mesa (NT DOE 1996c:4-169). Additional Native American resources such as archaeological sites, traditionally used plant and animal species, and rock art may be affected by both the construction and operation of the proposed facility. Consultation with interested parties during the American Indian Religious Freedom Act, Native American Graves Protection and Repatriation Act, and NHPA compliance process may identify some of these resources.

Construct New Plutonium Storage Facility

The new Pu storage facility would be situated in the northern portion of Frenchman Flat in Area 6, near the DAF. The operational land requirement for the proposed facility is 56 ha (138 acres). Some new construction would be necessary. Land to be disturbed during construction totals 58.5 ha (144 acres). A 1.6-km (1-mi) reduced-access buffer zone would be created around the facility.

[Text deleted.] In 1984, a Class III cultural resources survey was conducted across the 660-ha (1,610-acres) DAF site and no NRHP-eligible sites were identified. Although no resources were identified within the DAF project area, Frenchman Flat contains 49 sites that have been determined eligible for inclusion on the NRHP. Recorded prehistoric sites within Frenchman Flat include base and temporary camps, quarries, and lithic reduction areas. Identified historic resources include sites associated with nuclear testing and research. Additional unsurveyed lands necessary for the proposed facility may contain similar prehistoric or historic resources.

Impacts to resources could occur during construction of the proposed facility. Operation would not result in additional impact as it does not involve ground disturbance or increased activity. In addition, construction (but not operation) may have some impact on Late Pleistocene paleontological resources, should any exist within the acreage to be disturbed. a second and the second second

The CGTO has conducted surveys over portions of Frenchman Flat. The area is known to contain at least 20 plant species of importance to Native Americans. Additional project-specific consultations would be necessary to identify impacts to Native American resources resulting from the construction and operation of the proposed facility. Potential impacts include reduced access to traditional use areas and visual or auditory intrusions to 1 sacred space.

Collocation Alternative

17, 18 14 A

Modify Existing Tunnel Drifts and Construct New Material Handing Building at the P-Tunnel

Potential impacts under this option would be the same as the P-Tunnel discussion under the Consolidation a construction of the second sec Alternative.

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Construction of the proposed Pu and HEU storage facility would disturb 89.5 ha (221 acres) of land; operation would require 87 ha (215 acres). A-1:6-km (1-mi) reduced-access buffer zone would be created around the facility. The proposed location is contained within Area 6 near the DAF. Impacts to cultural and paleontological resources would be similar to those discussed under the new storage facility option of the Consolidation Alternative.

Subalternative Not Including Strategic Reserve and Weapon's Research and Development Materials

Under this subalternative, facility and other resource requirements will be almost the same as the Consolidation and Collocation Alternative. Therefore, impacts to cultural and paleontological resources would equal those previously discussed. [Text deleted.]

> ۱ 1 2 3 14

14

· . . . · ·

1

1 15 1.7

The stand of the second second

. one sept Mark

τ,

a tay station of

1. 5

543.2

4.2.2.8 Socioeconomics

「「日本」

I

Preferred Alternative: No Action Alternative

Under this alternative, the existing storage facility would remain operational. No new employment or inmigration of workers would be required.

Regional Economy Characteristics. Total employment in the REA is projected to grow 3.4 percent annually between 1995 and 2000, reaching 686,800 in the latter year. Slower growth is projected after the year 2000, when employment will increase 2 percent annually and reach approximately 1,236,600 in 2040. Unemployment in the REA was 6.1 percent in 1994 and is expected to remain at this level into the near future. Per capita income is projected to increase from approximately \$21,900 in 1995 to \$46,134 in 2040. No Action projections are presented in Table L.1–19.

Population and Housing. Population in the ROI is projected to total 990,700 in 1995 and reach 2,087,000 by 2040. Over the same period, the total number of housing units in the ROI is projected to increase from 403,700 to 850,500. No Action projections for population and housing are presented in Tables L.1–20 and L.1–21, respectively.

Community Services. Education, public safety, and health care characteristics are used to assess the level of community services in the NTS ROI. School enrollments are projected to increase from approximately 165,630 students in 1995 to 348,920 students by 2040. In 1994, the student-to-teacher ratio was 19.6:1. To maintain this level of service, the number of teachers in the ROI would need to increase from 8,466 in 1995 to 17,833 in 2040. No Action projections are presented in Tables L.1–22 and L.1–23.

The projected number of sworn police officers and firefighters serving ROI communities during the period 1995 to 2040 is shown in Tables L.1–24 and L.1–25, respectively. Under No Action, the number of sworn police officers is projected to increase from 1,946 in 1995 to 4,101 in 2040 to maintain the current service level of 2.0 officers per 1,000 persons. The number of firefighters in the ROI is estimated to increase from 1,553 in 1995 to 3,271 in 2040 to maintain the current service level of 1.6 firefighters per 1,000 persons.

Hospital occupancy rates are based on current capacity. Projections of hospital occupancy rates and the number of practicing physicians serving the ROI population between 1995 and 2040 are presented in Tables L.1–26 and L.1–27, respectively. Without expansion of existing capacity, hospital occupancy rates for the ROI would increase from 62 percent in 1995 to over 100 percent in 2040. If the 1994 physician-to-population ratio of 1.3 physicians per 1,000 persons is to be maintained, the total number of physicians would need to increase from 1,276 in 1995 to 2,704 in 2040.

Local Transportation. The worker population at NTS would not increase and could decrease. Any increases in traffic would be due to the projected growth in the area unrelated to DOE activities. [Text deleted.]

Consolidation Alternative

1

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Modification of the existing P-Tunnel at NTS to consolidate the storage of Pu would require 1,103 workers during the peak construction year and would generate over 2,100 jobs (direct and indirect) in the region. Operation of the facility would generate over 1,400 jobs (direct and indirect) in the region. Projections indicate that there would be sufficient labor available within the region to fill all direct and indirect jobs during both construction and operations.

1. 12.2

Regional Economy Characteristics. A total of 2,139 (1,103 direct and 1,036 indirect) jobs would be created in the region during the construction of the facility. Employment in the region would increase by less than 1 percent, and the unemployment rate would fall from the No Action projection of 6.1 percent to 5.8 percent. Per capita income would increase by less than 1 percent (Socio 1996a).

Operation of the facility would generate a total of 1,406 jobs (527 direct and 879 indirect) in the region, increasing regional employment by less than 1 percent. Operation workers would begin phasing in as construction nears completion. Unemployment would rise from 5.8 percent during peak construction to 5.9 percent during operation but remain below the No Action level of 6.1 percent (Socio 1996a). and the second second

Population, Housing, and Community Services. All newly created jobs would be filled by the resident labor force. Therefore, there would be no change to the region's population from the No Action projections. Accordingly, there would be minimal impacts to the demand for housing or community services in the region 't eas a result of the construction and operation of this facility (Socio 1996a).

あいろう いっとななななないのでのないない うちん

č

Local Transportation. During the peak construction year, workers would generate 2,118 vehicle trips per day. During operations, workers would generate 1,012 vehicle trips per day. These increases would not affect the level of service on the local road segments analyzed (Socio 1996a). • r = 5 3 an an the second se **!**^

Construct New Plutonium Storage Facility

Construction of a new facility for the consolidated storage of Pu at NTS would generate over 2,100 jobs (direct and indirect) within the REA. Operation would generate over 1,300 jobs (direct and indirect) in the region. Projections indicate that there would be sufficient available labor in the region to fill all of the direct and indirect jobs. [Text deleted.] 6.7.3

r -

1 1

and the factor of the second states of the second s

[Text deleted.]

Regional Economy Characteristics. During the construction period, the project would add a total of 2,122 (1,094 direct and 1,028 indirect) jobs to the regional economy that would be filled by the labor available within ' the region. The regional unemployment rate would decrease from 6.1 percent to 5.8 percent, while per capita income would increase minimally (much less than 1 percent) (Socio 1996a).

,

The consolidated storage facility at NTS would create 1,313 jobs (492 direct and 821 indirect) during full operation, increasing total employment in the REA by much less than 1 percent over the No Action projection. Available labor would fill all of the indirect positions. However, some workers would need to in-migrate to fill specialized direct employment requirements. Operation workers would begin phasing in as construction nears completion. Unemployment would rise from 5.8 percent during peak construction to 6.0 percent during operation but would remain below the No Action level of 6.1 percent. Per capita income would increase minimally (much less than 1 percent) (Socio 1996a).

Population, Housing, and Community Services. All newly created jobs would be filled by the resident labor force. Therefore, there would be no change to the region's population from the No Action projections. Accordingly, there would be minimal impact to the demand for housing or community services in the region as a result of the construction and operation of this facility. [Text deleted.]

[Text deleted.]

the straight 1 200 ·: . ه دور اي

Local Transportation. Construction workers would generate a projected 2,100 vehicle trips per day during peak construction activity. During operations, workers would generate 945 vehicle trips per day. These increases would not affect the level of service on the local road segments analyzed (Socio 1996a).

Collocation Alternative

I

I

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Modification of the existing P-Tunnel at NTS to consolidate the storage of all Pu and HEU would generate a total of over 2,500 jobs (direct and indirect) during the construction phase. Projections indicate that there would be sufficient labor available within the region to fill these positions. However, operations of the upgraded facility, which would generate over 1,700 total jobs (direct and indirect), would require some workers to relocate to the area to fill specialized employment requirements. Available labor within the REA would fill all of the indirect jobs created during the operation phase. The impacts on the region's economy, population, housing, community services, and local transportation are discussed below.

Regional Economy Characteristics. Modification of the existing P-Tunnel would require 1,297 workers during the peak construction year and would generate an additional 1,218 jobs in other industries within the REA. The available labor force within the region would be sufficient to fill all the direct and indirect jobs created. Employment in the REA would increase by less than 1 percent over No Action, and the unemployment rate would fall from the No Action projection of 6.1 percent to 5.8 percent. Per capita income would increase by less than 1 percent (Socio 1996a).

Operation of the facility would require 641 new employees and would generate an additional 1,069 indirect jobs within the REA. Available labor within the REA would fill all of the indirect positions, however, some workers would have to in-migrate to fill specialized direct employment requirements. Total employment and per capita income in the region would both increase by much less than 1 percent over No Action projections. Operation workers would begin phasing in as construction nears completion. Unemployment would rise from 5.8 percent during peak construction to 5.9 percent during operation but would remain below the No Action level of 6.1 percent (Socio 1996a).

Population and Housing. During full operation of the upgraded facility, in-migration is projected to increase population in the ROI by 73 over the No Action Alternative, a change of much less than 1 percent of total population. Projected vacancies in the housing stock would be sufficient to accommodate the slight increase in demand (Socio 1996a).

Community Services. Because there would be no in-migration associated with construction of the upgraded facility, and the population change resulting from operations is so small, the demand for community services would remain unchanged from No Action projections (Socio 1996a). [Text deleted.]

[Text deleted.]

Local Transportation. Construction workers would generate a projected 2,490 vehicle trips per day in the year 2001 (the peak construction year) under this alternative. Operation workers would generate 1,231 vehicle trips per day. These increases would not affect the level of service on the local road segments analyzed (Socio 1996a).

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Construction of a new facility to store all Pu and HEU would generate almost 2,200 jobs (direct and indirect) in the region. Operation of the facility would generate over 1,600 jobs. Projections indicate that workers would inmigrate to the REA to fill some of the direct jobs created during the operation of a new consolidated Pu and HEU storage facility at NTS. However, there would be sufficient available labor in the REA to fill both the

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

indirect jobs created during operation and all employment generated by the construction of the facility. The effects on the region's economy, population, housing, community services, and local transportation are discussed below.

1

1-1

Regional Economy Characteristics. Construction of the new consolidated Pu and HEU storage facility would generate 1,123 direct and 1,055 indirect jobs during peak construction at NTS. Total employment in the region would increase by less than 1 percent, and the unemployment rate would decrease from the No Action projection of 6.1 percent to 5.8 percent. Per capita income would increase by much less than 1 percent (Socio 1996a). 22 1 . . .

Operation of the proposed storage facilities would generate a total of 1,659 new jobs (622 direct and 1,037 indirect) in the REA, with some direct jobs filled by in-migrant workers. All of the indirect positions would be filled by available labor in the REA. Total employment in the region would increase by less than 1 percent over the No Action projection. Operation workers would begin phasing in as construction nears completion. Unemployment would rise from 5.8 percent during peak construction to 5.9 percent during operation but remain below the No Action level of 6.1 percent. Per capita income would increase much less than 1 percent (Socio 1996a).

Population and Housing. Although 22 workers would in-migrate to the REA during the operation phase of the proposed storage facilities, the resulting change to population would be less than a 1 percent increase over No Action. Projected housing vacancies would be sufficient to accommodate demand from the in-migrating · · population (Socio 1996a).

Community Services. Because there would be no in-migration associated with construction of the facility, and the population change from operations is so small, the demand for community services would remain unchanged from No Action projections (Socio 1996a).

[Text deleted.]

Local Transportation. Construction employees would generate 2,156 vehicle trips per day. Operation employees would generate an estimated 1,194 vehicle trips per day. These increases would not affect the level of service on the local road segments analyzed (Socio 1996a). 1. 1. 11 1. 11 1.

• ~ Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

and a second state of the second state of the second If strategic reserve and weapons R&D materials are not included in the storage requirements at NTS, there would be a small reduction in worker requirements for construction and operation of the facility. Therefore, the socioeconomic effects would be less than those options including strategic reserve and weapons R&D for the Consolidation Alternative and the Collocation Alternative. [Text deleted.] the first of the second s . .

a strategy and a stra - ショー コンジョン コーマン ふくほう ひかかれい ビジア nandela, en la produía, en los destrutores the set of the second by the second second

4.2.2.9 Public and Occupational Health and Safety

The assessments of potential radiological and chemical impacts associated with the storage alternatives at NTS are presented in this section. Summaries of radiological impacts from normal operations are presented in Tables 4.2.2.9–1 and 4.2.2.9–2 for the public and workers, respectively. Impacts from hazardous chemicals are presented in Table 4.2.2.9–3. Summaries of impacts associated with postulated accidents are given in Tables 4.2.2.9–4 through 4.2.2.9–7. Detailed results are presented in Appendix M.

Preferred Alternative: No Action Alternative

ţ

This section describes the radiological and hazardous chemical releases and their associated impacts resulting from normal operations involved with the sitewide NTS missions. The radiological and chemical source terms (releases) under the No Action Alternative are taken to be the same as for the existing baseline condition; the resulting impacts would be within applicable regulatory limits. For facility accidents, the risks and consequences are described in site safety documentation.

Normal Operation. The doses and potential health effects on the public and workers during normal operations are described below.

Radiological Impacts. The calculated annual dose to the average and maximally exposed members of the public from total site operation; the associated fatal cancer risks to these individuals from 50 years of operation; the dose to the population within 80 km (50 mi) from total site operation in the year 2030; and the projected number of fatal cancers in this population from 50 years of operation are presented in Table 4.2.2.9–1 under this alternative at NTS. The annual dose of 4.2×10^{-3} mrem to the MEI is within the radiological limits specified in NESHAPS (40 CFR 61, Subpart H) and DOE Order 5400.5. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 1.0×10^{-7} . This activity would be included in a program to ensure that doses to the public are ALARA. The annual dose of 3.7×10^{-3} person-rem to the population from 50 years of operation would be within the limit in proposed 10 CFR 834. The corresponding number of fatal cancers in this population from 50 years of operational doses from natural background radiation are included in the table.

Under the No Action Alternative, as shown in Table 4.2.2.9–2, the annual average dose to a noninvolved (No Action) site worker and the annual dose to the noninvolved (No Action) total site workforce would be 5.0 mrem and 3.0 person-rem, respectively. The associated risk of fatal cancer to the average worker from 50 years of total site operations would be 1.0×10^{-4} and the projected number of fatal cancers among all workers from 50 years of total site operations would be 0.060.

Hazardous Chemical Impacts. There would be no hazardous chemical impacts shown in Table 4.2.2.9–3 on the public or to workers resulting from the normal operation under No Action at NTS because no hazardous or other carcinogens are released.

Facility Accidents. Under the No Action Alternative, facilities would continue to operate in accordance with DOE safety orders, which ensure that the risk to the public of prompt fatalities due to accidents or cancer fatalities due to operations will be minimized. The safety to workers and the public from accidents at existing facilities is also controlled by Technical Safety Requirements specified in detail in SARs or a Basis for Interim Operations document prepared and maintained specifically for a facility or process within a facility. Under these controls, any change in approved operations or to facilities would cause a halt in operations until it can be established that worker and public safety has not been compromised.

	No Action	, i .	Consol	idation		Collocation
	3	Modify I	P-Tunnel	New Stor	age Facility	
Receptor	Total Site	Facilities	Total Site ^a	Facility	- Total Site ^a -	
Annual Dose to the Maximally Exposed Individual Member of the Public ^b		······	· · · ·		· · · ·	
Atmospheric release pathway (mrem)	4.2x10 ⁻³	5.6x10 ⁻⁶	4.2x10 ⁻³	1.3x10 ⁻⁶	4.2x10 ⁻³	c
Drinking water pathway (mrem)	0	0	O	0	0	c
Total liquid release pathway (mrem)	0	0	, 0	0 .	0_`	C .
Atmospheric and liquid release pathways combined (mrem)	4.2x10 ⁻³	5.6x10 ⁻⁶	4.2x10 ⁻³	1.3x10 ⁻⁶	4.2x10 ⁻³	* c 14 - 41
Percent of natural background ^d	1.3x10 ⁻³	1.8x10 ⁻⁶	1.3x10 ⁻³	4.2x10 ⁻⁷	1.3x10 ⁻³	C T,
50-year fatal cancer risk	1.0x10 ⁻⁷	1.4x10 ⁻¹⁰	1.0x10 ⁻⁷	3.2x10 ⁻¹¹	1.0x10 ⁻⁷	C
Population Dose Within 80 Kilometers for Year 2030			-			1
Atmospheric release pathway (person-rem)	3.7x10 ⁻³	[°] 1.7x10 ^{−6}	- 3.7x10 ⁻³	-2.6x10 ⁻⁶	3.7x10 ⁻³	م میں تو میں اور
Total Liquid Release Pathway (person-rem)	~~0 -	0, ,	0	0	0 ້_	Ċ,
Atmospheric and liquid release pathways combined	3.7x10 ⁻³	1.7x10 ⁻⁶	3.7x10 ⁻³	2.6x10 ⁻⁶	3.7x10 ⁻³	r C
(person-rem)	40.10-5	1.0.10-8	10-10-5	0.0.10-8	40-10-5	c
Forcent of natural background	4.0X10 ⁻⁵	1.8X10 °	4.0x10 ⁻⁵	2.8X10°	4.0x10 ⁻⁵	- + c
SU-year fatal cancers	9.3x10 ⁻⁵	4.3x10	9.3x10 ⁻⁵	6.5x10 °	9.3x10 ⁻⁵	
Annual Dose to the Average Individual Within 80 Kilometers ^f		¥ .	, _u		۰. ^۲	· · · · · · · · ·
Atmospheric and liquid release pathways combined (mrem)	1.3x10 ⁻⁴	5.8x10 ⁻⁸	1.3x10 ⁻⁴	8.8x10 ⁻⁸	1.3x10 ⁻⁴	c
50-year fatal cancer risk	3.1x10 ⁻⁹	1.4x10 ⁻¹²	3.1x10 ⁻⁹	2.2x10 ⁻¹²	3.1x10 ⁻⁹	c

	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	+	the states		
Table 4.2.2.9–1.	Potential Radiological Impacts to	the Public Du	ring Normal (Operation at Nevado	a Test
	Site—No Action and	l Storage Alteri	natives 🕔 🐇		

Includes impacts from No Action (Baseline) facilities. The location of the MEI may be different under No Action than for the a other alternatives. Therefore, the impacts may not be directly additive.

^b The applicable radiological limits for an individual member of the public from total site operations are 10 mrem per year from the air pathways as required by NESHAPS (40 CFR 61, Subpart H) under the CAA, 4 mrem per year from the drinking water pathway as required by the SDWA; and 100 mrem per year from all pathways combined. Refer to DOE Order 5400.5.

С The impacts for both the new storage facility and the new and modified P-Tunnel Drifts under the Collocation of Pu Storage would be virtually the same as for these two options under the Consolidation of Pu Storage. This is because the HEU contributes a negligible dose to the public.

^d The annual natural background radiation level at NTS is 313 mrem for the average individual; the population within 80 km in the year 2030 receives 9,190 person-rem.

^e For DOE activities, proposed 10 CFR 834 (see 58 FR 16268) would generally limit the potential annual population dose to 100 person-rem from all pathways combined, and would require an ALARA program.

[Text deleted.]

^f Obtained by dividing the population dose by the number of people projected to live within 80 km of NTS in 2030 (29,400). Source: Section M.2.

	Conso	lidation	Colle	ocation
Receptor	Modify P-Tunnel	New Storage Facility	Modify P-Tunnel	New Storage Facility
Involved Workforce ^a		;	·····	
Average worker dose (mrem/yr) ^b	262	258	262	264
50-year risk of fatal cancer	5.2x10 ⁻³	5.2×10^{-3}	5.2x10 ⁻³	5.3×10^{-3}
Total dose (person-rem/yr)	30	24	40	25
50-year fatal cancers	0.60	0.48	0.80	0.50
Noninvolved Workforce ^c			0.00	0.50
Average worker dose (mrem/yr) ^b	5.0	5.0	5.0	5.0
50-year risk of fatal cancer	1.0x10 ⁻⁴	1.0x10 ⁻⁴	1.0x10 ⁻⁴	1.0×10^{-4}
Total dose (person-rem/yr)	3.0	3.0	3.0	3.0
50-year fatal cancers	0.060	0.060	0.060	0.060
Total Site Workforce ^d			0.000	0.000
Dose (person-rem/yr)	33	27	43	28
50-year fatal cancers	0.66	0.54	0.86	0.56

 Table 4.2.2.9–2.
 Potential Radiological Impacts to Workers During Normal Operation at Nevada Test Site—Storage Alternatives

^a The involved worker is a worker associated with operations of the proposed action. The number of involved badged workers for the consolidation and collocation alternatives using the New Storage Facility would be 92 and 95, respectively. For storage in P-Tunnel, the number of involved badged workers would be 113 for the consolidation alternative and 152 for the collocation alternative. The maximum dose to an involved worker would be kept below 500 mrem per year. An effective ALARA program will ensure that exposure will be reduced to that level which is as low as reasonably achievable.

^b The radiological limit for an individual worker is 5000 mrem/year (10 CFR 835). However, DOE has also established an administrative control level of 2,000 mrem per year (DOE 1992t); the site must make reasonable attempts to maintain worker doses below this level.

^c The noninvolved worker is a worker onsite but not associated with operations of the proposed action. The projected number of noninvolved badged workers in 2005, and beyond, is 619. The Noninvolved Workforce is equivalent to the No Action workforce.

^d The impact to the total site workforce is the summation of the involved worker impact and the noninvolved worker impact. [Text deleted.]

Source: NT DOE 1996a and Section M.2.

いたいにないない たいとうちょう

.,

Table 4 2 2 9-3	Potential Hazardous Chemical Impacts to the Public and Workers During Normal Operation at Nevada Test S	d Workers During Normal Operation at Nevada Test Site—No Act	ACHO	n
10010 4.2.2.7-5.	and Storage Alternatives	- +	-	-

, -	· · · · · ·		Consoli	dation			Colle	ocation	
,		Modify P	Tunnel	New F		Modify P	-Tunnel	New F	acilities
- -	No Action	Facilities ^b	Total Site ^a	Facility ^b	Total Site ^a	Facilityb	Total Site ^a	Facility ^b	Total Site ^a
Maximally Exposed Individual (Public) Hazard Index ^c	0,	2.5x10 ⁻⁶	2.5x10 ⁻⁶	2.3x10 ⁻⁶	2.3×10 ⁻⁶	2.8x10 ⁻⁶	2.8x10 ⁻⁶	4.2x10 ⁻⁶	4.2x10 ⁻⁶
Cancer Risk ^d Worker Onsite Hazard Index ^e	0 0	4.1x10 ⁻⁹	4.1x10 ⁻⁹	4.1×10^{-9} 4.7×10^{-4}	4.1x10 ⁻⁹ 4.7x10 ⁻⁴	4.1×10^{-9} 5.6x10 ⁻⁴ 6.4x10 ⁻⁶	4.1x10 ⁻⁴ 5.6x10 ⁻⁴ 6.4x10 ⁻⁶	4.1x10 7.2x10 ⁻⁴ 6.4x10 ⁻⁶	7.2x10 ⁻⁴ 6.4x10 ⁻⁶

^a Total=Sum of the No Action plus the contributions of the above activity.

4-114

^b Contribution from the above activity only (for example, the amount of increase over the existing, No Action level at the site).

^e Hazard Index for MEI=Sum of Individual Hazard Quotients (Noncancerous health effects) for MEI.

^d Cancer Risk for MEI=(Emissions for 8-hr) x (0.286 [converts concentrations to doses]) x (Slope Factor [SF]).

^e Hazard Index for workers=Sum of Individual Hazard Quotients (Noncancerous health effects) for workers.

f Cancer risk for worker=(Emissions for 8-hr) x (0.286 [converts concentrations to doses]) x (0.237 [Fraction of year exposed]) x (0.571 [Fraction of lifetime working]) x (SF). Note: Where there are no known carcinogens among the hazardous chemicals emitted, there are no slope factors, therefore the calculated cancer risk value is 0. Source: Section M.3, Tables M.3.4-5 through M.3.4-9.

11

Earthquakes offer the greatest threat from natural phenomena. Available seismology studies indicate that active faults such as the Mine Mountain Fault, the Carpetbag Fault, the Yucca Flat Fault, and the Cane Spring Fault in the NTS vicinity are capable of generating earthquakes of up to 0.85 g (PX DOE 1996b:5-16). NTS has a natural background seismicity. The Cane Springs Fault, located 5 to 8 km (3 to 5 mi) south-southeast of the DAF area, has been identified as the most significant feature from the standpoint of seismic risk. However, a large portion of seismic events occurring near NTS may have been aftershocks from past nuclear explosions. The proposed storage area in P-Tunnel is only a few hundred feet away from the site of some past nuclear explosions. Since the P-Tunnel has survived these explosions without noticeable degradation, it is not reasonably foreseeable that the proposed storage area would be damaged by an earthquake. However, if the P-Tunnel collapses, the impact forces could breach some containers. The collapse would also seal the containers inside the tunnel, resulting in no or minimal short-term releases to the environment. Thus, the consequences to the public and workers are considered negligible.

Consolidation Alternative

This section includes a description of radiological and hazardous chemical releases and their associated impacts resulting from either normal operation or accidents involved with the new material handling building and modified P-Tunnel drifts and with the new consolidated Pu storage facility at NTS.

Normal operation under either consolidated storage option would result in impacts that are within applicable regulatory limits.

[Text deleted.]

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Normal Operation. There would be no radiological releases during the construction of a new material handling building or from modifying P-Tunnel drifts at NTS. Construction worker exposures to material potentially contaminated with radioactivity (for example, from construction activities involved with existing contaminated soil) would be limited to assure that doses are maintained ALARA. Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be both radiological and hazardous chemical releases to the environment as well as direct in-plant exposures. The resulting doses and potential health effects to the public and workers at NTS are described below.

Radiological Impacts. The radiological impacts to the public resulting from the normal operation of the modified P-Tunnel and the associated handling building are given in Table 4.2.2.9–1. The dose to the MEI due to annual storage operation in the P-Tunnel drifts and handling building would be 5.6×10^{-6} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 1.4×10^{-10} . The impacts to the average individual would be less. As a result of storage operations in the year 2030, the population dose would be 1.7×10^{-6} person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 4.3×10^{-8} .

The dose to the MEI of the public due to annual total site operations is within the radiological limits specified in NESHAPS (40 CFR 61, Subpart H) and DOE Order 5400.5, and would be 4.2×10^{-3} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 1.0×10^{-7} . The impacts to the average individual would be less. This activity would be included in a program to ensure that doses to the public are ALARA. As a result of total site operation in the year 2030, the population doses would all be within the limit in proposed 10 CFR 834 and would be 3.7×10^{-3} person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 9.3×10^{-5} .

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Doses to onsite workers due to normal operations are given in Table 4.2.2.9–2. Included are involved workers directly associated with the new handling building and modified P-Tunnel drifts, workers who are not involved with the new building and modified P-Tunnel drifts, and the entire workforce at NTS. All doses fall within regulatory limits and administrative control levels. The associated risks and numbers of fatal cancers among the different workers from 50 years of operation are included in the table. For the purposes of analyses only, this PEIS assumes that TRU and TRU mixed waste would be treated onsite to the current planning-basis WIPP WAC, and shipped to WIPP for disposal. This PEIS also assumes that LLW and mixed LLW would be treated and disposed of in accordance with current site practice. Also, this analysis assumes that hazardous waste would be treated and disposed of in accordance with current site practice.

Hazardous Chemical Impacts. Hazardous chemical impacts to the public and to the onsite worker resulting from the normal operations of consolidated storage in the P-Tunnel facility and the new handling building at NTS are presented in Table 4.2.2.9–3. The impacts from all site operations, including the upgraded storage facilities, are also included in this table. Total site impacts, which include the No Action impact plus the facility impact, are provided. All analyses to support the values presented in this table are provided in Section M.3.

The HI to the MEI of the public is 2.5×10^{-6} , and the cancer risk is 4.1×10^{-9} as a result of operation of consolidated storage in the P-Tunnel facility and handling building in the year 2030. The total HI and cancer risk from hazardous chemicals would remain constant over 50 years of operation, provided exposures remain the same. The total site operation, including the consolidated facility, would result in an HI of 2.5×10^{-6} and a cancer risk of 4.1×10^{-9} for the MEI in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

The HI to the onsite worker at P-Tunnel is 5.1×10^{-4} , and the cancer risk is 6.4×10^{-6} as a result of operation of consolidated storage in the year 2030. The total HI and cancer risk would remain constant over 50 years of operation, provided exposures remain the same. The total operation, including the consolidated facility, would result in an HI of 5.1×10^{-4} and a cancer risk of 6.4×10^{-6} for the onsite worker. This would be expected to remain constant as a result of 50 years of operation.

Facility Accidents. A set of potential accidents for modified P-Tunnel drifts for which there may be releases of Pu that may impact onsite workers and the offsite population has been postulated for the P-Tunnel. The accident consequences and risks to a worker located 1,000 m (3,280 ft) from the accident release point, the maximum offsite individual located at the site boundary, and the population located within 80 km (50 mi) of the accident release point are summarized in Table 4.2.2.9–4. For the set of accidents analyzed, the maximum number of cancer fatalities in the population within 80 km (50 mi) would be 5.3x10⁻⁴ at NTS for the truck bay fire accident scenario with a probability of 1.0x10⁻⁷ per year. The corresponding 50-year facility lifetime risk from the same accident scenario for the population, maximum offsite individual, and worker at 1,000 m (3,280 ft), would be 5.1x10⁻⁹ (that is, one fatality in about 1,000,000 years) at NTS for the PCV Penetration by Corrosion accident scenario with a probability of 0.064 per year. The corresponding maximum offsite individual and worker 50-year facility lifetime risks would be 6.9x10⁻⁶ and 4.0x10⁻⁵, respectively. Section M.5 presents additional facility accident scenario with a probability of 0.064 per year.

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

	Worker	at 1,000 m	Maximum O	ffsite Individual	Population	n to 80 km	
Accident Description	Risk of Cancer Fatality (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatality (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatalities (per 50 yr) ^a	Number of Cancer Fatalities ^c	Accident Frequency (per yr)
PCV puncture by forklift impact	8.8x10 ⁻⁸	2.9x10 ⁻⁶	1.5x10 ⁻⁸	5.0x10 ⁻⁷	1.1x10 ⁻⁷	3.7x10 ⁻⁶	6.0x10 ⁻⁴
PCV breach by firearms discharge	5.2x10 ⁻⁹	2.9x10 ⁻⁷	8.8x10 ⁻¹⁰	5.0x10 ⁻⁸	6.5x10 ⁻⁹	3.7x10 ⁻⁷	3.5x10 ⁻⁴
PCV penetration by corrosion	4.0x10 ⁻⁵	1.3x10 ⁻⁵	6.9x10 ⁻⁶	2.2x10 ⁻⁶	5.1x10 ⁻⁵	1.6x10 ⁻⁵	0.064
Truck bay fire	2.1x10 ⁻⁹	4.2x10 ⁻⁴	3.6x10 ⁻¹⁰	7.2x10 ⁻⁵	2.7x10 ⁻⁹	5.3x10 ⁻⁴	1.0x10 ⁻⁷
Spontaneous combustion	2.1x10 ⁻¹¹	5.9x10 ⁻⁷	3.5x10 ⁻¹²	1.0x10 ⁻⁷	2.6x10 ⁻¹¹	7.5x10 ⁻⁷	1.0x10 ⁻⁷
Explosion outside of vault	~ 2.1x10 ⁻¹¹	4.2x10 ⁻⁶	3.6x10 ⁻¹²	7.1x10 ⁻⁷	2.6x10 ⁻¹¹	5.3x10 ⁻⁶	1.0x10 ⁻⁷
Nuclear criticality	1.5x10 ⁻¹¹	3.1x10 ⁻⁶	3.1x10 ⁻¹²	6.1x10 ⁻⁷	3.2x10 ⁻¹²	- 6.4x10 ⁻⁷	1.0x10 ⁻⁷
Expected risk ^d	4.0x10 ⁻⁵	·	6.9x10 ⁻⁶	-	5.1x10 ⁻⁵	-	

Table 4.2.2.9-4. Consolidation Alternative (P-Tunnel) Accident Impacts at Nevada Test Site

^a The risk values are calculated by multiplying the probability of cancer fatality (for the worker at 1,000 m or the MEI) or the number of cancer fatalities (for the population to 80 km) by the accident frequency and the number of years of operation.

^b Increased likelihood of cancer fatality to a hypothetical individual (a single onsite worker at a distance of 1,000 m or the site boundary, whichever is smaller, or to a hypothetical individual in the offsite population located at the site boundary) if exposed to the indicated dose. The value assumes the incident has occurred.

^c Estimated number of cancer fatalities in the entire offsite population out to a distance of 80 km if exposed to the indicated dose. The value assumes the accident has occurred.

^d Expected risk is the sum of the risks for each accident over the 50-year lifetime of the facility.

Note: All values are mean values.

Source: Calculated using the source terms in Tables M.5.2.8.1-3 and M.5.2.8.1-4 and the MACCS computer code

4-117

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

Construct New Plutonium Storage Facility

Normal Operation. There would be no radiological releases during the construction of a new consolidated Pu storage facility at NTS. Construction worker exposures to material potentially contaminated with radioactivity (for example, from construction activities involved with existing contaminated soil) would be limited to assure that doses are maintained ALARA. Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be both radiological and hazardous chemical releases to the environment as well as direct in-plant exposures. The resulting doses and potential health effects on the public and workers at NTS are described below.

Radiological Impacts. Radiological impacts to the public resulting from the normal operation of the new consolidated Pu storage facility are presented in Table 4.2.2.9–1. The impacts from all site operations, including the new consolidated storage facility, are also given in the table. To put operational doses into perspective, comparisons with natural background radiation doses are included in the table.

The dose to the MEI due to annual storage facility operation would be 1.3×10^{-6} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 3.2×10^{-11} . The impacts to the average individual would be less. As a result of storage facility operation in the year 2030, the population dose would be 2.6×10^{-6} person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 6.5×10^{-8} .

The dose to the MEI due to annual total site operations is within the radiological limits specified in NESHAPS (40 CFR 61, Subpart H) and DOE Order 5400.5, and would be 4.2×10^{-3} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 1.0×10^{-7} . The impacts to the average individual would be less. This activity would be included in a program to ensure that doses to the public are ALARA. As a result of total site operation in the year 2030, the population dose would be within the limit in proposed 10 CFR 834 and would be 3.7×10^{-3} person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 9.3×10^{-5} .

Doses to onsite workers from normal operations are given in Table 4.2.2.9–2. Included are involved workers directly associated with the new consolidated Pu storage facility, workers who are not involved with the storage facility, and the entire workforce at NTS. All doses fall within regulatory limits and administrative control levels. The associated risks and numbers of fatal cancers among the different workers from 50 years of operations are included in the table.

Hazardous Chemical Impacts. Hazardous chemical impacts to the public and to the onsite worker resulting from the normal operations of the consolidated storage facilities at NTS are presented in Table 4.2.2.9–3. The impacts from all site operations, including the new consolidated storage facilities are also included in this table. Total site impacts, which include the No Action impact plus the facility are provided. All analyses to support the values presented in this table are provided in Section M.3.

The HI to the MEI of the public is 2.3×10^{-6} , and the cancer risk is 4.1×10^{-9} as a result of operation of the consolidated storage facilities in the year 2030. The HI and cancer risk from hazardous chemicals would remain

4-118

constant over 50 years of operation, because exposures would be expected to remain the same. The total site operation, including the consolidated facility, would result in an HI of 2.3×10^{-6} and a cancer risk of 4.1×10^{-9} for the MEI in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

The HI to the onsite worker would be 4.7×10^{-4} , and the cancer risk is 6.4×10^{-6} as a result of operation of the consolidated storage facilities in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, because exposures would be expected to remain the same. The total site operation, including the consolidated facility, would result in an HI of 4.7×10^{-4} and a cancer risk of 6.4×10^{-6} for the onsite worker in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

Facility Accidents. A set of potential accidents for consolidation of Pu in a new storage facility for which there may be releases of Pu that may impact onsite workers and the offsite population has been postulated. The accident consequences and risks to a worker located 1,000 m (3,280 ft) from the accident release point, the maximum offsite individual located at the site boundary, and the population located within 80 km (50 mi) of the accident release point are summarized in Table 4.2.2.9–5. For the set of accidents analyzed, the maximum number of cancer fatalities in the population within 80 km (50 mi) would be 0.027 at NTS for the beyond design basis earthquake accident scenario with an estimated probability of 1.0×10^{-7} per year (that is, probability of severe earthquake occurring is estimated to be about 1.0×10^{-5} , once in 100,000 years, multiplied by a damage and release probability of 0.01). The corresponding 50-year facility lifetime risk from the same accident scenario for the population, maximum offsite individual, and worker at 1,000 m (3,280 ft), would be 9.4×10^{-7} . 1.3×10^{-9} , and 7.3×10^{-8} , respectively. The maximum population 50-year facility lifetime risk would be 9.4×10^{-5} (that is, one fatality in about 540,000 years) at NTS for the PCV penetration by corrosion accident scenario with a probability of 0.064 per year. The corresponding maximum offsite individual and worker 50-year facility lifetime risks would be 9.1×10^{-7} and 3.9×10^{-5} , respectively. Section M.5 presents additional facility accident data and summary descriptions of the accident scenarios identified in Table 4.2.2.9–5.

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

Collocation Alternative

1

This section includes a description of radiological and hazardous chemical releases and the associated impacts resulting from either normal operation or accidents involved with the consolidation of Pu storage and collocation with HEU storage facilities at NTS. This storage would take place in either the modified P-Tunnel drifts or in a new collocated Pu and HEU storage facility.

Normal operation under either the modified P-Tunnel option or the new collocated storage facility option at NTS would result in impacts that are within applicable regulatory limits.

Involved workers, those that would work in the facilities associated with the proposed action, may be subject to injury and, in some cases, fatality as a result of potential accidents. [Text deleted.] The locations of workstations, number of workers, personnel protective features, engineered safety features, and other design details affect the extent of worker exposures to accidents. Certain accidents such as fires, explosions and criticality could cause fatalities to workers close to the accident. Prior to construction of a new or modification of an existing facility, DOE Orders require detailed safety analyses to assure that facility designs and operating procedures limit the number of workers in hazardous areas and minimize risk of injury or fatality in the event of an accident.

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Non's

, , , , , , , , , , , , , , , , , , , ,		· ·	-				
· · · · · · · · · · · · · · · · · · ·	Wor 1,00	ker at)0 m	Maximu Indiv	m Offsite 🤞 vidual	E Dopula 80	ition to km	and the second s
i i i i i i	"Risk of	Probability	Risk of	Probability	Risk of	Number of	· · · ·
	Cancer	of Cancer	Cancer	of Cancer	Cancer	Cancer	Accident
3	Fatality	. Fatality ^o	~ 'Fatality	[•] Fatality ^D	Fatalities :	Fatalities	Frequency
Accident Description .	(per 50 yr) ^a		(per 50 yr) ^a	fi to the second s	(per 50 yr) ^a	, 1	(per yr)
PCV puncture by forklift	9.0x10 ⁻⁸	·、3.0x10 ⁻⁶	2.1x10 ⁻⁹	.{ 7. 0x10 ⁻⁸	, 2.2x10⁻⁷ ,	7.2x10 ⁻⁶	6.0x10 ⁻⁴
PCV breach by firearms discharge	5.3x10 ⁻⁹	3.0x10 ⁻⁷	1.2x10 ⁻¹⁰	7.0x10 ⁻⁹	1.3x10 ⁻⁸	7.2x10 ⁻⁷	3.5x10 ⁻⁴
PCV penetration by corrosion	3.9x10 ⁻⁵	1.2x10 ⁻⁵	9.1x10 ⁻⁷	2.9x10 ⁻⁷	9.4x10 ⁻⁵	3.0x10 ⁻⁵	0.064
Vault fire and the second	- 3.8x10 ⁻⁸	7.6x10 ⁻³	7.3x10 ⁻¹⁰	1:5x10 ⁻⁴	7.6x10 ⁻⁸	· 0.015	1.0x10 ⁻⁷
Truck bay fire	2.1x10 ⁻⁹	4.2x10 ⁻⁴	4.9x10 ⁻¹¹	9.7x10 ⁻⁶	5.1x10 ⁻⁹ -	-1.0x10 ⁻³	- 1.0x10 ⁻⁷
Spontaneous combustion	2.1x10 ⁻¹¹	6.0x10 ⁻⁷	4.9x10 ⁻¹³	1.4x10 ⁻⁸	5.1x10 ⁻¹¹	1.5x10 ⁻⁶	7.0x10 ⁻⁷
Explosion in the vault	4.9x10 ⁻⁹	9.9x10 ⁻⁴	1.1x10 ⁻¹⁰	2.3x10 ⁻⁵	1.2x10 ⁻⁸	2.4x10 ⁻³	1.0x10 ⁻⁷
Explosion outside of vault	2.3x10 ⁻¹¹	4.5x10 ⁻⁶	5.2x10 ⁻¹³	1.0x10 ⁻⁷	5.4x10 ⁻¹¹	1.1x10 ⁻⁵	1.0x10 ⁻⁷
Nuclear criticality	1.5x10 ⁻¹¹	3.1x10 ⁻⁶	3.3x10 ⁻¹³	6.5x10 ⁻⁸	3.5x10 ⁻¹²	6.9x10 ⁻⁷	_1.0x10 ⁻⁷
Beyond design basis earthquake	7.3x10 ⁻⁸	0.015	1.3x10 ⁻⁹	2.6x10 ⁻⁴	1.4x10 ⁻⁷	0.027	1.0x10 ⁻⁷
Expected risk ^d	4.0x10 ⁻⁵	-	9.2x10 ⁻⁷		9.5x10 ⁻⁵	- .	

Table 4.2.2.9-5. Consolidation Alternative (New Storage Facility) Accident Impacts at Nevada Test Site

^a The risk values are calculated by multiplying the probability of cancer fatality (for the worker at 1,000 m or the maximum offsite individual) or the number of cancer fatalities (for the population to 80 km) by the accident frequency and the number of years of operation

^b Increased likelihood (or probability) of cancer fatality to a hypothetical individual (a single onsite worker at a distance of 1,000 m or the site boundary, whichever is smaller or to a hypothetical individual in the offsite population located at the site boundary) if exposed to the indicated dose. The value assumes the accident has occurred

^c Estimated number of cancer fatalities in the entire offsite population out to a distance of 80 km if exposed to the indicated dose. The value assumes the accident has occurred.

^d Expected risk is the sum of the risks for each accident over the 50-year lifetime of the facility. Note: All values are mean values.

Source: Calculated using the source terms in Tables M.5.2.1.1-5 and M.5.2.1.1-6 and the MACCS computer code.

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Normal Operation. There would be no radiological releases during the construction of a new, but larger (than for consolidation) material handling building or from modifying P-Tunnel drifts at NTS. Worker exposures to material potentially contaminated with radioactivity (for example, from construction activities involved with existing contaminated soil) would be limited to assure that doses are maintained ALARA. Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be both radiological and hazardous chemical releases to the environment as well as direct in-plant exposures. The resulting doses and potential health effects to the public and workers are described below.

Radiological Impacts. Radiological impacts to the public resulting from the normal operation of the modified P-Tunnel drifts and the associated handling building at NTS are included in the information presented in Table 4.2.2.9-1. The impacts from all site operations are also given in the table. To put operational doses into perspective, comparisons with natural background radiation doses are also included in the table. Similar information regarding radiological impacts to workers is given in Table 4.2.2.9-2. [Text deleted.]

The dose to the MEI due to annual storage operation in the P-Tunnel and handling building would be 5.6×10^{-6} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 1.4×10^{-10} .

4-120

I

I

The impacts to the average individual would be less. As a result of storage operations in the year 2030, the population dose would be 1.7×10^{-6} person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 4.3×10^{-8} .

The dose to the MEI due to annual total site operations is within the radiological limits specified in NESHAPS (40 CFR 61, Subpart H) and DOE Order 5400.5, and would be 4.2×10^{-3} mrem. From 50 years of operation, the corresponding risk of fatal cancer to this individual would be 1.0×10^{-7} . The impacts to the average individual would be less. This activity would be included in a program to ensure that doses to the public are ALARA. As a result of total site operation in the year 2030, the population dose would be within the limit in proposed 10 CFR 834 and would be 3.7×10^{-3} person-rem. The corresponding number of fatal cancers in this population due to 50 years of operation would be 9.3×10^{-5} .

Doses to onsite workers due to normal operations are given in Table 4.2.2.9–2. Included are involved workers directly associated with the modified P-Tunnel drifts and handling building, workers who are not involved with the modified P-Tunnel drifts and handling building, and the entire workforce at NTS. All doses fall within regulatory limits and administrative control levels. The associated risks and numbers of fatal cancers among the different workers from 50 years of operation are included in the table.

I

1

Hazardous Chemical Impacts. Hazardous chemical impacts to the public and to the onsite worker resulting from the normal operations of the collocated storage facility in the P-Tunnel at NTS are presented in Table 4.2.2.9–3.

The HI to the MEI of the public is 2.8×10^{-6} , and the cancer risk is 4.1×10^{-9} as a result of operation of the collocated storage facilities in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, because exposures would be expected to remain the same. The total site operation in P-Tunnel would result in an HI of 2.8×10^{-6} and a cancer risk of 4.1×10^{-9} for the MEI in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

The HI to the onsite worker at P-Tunnel would be 5.6×10^{-4} , and the cancer risk is 6.4×10^{-6} as a result of operation of the collocated storage facility in the P-Tunnel at NTS in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, because exposures would be expected to remain the same. The total site operation including the consolidated facility would result in an HI of 5.6×10^{-4} and a cancer risk of 6.4×10^{-6} for the onsite worker in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

Facility Accidents. A set of potential accidents for modified P-Tunnel and collocated Pu and HEU storage facilities for which there may be releases of Pu or uranium that may impact onsite workers and the offsite population has been postulated. The consequences and risks of potential accidents that release both Pu and HEU would be bounded by the impacts associated with Pu. The accident consequences and risks to a worker located 1,000 m (3,280 ft) from the accident release point, the maximum offsite individual located at the site boundary, and the population located within 80 km (50 mi) of the accident release point are summarized in Table 4.2.2.9–6 For the set of accidents analyzed, the maximum number of cancer fatalities in the population within 80 km (50 mi) would be 5.3×10^{-4} at NTS for the truckbay fire accident scenario with a probability of 1.0×10^{-7} per year. The corresponding 50-year facility lifetime risk from the same accident scenario for the population, maximum offsite individual, and worker at 1,000 m (3,280 ft), would be 2.7×10^{-9} , 3.6×10^{-5} (that is, one fatality in about 1,000,000 years) at NTS for the PCV penetration by corrosion accident scenario with a probability of 0.064 per year. The corresponding maximum offsite individual and worker 50-year facility lifetime risks would be 6.9×10^{-6} and 4.0×10^{-5} , respectively. Section M.5 presents additional facility accident data and summary descriptions of the accident scenarios identified in Table 4.2.2.9–6.

4

i i i i i i i i i i i i i i i i i i i	Worker at	1,000 m 😁	Maximum Of	Tsite Individual	Population	to 80 km 💦 📉	
Accident Description	Risk of Cancer Fatality C (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatality (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatalities (per 50 yr) ^a	Number of Cancer Fatalities ^c	Accident Frequency (per yr)
CV puncture by forklift impact.	8.8x10 ⁻⁸	2.9x10 ⁻⁶ ~ -	1.5x10 ⁻⁸	5.0x10 ⁻⁷	1.1x10 ⁻⁷	3.7x10 ⁻⁶	6.0x10 ⁻⁴
CV breach by firearms discharge	5.2x10 ⁻⁹	2.9x10 ⁻⁷	8.8x10 ⁻¹⁰	5.0x10 ⁻⁸	6.5x10 ⁻⁹	3.7x10 ⁻⁷	3.5x10 ⁻⁴
CV penetration by corrosion	4.0x10 ⁻⁵	1.3x10 ⁻⁵	6.9x10 ⁻⁶	2.2×10^{-6}	5.1x10 ⁻⁵	1.6x10 ⁻⁵	0.064
nick bay fire	2.1×10^{-9}	4.2×10^{-4}	3.6x10 ⁻¹⁰	7.2x10 ⁻⁵	2.7×10 ⁻⁹	5.3x10 ⁻⁴	1.0×10^{-7}
pontaneous combustion	2.1×10^{-11}	5.9x10 ⁻⁷	3.5x10 ⁻¹²	1.0x10 ⁻⁷	2.6x10 ⁻¹¹	7.5×10 ⁻⁷ -	1.0×10^{-7}
valorion outside of vault	2.1x10-11 × ·	4.2×10 ⁻⁶	3.6x10 ⁻¹²	7 1x10 ⁻⁷	2.6x10 ⁻¹¹	5 3x10 ⁻⁶	1.0×10^{-7}
uclear criticality	1.5x10 ⁻¹¹	3.1×10^{-6}	3.1×10^{-12}	6 1x10 ⁻⁷	32×10^{-12}	6.4×10^{-7}	1.0×10^{-7}
reported riskd	1.0.10-5	5.1710	60-10-6	· 0,1110	5.1×10 ⁻⁵	0.4770	iloxio,
the population to 80 km) by the accident ncreased likelihood of cancer fatality to ndividual in the offsite population locate Estimated number of cancer fatalities in	is frequency and the nui a hypothetical individ ed at the site boundary the entire offsite popu	mber of years of op dual (a single onsite y) if exposed to the ilation out to a dista	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex	nce of 1,000 m or the si the value assumes the in posed to the indicated	ite boundary, which cident has occurred dose. The value ass	or the number of ca lever is smaller, or to l. sumes the accident I	o a hypothetical has occurred.
the risk values are calculated by humph the population to 80 km) by the accident increased likelihood of cancer fatality to individual in the offsite population locate estimated number of cancer fatalities in expected risk is the sum of the risks for	is frequency and the number of the state of	mber of years of op dual (a single onsite y) if exposed to the ilation out to a dista e 50-year lifetime of	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex f the facility.	too m or the maximum nee of 1,000 m or the si te value assumes the in posed to the indicated	offsite individual) of ite boundary, which cident has occurred dose. The value ass	or the number of ca lever is smaller, or to l. sumes the accident l	o a hypothetical
the population to 80 km) by the accident increased likelihood of cancer fatality to individual in the offsite population locate stimated number of cancer fatalities in expected risk is the sum of the risks for te: All values are mean values.	is frequency and the number of the site boundary the entire offsite population accident over the site boundary the entire offsite population over the site boundary the site accident over the site boundary the site accident over the site boundary the site accident over the si	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ilation out to a dista e 50-year lifetime of	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex f the facility.	too m or the maximum nee of 1,000 m or the si te value assumes the in sposed to the indicated	offsite individual) of ite boundary, which cident has occurred dose. The value ass	ever is smaller, or to sumes the accident l	o a hypothetical
the population to 80 km) by the accident ncreased likelihood of cancer fatality to ndividual in the offsite population locate stimated number of cancer fatalities in expected risk is the sum of the risks for te: All values are mean values.	in Tables M.5.2.8.1–3	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ulation out to a dista e 50-year lifetime of 3 and M.5.2 8.1–4 a	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex f the facility. nd the MACCS co	00 m or the maximum nee of 1,000 m or the side value assumes the in posed to the indicated omputer code.	orisite individual) of ite boundary, which cident has occurred dose. The value ass	ever is smaller, or to lever is smaller, or to sumes the accident l	o a hypothetical
he population to 80 km) by the accident increased likelihood of cancer fatality to individual in the offsite population locate stimated number of cancer fatalities in xpected risk is the sum of the risks for ie: All values are mean values.	in Tables M.5.2.8.1–3	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ilation out to a dista e 50-year lifetime of 3 and M.5.2 8.1–4 a	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex f the facility. nd the MACCS co	00 m or the maximum nee of 1,000 m or the si te value assumes the in posed to the indicated omputer code.	offsite individual) of ite boundary, which cident has occurred dose. The value ass	or the number of ca lever is smaller, or to sumes the accident I	o a hypothetical
the risk values are carculated by humph he population to 80 km) by the accident increased likelihood of cancer fatality to adividual in the offsite population locate stimated number of cancer fatalities in expected risk is the sum of the risks for the: All values are mean values.	in Tables M.5.2.8.1-3	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ulation out to a dista e 50-year lifetime of 3 and M.5.2 8.1–4 a	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex the facility. nd the MACCS co	00 m or the maximum nee of 1,000 m or the si le value assumes the in sposed to the indicated omputer code.	offsite individual) of ite boundary, which cident has occurred dose. The value ass	ever is smaller, or to sumes the accident l	o a hypothetical
the risk values are calculated by humph he population to 80 km) by the accident nereased likelihood of cancer fatality to ndividual in the offsite population locate istimated number of cancer fatalities in expected risk is the sum of the risks for the All values are mean values. Ince: Calculated using the source terms	is frequency and the number of the site boundary the entire offsite popule each accident over the in Tables M.5.2.8.1–3	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ilation out to a dista e 50-year lifetime of and M.5.2 8.1-4 a	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex f the facility. nd the MACCS co	00 m or the maximum nee of 1,000 m or the si le value assumes the in posed to the indicated omputer code.	offsite individual) of ite boundary, which cident has occurred dose. The value ass	or the number of ca lever is smaller, or to sumes the accident I	o a hypothetical has occurred.
the risk values are calculated by humph the population to 80 km) by the accident increased likelihood of cancer fatality to adividual in the offsite population locate stimated number of cancer fatalities in expected risk is the sum of the risks for the: All values are mean values. Firce: Calculated using the source terms	is frequency and the number of the site boundary the entire offsite population accident over the in Tables M.5.2.8.1–3	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ilation out to a dista e 50-year lifetime of 3 and M.5.2 8.1–4 a	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex the facility. nd the MACCS co	00 m or the maximum nee of 1,000 m or the si te value assumes the in posed to the indicated omputer code.	orisite individual) of ite boundary, which cident has occurred dose. The value ass	or the number of ca lever is smaller, or to sumes the accident I	o a hypothetical
the risk values are calculated by humph the population to 80 km) by the accident nereased likelihood of cancer fatality to adividual in the offsite population locate stimated number of cancer fatalities in expected risk is the sum of the risks for the All values are mean values.	in Tables M.5.2.8.1-3	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ulation out to a dista e 50-year lifetime of 3 and M.5.2 8.1–4 at	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex the facility. nd the MACCS co	00 m or the maximum nee of 1,000 m or the si le value assumes the in sposed to the indicated omputer code.	orisite individual) of ite boundary, which cident has occurred dose. The value ass	ever is smaller, or to sumes the accident I	o a hypothetical has occurred.
the risk values are calculated by humph he population to 80 km) by the accident nereased likelihood of cancer fatality to ndividual in the offsite population locate istimated number of cancer fatalities in expected risk is the sum of the risks for te: All values are mean values. Firce: Calculated using the source terms	in Tables M.5.2.8.1–3	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ilation out to a dista e 50-year lifetime of and M.5.2 8.1–4 at	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex the facility. nd the MACCS co	00 m or the maximum nee of 1,000 m or the si le value assumes the in posed to the indicated omputer code.	offsite individual) of ite boundary, which cident has occurred dose. The value ass	ever is smaller, or to sumes the accident l	o a hypothetical has occurred.
the risk values are calculated by humph the population to 80 km) by the accident increased likelihood of cancer fatality to adividual in the offsite population locate stimated number of cancer fatalities in Expected risk is the sum of the risks for the All values are mean values. Since: Calculated using the source terms	ying the probability of frequency and the num a hypothetical individed at the site boundary the entire offsite popu each accident over the in Tables M.5.2.8.1–3	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ulation out to a dista e 50-year lifetime of and M.5.2 8.1–4 a	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex the facility.	00 m or the maximum nee of 1,000 m or the si te value assumes the in sposed to the indicated omputer code.	orisite individual) of ite boundary, which cident has occurred dose. The value ass	ever is smaller, or to sumes the accident l	o a hypothetical has occurred.
the risk values are calculated by humph he population to 80 km) by the accident nereased likelihood of cancer fatality to adividual in the offsite population locate stimated number of cancer fatalities in expected risk is the sum of the risks for te: All values are mean values. Firce: Calculated using the source terms	in Tables M.5.2.8.1–3	b cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ilation out to a dista e 50-year lifetime of 3 and M.5.2 8.1–4 at	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex the facility.	too m or the maximum nee of 1,000 m or the si le value assumes the in sposed to the indicated omputer code.	offsite individual) of ite boundary, which cident has occurred dose. The value ass	ever is smaller, or to sumes the accident I	o a hypothetical has occurred.
he population to 80 km) by the accident ncreased likelihood of cancer fatality to ndividual in the offsite population locate stimated number of cancer fatalities in Expected risk is the sum of the risks for te: All values are mean values. Frce: Calculated using the source terms	is frequency and the number of the entire offsite population of the entire offsite population and the entire offsite population accident over the in Tables M.5.2.8.1–3	of cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the alation out to a dista to 50-year lifetime of and M.5.2 8.1-4 and and M.5.2 8.1-4	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex the facility. nd the MACCS co	too m or the maximum nee of 1,000 m or the si le value assumes the in posed to the indicated omputer code.	orrsite individual) of ite boundary, which cident has occurred dose. The value ass	ever is smaller, or to sumes the accident l	o a hypothetical has occurred.
the population to 80 km) by the accident increased likelihood of cancer fatality to individual in the offsite population locate stimated number of cancer fatalities in expected risk is the sum of the risks for te: All values are mean values.	in Tables M.5.2.8.1–3	of cancer fatality (for mber of years of op dual (a single onsite y) if exposed to the ulation out to a dista e 50-year lifetime of and M.5.2 8.1–4 a	r the worker at 1,0 eration. worker at a distar indicated dose. Th nce of 80 km if ex the facility. nd the MACCS co	too m or the maximum nee of 1,000 m or the si revalue assumes the in posed to the indicated omputer code.	orisite individual) of ite boundary, which cident has occurred dose. The value ass	ever is smaller, or to sumes the accident I	o a hypothetical has occurred.

.

-

'.

.

. -~ ÷

. . .

.

•

.1

4 14 In T **m** . 11 m .

..... - ------

-

.

~

·.•

• •

۰,

. 7

-- 14 1

а. н . 5 89 ,

. .

č

-2.

.

.

.....

184. - 1³ - 14⁶ - 1 4 ~

•

3

ĵ,

4-122

.

-

: . .

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Normal Operation. There would be no radiological releases during the construction of a new collocated storage facility at NTS. Worker exposures to material potentially contaminated with radioactivity would be limited to assure that doses are maintained ALARA. Toward this end, construction workers would be monitored as appropriate. Limited hazardous chemical releases are anticipated as a result of construction activities. However, concentrations would be within the regulated exposure limits. During normal operation, there would be both radiological and hazardous chemical releases to the environment as well as direct in-plant exposures. The resulting doses and potential health effects to the public and workers are described below.

Radiological Impacts. Radiological impacts to the public resulting from the normal operation of the new collocated storage facility at NTS are included in the information presented in Table 4.2.2.9-1. The impacts from all site operations are also given in the table. To put operational doses into perspective, comparisons with natural background radiation doses are also included in the table. Similar information regarding radiological impacts to workers is given in Table 4.2.2.9-2. [Text deleted.]

The dose to the MEI from the annual storage operations in the new storage facility would be 1.3×10^{-6} mrem. For 50 years of operation, the corresponding risk of fatal cancer to this individual would be 3.2x10⁻¹¹. The impacts to the average individual would be less. As a result of storage operation in the year 2030, the population dose would be 2.6x10⁻⁶ person-rem. The corresponding number of fatal cancers in this population from 50 years of operation would be 6.5×10^{-8} .

Doses and associated health risks to the public from total site operations are virtually the same whether storage is in the new storage facility or in the modified P-Tunnel drifts (Table 4.2.2.9-1). This is because the storage operations contribute negligibly to the total offsite doses.

Doses to onsite workers due to normal operations are given in Table 4.2.2.9-2. All doses fall within regulatory limits and administrative control levels. The associated risks and numbers of fatal cancers among the different workers from 50 years of operation are included in the table. Dose to individual workers would be kept low by instituting badged monitoring and ALARA programs and also workers rotations. As a result of the implementation of these mitigation measures, the actual number of fatal cancers calculated would be lower for the operation of this facility.

Hazardous Chemical Impacts. Hazardous chemical impacts to the public and to the onsite worker resulting from the normal operations of the new consolidation of Pu storage and collocation with HEU storage facilities at NTS are presented in Table 4.2.2.9-3. The impacts from all site operations, including the consolidation of Pu storage and collocation with HEU storage facilities, are also included in this table. Total site impacts, which include the No Action impact plus the facility impacts, are provided. All analyses to support the values presented in this table are provided in Section M.3

- The HI to the MEI of the public is 4.2×10^{-6} , and the cancer risk is 4.1×10^{-9} as a result of operation of the new consolidation of Pu storage and collocation with HEU storage facilities in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, because exposures would be expected to remain the same. The total site operation, including the new facility, would result in an HI of 4.2x10⁻⁶ and a cancer risk of 4.1x10⁻⁹ for the MEI in the year 2030. This would be expected to remain constant as a result of 50 years of operation.
- The HI to the onsite worker would be 7.2×10^{-4} , and the cancer risk is 6.4×10^{-6} as a result of operation of the new consolidation of Pu storage and collocation with HEU storage facilities in the year 2030. The HI and cancer risk would remain constant over 50 years of operation, because exposures would be expected to remain the same. The total site operation, including the new facility, would result in an HI of 7.2x10⁻⁴ and a cancer risk of 6.4×10^{-6} for the onsite worker in the year 2030. This would be expected to remain constant as a result of 50 years of operation.

I

Facility Accidents. A set of potential accidents for a new storage facility for collocated Pu and HEU for which there may be releases of Pu or HEU that may impact onsite workers and the offsite population has been postulated. The consequences and risks of potential accidents that release both Pu and HEU would be bounded by the impacts associated with Pu. The accident consequences and risks to a worker located 1,000 m (3.280 ft)⁺ from the accident release point, the maximum offsite individual located at the site boundary, and the population located within 80 km (50 mi) of the accident release point are summarized in Table 4.2.2.9-7. For the set of accidents analyzed, the maximum number of cancer fatalities in the population within 80 km (50 mi) would be 0.027 at NTS for the beyond design basis earthquake accident scenario with an estimated probability of 1.0×10^{-7} per year (that is, probability of severe earthquake occurring is estimated to be about 1.0x10⁻⁵, once in 100,000 years, multiplied by a damage and release probability of 0.01). The corresponding 50-year facility lifetime risk from the same accident scenario for the population, maximum offsite individual, and worker at 1,000 m (3,280 ft), would be 1.4×10^{-7} , 1.3×10^{-9} , and 7.3×10^{-8} , respectively. The maximum population 50-year facility lifetime risk would be 9.4x10⁻⁵ (that is, one fatality in about 530,000 years) at NTS for the PCV penetration by corrosion accident scenario with a probability of 0.064 per year. The corresponding maximum offsite individual and worker 50-year facility lifetime risks would be 9.1x10⁻⁷ and 3.9x10⁻⁵, respectively. Appendix M.5 presents additional facility accident data and summary descriptions of the accident scenarios identified in Table 4.2.2.9-7.

21. 2005

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

If the strategic reserve and weapons R&D materials are not included, the impacts to the public and to workers from the accident-free storage activities would be reduced in proportion to the decrease in the amount of material stored. The impacts from total site operations would decrease slightly. The risks due to accidents would also tend to be lower.

	Worker	at 1,000 m	Maximum O	ffsite Individual	Populatio	n to 80 km	
Accident Description	Risk of Cancer Fatality (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatality (per 50 yr) ^a	Probability of Cancer Fatality ^b	Risk of Cancer Fatalities (per 50 yr) ^a	Number of Cancer Fatalities ^c	Accident Frequency (per yr)
PCV puncture by forklift	9.0x10-8	3.0x10 ⁻⁶	2.1x10 ⁻⁹	7.0x10 ⁻⁸	2.2x10 ⁻⁷	7.2x10 ⁻⁶	6.0x10 ⁻⁴
PCV breach by firearms discharge	5.3x10-9	3.0x10 ⁻⁷	1.2x10 ⁻¹⁰	7.0x10 ⁻⁹	1.3x10 ⁻⁸	7.2x10 ⁻⁷	3.5x10 ⁻⁴
PCV penetration by corrosion	3.9x10-5	1.2x10 ⁻⁵	9.1x10 ⁻⁷	2.9x10 ⁻⁷	9.4x10 ⁻⁵	3.0x10 ⁻⁵	0.064
Vault fire	3.8x10-8	7.6x10 ⁻³	7.3x10 ⁻¹⁰	1.5x10 ⁻⁴	7.6x10 ⁻⁸	0.015	1.0x10 ⁻⁷
Truck bay fire	2.1x10-9	4.2x10 ⁻⁴	4.9x10 ⁻¹¹	9.7x10 ⁻⁶	5.1x10 ⁻⁹	1.0x10 ⁻³	1.0x10 ⁻⁷
Spontaneous combustion	2.1x10-11	6.0x10 ⁻⁷	4.9x10 ⁻¹³	1.4x10 ⁻⁸	5.1x10 ⁻¹¹	1.5x10 ⁻⁶	7.0x10 ⁻⁷
Explosion in the vault	4.9x10-9	9.9x10 ⁻⁴	1.1x10 ⁻¹⁰	2.3x10 ⁻⁵	1.2x10 ⁻⁸	2.4x10 ⁻³	1.0x10 ⁻⁷
Explosion outside the vault	2.3x10-11	4.5x10 ⁻⁶	5.2x10 ⁻¹³	1.0x10 ⁻⁷	-5.4x10 ⁻¹¹	1.1x10 ⁻⁵	1.0x10 ⁻⁷
Nuclear criticality	1.5x10-11	3.1x10 ⁻⁶	3.3x10 ⁻¹³	6.5x10 ⁻⁸	3.5x10 ⁻¹²	6.9x10 ⁻⁷	1.0x10 ⁻⁷
Beyond evaluation basis earthquake	7.3x10-8	0.014	1.3x10 ⁻⁹	2.6x10 ⁻⁴	1.4x10 ⁻⁷	0.027	1.0x10 ⁻⁷
Expected risk ^d	4.0x10-5	-	9.2x10 ⁻⁷		9.5x10 ⁻⁵	<u> </u>	-

Table 4.2.2.9–7. Collocation Alternative (New Plutonium and Highly Enriched Uranium Facilities) Accident Impacts at Nevada Test Site

^a The risk values are calculated by multiplying the probability of cancer fatality (for the worker at 1,000 m or the maximum offsite individual) or the number of cancer fatalities (for the population to 80 km) by the accident frequency and the number of years of operation.

^b Increased likelihood (or probability) of cancer fatality to a hypothetical individual (a single onsite worker at a distance of 1,000 m or the site boundary, whichever is smaller or to a hypothetical individual in the offsite population located at the site boundary) if exposed to the indicated dose. The value assumes the accident has occurred.

^c Estimated number of cancer fatalities in the entire offsite population out to a distance of 80 km if exposed to the indicated dose. The value assumes the accident has occurred.

^d Expected risk is the sum of the risks over the 50-year lifetime of the facility. Note: All values are mean values.

Source: Calculated using the source terms in Tables M.5.2.2.1-3 and M.5.2.2.1-4 and the MACCS computer code.

1

1000

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

F.

4.2.2.10 Waste Management

This section summarizes the impacts on waste management at NTS under No Action and for each of the longterm storage alternatives. There is no spent nuclear fuel or HLW associated with Pu or HEU storage. Table 4.2.2.10-1 lists the projected sitewide waste generation rates and treatment, storage, and disposal capacities under No Action for 2005. Projections for No Action were derived from the most recent applicable environmental data, with the appropriate adjustments made for those changing operational requirements where the volumes of wastes generated are identifiable. The projections do not include waste from future, yet uncharacterized, environmental restoration activities. The projections for No Action could change significantly depending on the decisions resulting from the Waste Management PEIS or the NTS Site-Wide EIS. Table 4.2.2.10-2 provides the estimated incremental operational waste volumes projected to be generated at NTS as a result of the various storage alternatives prior to treatment. Some of the waste values described in this section are different than the waste values in the table. For those values that differ (for example LLW), the table gives waste generated pre-treatment values and the text discusses post-treatment values (indicated as after treatment and volume reduction). The waste volumes generated from the various storage alternatives that were added to the No Action projection and the resultant waste effluent used for the waste impact analysis are shown in Section E.3.1. Facilities that would support the storage of Pu and/or HEU would treat and package all generated waste into forms that would enable staging and/or disposal in accordance with RCRA and other applicable statutes as outlined in Section E.1.2. Depending in part on decisions in waste-type-specific RODs for the Waste Management PEIS and in subsequent RODs and NEPA documents, wastes could be treated and disposed of onsite or at regionalized or centralized DOE sites. For purposes of analyses only, this PEIS assumes that TRU and mixed TRU waste would be treated onsite to the current planning-basis WIPP WAC, and shipped to WIPP for disposal. This PEIS also assumes that LLW, mixed LLW, hazardous, and nonhazardous wastes would be treated and disposed of in accordance with current site practice.

がらい

Preferred Alternative: No Action Alternative

Under this alternative, TRU, low-level, mixed, hazardous, and nonhazardous wastes would continue to be managed from the missions outlined in Section 3.3. The disposal of waste received from offsite would not involve treatment at NTS, since this waste must be treated, packaged, and certified to NTS WAC before being shipped to NTS for disposal. NTS has retrievable storage of TRU waste awaiting shipment to a Federal repository. Although there would be no generation of TRU waste onsite, mixed TRU waste from LLNL would continue to be stored at NTS. Solid mixed LLW would be stored and treated in accordance with the NTS Site Treatment Plan. Hazardous waste would be accumulated, then shipped offsite for treatment and disposal at commercial RCRA-permitted facilities. Nonhazardous and sanitary wastes would be treated and disposed of locally in facilities located within the separate activity areas onsite.

Consolidation Alternative

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Construction and operation of a consolidated Pu storage facility using P-Tunnel would have an impact on existing NTS waste management activities, increasing the generation of TRU, low-level, mixed, hazardous, and nonhazardous wastes. Waste generated during construction would consist of wastewater and solid nonhazardous and hazardous wastes. The nonhazardous waste would be disposed of as part of the construction project by the contractor, and the hazardous waste would be shipped to commercial RCRA-permitted treatment and disposal facilities. No soil contaminated with hazardous material or radioactive constituents is expected to be generated during construction. However, if any was generated it would be managed in accordance with site practice and all applicable Federal and State regulations.

After treatment and volume reduction, approximately 5 m^3 (7 yd³) of TRU waste and 4 m^3 (5 yd³) of mixed TRU waste from leaded gloves, windows, and contaminated lead shielding would be treated and packaged to

4-126

Category	Annual Generation (m ³)	Treatment Method	Treatment Capacity	Storage Method	Storage Capacity (m ³)	Disposal Method	Disposal Capacity (m ³)
Mixed Transuranic	None	None	None	Containers on asphalt pads	1,208 ^a	To WIPP or alternate facility	NA
Low-Level	Dependent on	Not determined	None	None	NA	NA .	NA
Solid	15,000 ^b	None	None	None	. NA	Shallow burial	650,000 ^c
Mixed Low-Level	·		•		•	Ξ	
Liquid	None	None	NA	None	NA	NA	NA
Solid	50 ^d	None ·	NA	Containers on asphalt pads	Included in mixed TRU	Shallow land burial ^e	90,626 ^f
Hazardous				- ,			
Liquid	Included in solid	Contracted offsite	None	RCRA-permitted pad	Included in solid	Contracted offsite	NA
Solid	212	Contracted offsite	None	RCRA-permitted pad	61.6 ^g	Contracted offsite	NA
Nonhazardous (Sanitary)							
Liquid	Included in solid	Septic fields	As required	None	NA	Septic fields	As required
Solid	2,120	None	None	NA	NA	Landfill onsite	As required
Nonhazardous (Other)						•	
Liquid	Included in sanitary	Septic fields	As required	None	NA	Septic fields	As required
Solid	76,500	None	None	NA	NA	NA	As required

Table 4.2.2.10–1. Projected Waste Management Under No Action (2005) at Nevada Test Site

^a TRU waste pad and cover building, Area 5. Storage area is 1,765 m². Volume estimate made assuming 50 percent of area is available due to aisle space and drums are stacked two high.

^b Depending on ROD from the sitewide EIS, additional solid LLW could be received.

^c Area 3 and 5. Additional acreage available for expansion

^d Depending on ROD from the sitewide EIS, additional mixed LLW could be generated from onsite environmental restoration activities and receipts from offsite for disposal.

• As required to meet RCRA requirements.

f Remaining capacity Pit 3, Area 5 RWMS.

^g Total capacity for solid and liquid hazardous wastes stored at the Hazardous Waste Storage Unit in Area 5.

Source: NTS 1996a:1.

127

			1 * +		Co	nsolidat	ion		۴. ę.	· C	collocatio	n			
	Category		No Action ^a (m ³)	Modify (1	P-Tunnel ^t m ³))	. New Facili (m ³)	ty ^b	Modify	P-Tunnel (m ³)	b	New Fa	cility ^b		
Τ	ransuranic 🔅	•						~	,		•		-		
	Liquid	· 1 1	None	. (0.02 ^c	•	0.02 ^c		د بر - د بر ا	0.02 ^c		0.0)2° :		
-	Solid		None	i i	0		10		i آ آ آ آ	0	· ,	, 10	· ·		
N	lixed Transuranic	•	· · · · ·				m '			1 n [1	~	·	•	•	
-	Liquid		None	- f	0		~ 0			0 🗄 🗍	·	0			
	Solid	•	- None		4	-	4		5.000 -	4`		- 4	-	,	
L	ow-Level	- 3				-			-	r.			-		~5
	Liquid	Der	pendent on restoration	2	c		2°			2.1 ^c		2.1	lc .		
,	-	. n	activities	*			ŗ				•			· •	¥ ⁶
ر - سو	Solid	,	15,000	1,26	0		1,260		1,30)0 ⁰ , ,		1,300	، بر بر	. '	
Ň	lixed Low-level	1 i - 2-	77. T. 2	•		:*	-		· · · ·	"		- *_	、"、		
	Liquid	`_• , ;	Included in solid	-	0.2 ·		· 0.2		· · · ·	0.2		··· 0.2	2		
	Solid) <u>-</u> } -	50	6	5	-	- 65		, i	5 6 • • • •		66	,		
۰H	lazardous	· · ·		·			·		-		,	•			
~~ \ •	Liquid	•	Included in solid		2	_ (-> 1 , -> 1	· · 2	-	· · · · · ·	2	• • •	2			
-	Solid	· · · · ·	212	1 4 1 4 1 6	2	•	₹2	• •	* , `	2		2	• •		
7	onhazardous (Sanitar	•v)		* • , •	、 、	· · ·				• /	<u> </u>		^ ()	~>	
•	Liquid	, , , , , , , , , , , , , , , , , , ,	Included in solid	135.00	0	'	114.000	- -	189.0	00 ÷ =	<u>.</u> -	153,000	• - •	14	
•	Solid	۰. ۱	2.120	1.62	0	•	1.500		1.9	50	· ·	1,900			
ر م	Ionhazardous (Other)	1.		-,		`			, - •-	1	<u> </u>		5		يب ب
.	Liquid	The second	cluded in sanitary	Included in	n sanitarv	Inc	luded in san	itary	Included	in sanitarv	In	cluded in s	sanitary		
•	Solid		76 500	2.00	0 d	、	1.900 ^d	· · · · · ·	2.5	00 ^d		2.400	d 🚬 🌷	`~	
<u> </u>		<u> </u>				r > 5	-					-			
а ,	The No Action waste volu	mes are fro	om Table 4.2.2.10–1.		antion E 2	۱ (Theblac	. ' ` E2122E2	122 62	" 12.0 and "	5 2122)	Worte eff	uente (that {	ic ofter		
Ū	incremental generated was	ste volumes	ch are used in the narratives a	re found in a e description	of the imp	acts are al	so provided in	these table	S.	د	waste em		is, alter		
c	Liquid TRU and LLW wor	uld be treat	ed and solidified prior to	disposal.		•	• • • •		4.1	- t		. : .	i	· .	
d	Recyclable wastes.	1 - 3	· · · · ·	- - -					, <i>(</i> .,			_	4 4 m.		
	چېنې د د د			1	۰ ۰	Ξ, ΄	- ,	به بدي اد		r • • •	***	• • •	•		
		í.,		· • ,	· ·	- 1	. ,	۰,		- 		,			
	•		······································	3 y T		, -				•	۰ ر ۲	· ·	4 % ***	*	
ĺ.		- 4.46	· · · · · · · ·		÷ ′-,		<u>ر</u> ، ۱		- ``*	· · ,	, , , , , , , , , , , , , , , , , , ,		6		
		· ·			- /.			, •	n hew	+	7	· • .	•	•	

Entry distance in the second second

 Table 4.2.2.10–2.
 Estimated Annual Generated Waste Volumes at Nevada Test Site—No Action (2005) and Net Incremental for Storage

 Alternatives

4-128

• ···

11

meet the current planning-basis WIPP WAC or alternative treatment level. While awaiting shipment to WIPP (depending on decisions made in the ROD associated with the supplemental EIS for the proposed continued phased development of WIPP for disposal of TRU waste), the TRU and mixed TRU waste would be stored at the TRU Pad Waste Storage in Area 5. One additional truck shipment per year would be required to transport these wastes to WIPP.

Following treatment and volume reduction, approximately 630 m^3 (824 yd^3) of LLW from solidified liquid LLW (such as decontamination solutions), protective clothing, HEPA filters, glovebox gloves, and decontamination equipment and materials would require disposal in the Area 5 or Area 3 RWMS. Assuming a land usage of 6,000 m³/ha ($3,200 \text{ yd}^3/\text{acre}$), this would require approximately 0.1 ha/year (0.3 acres/year) of LLW disposal area. The 0.2 m³ (50 gal) of liquid and 65 m³ (85 yd^3) of solid mixed LLW would be treated and disposed of in accordance with the NTS Site Treatment Plan through the use of existing and planned facilities. The 2 m³ (476 gal) of liquid and 2 m³ (3 yd^3) of solid hazardous wastes would have a minimal impact on waste management activities at NTS. Existing facilities at the Area 5 Hazardous Waste Storage Unit are adequate to stage the increase in hazardous waste while awaiting shipment to an offsite commercial RCRA-permitted treatment and disposal facility. New sanitary lagoons would be required to treat the 135,000 m³ (35,600,000 gal) of nonhazardous liquid wastes. After volume reduction, 810 m^3 (1,060 yd³) of solid nonhazardous waste would require disposal at one of the onsite landfills.

Construct New Plutonium Storage Facility

Construction and operation of a consolidated Pu storage facility would have an impact on existing waste management activities identical to that described above for the P-Tunnel facility, with the following exceptions. Construction of sanitary, utility, and process wastewater treatment systems would be required to treat approximately 114,000 m³ (30,100,000 gal) of liquid nonhazardous waste. After volume reduction, 750 m³ (981 yd³) of solid nonhazardous waste would require disposal at one of the onsite facilities.

Collocation Alternative

1

1

1

Modify Existing Tunnel Drifts and Construct New Material Handling Building at the P-Tunnel

Construction and operation of a consolidated Pu storage facility collocated with HEU storage using the P-Tunnel would have an impact on existing NTS waste management activities, increasing the generation of TRU, low-level, mixed, hazardous, and nonhazardous wastes. Waste generated during construction would consist of wastewater and solid nonhazardous and hazardous wastes. The nonhazardous waste would be disposed of as part of the construction project by the contractor, and the hazardous waste would be shipped to commercial RCRA-permitted treatment and disposal facilities. No soil contaminated with hazardous material or radioactive constituents is expected to be generated during construction. However, if any was generated it would be managed in accordance with site practice and all applicable Federal and State regulations.

Since there is no TRU or mixed TRU wastes associated with HEU storage, the impacts from TRU and mixed TRU wastes are identical to those identified in the consolidated Pu storage alternative. The sources of waste are similar to those of the consolidated Pu storage facility, except the source of radioactive contamination from the HEU storage is uranium.

Following treatment and volume reduction, approximately 630 m³ (824 yd³) of LLW contaminated with Pu and 20 m³ (26 yd³) contaminated with uranium would require disposal in the Area 5 or Area 3 RWMS. Assuming a land usage of 6,000 m³/ha (3,200 yd³/acre), this would require approximately 0.1 ha/yr (0.3 acre/yr) of LLW disposal area. The 0.2 m³ (55 gal) of liquid mixed LLW and 66 m³ (86 yd³) of solid mixed LLW would be treated and disposed of in accordance with the NTS Site Treatment Plan through the use of existing and planned facilities. The 2 m³ (528 gal) of liquid and 2 m³ (3 yd³) of solid hazardous wastes would have a minimal impact on waste management activities at NTS. Existing facilities at the Area 5 Hazardous Waste Accumulation Site

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

1. T . T. are adequate to stage the increase in hazardous waste while awaiting shipment to an offsite commercial RCRApermitted treatment and disposal facility. New sanitary lagoons would be required to treat the 189,000 m³ (49,900,000 gal) of liquid nonhazardous waste. After volume reduction, 980 m³ (1,280 yd³) of solid nonhazardous waste would require disposal at one of the onsite landfills. .

5

Construct New Plutonium and Highly Enriched Uranium Storage Facilities 1 . . .

Construction and operation of a consolidated Pu storage facility collocated with HEU storage would have an impact on existing NTS waste management activities identical to that described above for the modified P-Tunnel facility, with the following exceptions. Construction of sanitary, utility, and process wastewater treatment systems would be required to treat approximately 153,000 m³ (40,500,000 gal) of nonhazardous waste. After volume reduction, 950 m³ (1,240 yd³) of solid nonhazardous waste would require disposal at one of the onsite landfills.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

The exclusion of strategic reserve or weapons R&D materials would reduce the amount of operational waste volumes shown in Table 4.2.2.10-2 for the Consolidation Alternative and the Collocation Alternative. The decrease would be proportional to the amount of material excluded. [Text deleted.]

> . 1 1 . 1 . . · , , · 11 7 , j. . (1.1 11a la statue d'al composition de la seconda de la second - , We and the State of the Mar all a 2011年1月1日(1月11日)。 1911年1日 - 1月1日(1月11日) 1911年1日 - 1月1日(1月11日)

ななないとないであるとうないであるというないのであるない

4.2.3 IDAHO NATIONAL ENGINEERING LABORATORY

A listing of the proposed long-term storage alternatives, subalternatives, and related actions, including the No Action Alternative at INEL, is provided below. The potential impacts of implementing these alternatives and related actions at INEL are described for: land resources, site infrastructure, air quality and noise, water resources, geology and soils, biological resources, cultural and paleontological resources, socioeconomics, public and occupational health and safety, and waste management. The specific long-term storage alternatives proposed for INEL are the Upgrade Alternative, the Consolidation Alternative, and the Collocation Alternative.





Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

4.2.3.1 Land Resources

, ,**,** ,

,

1.5

Preferred Alternative: No Action Alternative

Under this alternative, Pu storage would continue at the current interim storage locations at the ICPP and at ANL-W in the ZPPR and FMF vaults in stabilized form pursuant to DNFSB Recommendation 94-1. The ongoing (no new action) activities conform with present and future land-use plans, policies, and controls. Therefore, no effects to land use or visual resources would be anticipated at INEL beyond those of existing and future activities that are independent of the proposed action.

1.

Star & de

- 6

fai P

111

1 -

1: 11

Upgrade Alternative * , ** .

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative 1.1.15

Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium Storage

Long-term storage of the existing inventory of INEL Pu material would be accommodated at ANL-W on the INEL site. The proposed facility would be a modification of the FMF on the ANL-W and construction of a new material handling building. Construction laydown area and the operating facility would be situated on previously disturbed land entirely within an upgraded protected area of ANL-W totalling approximately 9 ha (22 acres) and would not create any newly disturbed area. A buffer zone exists between ANL-W operations and the INEL site boundary. [Text deleted.] 1 1 4

Land Use. Upgrading existing storage facilities and constructing a new material handling building at ANL-W would have no direct land-use effect during construction or operations. Existing land use would not change and would conform with site development and facility utilization plans. As discussed in Section 4.2.3.8, no in-migration of workers would be required during the construction and operation phases. No indirect effects to offsite land use would be anticipated.

Construction and operation would not affect other land uses at INEL or special status lands. Construction and operation would not be in conflict with land-use plans, policies, and controls of adjacent counties and the city of Idaho Falls since they do not address the potential site. •3 <u>1</u> • . John Merthalt

1 1

6 2 2 3 4

The in

Visual Resources. [Text deleted.] Construction and operation of the facilities would be compatible with the industrial landscape character of ANL-W. The current VRM Class 5 designation of ANL-W would not change.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium (a) a state for the second state of the state of the second sta Storage 1. 1. 1. 1 12 1 a francisk men a

The FMF would be modified, and a material handling building would be constructed at ANL-W to accommodate INEL, RFETS, and LANL Pu material. Land area requirements during construction and operations would be equal to the Upgrade Without RFETS Pu or LANL Pu Subalternative (that is, protected area). Direct and indirect effects on land resources during construction and operations would be similar to the Upgrade Alternative, Upgrade Without RFETS Pu or LANL Pu Subalternative.

Consolidation Alternative

Construct New Plutonium Storage Facility

All the Pu within the scope of the PEIS would be stored at a new storage facility to be constructed at INEL within the Prime Development Zone of the Sitewide Area near the ICPP. Land disturbance would be 58.5 ha (144 acres) during construction of which 56 ha (138 acres) would be used during operations. A buffer zone would be provided between operations and the INEL site boundary. Pu storage in existing DOE storage facilities would be phased out.

Land Use. Consolidating the storage of Pu at INEL would be situated on undisturbed land. However, the proposed site is within designated prime development land pursuant to the current *Idaho National Engineering Laboratory Site Development Plan* (IN DOE 1994d:8-4) although existing land use would change. As discussed in Section 4.2.3.8, expected vacancies and historic housing construction rates indicate that sufficient housing would be available to accommodate the estimated in-migration of workers during the construction and operational phases. Therefore, indirect effects to offsite land use would not be anticipated.

Construction and operation would not be in conflict with land-use plans, policies, and controls of adjacent counties and the city of Idaho Falls since they do not address the potential site. Construction and operation would not affect other land uses at INEL or special status lands. No onsite grazing permits would be affected. No prime farmlands exist onsite.

Visual Resources. [Text deleted.] Construction and operation would be compatible with the industrial character of INEL's developed areas, which consist of large industrial facilities and stack plumes. The current VRM Class 5 designation of the proposed site would not change.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

All the Pu and HEU within the scope of this PEIS would be collocated at a primary new storage plant at INEL, on undisturbed land in the Prime Development Zone of the Sitewide Area near the ICPP. Land disturbance would be 89.5 ha (221 acres) during construction of which 87 ha (215 acres) would be used during operation. A buffer zone would be provided between operations and the INEL site boundary. Pu and HEU storage in existing DOE storage facilities would be phased out. Direct and indirect effects on land resources would be similar to those described under the Consolidation Alternative.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Under this subalternative, land effects during construction and operation would be almost the same in extent and magnitude to the Upgrade with All or Some RFETS Pu and LANL Pu Subalternative, Consolidation Alternative, and Collocation Alternative because the facility would be almost the same. However, because the smaller quantity of material would require smaller facilities, it is likely that less land area would be disturbed during construction and used during operations. [Text deleted.]

Phaseout

No new construction or upgrade of existing facilities would occur under phaseout of the Pu mission. INEL Pu material would be moved out of ANL-W to the Consolidation or Collocation site or to disposition. Potential impacts on visual resources could occur if facilities are not maintained.

[Text deleted.]

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

4.2.3.2 Site Infrastructure

The INEL infrastructure would be capable of supporting any of the storage alternatives without major modifications to the existing infrastructure. A comparison of site infrastructure and facilities resource needs for the various storage alternatives is shown in Table 4.2.3.2–1.

0 0

Preferred Alternative: No Action Alternative

INEL would continue Pu storage at ANL-W. No change to the baseline infrastructure is anticipated, and no additional environmental impacts would be expected.

Upgrade Alternative

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing and Construct New Argonne National Laboratory–West Facilities for Continued Plutonium Storage

Modifying the existing storage facility plus building a new facility to accommodate long-term storage of existing quantities of Pu at INEL would have minimal impact on site infrastructure. Data for construction are presented in Appendix C. [Text deleted.] As shown-in-Table 4.2.3.2–1, the INEL infrastructure would be capable of supporting the modification of the existing storage facility without major improvements. Adequate electrical energy is available from the regional power grid. [Text deleted.] All infrastructure requirements are within site capacities.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing and Construct New Argonne National Laboratory–West Facilities for Continued Plutonium Storage

To accommodate material currently at the site, plus the relocation of material from RFETS and LANL, the ANL-W storage capacity would be upgraded by constructing a new 2,550 m² (27,400 ft²) Material Handling Building to augment modified existing support buildings and by using balance of plant facilities. Building 704 would be modified for an additional 4,100 storage positions above that of the baseline upgrade, while Buildings 774 and 775 would be modified for storage support. The buildings would be interconnected by a new material transfer access corridor.

Construction for upgrading the existing storage facility, plus building a new facility to accommodate long-term storage of existing quantities of Pu, plus material relocated from RFETS and LANL would have minimal impact on site infrastructure. Data for construction are presented in Appendix C. As shown in Table 4.2.3.2–1, additional electrical energy, and peak load would be required to operate the facility. The INEL infrastructure would be capable of supporting the modification of the existing storage facility, plus building a new facility to accommodate long-term storage of existing quantities of Pu, plus material relocated from RFETS and LANL without major improvements. Adequate electrical energy is available from the regional power grid. [Text deleted.] All infrastructure requirements are within site capacities.

Since impacts associated with relocating all of the RFETS Pu and LANL Pu material to INEL for long-term storage are minimal for construction and can be managed for operations, relocating only a portion of this material to INEL would result in minimal impacts on the site infrastructure as well. Additional annual electrical energy requirements would be proportionately less than the 700 MWh/yr required for storage of the full amount of RFETS Pu and LANL Pu material, depending on the amount of material relocated to the site.

4-134
- •	Transportation		Electrical		Fuel			
Alternative	Roads (km)	Railroads (km)	Energy (MWh/yr)	Peak Load (MWe)	Oil (l/yr)	Natural Gas (m ³ /yr)	Coal (t/yr)	
No Action								
Site availability	445	48	394,200	124	16,000,000	0	11,340	
Projected usage	445	48	232,500	42	5,820,000	0	11,340	
Upgrade (without RFETS Pu or LANL Pu material)								
Projected usage with upgraded facility	445	48	236,300	43	6,460,000	0	0	
Amount required in excess to site availability	0	0	0	0	· 0	0	0	
Upgrade (with RFETS Pu and LANL Pu material)				-				
Projected usage with upgraded facility	445	48	237,000	43	6,550,000	0	0	
Amount required in excess to site availability	0	0	0	0	0	0	0	
Consolidation								
Projected usage with consolidated facility	450	53	277,500	50	5,940,000	0	22,340	
Amount required in excess to site availability	ব	<5	0	0	0	0	11,000	
Collocation				6		\$		
Projected usage with consolidated and new collocated facilities	450	53	290,500	52	5,960,000	0	25,340	
Amount required in excess to site availability	ব	ব	0	0	0	0	14,000	
Phaseout								
Projected usage without storage facility	445	48	228,700	41	5,820,000	0	11,340	
Amount required in excess to site availability	0	0	0	0	0	0	0	

 Table 4.2.3.2–1.
 Site Infrastructure Changes Required for Operation at Idaho National Engineering Laboratory (Annual)—

 No Action (2005) and Storage Alternatives

NE-TAS TRANSPORT

المحاصلين والمشاعد والمستكانية

.

Source: Modified from DOE 1995j; INEL 1993a:5.

· 1.

Same ,

a set a set of a set of the set o

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Consolidation Alternative

Construct New Plutonium Storage Facility

Under this alternative, all the Pu within the scope of this PEIS would be stored at a new storage facility located at INEL. Construction requirements would constitute a small change in resource requirements at INEL. Since coal availability is governed by usage and not by site storage capacity, the additional coal required could be procured through contractual means. Impacts on the site infrastructure would be negligible. The INEL infrastructure would be capable of supporting operation of the consolidated Pu storage facility without major modifications to the existing infrastructure. As shown in Table 4.2.3.2–1, less than 5 km (3 mi) of roads and less than 5 km (3 mi) of railroad lines would need to be added to the site. Adequate electrical energy is available from the regional power grid.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Under this alternative, all the HEU within the scope of this PEIS would be stored at INEL in a new storage facility, collocated with the consolidated Pu storage facility. Construction requirements would constitute a small change in resource requirements for INEL. Since coal availability is governed by usage and not by site storage capacity, the additional coal required could be procured through contractual means. As shown in Table 4.2.3.2–1, less than 5 km (3 mi) of roads and less than 5 km (3 mi) of railroad lines would need to be added to the site. The INEL infrastructure would be capable of supporting operation of the consolidated and collocated facilities without major modifications to the existing infrastructure. Adequate electrical energy is available from the regional power grid.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Since the existing INEL site infrastructure would be fully capable of supporting construction/modification and operation of facilities for the Upgrade With All or Some RFETS Pu and LANL Pu, Consolidation of Pu, and Collocation of Pu and HEU Alternatives, constructing and operating such facilities without including provisions for storage of strategic reserve and weapons R&D materials could be accommodated as well. Expected reductions in amounts of annual electrical energy requirements for the various storage facilities are the only site infrastructure changes expected if this subalternative is chosen because electric usage is dependant on the amount of material. [Text deleted.]

Phaseout

Because of the relatively small amounts of Pu located on the site, INEL storage operations at ANL-W would be phased out with minimal impact on the site infrastructure.

n is a true to a standard of an and the second of a second of the second of

4.2.3.3 Air Quality and Noise

Construction and operation activities associated with the No Action Alternative and the proposed storage alternatives would generate criteria and toxic/hazardous pollutants. To evaluate the air quality impacts at INEL, criteria and toxic/hazardous concentrations from the No Action Alternative and the proposed storage alternatives are compared with Federal and State standards and guidelines. Impacts from radiological airborne emissions are described in Section 4.2.3.9.

In general, all of the proposed storage facilities would emit the same types of air pollutants during construction. It is expected emissions would not exceed Federal, State, or local air quality regulations. PM_{10} and TSP concentrations will be increased, especially during peak construction periods.

The principal sources of emissions during construction include the following:

- Fugitive dust from land clearing, site preparation, excavation, and wind erosion of exposed ground surfaces
- Exhaust and road dust generated by construction equipment, vehicles delivering construction materials, and vehicles carrying construction workers

During operation, impacts from each of the individual storage facilities with respect to the concentrations of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Table 4.2.3.3–1 presents the estimated pollutant concentrations for each of the fissile materials storage alternatives, indicating little difference between alternatives with respect to impacts on air quality.

Emission rates attributed to operation of the proposed storage facilities are presented in Tables F.1.3–1 to F.1.3–3. [Text deleted.] Air pollutant emission sources associated with operations include the following:

- Operation of boilers for space heating
- Operation of diesel generators and periodic testing of emergency diesel generators
- Exhaust and road dust generated by vehicles delivering supplies and bringing employees to work
- Toxic/hazardous pollutant emissions from facility processes

Noise impacts during either construction or operation are expected to be low. Air quality and noise impacts for each storage alternative are described separately. Supporting data for the air quality and noise analysis are presented in Appendix F.

AIR QUALITY

An analysis was conducted of the potential air quality impacts of emissions from each of the storage alternatives as described in Section 4.1.3.

Section 176 (c) of the 1990 CAA amendments requires that all Federal actions conform with the applicable state implementation plan. EPA has implemented rules that establish the criteria and procedures governing the determination of conformity for all Federal actions in nonattainment and maintenance areas. These are discussed in Section 4.1.3. The attainment status of the area in which INEL is located is discussed in Section 3.4.3. Since

1	Averaging	Most Stringent Regulations or Guidelines ^a (ug/m ³)	No Action (μg/m ³)	Upgrade (µg/m ³)	Consolidation (µg/m ³)	Collocation (µg/m ³)
Pollutant	a a c ign y c					
Carbon monoxide	8-hour	10,000 ^b	284	284.36	285.4 617.4	- 285.6 618.1
11 P (11 7 12P	1-hour	40,000	614	014.91	0.001	0.001
Lead The reaction	Calendar Quarter	1.5°.	0 001	0.001	4.73	4 91
Nitrogen dioxide	Annual	100 ⁰	4	4.02	4.75 c	c
Ozone	1-hour . "	235 ^b ,	C		5 OF 1	s ns i l
Particulate matter less than or equal	Annual	50 ^b	5	5.01	5.05	5.00
to10 microns in diameter	, r* *	h	· ,	PO 14	80.98	81.17
· · · · · · · · · · · · · · · · · · ·	24-hour	150	80	80.14	7.25	7.53
Sulfur dioxide	Annual	80 ⁰	6	0.04	159.5	163.7
r - r - r - r	24-hour	365	135	135.71	130.3 ·	704.2
	3-hour	1,300	579	582.07	001.5	1 U T. 20
Mandated by Idaho	11.177		4		5 05	5.06
Total suspended particulate	Annual	60 ^a .	5	5.1	0.00 ⁴	81.17
7 2 3 [°] 4	24-hour	150 ^d	80	80.4	80.98	61.17
Hazardous and Other Toxic	• • • • • 1	3	6 Y	C	r	1 . 1 . 2.
Compounds	* 1251 +	0.458	0.011	0.011	0.011+**	0.011 📬
Acetaldehyde	Annual	0.45	0.011	0.01) >	60	6.0
Ammonia	Annual	180°	6.0	0.0	0.00009	0.00009
Arsenic	Annual	0.00023	0.00009	0.0009	0.00002	0.029
Benzene	Annual	0.12	0.029	0.029	0.025	0.001
1,3-Butadiene	Annual	···· - 0.0036 ^e	0.001	0.001		0.0060
Carbon tetrachloride	Annual 🔡 🕫	0.067 ^e	0.0060	0.0000		<0.018
Chlorine	Annual	30 ^e	•	<0.015	<0.01°	0 00040
Chloroform	Annual	0.043 ^e	0.00040	0.00040	0.00040	0.000-10 0 7
Cyclopentane	Annual	17,000 ^e	2.7	2.7	2.7	0.012
Formaldehyde	Annual	0.077°	0.012	0.012	- 0.012	0.012
Hexavalent chromium	Annual	~ 0.000083 ^e	0.00006	0.00006	0.0000	0.0000

Table 4.2.3.3–1.Estimated Operational Concentrations of Pollutants at Idaho National Engineering Laboratory and Comparison With MostStringent Regulations or Guidelines—No Action (2005) and Storage Alternatives

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Pollutant	Averaging Time	Most Stringent Regulations or Guidelines ^a (µg/m ³)	No Action (µg/m ³)	Upgrade (µg/m ³)	Consolidation (µg/m ³)	Collocation (µg/m ³)
Hazardous and Other Toxic Compounds (Continued)						
Hydrogen chloride	Annual	7.5 ^e	0.98	0.98	0.98	0.98
Hydrazine	Annual	0.00034 ^e	0.000001	0.000001	0.000004	0.000004
Mercury	Annual	1 ^e	0.042	0.042	0.042	0.042
Methylene chloride	Annual	0.24 ^e	0.006	0.006	0.006	0.006
Naphthalene	Annual	500 ^e	18	18	18	18
Nickel	Annual	0.0042 ^e	0.0027	0.0027	0.0027	0.0027
Nitric acid	Annual	50 ^e	0.64	0.64	0.64	0.64
Perchloroethylene	Annual	2.1 ^e	0.11	0.11	0.11	0.11
Phosphorus	Annual	1 ^e	0.30	0.30	0.30	0.30
Phosphoric acid	Annual	10 ^e	f	<0.01 ^g	<0.01 ^g	<0.01 ^g
Potassium hydroxide	Annual	20 ^e	0.20	0.20	0.20	0.20
Proprionaldehyde	Annual	4.3 ^e	0.30	0.30	0.30	0.30
Styrene	Annual	1,000 ^e	1.3	1.3	1.3	1.3
Sulfuric acid	Annual	10 ^e	f	<0.01 ^g	<0.01 ^g	<0.01 ^g
Toluene	Annual	3,750 ^e	370	370	370	370
Trichloroethylene	Annual	0.077 ^e	0.00097	0.00097	[~] 0.00097	0.00097
Trimethylbenzene	Annual	1,230 ^e	100	100	100	100
Trivalent chromium	Annual	5 ^e	0.036	0.036	0.036	0.036

Table 4.2.3.3–1.	Estimated Operational Concentrations of Pollutants at Idaho National Engineering Laboratory and Comparison With Most
- ,	Stringent Regulations or Guidelines—No Action (2005) and Storage Alternatives—Continued

^a The more stringent of the Federal and State standard is presented if both exist for the averaging time.

^b Federal and State standard.

^c Ozone, as a criteria pollutant, is not directly emitted or monitored by the candidate site. See Section 4.1.3 for a discussion of ozone-related issues.

^d State standard or guideline.

• Acceptable air concentrations listed in Rules for the Control of Air Pollution in Idaho apply only to new (not existing) sources and are used here only as reference levels.

^f No sources of this pollutant have been identified.

^g The concentration represents the alternative contribution only.

Note: Concentrations are based on site contribution, including concentrations from ongoing activities (No Action), and do not include the contribution from non-facility sources (for example, traffic).

Source: 40 CFR 50; DOE 1995v; DOE 1996e; DOE 1996f; FDI 1996a:1; ID DHW 1995a; ID DHW 1995b; IN DOE 1996a.

. - .*

4-139

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

1 7 ... the area is considered to be an attainment area for criteria pollutants, the proposed actions at this site do not require that a conformity analysis be performed.

40231

and the states and the second

Preferred Alternative: No Action Alternative

This alternative utilizes estimated air emissions data from total site operations at INEL assuming continuation of site missions as described in Section 3.4. These data reflect conservative estimates of criteria and toxic/hazardous emissions at INEL. The emission rates for the criteria and toxic/hazardous pollutants for No Action are presented in Table F.1.2.4-1. Table 4.2.3.3-1 presents the No Action concentrations. During dry and windy conditions, a increased PM₁₀ and TSP concentrations may occur due to ongoing construction associated with other activities (that are outside of the scope of this PEIS) under the No Action Alternative. Concentrations of all other criteria and toxic/hazardous air pollutants at the site boundary or public access highways are expected to remain within applicable Federal, State, and local ambient air quality standards.

Upgrade Alternative

et a car

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

, *** y ² ** 2 * *

Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium Storage

Particulate matter and TSP concentrations are expected to increase during the peak construction period, particularly during dry and windy conditions. Appropriate control measures would be followed to minimize pollutant concentrations during construction. It is expected that concentrations of all pollutants at the site boundary or public access highways would remain within applicable Federal and State ambient air quality / standards during construction of new and modified facilities.

During operation, concentrations of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Estimated pollutant concentrations, attributable to increased operations associated with this storage alternative, plus the No Action concentrations, are presented in Table 4.2.3.3-1. Concentrations of air pollutants are expected to be the same with or without the RFETS and LANL material.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative a sufficiency of the second commencies and the active second second second second second second second second s Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium and the second and the Storage

Air quality impacts for construction and operation for this subalternative are expected to be similar to those previously described for the Upgrade Alternative for INEL.

and the state of the Consolidation Alternative the state of the second strand of the state of the second state Construct New Plutonium Storage Facility struct New Plutonium Storage Facility is the state of the

In addition to the types of sources of emissions during construction associated with the No Action and upgrade of storage facilities, fugitive dust resulting from the operation of a concrete batch plant would be an additional emission source associated with a new facility. PM_{10} and TSP concentrations may increase during the peak construction period for a new facility, particularly during dry and windy conditions. Appropriate control measures would be followed to minimize pollutant concentrations during construction. It is expected that

4-140

construction period for a new facility, particularly during dry and windy conditions. Appropriate control measures would be followed to minimize pollutant concentrations during construction. It is expected that concentrations of all pollutants at the site boundary or public access highways would remain within applicable Federal and State ambient air quality standards during construction.

During operation, impacts of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Estimated pollutant concentrations attributable to increased operations associated with this storage alternative, plus the No Action concentrations, are presented in Table 4.2.3.3–1.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

The Pu and HEU Collocation Alternative would be located in the same area as the consolidation Pu facility and would have similar air quality impacts, with the following exceptions. During operation, emissions would be higher, as shown in Appendix F. Impacts of criteria and toxic/hazardous air pollutants are predicted to be in compliance with Federal, State, and local air quality regulations or guidelines. Estimated pollutant concentrations attributable to increased operations associated with this storage alternative, plus the No Action concentrations, are presented in Table 4.2.3.3–1.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Air quality impacts for construction and operation for this subalternative are expected to be similar to those previously described for the Upgrade With All or Some RFETS and LANL Pu Subalternative, the Consolidation Alternative, and the Collocation Alternative. [Text deleted.]

Phaseout

Phaseout of existing Pu inventories as a result of consolidating Pu at another site is expected to result in a small reduction in air pollutant concentrations from the No Action concentrations and would be in compliance with Federal and State standards.

NOISE

The location of the proposed storage facilities relative to the site boundary and sensitive receptors was examined to evaluate the potential for onsite and offsite noise impacts. Noise sources during construction may include heavy construction equipment and increased traffic. Increased traffic would occur onsite and along offsite local and regional transportation routes used to bring construction materials and workers to the site.

Preferred Alternative: No Action Alternative

Nontraffic noise sources associated with continued storage and other ongoing missions are the same as described in Chapter 3. The continuation of operations at INEL would result in no appreciable change in traffic noise and onsite operational noise sources from current levels. Nontraffic noise sources are located at a sufficient distance from offsite areas that the contribution to offsite noise levels would continue to be small. Due to the size of the site, noise emissions from construction equipment and operations activities would not be expected to cause annoyance to the public. Some noise sources may be located close enough to onsite noise sensitive areas to result in impacts, such as disturbance of wildlife.

33

Į,

こうなない べきおねのたいなない どうこう

Upgrade, Consolidation, and Collocation Alternatives

Nontraffic noise sources associated with the storage Upgrade Alternative would be similar to those for existing facilities as discussed in Chapter 3. Nontraffic, operational noise sources associated with the storage alternatives include existing or additional equipment and machines (cooling systems, vents, motors, and material handling equipment). These noise sources would be located at a sufficient distance from offsite areas that the contribution to offsite noise levels would be small. Due to the size of the site, noise emissions from construction equipment and operations activities would not be expected to cause annoyance to the public. Some noise sources may result in impacts, such as disturbance of wildlife.

Y , Marson is

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Noise impacts for construction and operations for this option are expected to be almost the same as those previously described for the Consolidation Alternative and the Collocation Alternative because noise impacts are based on the use of the facility and not the size. [Text deleted.]

Phaseout

A reduction in noise levels associated with facility operations may result from the phaseout of storage facilities.

. *¹. 5. . . .

こうとやいながくとうかいるななななない かざ ちょうていないないないない

and the second second

111 112 12 211 1 1201 ÷., alainer, E 10 - 10 - 2 1 1 merily to - '|: • 5.5 8 · · · · · · · · · 1 -£ . Set in your on the arrent of the fiber of the 1 - 2 to me that you is post the instant set The second and the second second second second - i \$ 17 real of the second destruction is a second 151 and the second products of the second s and the state of the

4.2.3.4 Water Resources

Construction and operation of the proposed long-term storage facilities at INEL would affect water resources. No surface water would be withdrawn for construction or for normal operations. Instead, groundwater from the Snake River Plain Aquifer would be used, which is a sufficient source. Water requirements for normal operation for all storage options would fall within INEL's current allotment (43,000 million l/yr [11,360 million gal/yr]). The site proposed for the upgraded storage facilities would be outside the floodplain that could result from failure of MacKay Dam during a probable maximum flood. The site proposed for the Consolidation or Collocation Alternative, however, falls within that floodplain. During construction, treated sanitary wastewater would be discharged to lined evaporation ponds. While the potential impacts on surface water during the construction phase would result from erosion and sedimentation of drainage channels, the relatively dry climate and application of appropriate controls should preclude these potential impacts. No wastewater would be discharged on surface waters during operation of the facilities, nor would there be impacts on surface water quality from these activities. All wastewater would be treated and recycled for cooling system makeup. Stormwater runoff would be collected and treated, if necessary, before discharge to natural drainage channels. [Text deleted.] Table 4.2.3.4–1 presents No Action water resources uses and discharges and the potential changes to water resources at INEL resulting from the long-term storage alternatives.

Preferred Alternative: No Action Alternative

Surface Water. A description of the activities that would continue at INEL is provided in Section 3.4. Treated wastewater discharged to evaporation/infiltration ponds is expected to continue at a rate of 540 million l/yr (142.7 million gal/yr).

Groundwater. Under this alternative, no additional impacts on groundwater resources are anticipated. Current groundwater usage of 7,949 million l/yr (2,100 million gal/yr) is anticipated to decrease to 7,570 million l/yr (2,000 million gal/yr) by 2005. Existing tritium plumes in groundwater and in perched groundwater are expected to continue to migrate southwest slowly. Studies show that water withdrawals could change the existing plumes' southwesterly direction to the east.

Upgrade Alternative

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium Storage

Surface Water. There are no unique construction characteristics associated with water requirements and discharges from this alternative. No surface water would be withdrawn for any construction or operation activities associated with any of the proposed upgraded Pu storage facilities. Therefore, there would be no impacts on surface water availability. Nonhazardous wastewater generated during construction and operation of the upgraded Pu storage facilities would be diverted to either the sanitary or industrial waste treatment ponds, where it would be allowed to evaporate into the atmosphere and percolate into the subsurface. It is expected that a total of approximately 4.0 million I/yr (1.1 million gal/yr) of nonhazardous wastewater treatment ponds, where it would undergo aerobic and anaerobic treatment and then be allowed to evaporate to the atmosphere and percolate into the subsurface to the atmosphere and percolate into the subsurface under NPDES permit requirements.

÷ .

a. . .

		Upgı	rade	_			
Affected Resource Indicator	No Action	Without RFETS or LANL Material	With RFETS and LANL Material	Consolidation	Collocation	Phaseout	_
Water Source	Ground	Ground	Ground	Ground	Ground	Ground	
Construction							
Water Availability and Use							
Total water requirement (million l/yr)	NA ^a	9.7	12.5	85	104.7	0	
Percent increase in projected water use ^b	NA ^a	0.1	0.2	1.1	1.4	0	
Water Quality				_		•	را ال
Total wastewater discharge (million l/yr)	NA ^a	4.0	6.1	7.8	12.8	0	
Percent change in wastewater discharge ^c	NA ^a	0.7	1.1	1.4	2.4	0	* ¥jao * ~
Operation							
Water Availability and Use	···						
Total water requirement (million l/yr)	7,570	17	22	66	87	0	
Percent increase in projected water used	· · ·	····· 0.2	0.3	0.9	1.2	0	
Water Quality	, ¹ • ¹ •			••••	• • • •	• •	
						_	
Total wastewater discharge (million l/yr)	540	0	0	0	Sterrer 0 - 14	0	`
Total wastewater discharge (million l/yr) Percent change in wastewater discharge	540 0	0 ,0	0	· · · · · · · · · · · · · · · · · · ·	37777 0 - 14 149 ² - 10	··· · 0 (2)	1
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^e	540 0	0 ,, 0	0 0	· · · · · · · · · · · · · · · · · · ·	Stearn Orige Alfaber 6 Charles Alfaber 6		۲ ۲
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c	540 0	0	0 0		Contraction 0 - 14 1412 - 10 - 11 1412 - 10 - 11 1412 - 11 141		۰ ۱۰
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^e	540 0	0	· · · · · · · · · · · · · · · · · · ·		Charles Oral and Altable of Oral and Altable o		
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c	540 0	0	· · · · · · · · · · · · · · · · · · ·		Sterrer 0 - 13 1992 - 10 2013 - 10 2014 - 10 2		
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^e	540	0 ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			Conne Origi 1992 - Conne 2002 - Conne 2003 - Conne Conne 2003 - Conne 2		
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^e	540 0						
Total wastewater discharge (million l/yr) Percent change in wastewater discharge	540				Control 0 - 13 14122 - 10 1 - 12 1 - 12 1 - 12 2 - 12 2 - 12 1 - 12 2 - 12 1		۲
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c	540 , 0				Constraint Ore 19 1993 - 20 - 20 2013 - 20 2014 - 20		
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c	540 0				Connection 0 - 13 1 1 12 2 2 0 0 - 13 2 1 12 2 2 0 0 - 14 2 1 12 2 2 0 0 - 14 2 1 12 2 0 - 14 2 1 12 - 14 2 1 12 2 0 - 14 2 1 12 - 14 2		۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c	540 0						
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c	540						
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c	540 , 0						
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c	540						
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c	540						
Total wastewater discharge (million l/yr) Percent change in wastewater discharge ^c							

Î

1. 2

 Table 4.2.3.4–1. No Action and Potential Changes to Water Resources at Idaho National Engineering Laboratory—No Action (2005) and Storage Alternatives—Continued

		Upg	rade			*	
Affected Resource Indicator	No Action	Without RFETS or LANL Material	With RFETS and LANL Material	Consolidation	Collocation	Phaseout	
Floodplain							
Is action in 100-year floodplain?	NA	No	No	No	No	No	
Is critical action in 500-year floodplain?	NA	No	No	Uncertain	Uncertain	No	

* See operations section of table for No Action water data.

^b Percent increases in projected water use during construction at INEL are calculated by dividing No Action water requirements (7,570 million l/yr with that for each storage option: upgrade existing storage facility without RFETS or LANL material (9.7 million l/yr), Pu upgrade with all RFETS and LANL material (12.5 million l/yr), consolidate Pu storage facility (85 million l/yr), collocate Pu and HEU storage facility (104.7 million l/yr), and storage phaseout (0 l/yr).

^c Percent changes in wastewater discharge during construction at INEL are calculated by dividing No Action wastewater discharges (540 million l/yr) with that for each storage option: upgrade existing storage facility without RFETS or LANL material (4.0 million l/yr), Pu upgrade with all RFETS and LANL material (6.1 million l/yr), consolidate Pu storage facility (7.8 million l/yr), collocate Pu and HEU storage facility (12.8 million l/yr), and storage phaseout (0 l/yr).

^d Percent increases in projected water use during operation at INEL are calculated by dividing No Action water requirements (7,570 million l/yr) with that for each storage option: upgrade existing storage facility without RFETS or LANL material (17 million l/yr), Pu storage upgrade with all RFETS and LANL material (22 million l/yr), consolidate Pu storage facility (66 million l/yr), collocate Pu and HEU storage facility (87 million l/yr), and storage phaseout (0 l/yr).

Present changes in wastewater discharged during operation at INEL are calculated by dividing No Action wastewater discharges (540 million l/yr) with that for each storage option: upgrade existing storage facility without RFETS or LANL material (0 l/yr), Pu storage upgrade with all RFETS and LANL material (0 l/yr), consolidate Pu storage facility (0 l/yr), collocate Pu and HEU storage facility (0 l/yr), and storage phaseout (0 l/yr).

Note: NA=not applicable. Construction impacts are considered to be temporary, lasting only throughout the construction period. Impacts from operations would occur continuously. During operations wastewater will be recycled.

Source: DOE 1996e; DOE 1996f; IN DOE 1996a; INEL 1995a:1.

٠.

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

[Text deleted.]

During operation, utility, process, and sanitary wastewater for the upgraded Pu storage facilities would be diverted to the sanitary waste treatment ponds, where it would undergo aerobic and anaerobic treatment and then be allowed to evaporate into the atmosphere and percolate into the subsurface. Similarly, cooling system blowdown and stormwater runoff would be diverted to the industrial waste treatment ponds and ANL-W sewage lagoons, where it would be allowed to evaporate or percolate. Industrial and sanitary wastewater treatment pond water is monitored for the parameters specified in the site-specific NPDES permit. If evaporation pond capacity is limited, uncontaminated effluents would be discharged to natural drainage channels. Contaminated effluents would be diverted to, and treated in, the liquid radioactive waste treatment system before disposal.

W. 2.

No construction would occur in areas delineated as 100-year floodplains. The proposed site is also located above the maximum probable flood elevation, which is higher than the 500-year flood elevation. The closest large surface water body, the Big Lost River, is located approximately 16 km (10 mi) west of the proposed site. Because INEL is in a region where flash floods could occur, the facilities would be designed to withstand such flooding.

Groundwater. All water required for construction and operation would be supplied from groundwater from the Snake River Plain Aquifer. Construction water requirements for the upgraded Pu storage upgrade are small relative to INEL's total usage. As shown in Table 4.2.3.4–1, upgrading the ANL-W facilities would require approximately 9.7 million l/yr (2.6 million gal/yr) of water, which represents a 0.1-percent increase over the projected annual groundwater usage. Annual groundwater-requirements for operation of the proposed facilities are estimated to be approximately 17 million 1 (4.5 million gal), which represents a 0.2-percent increase over the projected No Action groundwater usage. This small increase in overall demand should cause minimal impacts. This would increase the total projected amount to be pumped at INEL to under 18 percent of the total allotment.

Construction and operation of the proposed upgraded Pu storage facilities would not result in direct discharges to groundwater. Treated wastewater discharged to evaporation/infiltration ponds, however, would percolate downward into the groundwater. The water would be monitored and would not be discharged into the ponds until contaminant levels are within the limits specified. Impacts on groundwater quality are therefore not expected. In addition, other factors contributing to a lessening of potential impacts on groundwater are the combined effects of a deep water table, low discharge volumes, and high evaporation rates. Therefore, the tritium contamination problem in the Snake River Plain Aquifer, as identified in Section 3.4.4, would not be exacerbated by any of the long-term storage alternatives.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium 35 Storage 77

The Pu storage upgrade using all or some RFETS Pu and LANL Pu material at INEL would increase water discharges during construction by 6.1 million l/yr (1.6 million gal/yr), or 1.1 percent over the projected No Action discharge during construction. During operations, wastewater would be recycled. All other wastewater requirements and floodplain issues of the Pu storage upgrade with RFETS Pu and LANL Pu material are similar to those of the Pu consolidated option. During construction, Pu storage upgrade using RFETS Pu and LANL Pu material would require 12.5 million l/yr (3.3 million gal/yr), or a 0.2-percent increase over projected No Action water use. During operations, 22 million l/yr (5.8 million gal/yr) of water would be required, or a 0.3-percent increase over projected No Action water use.

Water resources impacts for construction and operation upgrading with some RFETS Pu and LANL Pu material are expected to be similar to, but less than, those previously described for the other storage options at INEL.

4-146

Consolidation Alternative

Construct New Plutonium Storage Facility.

The new consolidated Pu storage facility would be located just outside the ICPP area of INEL. The impacts associated with it are the same as those discussed above for the upgrade of the existing Pu storage area, with the following exceptions.

- . · · ·

. .

54

Sanitary wastewater quantities generated during construction of this alternative would be approximately 7.8 million l/yr (2.1 million gal/yr). These effluents would be discharged to evaporation/infiltration ponds. No impacts are expected. Surface water would not be used for this option, so no impacts on surface water availability would be expected. The groundwater requirements of this option are slightly greater than those for the previous option. This option would require approximately 85 million l/yr (22.5 million gal/yr) and 66 million l/yr (17.4 million gal/yr) of water for construction and operation, respectively. These additional requirements represent 1.1- and 0.9-percent increases, respectively, in the projected No Action annual withdrawals from the Snake River Plain Aquifer and should not cause any impacts on groundwater availability.

The proposed site for this facility falls within the estimated floodplain that could result from failure of the MacKay Dam during a maximum flood, which would be greater than the 500-year flood.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

The new consolidated and collocated storage facilities would be located in the same area as the new storage facility, just outside the ICPP area of INEL. The impacts associated with it are the same as those discussed above, with the following exceptions.

Sanitary wastewater quantities generated during construction and operation of this option would be greater than for the previous option and are approximately 12.8 million l/yr (3.4 million gal/yr). These effluents would be discharged to evaporation/infiltration ponds. During operations, wastewater will be recycled. No impacts are expected. Groundwater requirements during construction and operation of this option would be slightly greater than those for the new Pu storage facility. This option would require approximately 104.7 million l/yr (27.7 million gal/yr) and 87 million l/yr (23 million gal/yr) for construction and operation, respectively. These additional requirements represent 1.4- and 1.2-percent increases, respectively, in the projected No Action annual groundwater withdrawals. These small increases boost the total projected groundwater withdrawal to a maximum of 17.8 percent of the groundwater allotment; there should be no impact on groundwater availability.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Water resource impacts for construction and operation for this option are expected to be slightly less than those previously described for the Pu consolidated and Pu and HEU collocated storage alternatives at INEL because of the reduction in the amount of material. [Text deleted.]

Phaseout

× .

If the current Pu storage mission at INEL was phased out, groundwater withdrawals from the Snake River Plain Aquifer and nonhazardous wastewater discharge to evaporation/percolation ponds would decrease by negligible quantities. No noticeable impacts would occur or be alleviated due to these decreases.

[Text deleted.]

Storage and Disposition of Weapons-Usable Fissile Materials Final PEIS

Geology and Soils 4.2.3.5

Construction and operation of the alternatives at INEL would have no impact on the geologic resources. A moderate seismic risk exists, but would be considered in the design of the proposed alternatives. The existing seismic risk does not preclude the safe construction and operation of the proposed alternative facilities. The facilities would be designed for earthquake-generated ground accelerations, in accordance with DOE O 420.1, Facility Safety. Because there are no known capable faults at INEL, the potential for ground rupture as a result of an earthquake during the life of a proposed alternative is minimal; ground shaking is more likely. Intensities of approximately VII on the MMI scale are possible but would not affect newly designed facilities. Human health effects from accidents initiated by natural phenomena (for example, earthquakes) are discussed in Section 4.2.3.9. Volcanic activity is improbable during the life of the alternatives and is not anticipated to affect the construction and operation of the alternatives. Lava extrusions could recur with a recurrence probability at approximately once in every 3,000 years. Precursors, such as shallow earthquakes, gas venting activity, and an increase in groundwater temperatures can provide advance warning of most eruption of this type; no such activity is currently indicated at INEL. It is highly unlikely that landslides, sinkhole development, or other nontectonic events would affect project activities. Slopes and underlying foundation materials are generally considered stable. Geologic resources at INEL consist of surficial sand, gravel, or clay deposits that have low economic value. New construction may increase the use of these materials, but because large volumes of these materials are present, the effect to the geologic resource is anticipated to be negligible.

- end Alt - to go and - to g

Impacts to the geologic and soil resources occur during, or as a result of, ground-disturbing construction activities. Construction of the alternatives may involve ground-disturbing activities that could affect the soil resources. The amount of land disturbed is specified below for each alternative. Impacts to the soil resource depend on the specific soil units in the disturbed area, the extent of land-disturbing activities, and the amount of soil disturbed. Control measures would be employed to minimize soil erosion. Within INEL, the soil erosion potential is directly related to the amount of land disturbed because soil and climatic conditions are similar throughout the site.

[Text deleted.]

Under the No Action Alternative, DOE would continue current and ongoing activities at INEL. There would be no ground-disturbing activities beyond those associated with existing and future site improvements. Because no new construction and the associated ground disturbance for potential soil erosion would occur, the No Action Alternative would have no effect on the soil resources at the site.

Upgrade Alternative to the data of the second state of the second

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium Storage

No apparent direct or indirect effects on the geologic resource are anticipated because neither facility construction and operation activity nor site infrastructure improvements will restrict access to potential geologic resources. Design of the facilities would ensure that they would not be affected by potentially hazardous geologic conditions.

Construction activities will occur completely on previously disturbed land, as described in Section 4.2.3.1, and involve land disturbance of approximately 9 ha (22 acres). Soil disturbance would occur primarily from ground-

4-148

2 . .

disturbing construction activities (foundation preparation) and activities associated with building construction laydown areas that can expose the soil profile and lead to a possible increase in soil erosion as a result of wind and water action. Soil loss would depend on the frequency and severity of rain, wind velocities (increases in wind velocity and duration increase potential soil erosion), and the size, location, and duration of grounddisturbing activities.

Net soil disturbance during operations would be considerably less than during construction because areas temporarily used for construction laydown would be restored. Although stormwater runoff and wind action could occur during operation, they are anticipated to be minimal. [Text deleted.]

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium Storage

Construction and operation effects on geological and soil resources would be the same as those discussed previously for the upgrade without RFETS or LANL Pu Subalternative, because the inclusion or exclusion of RFETS and LANL Pu material would not change the amount of land disturbed during construction.

Consolidation Alternative

ためためでいたかないというないのないないない

Construct New Plutonium Storage Facility

No apparent direct or indirect effects on the geologic resources are anticipated, because neither facility construction and operational activities nor site infrastructure improvements will limit access to potential geologic resources.

[Text deleted.] Additional soil impacts would be expected from the construction of the storage facility, which will occur completely on undisturbed land, as described in Section 4.2.3.1. Approximately 58.5 ha (144 acres) would be disturbed for the Consolidation Alternative, affecting the soil profile and leading to a possible increase in erosion. Analysis in this section is the same as that provided for the Upgrade Alternative.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

No apparent direct or indirect effects on the geologic resource are anticipated, because neither facility construction and operational activities nor site infrastructure improvements will restrict access to potential geologic resources.

[Text deleted.] Construction of the storage facilities would occur on undisturbed land as described in Section 4.2.3.1. However, additional soil impacts would be anticipated because this alternative has the largest construction and land use requirements. During construction, approximately 89.5 ha (221 acres) would be disturbed for the new facilities, affecting the soil profile and leading to a possible temporary increase in erosion as a result of stormwater runoff and wind action. Soil impacts during operation are expected to be minimal. Analysis in this section is the same as that provided for the Upgrade Alternative.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

Exclusion of strategic reserve and weapons R&D materials would give almost the same effects to the soil resources for the No Action Alternative, the Upgrade Alternative, the Consolidation Alternative, and the Collocation Alternative. By excluding these materials the size of a facility would be similar, thus not changing the amount of land disturbed by construction activities. No effect to the geologic resource is anticipated as a result of this subalternative.

Phaseout

The phaseout of storage capacity would have no apparent effects on the geologic resources. However, phaseout could result in beneficial effects on the soils of the area. Hazardous, radioactive, and mixed waste sources would be eliminated from the area, thus decreasing the potential for future soil contamination.

[Text deleted.]

and the second second

(a) A strateging a strateging of the MCM strateging of the St

75

The second se

- ここでながらした。 ほうしょう かんした 手 部にかっ 脱液 しゅうたい シューション ション・ション

Biological Resources 4.2.3.6

Preferred Alternative: No Action Alternative

. .

and a set Under this alternative, the Pu storage mission described in Section 2.2.3 would continue at INEL. These activities would result in no appreciable change to current conditions of biological resources at INEL, as described in Section 3.4.6.

Upgrade Alternative

Upgrade Without Rocky Flats Environmental Technology Site Plutonium or Los Alamos National Laboratory Plutonium Subalternative

Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium Storage

Upgrading existing Pu storage facilities at the ANL-W area of INEL would cause minimal disturbance to biological resources. This is because all activities, including some new construction, would take place within an area that is currently disturbed. Noise associated with construction could cause some temporary disturbance to wildlife, but this impact would be minimal since animals living adjacent to the current facility would have already adapted to its presence. Impacts on wetlands and aquatic resources would not occur since these resources are not found in the upgrade area, and all discharges would be to existing evaporation ponds. Since the upgrade would take place within a developed area, impacts on threatened and endangered species would not be expected.

Upgrade With All or Some Rocky Flats Environmental Technology Site Plutonium and Los Alamos National Laboratory Plutonium Subalternative

Modify Existing and Construct New Argonne National Laboratory-West Facilities for Continued Plutonium Storage

Upgrading with all or some of the RFETS and LANL materials stored at INEL would not be expected to change impacts on biological resources from those described for the Upgrade without RFETS Pu or LANL Pu Subalternative.

Consolidation Alternative

ł

I

Construct New Plutonium Storage Facility

Under this alternative, Pu reserves would be consolidated in a new storage facility at INEL. Impacts on terrestrial resources, wetlands, aquatic resources, and threatened and endangered species are discussed below.

Terrestrial Resources. Construction of the Consolidated Pu storage facility would result in the disturbance of 58.5 ha (144 acres) of terrestrial resources, or less than 0.03 percent of INEL. This includes areas on which plant facilities would be constructed, as well as areas revegetated following construction. Vegetation within the proposed site would be destroyed during land-clearing operations. Big sagebrush is the dominant plant within the proposed site. Plant communities in which big sagebrush is the dominant overstory species are well represented on INEL, but they are relatively uncommon regionally because of widespread conversion of shrubsteppe habitats to agriculture.

Construction of the Pu storage facility would affect animal populations. Less-mobile animals within the project area, such as reptiles and small mammals, would not be expected to survive. Construction activities and noise would cause larger mammals and birds in the construction and adjacent areas to move to similar habitat nearby. If the area to which they moved was below its carrying capacity, these animals would be expected to survive. However, if the area was already supporting the maximum number of individuals, the additional animals would compete for limited resources, which could lead to habitat degradation and eventual loss of the excess population. Because pronghorn use of the proposed site is relatively low, the facility should not have a lasting impact on these species. Nests and young animals living within the proposed site may not survive. The site would be surveyed as necessary for the nests of migrating birds prior to construction. Areas disturbed by construction but not occupied by facility structures would be of minimal value to wildlife because they would be maintained as landscaped areas.

الكلية يتحد م

18-17-11

Activities associated with facility operations, such as noise and human activity, could affect wildlife living immediately adjacent to the Pu storage facility. These disturbances may cause some species to move from the area. Disturbance to wildlife living adjacent to the facility would be minimized by preventing workers from entering undisturbed areas. Impacts on vegetation from salt drift would not occur since dry cooling systems would be used.

Wetlands. Construction and operation of the Pu storage facility would not affect wetlands since there are no wetlands on the proposed site. Wetlands associated with the Big Lost River are located 2.4 km (1.5 mi) from the site, so impacts on these wetlands are not expected.

Aquatic Resources. Construction and operation of the Pu storage facility would not affect aquatic resources since there are no surface water bodies on the proposed site. The nearest surface water body is in the Big Lost River, which is located 2.4 km (1.5 mi) from the site. Temporary aquatic habitat may develop in evaporation and retention ponds, as well as in natural channels in the immediate vicinity of NPDES-permitted outfalls.

Threatened and Endangered Species. It is unlikely that federally listed threatened or endangered species would be affected by construction of the Pu storage facility on INEL, but several State-status species may be affected. Up to 58.5 ha (144 acres) of habitat would be lost. Burrows and foraging habitat for the pygmy rabbit would be lost. Bat species such as the Townsend's western big-eared bat may roost in caves and forage throughout the proposed site. One State-listed sensitive plant species could potentially be affected by construction of the facility. The plant species, treelike oxytheca, has been collected at eight sites on INEL and at only two other sites in Idaho (IN DOE 1984a:34,36). If present, individual plants of this species could be destroyed during land-clearing activities. Preactivity surveys would be conducted as appropriate prior to construction to determine the presence of these species in the area to be disturbed. Consultation with USFWS and State agencies would be conducted at the site-specific level, as appropriate.

During operation of the new facilities, several bat species could forage at evaporation and stormwater retention ponds. No impacts on threatened and endangered species are expected due to facility operation.

Collocation Alternative

Construct New Plutonium and Highly Enriched Uranium Storage Facilities

Under this alternative, consolidated Pu reserves would be stored with HEU inventories in a new collocated storage facility(s) at INEL. Construction and operation of collocated storage facilities at INEL would have effects on biological resources similar to, but somewhat greater than, those described for the consolidated storage facility. Construction of the collocated storage alternative would disturb 89.5 ha (221 acres) of habitat.

Subalternative Not Including Strategic Reserve and Weapons Research and Development Materials

The exclusion of strategic reserve and weapons R&D materials would have almost the same effects to the Upgrade With All or Some RFETS and LANL Pu Subalternative, the Consolidation Alternative, and the

4-152

11

1

E

Collocation Alternative. The size of the facility would be similar and would not reduce the amount of habitat disturbance, and the potential impacts on biological resources would be similar. [Text deleted.]

Phaseout

The phaseout of Pu storage facilities at INEL would not be expected to affect biological resources, although increased human activity could temporarily disturb some wildlife species in the vicinity of the site.